UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

x QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT 1934

For the quarterly period ended September 30, 2018

0 TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission File Number: 001-37806

Twilio Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of

incorporation or organization)

26-2574840 (I.R.S. Employer Identification Number)

375 Beale Street, Suite 300 San Francisco, California 94105 (Address of principal executive offices) (Zip Code)

(415) 390-2337

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes x No o

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes x No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer x Non-accelerated filer o Accelerated filer o Smaller reporting company o Emerging growth company o

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. o

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No x

As of October 31, 2018, 79,190,697 shares of the registrant's Class A common stock and 19,666,380 shares of registrant's Class B common stock were outstanding.

Quarterly Report on Form 10-Q

For the Three Months Ended September 30, 2018

TABLE OF CONTENTS

	PART I — FINANCIAL INFORMATION	Page
Item 1.	Financial Statements (unaudited)	4
	Condensed Consolidated Balance Sheets as of September 30, 2018 and December 31, 2017	4
	Condensed Consolidated Statements of Operations for the Three and Nine Months Ended September 30, 2018 and 2017	5
	Condensed Consolidated Statements of Comprehensive Loss for the Three and Nine Months Ended September 30, 2018 and 2017	6
	Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2018 and 2017	7
	Notes to Condensed Consolidated Financial Statements	8
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	30
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	44
<u>Item 4.</u>	Controls and Procedures	44
	PART II - OTHER INFORMATION	
<u>Item 1.</u>	Legal Proceedings	46
Item 1A.	Risk Factors	47
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	75
<u>Item 3.</u>	Defaults Upon Senior Securities	75
<u>Item 4.</u>	Mine Safety Disclosures	75
<u>Item 5.</u>	Other Information	75
<u>Item 6.</u>	Exhibits	76
	Signatures	78
	1	

Special Note Regarding Forward-Looking Statements

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), which statements involve substantial risks and uncertainties. Forward-looking statements generally relate to future events or our future financial or operating performance. In some cases, you can identify forward-looking statements because they contain words such as "may," "will," "should," "expects," "plans," "anticipates," "could," "intends," "target," "projects," "contemplates," "believes," "estimates," "predicts," "potential" or "continue" or the negative of these words or other similar terms or expressions that concern our expectations, strategy, plans or intentions. Forward-looking statements contained in this Quarterly Report on Form 10-Q include, but are not limited to, statements about:

- our future financial performance, including our revenue, cost of revenue, gross margin and operating expenses, ability to generate positive cash flow and ability to achieve and sustain profitability;
- · the impact and expected results from changes in our relationship with our larger customers;
- · the sufficiency of our cash and cash equivalents to meet our liquidity needs;
- · anticipated technology trends, such as the use of and demand for cloud communications;
- · our ability to continue to build and maintain credibility with the global software developer community;
- · our ability to attract and retain customers to use our products;
- · our ability to attract and retain enterprises and international organizations as customers for our products;
- · our ability to form and expand partnerships with independent software vendors and system integrators;
- · the evolution of technology affecting our products and markets;
- · our ability to introduce new products and enhance existing products;
- · our ability to optimize our network service provider coverage and connectivity;
- · our ability to pass on our savings associated with our platform optimization efforts to our customers;
- · our ability to successfully enter into new markets and manage our international expansion;
- the attraction and retention of qualified employees and key personnel;
- · our ability to effectively manage our growth and future expenses and maintain our corporate culture;
- · our anticipated investments in sales and marketing and research and development;
- · our ability to maintain, protect and enhance our intellectual property;
- · our ability to successfully defend litigation brought against us;
- · our ability to service the interest on our convertible notes and repay such notes, to the extent required;
- our ability to comply with modified or new laws and regulations applying to our business, including GDPR, the California Consumer Privacy Act of 2018 and other privacy regulations that may be implemented in the future; and
- · our ability to successfully integrate and realize the benefits of our past or future strategic acquisitions or investments, including our proposed acquisition of SendGrid, Inc.

We caution you that the foregoing list may not contain all of the forward-looking statements made in this Quarterly Report on Form 10-Q.

You should not rely upon forward-looking statements as predictions of future events. We have based the forward-looking statements contained in this Quarterly Report on Form 10-Q primarily on our current expectations and projections about future events and trends that we believe may affect our business, financial condition, results of operations and prospects. The outcome of the events described in these forward-looking statements is subject to risks, uncertainties and other factors described in Part II, Item 1A, "Risk Factors" and elsewhere in this Quarterly Report on Form 10-Q. Moreover, we operate in a very competitive and rapidly changing environment. New risks and uncertainties emerge from time to time and it is not possible for us to predict all risks and uncertainties that could have an impact on the forward-looking statements contained in this Quarterly Report on Form 10-Q. We cannot assure you that the results, events and circumstances reflected in the forward-looking statements will be achieved or occur, and actual results, events or circumstances could differ materially from those described in the forward-looking statements.

The forward-looking statements made in this Quarterly Report on Form 10-Q relate only to events as of the date on which the statements are made. We undertake no obligation to update any forward-looking statements made in this Quarterly Report on Form 10-Q to reflect events or circumstances after the date of this Quarterly Report on Form 10-Q to reflect new information or the occurrence of unanticipated events, except as required by law. We may not actually achieve the plans, intentions or expectations disclosed in our forward-looking statements and you should not place undue reliance on our forward-looking statements. Our forward-looking statements do not reflect the potential impact of any future acquisitions, mergers, dispositions, joint ventures or investments we may make.

PART I - FINANCIAL INFORMATION

Item 1. Financial Statements

TWILIO INC.

Condensed Consolidated Balance Sheets

(In thousands)

(Unaudited)

	A	as of September 30, 2018	As of	f December 31, 2017
ASSETS				
Current assets:				
Cash and cash equivalents	\$	469,132	\$	115,286
Short-term marketable securities		276,221		175,587
Accounts receivable, net		80,159		43,113
Prepaid expenses and other current assets		25,237		19,279
Total current assets		850,749		353,265
Restricted cash		20,182		5,502
Property and equipment, net		59,205		50,541
Intangible assets, net		29,784		20,064
Goodwill		37,106		17,851
Other long-term assets		6,771		2,559
Total assets	\$	1,003,797	\$	449,782
LIABILITIES AND STOCKHOLDERS' EQUITY				
Current liabilities:				
Accounts payable	\$	12,842	\$	11,116
Accrued expenses and other current liabilities		93,894		53,614
Customer deposits		8,483		_
Deferred revenue		10,208		13,797
Total current liabilities		125,427		78,527
Convertible senior notes, net		428,778		_
Other long-term liabilities		15,239		11,409
Total liabilities		569,444		89,936
Commitments and contingencies (Note 11)				
Stockholders' equity:				
Preferred stock		_		_
Class A and Class B common stock		97		94
Additional paid-in capital		757,392		608,165
Accumulated other comprehensive income		1,380		2,025
Accumulated deficit		(324,516)		(250,438)
Total stockholders' equity		434,353		359,846
Total liabilities and stockholders' equity	\$	1,003,797	\$	449,782

See accompanying notes to condensed consolidated financial statements.

Condensed Consolidated Statements of Operations

(In thousands, except share and per share amounts)

(Unaudited)

	Three Months Ended September 30,					Nine Months Ended September 30,			
		2018		2017		2018	_	2017	
Revenue	\$	168,895	\$	100,542	\$	445,765	\$	283,784	
Cost of revenue		77,031		48,254		204,553		127,873	
Gross profit		91,864		52,288		241,212		155,911	
Operating expenses:									
Research and development		42,340		31,674		119,727		87,910	
Sales and marketing		45,949		25,778		116,520		73,047	
General and administrative		28,608		18,867		76,213		40,810	
Total operating expenses		116,897		76,319		312,460		201,767	
Loss from operations		(25,033)		(24,031)		(71,248)		(45,856)	
Other income (expenses), net		(1,939)		1,000		(3,172)		1,969	
Loss before provision for income taxes		(26,972)		(23,031)		(74,420)		(43,887)	
Provision for income taxes		(84)		(422)		(371)		(902)	
Net loss attributable to common stockholders	\$	(27,056)	\$	(23,453)	\$	(74,791)	\$	(44,789)	
Net loss per share attributable to common stockholders, basic and diluted	\$	(0.28)	\$	(0.25)	\$	(0.78)	\$	(0.49)	
Weighted-average shares used in computing net loss per share attributable to common									
stockholders, basic and diluted		98,019,629		92,156,768		96,359,437		90,543,087	

See accompanying notes to condensed consolidated financial statements.

Condensed Consolidated Statements of Comprehensive Loss

(In thousands)

(Unaudited)

	 Three Months Ended September 30,				Nine Months Ended September 30,				
	2018		2017		2018	_	2017		
Net loss	\$ (27,056)	\$	(23,453)	\$	(74,791)	\$	(44,789)		
Other comprehensive income (loss):									
Unrealized gain (loss) on marketable securities	291		(44)		126		(238)		
Foreign currency translation	(873)		793		(771)		2,274		
Total other comprehensive income (loss)	 (582)		749		(645)		2,036		
Comprehensive loss attributable to common stockholders	\$ (27,638)	\$	(22,704)	\$	(75,436)	\$	(42,753)		

See accompanying notes to condensed consolidated financial statements.

Condensed Consolidated Statements of Cash Flows

(In thousands)

(Unaudited)

		Nine Months Ended September 30,		
		2018	2017	
CASH FLOWS FROM OPERATING ACTIVITIES:			As Adjusted	
Net loss	\$	(74,791) \$	(44,789)	
Adjustments to reconcile net loss to net cash provided by (used in) operating activities:			,	
Depreciation and amortization		18,246	13,406	
Net amortization of investment premium and discount		(845)	153	
Amortization of debt issuance costs		657	-	
Accretion of debt discount on convertible senior notes		7,717	_	
Stock-based compensation		61,287	35,973	
Amortization of deferred commissions		885	_	
Provision for doubtful accounts		2,626	407	
Write-off of internally developed software and intangible assets		1,687	96	
Gain on lease termination		_	(295)	
Changes in operating assets and liabilities:				
Accounts receivable		(39,643)	(9,173)	
Prepaid expenses and other current assets		(6,600)	(4,947	
Other long-term assets		(3,681)	(1,512)	
Accounts payable		1,641	1,411	
Accrued expenses and other current liabilities		39,732	(1,454	
Customer deposits		8,482	_	
Deferred revenue		(3,390)	3,364	
Long-term liabilities		(1,177)	306	
Net cash provided by (used in) operating activities		12,833	(7,054)	
CASH FLOWS FROM INVESTING ACTIVITIES:				
Purchases of marketable securities		(213,533)	(280,569)	
Maturities of marketable securities		113,497	87,325	
Capitalized software development costs		(15,276)	(12,281)	
Purchases of property and equipment		(3,048)	(8,613)	
Purchases of intangible assets		(380)	(206	
Acquisitions, net of cash acquired		(29,662)	(22,621)	
Net cash used in investing activities		(148,402)	(236,965)	
CASH FLOWS FROM FINANCING ACTIVITIES:		(1.10,102)	(,,,,,,,,	
Proceeds from issuance of convertible senior notes		550,000	_	
Payment of debt issuance costs		(12,877)	_	
Purchase of capped call		(58,465)		
Payments of costs related to public offerings			(430)	
Proceeds from exercises of stock options		22,578	22,504	
Proceeds from shares issued under ESPP		4,474	7.404	
Value of equity awards withheld for tax liabilities		(1,720)	(476)	
Net cash provided by financing activities		503,990	29.002	
Effect of exchange rate changes on cash, cash equivalents and restricted cash		105	88	
NET INCREASE (DECREASE) IN CASH, CASH EQUIVALENTS AND RESTRICTED CASH		368,526	(214,929)	
CASH, CASH EQUIVALENTS AND RESTRICTED CASH—Beginning of period		120,788	314,280	
CASH, CASH EQUIVALENTS AND RESTRICTED CASH—End of period	\$	489,314 \$	99.351	
	*			
Cash paid for income taxes	\$	544 \$	489	
NON-CASH INVESTING AND FINANCING ACTIVITIES:				
Purchases of property, equipment and intangible assets, accrued but not paid	\$	785 \$	124	
Stock-based compensation capitalized in software development costs	\$	4,434 \$	2,712	
Acquisition holdback	\$	2,000 \$		
Debt offering costs, accrued but not paid	\$	103 \$		
Debt offering costs, accruca but hot paid	\$	103 \$		

See accompanying notes to condensed consolidated financial statements.

Notes to Condensed Consolidated Financial Statements

(Unaudited)

1. Organization and Description of Business

Twilio Inc. (the "Company") was incorporated in the state of Delaware on March 13, 2008. The Company is the leader in the Cloud Communications Platform category and enables developers to build, scale and operate real-time communications within their software applications via simple-to-use Application Programming Interfaces ("API"). The power, flexibility, and reliability offered by the Company's software building blocks empower entities of virtually every shape and size to build world-class engagement into their customer experience.

The Company's headquarters are located in San Francisco, California, and the Company has subsidiaries in the United States, the United Kingdom, Estonia, Ireland, Colombia, Germany, Hong Kong, Singapore, Bermuda, Spain, Sweden, Australia and Czech Republic.

2. Summary of Significant Accounting Policies

(a) Basis of Presentation

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with generally accepted accounting principles in the United States of America ("U.S. GAAP") and applicable rules and regulations of the Securities and Exchange Commission ("SEC") regarding interim financial reporting. Certain information and note disclosures normally included in the financial statements prepared in accordance with U.S. GAAP have been condensed or omitted pursuant to such rules and regulations. Therefore, these condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes included in our Annual Report on Form 10-K filed with the SEC on March 1, 2018 ("Annual Report").

The condensed consolidated balance sheet as of December 31, 2017, included herein, was derived from the audited financial statements as of that date, but may not include all disclosures including certain notes required by U.S. GAAP on an annual reporting basis.

In the opinion of management, the accompanying unaudited condensed consolidated financial statements reflect all normal recurring adjustments necessary to present fairly the financial position, results of operations, comprehensive loss and cash flows for the interim periods, but are not necessarily indicative of the results of operations to be anticipated for the full year 2018 or any future period.

(b) Principles of Consolidation

The condensed consolidated financial statements include the Company and its wholly owned subsidiaries. All significant intercompany balances and transactions have been eliminated.

(c) Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. These estimates are used for, but not limited to, revenue allowances and returns; recoverability of long-lived and intangible assets; capitalization and useful life of the Company's capitalized internal-use software development costs; fair value of acquired intangible assets and goodwill; accruals and contingencies. Estimates are based on historical experience and on various assumptions that the Company believes are reasonable under current circumstances. However, future events are subject to change and best estimates and judgments may require further adjustments; therefore, actual results could differ materially from those estimates. Management periodically evaluates such estimates and they are adjusted prospectively based upon such periodic evaluation.

(d) Concentration of Credit Risk

Financial instruments that potentially expose the Company to a concentration of credit risk consist primarily of cash, cash equivalents, marketable securities, restricted cash and accounts receivable. The Company maintains cash, cash equivalents, marketable securities and restricted cash with financial institutions that management believes are financially sound and have minimal credit risk exposure although the balances will exceed insured limits.

The Company sells its services to a wide variety of customers. If the financial condition or results of operations of any significant customers deteriorate substantially, operating results could be adversely affected. To reduce credit risk, management performs ongoing credit evaluations of the financial condition of significant customers. The Company does not require collateral from its credit customers and maintains reserves for estimated credit losses on customer accounts when considered necessary. Actual credit losses may differ from the Company's estimates. During the three and nine months ended September 30, 2018 and 2017, respectively, there was no customer organization that accounted for more than 10% of the Company's total revenue.

As of September 30, 2018 and December 31, 2017, no customer organization represented more than 10% of the Company's gross accounts receivable.

(e) Significant Accounting Policies

Effective January 1, 2018, the Company adopted Accounting Standards Codification ("ASC") 606, "*Revenue from Contracts with Customers*", which replaced the existing revenue recognition guidance, ASC 605, and outlines a single set of comprehensive principles for recognizing revenue under U.S. GAAP. Among other things, ASC 606 requires entities to assess the products or services promised in contracts with customers at contract inception to determine the appropriate unit at which to record revenue, which is referred to as a performance obligation. Revenue is recognized when control of the promised products or services is transferred to customers at an amount that reflects the consideration to which the entity expects to be entitled to in exchange for those products or services.

The Company adopted ASC 606 using the modified retrospective method with cumulative catch-up adjustment to the opening retained earnings as of January 1, 2018. Results for reporting periods beginning after December 31, 2017 are presented under ASC 606, while prior period amounts are not adjusted and continue to be reported in accordance with the Company's historical accounting policies prior to adoption. In adopting the standard, the Company elected to apply the new guidance only to those contracts which were not completed as of the date of the adoption.

The impact of adopting the new standard on the Company's consolidated financial statements was insignificant. The Company recorded a net cumulative catch-up adjustment to the beginning retained earnings as of January 1, 2018, of \$0.7 million.

The primary impact relates to the deferral of incremental commission costs of obtaining new contracts. Under ASC 605, the Company deferred only direct and incremental commission costs to obtain a contract and amortized those costs on a straight-line basis over the term of the related subscription contract. Under the new standard, the Company defers all incremental commission costs to obtain the contract and amortizes these costs on a straight-line basis over the expected term of benefit of the underlying asset, which was determined to be five years.

The impact on the Company's revenue recognition policies was insignificant. Prior to the adoption of ASC 606, the Company recognized the majority of its revenue according to the usage by its customers in the period in which that usage occurred. ASC 606 continuous transfer of control to the customer. The impact on the Company's balance sheet presentation includes presenting customer refundable prepayments as customer deposit liabilities, whereas under ASC 605 these were included in deferred revenues.

There was not a significant tax impact to the Company's consolidated statements of operations and consolidated balance sheet relating to the adoption of the new standard as there is a full valuation allowance due to the Company's history of continued losses.

Revenue Recognition

Revenue is recognized upon transfer of control of promised products or services to customers in an amount that reflects the consideration the Company expects to receive in exchange for those products or services. The Company enters into contracts that can include various combinations of products and services, which are generally capable of being distinct and accounted for as separate performance obligations. Revenue is recognized net of allowances for credits and any taxes collected from customers, which are subsequently remitted to governmental authorities.

The Company determines revenue recognition through the following steps:

- · Identification of the contract, or contracts, with a customer;
- Identification of the performance obligations in the contract;
- · Determination of the transaction price;
- Allocation of the transaction price to the performance obligations in the contract; and
- Recognition of revenue when, or as, the Company satisfies a performance obligation.

Nature of Products and Services

The Company's revenue is primarily derived from usage-based fees earned from customers accessing the Company's enterprise cloud computing services. Platform access is considered a monthly series comprising one performance obligation and usage-based fees are recognized as revenue in the period in which the usage occurs. In each of the three and nine months ended September 30, 2018, the revenue from usage-based fees represented 83% of total revenue. In the three and nine months ended September 30, 2017, the revenue from usage-based fees represented 82% and 83% of total revenue, respectively.

Subscription-based fees are derived from certain term-based contracts, such as with the sales of telephone numbers, short codes and customer support. Term-based contracts revenue is recognized on a ratable basis over the contractual term of the arrangement beginning on the date that the service is made available to the customer. In each of the three and nine months ended September 30, 2018, the revenue from term-based fees represented 17% of total revenue. In the three and nine months ended September 30, 2017, the revenue from term-based fees represented 18% and 17% of total revenue, respectively.

No significant judgments are required in determining whether products and services are considered distinct performance obligations and should be accounted for separately versus together, or to determine the stand-alone selling price ("SSP").

Refer to Note 10, Revenue by Geographic Area, for additional disaggregation.

The Company's arrangements do not contain general rights of return. However, credits may be issued on a case-by-case basis. The contracts do not provide customers with the right to take possession of the software supporting the applications. Amounts that have been invoiced are recorded in accounts receivable and in revenue or deferred revenue depending on whether the revenue recognition criteria have been met.

The reserve for sales credits is included in accounts receivable and is calculated based on historical trends and any specific risks identified in processing transactions. Changes in the reserve are recorded against revenue.

Deferred Revenue and Customer Deposits

Deferred revenue is recorded when cash payments are received in advance of future usage on non-cancellable contracts. Customer refundable prepayments are recorded as customer deposits. During the three and nine months ended September 30, 2018, the Company recognized \$3.9 million and \$11.1 million of revenue, respectively, that was included in the deferred revenue balance, as adjusted for ASC 606 on January 1, 2018.

Deferred Sales Commissions

The Company records an asset for the incremental costs of obtaining a contract with a customer, for example, sales commissions that are earned upon execution of contracts. The Company uses the portfolio method to recognize the amortization expense related to these capitalized costs related to initial contracts, upsells and renewals, and such expense is recognized over the estimated period of benefit of the capitalized costs of obtaining a contract with a customer 30, 2018 were \$7.7 million and are included in prepaid expenses and other current and long-term assets in the accompanying condensed consolidated balance sheet. Amortization of these assets was \$0.4 million and \$0.9 million in the three and nine months ended September 30, 2018, respectively, and is included in sales and marketing expense in the accompanying condensed consolidated statement of operations.

Other than adoption of ASC 606, there were no changes to our significant accounting policies as described in our Annual Report.

(f) Restricted Cash

In November 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") No. 2016-18, "*Statement of Cash Flows (Topic 230)* — *Restricted Cash*". This standard provides guidance on the presentation of restricted cash and restricted cash equivalents in the statement of cash flows. Restricted cash and restricted cash equivalents should be included with cash and cash equivalents when reconciling the beginning-of-period and end-of-period amounts shown on the statements of cash flows. The amendments of this ASU should be applied using a retrospective transition method and are effective for reporting periods beginning after December 15, 2017. The Company adopted ASU 2016-18 in the first quarter of 2018 and applied the guidance retrospectively to the prior period's condensed consolidated statement of cash flows with the following impact (in thousands):

	Nine Mon Septembe	
	 As Originally Reported	As Adjusted
Cash used in investing activities	\$ (235,795)	\$ (236,965)
Cash, cash equivalents and restricted cash — beginning of period	\$ 305,665	\$ 314,280
Cash, cash equivalents and restricted cash — end of period	\$ 91,906	\$ 99,351

Other than the revised statement of cash flows presentation of restricted cash, the adoption of ASU 2016-18 did not have an impact on the Company's financial position and results of operations.

(g) Recently Issued Accounting Guidance, Not yet Adopted

In August 2018, the FASB issued ASU 2018-15, "Intangibles-Goodwill and Other-Internal-Use Software (Subtopic 350-40): Customer's Accounting for Implementation Costs Incurred in a Cloud Computing Arrangement That is a Service Contract". This standard aligns the requirements for capitalizing implementation costs incurred in a hosting arrangement that is a service contract with the requirements for capitalizing implementation costs incurred to develop or obtain internal-use software. The standard is effective for interim and annual periods beginning after December 15, 2019, with early adoption permitted. The Company is currently evaluating the impact of the adoption of this standard on its consolidated financial statements.

In August 2018, the FASB issued ASU 2018-13, "Fair Value Measurement (Topic 820) Disclosure Framework-Changes to the Disclosure Requirements for Fair Value Measurement". The amendments under ASU 2018-13 remove, add and modify certain disclosure requirements on fair value measurements in ASC 820. The amendments are effective for interim and annual reporting periods beginning after December 15, 2019. The Company is currently evaluating the impact the new standard will have on its consolidated financial statements.

In July 2018, the FASB issued ASU 2018-09, "Codification Improvements", which does not prescribe any new accounting guidance, but instead makes minor improvements and clarifications of several different FASB ASC areas based on comments and suggestions made by various stakeholders. Certain updates are applicable immediately while others provide for a transition period to adopt as part of the next fiscal year beginning after December 15, 2018. The Company is currently evaluating this guidance to determine the impact it may have on its consolidated financial statements.

In June 2018, the FASB issued ASU 2018-07, "Improvements to Nonemployee Share-Based Payment Accounting", which aligns the measurement and classification for share-based payments to nonemployees with the accounting guidance for share-based payments to employees. Among other requirements, the measurement of non-employee awards will now be fixed at the grant date, rather than remeasured at every reporting date. This guidance is effective for interim and annual reporting periods beginning after December 15, 2018, with early application permitted. The Company will adopt this guidance upon its effective date. The Company does not expect the adoption of this guidance to have a material impact on the Company's consolidated financial statements.

In January 2017, the FASB issued ASU 2017-04, "Simplifying the Test for Goodwill Impairment", which removes the second step of the goodwill impairment test that requires a hypothetical purchase price allocation. A goodwill impairment will now be the amount by which a reporting unit's carrying value exceeds its fair value, not to exceed the carrying amount of goodwill. This guidance is effective prospectively for interim and annual reporting periods beginning after December 15, 2019. The Company will adopt this guidance upon its effective date. The Company does not expect the adoption of this guidance to have a material impact on the Company's consolidated financial statements.

In June 2016, the FASB issued ASU 2016-13, "Financial Instruments—Credit Losses: Measurement of Credit Losses on Financial Instruments", which changes the impairment model for most financial assets. The new model uses a forward-looking expected loss method, which will generally result in earlier recognition of allowances for losses. ASU 2016-13 is effective for annual and interim periods beginning after December 15, 2019 and early adoption is permitted for annual and interim periods beginning after December 15, 2018. The Company is evaluating the impact of this guidance on its consolidated financial statements.

In February 2016, the FASB issued ASU 2016-02, "*Leases*", which was further clarified by ASU 2018-10, "*Codification Improvements to Topic 842, Leases*", and ASU 2018-11, "*Leases - Targeted Improvements*", both issued in July 2018. ASU 2016-02 affects all entities that lease assets and will require lessees to recognize a lease liability and a right-of-use asset for all leases (except for short-term leases that have a duration of less than one year) as of the date on which the lessor makes the underlying asset available to the lessee. For lessors, accounting for leases is substantially the same as in prior periods. ASU 2016-02. The amendment affects narrow aspects of ASU 2016-02 related to the implicit rate in the lease, impairment of the net investment in the lease, lessee reassessment of lease classification, lessor reassessment of lease term and purchase options, variable payments that depend on an index or rate and certain transition adjustments. ASU 2018-11 adds a transition option for all entities can opt to continue to apply the legacy guidance in ASC 840, "*Leases*", including its disclosure requirements, in the comparative periods presented in the year the yaopt the new leases standard. Entities that elect this transition option will still be required to adopt the new leases standard using the modified retrospective transition method required by the standard, but they will recognize a cumulative-effect adjustment to the opening balance of retained earnings in the period of adoption rather than in the earliest period presented. The practical expedient provides lessors with an option to not separate the non-lease

components from the associated lease components when certain criteria are met and requires them to account for the combined component in accordance with the revenue recognition standard in ASC 606 if the associated non-lease components are the predominant components. The new standards are effective for fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. Early adoption is permitted. For leases existing at, or entered into after the beginning of the earliest comparative period presented in the financial statements, lesses and lessors must apply a modified retrospective transition approach. While the Company expects the adoption of these standards to result in an increase to the reported assets and liabilities, the Company has not yet determined the full impact that the adoption of this standard will have on its consolidated financial statements.

3. Fair Value Measurements

The following tables provide the financial assets measured at fair value on a recurring basis as of September 30, 2018 and December 31, 2017 (in thousands):

	Amortized Cost or Carrying Value		Gross Unrealized Gains		Gross Unrealized Losses Less Than 12 Months	1	Gross Unrealized Losses More Than 12 Months		F	ue Hierarchy as of ember 30, 2018 Level 2		Level 3		Aggregate Fair Value
Financial Assets:	 									 				
Cash and cash equivalents:														
Money market funds	\$ 432,040	\$		\$	—	\$		\$	432,040	\$ —	\$		\$	432,040
Commercial paper	 9,979		_		_		_		_	 9,979		_		9,979
Total included in cash and cash	 	_				_		_	<u> </u>		_	<u> </u>	_	
equivalents	 442,019		_		_		_		432,040	 9,979				442,019
Marketable securities:	 													
U.S. Treasury securities	59,858		—		(14)		(54)		59,790	—		—		59,790
Corporate debt securities and														
commercial paper	 216,835		_		(174)		(230)		_	216,431		_		216,431
Total marketable securities	 276,693		_		(188)		(284)		59,790	216,431		_		276,221
Total financial assets	\$ 718,712	\$	_	\$	(188)	\$	(284)	\$	491,830	\$ 226,410	\$	_	\$	718,240
_	Amortized Cost or Carrying Value		Gross Unrealized Gains		Gross Unrealized Losses Less Than 12 Months	1	Gross Unrealized Losses More Than 12 Months		F: Level 1	ue Hierarchy as of mber 31, 2017 Level 2		Level 3		Aggregate Fair Value
Financial Assets:	Cost or Carrying		Unrealized		Unrealized Losses Less Than	1	Unrealized Losses More Than			mber 31, 2017		Level 3	_	
Cash and cash equivalents:	Cost or Carrying Value		Unrealized	_	Unrealized Losses Less Than	1	Unrealized Losses More Than	_	Level 1	mber 31, 2017		Level 3		Fair Value
Cash and cash equivalents: Money market funds	Cost or Carrying	\$	Unrealized	\$	Unrealized Losses Less Than	1	Unrealized Losses More Than	\$		mber 31, 2017	\$	Level 3	\$	
Cash and cash equivalents: Money market funds Total included in cash and cash	Cost or Carrying Value 95,432	\$	Unrealized	\$	Unrealized Losses Less Than	1	Unrealized Losses More Than	\$	Level 1 95,432	mber 31, 2017	\$	Level 3	\$	Fair Value 95,432
Cash and cash equivalents: Money market funds Total included in cash and cash equivalents	Cost or Carrying Value	\$	Unrealized	\$	Unrealized Losses Less Than	1	Unrealized Losses More Than	\$	Level 1	mber 31, 2017	\$	Level 3	\$	Fair Value
Cash and cash equivalents: Money market funds Total included in cash and cash equivalents Marketable securities:	Cost or Carrying Value 95,432 95,432	\$	Unrealized	\$	Unrealized Losses Less Than 12 Months	1	Unrealized Losses More Than	\$	Level 1 95,432 95,432	mber 31, 2017	<u>\$</u>	Level 3	\$	Fair Value 95,432 95,432
Cash and cash equivalents: Money market funds Total included in cash and cash equivalents Marketable securities: U.S. Treasury securities	Cost or Carrying Value 95,432	<u>\$</u>	Unrealized	\$	Unrealized Losses Less Than	1	Unrealized Losses More Than	\$	Level 1 95,432	mber 31, 2017	<u>\$</u>	<u>Level 3</u>	<u>\$</u>	Fair Value 95,432
Cash and cash equivalents: Money market funds Total included in cash and cash equivalents Marketable securities: U.S. Treasury securities Corporate debt securities and	Cost or Carrying <u>Value</u> 95,432 95,432 59,962	\$	Unrealized	\$	Unrealized Losses Less Than 12 Months 	1	Unrealized Losses More Than	\$	Level 1 95,432 95,432	mber 31, 2017 Level 2 	<u>\$</u>	Level 3	\$	Fair Value 95,432 95,432 59,746
Cash and cash equivalents: Money market funds Total included in cash and cash equivalents Marketable securities: U.S. Treasury securities Corporate debt securities and commercial paper	Cost or Carrying <u>Value</u> 95,432 95,432 59,962 116,223	\$	Unrealized	\$	Unrealized Losses Less Than 12 Months — — — — (216) (382)	1	Unrealized Losses More Than	\$	Level 1 95,432 95,432 59,746	mber 31, 2017 Level 2 	\$	Level 3	\$	Fair Value 95,432 95,432 59,746 115,841
Cash and cash equivalents: Money market funds Total included in cash and cash equivalents Marketable securities: U.S. Treasury securities Corporate debt securities and	Cost or Carrying <u>Value</u> 95,432 95,432 59,962	\$	Unrealized	\$	Unrealized Losses Less Than 12 Months 	1	Unrealized Losses More Than	\$	Level 1 95,432 95,432	mber 31, 2017 Level 2 	\$	Level 3	\$	Fair Value 95,432 95,432 59,746

As of September 30, 2018, the fair value of the 0.25% convertible senior notes due 2023 (the "Notes"), as further described in Note 8 below, was approximately \$744.4 million. The fair value was determined based on the closing price for the Notes on the last trading day of the reporting period and is considered as Level 2 in the fair value hierarchy.

As the Company views its marketable securities as available to support current operations, it has classified all available for sale securities as short-term. As of September 30, 2018, for fixed income securities that were in unrealized loss positions, the Company has determined that (i) it does not have the intent to sell any of these investments, and (ii) it is not more likely than not that it will be required to sell any of these investments before recovery of the entire amortized cost basis. In addition, as of September 30,

2018, the Company anticipates that it will recover the entire amortized cost basis of such fixed income securities and has determined that no other-than-temporary impairments associated with credit losses were required to be recognized during the three and nine month ended September 30, 2018. Interest earned on marketable securities was \$0.8 million and \$2.2 million in the three and nine months ended September 30, 2017, respectively. The interest is recorded as other income (expense), net, in the accompanying condensed consolidated statements of operations.

The following table summarizes the contractual maturities of marketable securities as of September 30, 2018 and December 31, 2017 (in thousands):

		As of September 30, 2018				As of December 31, 2017			
	Amortized Cost				Amortized Cost		Aggregate Fair Value		
Financial Assets:	_								
Less than one year	\$	270,692	\$	270,276	\$	108,584	\$	108,360	
One to two years		6,001		5,945		67,601		67,227	
Total	\$	276,693	\$	276,221	\$	176,185	\$	175,587	

4. Property and Equipment

Property and equipment consisted of the following (in thousands):

	As Septem 20	ber 30,	As of December 31, 2017
Capitalized internal-use software development costs	\$	67,287	\$ 49,177
Leasehold improvements		14,600	14,246
Office equipment		12,117	9,652
Furniture and fixtures		2,310	1,976
Software		2,430	1,675
Total property and equipment		98,744	 76,726
Less: accumulated depreciation and amortization		(39,539)	(26,185)
Total property and equipment, net	\$	59,205	\$ 50,541

Depreciation and amortization expense was \$5.0 million and \$13.5 million for the three and nine months ended September 30, 2018, respectively, and \$3.4 million and \$9.3 million for the three and nine months ended September 30, 2017, respectively.

The Company capitalized \$6.7 million and \$19.8 million in internal-use software development costs in the three and nine months ended September 30, 2018, respectively, and \$5.5 million and \$15.0 million in the three and nine months ended September 30, 2017, respectively. Of this amount, stock-based compensation expense was \$1.5 million and \$4.4 million in the three and nine months ended September 30, 2018, respectively, and \$1.2 million and \$2.8 million in the three and nine months ended September 30, 2017, respectively.

Amortization of capitalized software development costs was \$3.6 million and \$9.3 million in the three and nine months ended September 30, 2018, respectively, and \$2.2 million and \$5.9 million in the three and nine months ended September 30, 2017, respectively.

5. Business Combinations

Fiscal 2018 Acquisitions

Ytica.com a.s.

In September 2018, the Company acquired all outstanding shares of Ytica.com a.s. ("Ytica"), a developer and provider of a contact center reporting and analytics based in the Czech Republic, for a total purchase price of \$21.8 million, paid in cash, of which \$3.2 million was held in escrow with a term of 18 months.

Additionally, the Company granted 47,574 restricted stock units of the Company's Class A common stock to a former shareholder of Ytica that had a value of \$3.6 million and is subject to vesting over a period of three years. The Company is recording stock-based compensation expense as the shares are vesting.

The acquisition was accounted for as a business combination and the total purchase price was allocated to the net tangible and intangible assets and liabilities based on their fair values on the acquisition date and the excess was recorded as goodwill. The acquired entity's results of operations have been included in the condensed consolidated financial statements of the Company from the date of acquisition.

The following table presents the preliminary purchase price allocation recorded in the Company's condensed consolidated balance sheet as of September 30, 2018 (in thousands):

	Total
Net tangible assets	\$ (1,036)
Intangible assets ⁽¹⁾	9,920
Goodwill ⁽²⁾	12,873
Total purchase price	\$ 21,757

⁽¹⁾ Identifiable finite-lived intangible assets were comprised of the following (in thousands):

		Total	Estimated life (in years)
Developed technology	\$	9,090	(iff years)
Customer relationships	ψ	830	2
Total intangible assets acquired	\$	9.920	-
		- 3- = 0	

(2) The goodwill is primarily attributable to the future cash flows to be realized from the acquired technology platform as well as operational synergies. The Company intends to file elections that make the goodwill deductible for U.S. tax purposes.

The Company acquired a net deferred tax liability of \$1.9 million in this business combination that is included in long-term liabilities in the accompanying condensed consolidated balance sheet.

The estimated fair value of the intangible assets acquired was determined by the Company, and the Company considered or relied in part upon a valuation report of a third-party expert. The Company used an income approach to estimate the fair values of the identifiable intangible assets.

The Company incurred cost related to this acquisition of \$0.5 million that were expensed as incurred and recorded in general and administrative expenses in the accompanying condensed consolidated statement of operation.

Pro forma results of operations for this acquisition are not presented as the financial impact to the Company's condensed consolidated financial statements is immaterial.

Core Network Dynamics GmbH

In August 2018, the Company acquired all outstanding shares of Core Network Dynamics GmbH ("CND"), a developer and provider of a complete software mobile network infrastructure based in Germany, for a total purchase price of \$11.1 million, paid in cash, of which \$2.0 million was withheld by the Company for a term of 18 months.

Additionally, the Company granted 35,950 restricted stock units of the Company's Class A common stock to a former shareholder of CND that had a value of \$2.2 million and is subject to vesting over a period of three years. The Company is recording a stock-based compensation expense as the shares are vesting.

The acquisition was accounted for as a business combination and the total purchase price was allocated to the net tangible and intangible assets and liabilities based on their fair values on the acquisition date and the excess was recorded as goodwill. The acquired entity's results of operations have been included in the condensed consolidated financial statements of the Company from the date of acquisition.

The following table presents the preliminary purchase price allocation recorded in the Company's condensed consolidated balance sheet as of September 30, 2018 (in thousands):

	Total
Net tangible liabilities	\$ (244)
Intangible assets (1)	4,500
Goodwill ⁽²⁾	6,800
Total purchase price	\$ 11,056

⁽¹⁾ Identifiable finite-lived intangible assets were comprised of the following (in thousands):

		Estimated life
	Total	(in years)
Developed technology	\$ 3,910	4
Customer relationships	590	0.5
Total intangible assets acquired	\$ 4,500	

(2) The goodwill is primarily attributable to the future cash flows to be realized from the operating synergies between the acquired technology platform and the Company's Programmable Wireless products. The Company intends to file elections that make the goodwill deductible for U.S. tax purposes.

The Company acquired a net deferred tax liability of \$1.2 million in this business combination that is included in long-term liabilities in the accompanying condensed consolidated balance sheet.

The estimated fair value of the intangible assets acquired was determined by the Company, and the Company considered or relied in part upon a valuation report of a third-party expert. The Company used a replacement cost approach to estimate the fair values of the identifiable intangible assets.

The Company incurred cost related to this acquisition of \$0.7 million that were expensed as incurred and have been recorded in general and administrative expenses in the accompanying condensed consolidated statement of operation.

Pro forma results of operations for this acquisition are not presented as the financial impact to the Company's consolidated financial statements is immaterial.

Fiscal 2017 Acquisitions

Beepsend AB

In February 2017, the Company completed its acquisition of Beepsend AB, a messaging provider based in Sweden, specializing in messaging and SMS solutions, for a total purchase price of \$23.0 million, paid in cash, of which \$5.0 million was held in escrow with a term of 18 months, which was fully released at the escrow due date.

Additionally, the Company deposited \$2.0 million into a separate escrow that was subject to future service conditions and was recorded ratably into the compensation expense as the services were rendered. As of September 30, 2018, the remaining balance in the escrow was insignificant.

The acquisition was accounted for as a business combination and the total purchase price was allocated to the net tangible and intangible assets and liabilities based on their fair values on the acquisition date and the excess was recorded as goodwill. The acquired entity's results of operations have been included in the consolidated financial statements of the Company from the date of acquisition.

The following table presents the purchase price allocation, as adjusted, recorded in the Company's condensed consolidated balance sheet as of March 31, 2017 (in thousands):

	Total
Net tangible liabilities	\$ (3,575)
Intangible assets ⁽¹⁾	13,700
Goodwill ⁽²⁾	12,837
Total purchase price	\$ 22,962

⁽¹⁾ Identifiable finite-lived intangible assets were comprised of the following (in thousands):

	Fotal	Estimated life (in years)
Developed technology	\$ 5,000	4
Customer relationships	6,100	7 - 8
Supplier relationships	2,600	5
Total intangible assets acquired	\$ 13,700	

(2) The goodwill is primarily attributable to the future cash flows to be realized from the acquired technology platform, existing customer and supplier relationships as well as operational synergies. Goodwill is deductible for tax purposes.

The Company acquired a net deferred tax liability of \$2.6 million in this business combination that is included in long-term liabilities in the accompanying condensed consolidated balance sheets.

The estimated fair value of the intangible assets acquired was determined by the Company, and the Company considered or relied in part upon a valuation report of a third-party expert. The Company used the income approach to estimate the fair values of the identifiable intangible assets.

The Company incurred costs related to this acquisition of \$0.7 million, of which \$0.3 million and \$0.4 million were incurred during fiscal years 2017 and 2016, respectively. All acquisition-related costs were expensed as incurred and have been recorded in general and administrative expenses in the accompanying condensed consolidated statements of operations.

Pro forma results of operations for this acquisition are not presented as the financial impact to the Company's consolidated financial statements is immaterial.

6. Goodwill and Intangible Assets

Goodwill

Goodwill balance as of September 30, 2018 and December 31, 2017 was as follows (in thousands):

	Total
Balance as of December 31, 2017	\$ 17,851
Goodwill additions related to 2018 acquisitions	19,673
Effect of exchange rate	(418)
Balance as of September 30, 2018	\$ 37,106

Intangible assets

Intangible assets consisted of the following (in thousands):

	 As of September 30, 2018 Accumulated				
	 Gross		Amortization		Net
Amortizable intangible assets:					
Developed technology	\$ 27,679	\$	(8,689)	\$	18,990
Customer relationships	8,245		(1,800)		6,445
Supplier relationships	2,738		(850)		1,888
Trade name	60		(60)		_
Patent	2,325		(159)		2,166
Total amortizable intangible assets	 41,047		(11,558)		29,489
Non-amortizable intangible assets:	 32				32
Domain names					
Trademarks	263		_		263
Total	\$ 41,342	\$	(11,558)	\$	29,784
	17				

	 As of December 31, 2017				
	 Gross		Accumulated Amortization		Net
Amortizable intangible assets:	 				
Developed technology	\$ 14,941	\$	(5,476)	\$	9,465
Customer relationships	7,159		(1,006)		6,153
Supplier relationships	2,881		(500)		2,381
Trade name	60		(60)		—
Patent	1,878		(108)		1,770
Total amortizable intangible assets	 26,919		(7,150)		19,769
Non-amortizable intangible assets:					
Domain names	32		_		32
Trademarks	263		_		263
Total	\$ 27,214	\$	(7,150)	\$	20,064

Amortization expense was \$1.9 million and \$4.7 million for the three and nine months ended September 30, 2018, respectively, and \$1.5 million and \$4.2 million for the three and nine months ended September 30, 2017, respectively

Total estimated future amortization expense was as follows (in thousands):

	As of September 30, 2018
2018 (remaining three months)	\$ 3,272
2019	8,704
2020	6,007
2021	4,592
2022	3,270
Thereafter	3,644
Total	\$ 29,489

7. Accrued Expenses and Other Liabilities

Accrued expenses and other current liabilities consisted of the following (in thousands):

	Septem	As of September 30, 2018		As of cember 31, 2017
Accrued payroll and related	\$	11,423	\$	4,898
Accrued bonus and commission		8,170		4,777
Accrued cost of revenue		23,023		10,876
Sales and other taxes payable		22,746		20,877
ESPP contributions		4,536		1,338
Deferred rent		1,334		1,048
Accrued other expense		22,662		9,800
Total accrued expenses and other current liabilities	\$	93,894	\$	53,614

Other long-term liabilities consisted of the following (in thousands):

	Septer	s of <u>nber 30,</u> 018	As of December 31, 2017		
Deferred rent	\$	7,557	\$	8,480	
Deferred tax liability, net		4,852		2,452	
Acquisition Holdback		2,000		_	
Accrued other expenses		830		477	
Total other long-term liabilities	\$	15,239	\$	11,409	

8. Convertible Senior Notes and Capped Call Transactions

In May 2018, the Company issued \$550.0 million aggregate principal amount of 0.25% convertible senior notes due 2023 in a private placement, including \$75.0 million aggregate principal amount of such Notes pursuant to the exercise in full of the over-allotment options of the initial purchasers (collectively, the "Notes"). The interest on the Notes is payable semi-annually in arrears on June 1 and December 1 of each year, beginning on December 1, 2018.

The Notes may bear special interest under specified circumstances relating to the Company's failure to comply with its reporting obligations under the indenture relating to the issuance of Notes (the "indenture") or if the Notes are not freely tradeable as required by the indenture. The Notes will mature on June 1, 2023, unless earlier repurchased or redeemed by the Company or converted pursuant to their terms. The total net proceeds from the debt offering, after deducting initial purchaser discounts and debt issuance costs, paid or payable by us, were approximately \$537.0 million.

Each \$1,000 principal amount of the Notes is initially convertible into 14.1040 shares of the Company's Class A common stock par value \$0.001, which is equivalent to an initial conversion price of approximately \$70.90 per share. The conversion rate is subject to adjustment upon the occurrence of certain specified events but will not be adjusted for any accrued and unpaid special interest. In addition, upon the occurrence of a make-whole fundamental change, as defined in the indenture, the Company will, in certain circumstances, increase the conversion rate by a number of additional shares for a holder that elects to convert its Notes in connection with such make-whole fundamental change or during the relevant redemption period.

Prior to the close of business on the business day immediately preceding March 1, 2023, the Notes may be convertible at the option of the holders only under the following circumstances:

(1) during any calendar quarter commencing after September 30, 2018, and only during such calendar quarter, if the last reported sale price of the Class A common stock for at least 20 trading days (whether or not consecutive) in a period of 30 consecutive trading days ending on, and including, the last trading day of the immediately preceding calendar quarter is more than or equal to 130% of the conversion price on each applicable trading day;

(2) during the five business days period after any five consecutive trading day period in which, for each trading day of that period, the trading price per \$1,000 principal amount of Notes for such trading day was less than 98% of the product of the last reported sale price of the Class A common stock and the conversion rate on each such trading day;

(3) upon the Company's notice that it is redeeming any or all of the Notes; or

(4) upon the occurrence of specified corporate events.

On or after March 1, 2023, until the close of business on the second scheduled trading day immediately preceding the maturity date, holders of the Notes may, at their option, convert all or a portion of their Notes regardless of the foregoing conditions.

Upon conversion, the Company will pay or deliver, as the case may be, cash, shares of Class A common stock, or a combination of cash and shares of Class A Common Stock, at the Company's election. It is the Company's current intent to settle the principal amount of the Notes with cash.

During the three months ended September 30, 2018, the conditions allowing holders of the Notes to convert were not met. The Company may redeem the Notes, in whole or in part, at its option, on or after June 1, 2021 but before the 35th scheduled trading day before the maturity date, at a cash redemption price equal to 100% of the principal amount of the Notes to be redeemed, plus accrued and unpaid interest, if any, if the last reported sale price of the Class A Common Stock has been at least 130% of the conversion price then in effect for at least 20 trading days (whether or not consecutive) during any 30 consecutive trading days ending on, and including, the trading day immediately before the date the redemption notices were sent; and the trading day immediately before such notices were sent.

No sinking fund is provided for the Notes. Upon the occurrence of a fundamental change (as defined in the indenture) prior to the maturity date, holders may require the Company to repurchase all or a portion of the Notes for cash at a price equal to 100% of the principal amount of the Notes to be repurchased, plus any accrued and unpaid interest to, but excluding, the fundamental change repurchase date.

The Notes are senior unsecured obligations and will rank senior in right of payment to any of the Company's indebtedness that is expressly subordinated in right of payment to the Notes; equal in right of payment with the Company's existing and future liabilities that are not so subordinated; effectively subordinated to any of the Company's secured indebtedness to the extent of the value of the assets securing such indebtedness; and structurally junior to all indebtedness and other liabilities (including trade payables) of current or future subsidiaries of the Company.

The foregoing description is qualified in its entirety by reference to the text of the indenture and the form of 0.25% convertible senior notes due 2023, which were filed as exhibits to our Quarterly Report on Form 10-Q for the quarter ended June 30, 2018 and are incorporated herein by reference.

In accounting for the issuance of the Notes, the Company separated the Notes into liability and equity components. The carrying amount of the liability component was calculated by measuring the fair value of a similar debt instrument that does not have an associated convertible feature. The carrying amount of the equity component representing the conversion option was \$119.4 million and was determined by deducting the fair value of the liability component from the par value of the Notes. The equity component is not remeasured as long as it continues to meet the conditions for equity classification. The excess of the principal amount of the liability component over its carrying amount, or the debt discount, is amortized to interest expense at an effective interest rate over the contractual terms of the Notes.

In accounting for the transaction costs related to the Notes, the Company allocated the total amount incurred to the liability and equity components of the Notes based on the proportion of the proceeds allocated to the debt and equity components. Issuance costs attributable to the liability component were approximately \$10.2 million, were recorded as an additional debt discount and are amortized to interest expense using the effective interest method over the contractual terms of the Notes. Issuance costs attributable to the equity component were netted with the equity component in stockholders' equity.

The net carrying amount of the liability component of the Notes was as follows (in thousands).

	As of September 30, 2018	
Principal	\$ 550,00	0
Unamortized discount	(111,71	7)
Unamortized issuance costs	(9,50	5)
Net carrying amount	\$ 428,77	8

The net carrying amount of the equity component of the Notes was as follows (in thousands):

	\$ As of September 30, 2018
Proceeds allocated to the conversion options (debt discount)	\$ 119,435
Issuance costs	(2,819)
Net carrying amount	\$ 116,616

The following table sets forth the interest expense recognized related to the Notes (in thousands):

	Three Months Ended September 30, 2018		Nine Months Ended September 3(2018),
Contractual interest expense	\$	344	\$	508
Amortization of debt issuance costs		446		657
Amortization of debt discount		5,233		7,717
Total interest expense related to the Notes	\$	6,023	\$	8,882

In connection with the offering of the Notes, the Company entered into privately-negotiated capped call transactions with certain counterparties (the "capped calls"). The capped calls each have an initial strike price of approximately \$70.90 per share, subject to certain adjustments, which corresponds to the initial conversion price of the Notes. The capped calls have initial cap prices of \$105.04 per share, subject to certain adjustments, approximately 7,757,200 shares of Class A Common Stock. The capped calls are generally intended to reduce or offset the potential dilution to the Class A Common Stock upon any conversion of the Notes with such reduction or offset, as the case may be, subject to a cap based on the cap price. The capped calls expire on the earlier of (i) the last day on which any convertible securities remain outstanding and (ii) June 1, 2023, subject to earlier exercise. The capped calls are subject to either adjustment or termination upon the occurrence of specified extraordinary events affecting the Company, including a merger event, a tender offer, and a nationalization, insolvency or delisting involving the Company. In addition, the capped calls are subject to certain specified additional disruption events that may give rise to a termination of the capped calls, including changes in law, insolvency filings, and hedging disruptions. The capped call transactions are recorded in stockholders' equity and are not accounted for as derivatives. The net cost of \$58.5 million incurred to purchase the capped call transactions was recorded as a reduction to additional paid-in capital in the accompanying condensed consolidated balance sheet.

9. Supplemental Balance Sheet Information

A roll-forward of the Company's reserves is as follows (in thousands):

(a) Allowance for doubtful accounts:

	_	Three Months Ended September 30,				Nine Months Ended September 30,				
		2018		2017		2018		2017		
Balance, beginning of period	\$	2,536	\$	923	\$	1,033	\$	1,076		
Additions		1,111		125		2,626		407		
Write-offs		(5)		—		(17)		(435)		
Balance, end of period	\$	3,642	\$	1,048	\$	3,642	\$	1,048		

b) Sales credit reserve:

	Three Months En	ded Septeml	ber 30,		hs Ended ber 30,			
	 2018		2017		2018		2017	
Balance, beginning of period	\$ 2,625	\$	734	\$	1,761	\$	544	
Additions	1,651		104		4,301		1,076	
Deductions against reserve	(773)		(238)		(2,559)		(1,020)	
Balance, end of period	\$ 3,503	\$	600	\$	3,503	\$	600	

10. Revenue by Geographic Area

Revenue by geographic area is based on the IP address at the time of registration. The following table sets forth revenue by geographic area (dollars in thousands):

		nths Ended 1ber 30,		 Nine Months Ended September 30,			
	 2018		2017	2018		2017	
Revenue by geographic area:	 						
United States	\$ 125,697	\$	76,713	\$ 335,575	\$	221,914	
International	43,198		23,829	110,190		61,870	
Total	\$ 168,895	\$	100,542	\$ 445,765	\$	283,784	
Percentage of revenue by geographic area:							
United States	74%		76%	75%		78%	
International	26%		24%	25%		22%	

Long-lived assets outside the United States were not significant.

11. Commitments and Contingencies

(a) Lease Commitments

The Company entered into various non-cancelable operating lease agreements for its facilities that expire over the next six years. Certain operating leases contain provisions under which monthly rent escalates over time. When lease agreements contain escalating rent clauses or free rent periods, the Company recognizes rent expense on a straight-line basis over the term of the lease.

In September 2018, the Company entered into a sub-lease agreement ("Sub-lease") for a total of 259,416 rentable square feet of office space at 101 Spear Street in San Francisco, California. The Sub-lease covers several floors for which the terms commence on December 1, 2018 and April 1, 2020, and will be expiring at various dates between March 2025 and June 2028. The lease payments will range from \$0.8 million per months to \$2.2 million per month. The Company secured its lease obligation with a \$14.7 million letter of credit, which it designated as restricted cash on its balance sheet as of September 30, 2018. The Company intends for this location to be its headquarters at some time in 2019.

Rent expense was \$2.2 million and \$6.4 million for the three and nine months ended September 30, 2018, respectively, and \$2.1 million and \$6.1 million for the three and nine months ended September 30, 2017, respectively.

Future minimum lease payments under non-cancelable operating leases were as follows (in thousands):

Year Ending December 31:	As of September 30, 2018
2018 (remaining three months)	\$ 3,446
2019	21,195
2020	27,798
2021	30,209
2022	29,735
Thereafter	113,211
Total minimum lease payments	\$ 225,594

Additionally, in the three and nine months ended September 30, 2018, the Company entered into several non-cancellable vendor agreements with a term from one to two years for total purchase commitments of \$2.2 million and \$10.0 million, respectively.

(b) Legal Matters

On April 30, 2015, Telesign Corporation, or Telesign, filed a lawsuit against the Company in the United States District Court, Central District of California ("Telesign I"). Telesign alleges that the Company is infringing three U.S. patents that it holds: U.S. Patent No. 8,462,920 ("920"), U.S. Patent No. 8,687,038 ("038") and U.S. Patent No. 7,945,034 ("034"). The patent infringement allegations in the lawsuit relate to the Company's Account Security products, its two-factor authentication use case and an API tool to find information about a phone number.

On March 28, 2016, Telesign filed a second lawsuit against the Company in the United States District Court, Central District of California ("Telesign II"), alleging infringement of U.S. Patent No. 9,300,792 ('792") held by Telesign. The '792 patent is in the same patent family as the '920 and '038 patents asserted in Telesign I. On March 8, 2017, in response to a petition by the Company, the U.S. Patent and Trademark Office ("PTO") issued an order instituting the *inter partes* review for the '792 patent. On March 6, 2018, the PTO found all claims challenged by Twilio in the *inter partes* review unpatentable. On March 15, 2017, Twilio filed a motion to consolidate and stay related cases pending the conclusion of the '792 patent *inter partes* review, which the court granted. The Central District of California court lifted the stay on April 13, 2018. The court transferred the cases to the United States District Court, Northern District of California. With respect to each of the patents asserted in the now-consolidated Telesign I and Telesign II cases ("Telesign Juff"), the consolidated complaint seeks, among other things, to enjoin the Company from allegedly infringing the patents, along with damages for lost profits. On August 14, 2018, the Company filed a motion seeking a judgment that all asserted claims are invalid under 35 U.S.C. 101. On October 19, 2018, the court granted the Company's motion and entered judgment in the Company's favor. On November 8, 2018, Telesign filed a notice of appeal. Based on final judgment being entered by the district court in favor of the Company, the Company does not believe a loss is reasonably possible or estimable.

On December 1, 2016, the Company filed a patent infringement lawsuit against Telesign in the United States District Court, Northern District of California ("Telesign III"), alleging indirect infringement of United States Patent No. 8,306,021 ("021"), United States Patent No. 8,837,465 ("465"), United States Patent No. 8,755,376 ("376"), United States Patent No. 8,736,051 ("051"), United States Patent No. 8,737,962 ("962"), United States Patent No. 9,270,833 ("833"), and United States Patent No. 9,226,217 ("217"). Telesign filed a motion to dismiss the complaint on January 25, 2017. In two orders, issued on March 31, 2017 and April 17, 2017, the Court granted Telesign's motion to dismiss with respect to the '962, '833, '051 and '217 patents, but denied Telesign's motion to dismiss as to the '021, '465 and '376 patents. On August 23, 2017, Telesign relations for *inter partes* review of the '021, '465, and '376 patents. The Court patent and granted Telesign's petitions for *inter partes* review of the '465 and '376 patents. Telesign III is currently stayed pending resolution of the *inter partes* reviews of the '465 and '376 patents. The Company is seeking a judgment of infringement, a judgment of willful infringement, and injunctive relief, enhanced damages, and an award of costs and expenses against Telesign.

On February 18, 2016, a putative class action complaint was filed in the Alameda County Superior Court in California, entitled Angela Flowers v. Twilio Inc. The complaint alleges that the Company's products permit the interception, recording and disclosure of communications at a customer's request and are in violation of the California Invasion of Privacy Act. The complaint seeks injunctive relief as well as monetary damages. On May 27, 2016, the Company filed a demurrer to the complaint. On August 2, 2016, the court issued an order denying the demurrer in part and granting it in part, with leave to amend by August 18, 2016 to address any claims under California's Unfair Competition Law. The plaintiff opted not to amend the complaint. Following a period of discovery, the plaintiff filed a motion for class certification on September 20, 2017. On January 2, 2018, the court issued an order

granting in part and denying in part the plaintiff's class certification motion. The court certified two classes of individuals who, during specified time periods, allegedly sent or received certain communications involving the accounts of three of the Company's customers that were recorded. The parties held a mediation on August 29, 2018. Following further discussions in coordination with the mediator, on September 27, 2018, the parties reached an agreement in principle to settle the case. The parties are preparing a long-form settlement agreement, and a preliminary class settlement approval hearing has been scheduled for January 15, 2019. The Company currently estimates its potential liability in the Flowers matter to be \$1.5 million and has reserved this amount in the accompanying condensed consolidated balance sheet as of September 30, 2018.

The Company intends to vigorously defend itself against these lawsuits and believes it has meritorious defenses to each matter in which it is a defendant.

In addition to the litigation matters discussed above, from time to time, the Company is a party to legal action and subject to claims that arise in the ordinary course of business. The claims are investigated as they arise and loss estimates are accrued, when probable and reasonably estimable. While it is not feasible to predict or determine the ultimate outcome of these matters, the Company believes that these legal proceedings will not have a material adverse effect on its financial position or results of operations.

Legal fees and other costs related to litigation and other legal proceedings are expensed as incurred and are included in general and administrative expenses in the accompanying condensed consolidated statements of operations.

(c) Indemnification Agreements

The Company has signed indemnification agreements with all of its board members and executive officers. The agreements indemnify the board members and executive officers from claims and expenses on actions brought against the individuals separately or jointly with the Company for certain indemnifiable events. Indemnifiable events generally mean any event or occurrence related to the fact that the board member or the executive officer was or is acting in his or her capacity as a board member or an executive officer for the Company or was or is acting or representing the interests of the Company.

In the ordinary course of business, the Company enters into contractual arrangements under which it agrees to provide indemnification of varying scope and terms to business partners and other parties with respect to certain matters, including, but not limited to, losses arising out of the breach of such agreements, intellectual property infringement claims made by third parties and other liabilities relating to or arising from the Company's various products, or its acts or omissions. In these circumstances, payment may be conditional on the other party making a claim pursuant to the procedures specified in the particular contract. Further, the Company's obligations under these agreements may be limited in terms of time and/or amount, and in some instances, the Company may have recourse against third parties for certain payments. The terms of such obligations may vary.

As of September 30, 2018 and December 31, 2017, no amounts were accrued.

(d) Other Taxes

The Company conducts operations in many tax jurisdictions throughout the United States. In many of these jurisdictions, non-income-based taxes, such as sales and use and telecommunications taxes are assessed on the Company's operations. Prior to March 2017, the Company had not billed nor collected these taxes from its customers and, in accordance with U.S. GAAP, recorded a provision for its tax exposure in these jurisdictions when it was both probable that a liability had been incurred and the amount of the exposure could be reasonably estimated. These estimates included several key assumptions including, but not limited to, the taxability of the Company's services, the jurisdictions in which its management believed it has nexus, and the sourcing of revenues to those jurisdictions. Starting in March 2017, the Company began collecting these taxes from customers in certain jurisdictions, and since then has expanded the number of jurisdictions where these taxes are being collected. Effective January 2018, the Company is no longer recording a provision for its exposure in that jurisdiction. The Company expects to continue to expand the number of jurisdictions where these taxes will be collected in the future. Simultaneously, the Company was and continues to be in discussions with certain states regarding its prior state sales and other taxes, if any, that the Company may owe.

During 2017, the Company revised its estimates of its tax exposure based on settlements reached with various states indicating that certain revisions to the key assumptions including, but not limited to, the sourcing of revenue and the taxability of the Company's services were appropriate. In the year ended December 31, 2017, the total impact of these changes on the net loss attributable to common stockholders was a reduction of \$13.4 million. As of September 30, 2018 and December 31, 2017, the liability recorded for these taxes was \$22.7 million and \$20.9 million, respectively.

On June 21, 2018, the U.S. Supreme Court issued an opinion in *South Dakota v. Wayfair* permitting a state to require a seller to collect sales tax even if the seller has no physical presence in that state. This opinion reversed a prior U.S. Supreme Court opinion requiring a physical presence by the seller. The Company has evaluated the states currently requiring an out of state seller to collect sales tax and have identified the states that impose sales tax on its applicable products.

In the event other jurisdictions challenge management's assumptions and analysis, the actual exposure could differ materially from the current estimates.

12. Stockholders' Equity

(a) Preferred Stock

As of September 30, 2018 and December 31, 2017, the Company had authorized 100,000,000 shares of preferred stock, par value \$0.001, of which no shares were issued and outstanding.

(b) Common Stock

As of September 30, 2018 and December 31, 2017, the Company had authorized 1,000,000,000 shares of Class A common stock and 100,000,000 shares of Class B common stock, each par value \$0.001 per share. As of September 30, 2018, 78,999,646 shares of Class A common stock and 19,727,396 shares of Class B common stock were issued and outstanding. As of December 31, 2017, 69,906,550 shares of Class A common stock and 24,063,246 shares of Class B common stock were issued and outstanding.

. .

. .

The Company had reserved shares of common stock for issuance as follows:

	As of September 30, 2018	As of December 31, 2017
Stock options issued and outstanding	8,567,152	10,710,427
Nonvested restricted stock units issued and outstanding	7,368,918	5,665,459
Class A common stock reserved for Twilio.org	635,014	635,014
Stock-based awards available for grant under 2016 Plan	10,773,682	10,200,189
Stock-based awards available for grant under 2016 ESPP	3,216,460	2,478,343
Class A common stock reserved for the convertible senior notes	10,472,165	_
Total	41,033,391	29,689,432

13. Stock-Based Compensation

2008 Stock Option Plan

The Company granted options under its 2008 Stock Option Plan (the "2008 Plan"), as amended and restated, until June 22, 2016, when the plan was terminated in connection with the Company's IPO. Accordingly, no shares are available for future issuance under the 2008 Plan. The 2008 Plan continues to govern outstanding equity awards granted thereunder.

2016 Stock Option Plan

The Company's 2016 Stock Option and Incentive Plan (the "2016 Plan") became effective on June 21, 2016. The 2016 Plan provides for the grant of ISOs, NSOs, restricted stock, RSUs, stock appreciation rights, unrestricted stock awards, performance share awards, dividend equivalent rights and cash-based awards to employees, directors and consultants of the Company. A total of 11,500,000 shares of the Company's Class A common stock were initially reserved for issuance under the 2016 Plan. These available shares automatically increase each January 1, beginning on January 1, 2017, by 5% of the number of shares of the Company's Class A and Class B common stock outstanding on the immediately preceding December 31, or such lesser number of shares as determined by the Company's compensation committee. On January 1, 2018, the shares available for grant under the 2016 Plan were automatically increased by 4,698,490 shares.

Under the 2016 Plan, the stock options are granted at a price per share not less than 100% of the fair market value per share of the underlying common stock on the date of grant. Under both plans, stock options generally expire 10 years from the date of grant and vest over periods determined by the board of directors. The vesting period for new-hire options and restricted stock units is generally a four-year term from the date of grant, at a rate of 25% after one year, then monthly or quarterly, respectively, on a straight-line basis thereafter. In July 2017, the Company began granting restricted stock units to existing employees that vest in equal quarterly installments over a four year service period.



2016 Employee Stock Purchase Plan

The Company's Employee Stock Purchase Plan ("2016 ESPP") became effective on June 21, 2016. A total of 2,400,000 shares of the Company's Class A common stock were initially reserved for issuance under the 2016 ESPP. These available shares will automatically increase each January 1, beginning on January 1, 2017, by the lesser of 1,800,000 shares of the common stock, 1% of the number of shares of the Company's Class A and Class B common stock outstanding on the immediately preceding December 31 or such lesser number of shares as determined by the Company's compensation committee. On January 1, 2018, the shares available for grant under the 2016 Plan were automatically increased by 939,698 shares.

The 2016 ESPP allows eligible employees to purchase shares of the Company's Class A common stock at a discount of up to 15% through payroll deductions of their eligible compensation, subject to any plan limitations. Except for the initial offering period, the 2016 ESPP provides for separate six-month offering periods beginning in May and November of each fiscal year, starting in May 2017.

On each purchase date, eligible employees purchase the Company's stock at a price per share equal to 85% of the lesser of (i) the fair market value of the Company's Class A common stock on the offering date or (ii) the fair market value of the Company's Class A common stock on the purchase date.

In the three months ended September 30, 2018 and 2017, no shares of Class A common stock were purchased under the 2016 ESPP, and 127,327 shares are expected to be purchased in November 2018. As of September 30, 2018, total unrecognized compensation cost related to the 2016 ESPP was \$0.5 million, which will be amortized over a weighted-average period of 0.1 years.

Stock-based awards activity under the 2008 Plan and 2016 Plan was as follows:

Stock Options

	Number of options outstanding	Weighted- average exercise price (per share)	Weighted- average remaining contractual term (in years)	Aggregate intrinsic value (in thousands)
Outstanding options as of December 31, 2017	10,155,427	\$ 10.31	7.12	\$ 145,763
Granted	1,170,118	34.73		
Exercised	(2,998,329)	7.53		
Forfeited and cancelled	(315,064)	16.03		
Outstanding options as of September 30, 2018	8,012,152	\$ 14.69	7.03	\$ 573,604
Options vested and exercisable as of September 30, 2018	4,616,492	\$ 9.07	6.21	\$ 356,433

Aggregate intrinsic value represents the difference between the fair value of the Company's Class A common stock as reported on the New York Stock Exchange and the exercise price of outstanding "in-themoney" options. The aggregate intrinsic value of stock options exercised was \$56.2 million and \$130.7 million during the three and nine months ended September 30, 2018, respectively, and \$18.6 million and \$119.3 million during the three and nine months ended September 30, 2017, respectively.

The total estimated grant date fair value of options vested was \$4.6 million and \$17.6 million during the three and nine months ended September 30, 2018, respectively, and \$3.0 million and \$12.2 million during the three and nine months ended September 30, 2017, respectively.

The weighted-average grant-date fair value per share of options granted was \$28.84 and \$16.04 during the three and nine months ended September 30, 2018, respectively, and \$13.48 during the nine months ended September 30, 2017, respectively. No options were granted in the three months ended September 30, 2017.

On February 28, 2017, the Company granted a total of 555,000 shares of performance-based stock options in three distinct awards to an employee with grant date fair values of \$13.48, \$10.26 and \$8.41 per share for a total grant value of \$5.9 million. The first half of each award vests upon satisfaction of a performance condition and the remainder vests thereafter in equal monthly installments over a 24-month period. The achievement window expires after 4.3 years from the date of grant and the stock options expire seven years after the date of grant. The performance condition for the first award was satisfied on March 31, 2018.

The stock options are amortized over a derived service period, as adjusted, of 3.1 years, 4.4 years and 4.9 years, respectively. The stock options value and the derived service period were estimated using the Monte-Carlo simulation model. The following table summarizes the details of the performance options:

	Number of options outstanding	Weighted- average exercise price (per share)	Weighted- average remaining contractual term (in years)	Aggregate intrinsic value (in thousands)
Outstanding options as of December 31, 2017	555,000	\$ 31.72	6.00	\$ _
Granted	—	—		
Exercised	—	—		
Forfeited and cancelled	—	—		
Outstanding options as of September 30, 2018	555,000	\$ 31.72	5.41	\$ 30,281
Options vested and exercisable as of September 30, 2018	92,500	\$ 31.72	5.41	\$ 5,047

As of September 30, 2018, total unrecognized compensation cost related to nonvested stock options was \$36.2 million, which will be amortized over a weighted-average period of 2.1 years.

Restricted Stock Units

	Number of awards outstanding	Weighted- average grant date fair value (per share)	Aggregate intrinsic value (in thousands)
Nonvested RSUs as of December 31, 2017	5,665,459	\$ 29.29	\$ 133,648
Granted	3,930,656	40.52	
Vested	(1,607,997)	29.31	
Forfeited and cancelled	(619,200)	30.78	
Nonvested RSUs as of September 30, 2018	7,368,918	\$ 35.16	\$ 635,766

As of September 30, 2018, total unrecognized compensation cost related to nonvested RSUs was \$238.9 million, which will be amortized over a weighted-average period of 3.1 years.

Valuation Assumptions

The fair value of employee stock options was estimated on the date of grant using the following assumptions in the Black-Scholes option pricing model:

	Three Months E September 3		ths Ended iber 30,				
	 2018	2018 2017 20			2018		
Employee Stock Options:							
Fair value of common stock	\$ 62.66	_	\$	33.01-\$62.66	\$	24.77 -\$31.96	
Expected term (in years)	6.08	_		6.08		6.08	
Expected volatility	43.6%	_		43.6%-44.2%		46.1%-47.6%	
Risk-free interest rate	2.8%	_		2.7%-2.9%		1.9%-2.1%	
Dividend rate	0%	-		0%		0%	
Employee Stock Purchase Plan:							
Expected term (in years)	_	0.5		0.5		0.5	
Expected volatility	_	33.2%		39.8%		33.2%	
Risk-free interest rate		1.1%		2.1%		1.1%	
Dividend rate	—	0%		0%		0%	
		26					

The following assumptions were used in the Monte Carlo simulation model to estimate the fair value and the derived service period of the performance options:

Asset volatility	40%
Equity volatility	45%
Discount rate	14%
Stock price at grant date	\$ 31.72

In May 2017, the FASB issued ASU 2017-09, "Compensation-Stock Compensation (Topic 718), Scope of Modification Accounting", ASU 2017-09 clarifies when changes to the terms or conditions of a share-based payment award must be accounted for as modifications. ASU 2017-09 allows companies to make certain changes to awards, such as vesting conditions, without accounting for them as modifications. It does not change the accounting for modifications. ASU 2017-09 should be applied prospectively to awards modified on or after the adoption date. The Company adopted ASU 2017-09 in the first quarter of 2018. The adoption of this guidance did not have an impact on the Company's financial position, results of operations or cash flows.

Stock-Based Compensation Expense

The Company recorded total stock-based compensation expense as follows (in thousands):

	Three Months Ended Nine Months Ended September 30, September 30,						
	2018		2017		2018		2017
Cost of revenue	\$ 284	\$	180	\$	772	\$	460
Research and development	10,879		6,493		28,500		16,687
Sales and marketing	5,246		2,603		14,154		6,961
General and administrative	6,332		4,912		17,861		11,865
Total	\$ 22,741	\$	14,188	\$	61,287	\$	35,973

14. Net Loss Per Share Attributable to Common Stockholders

Basic and diluted net loss per common share is presented in conformity with the two-class method required for participating securities.

Class A and Class B common stock are the only outstanding equity in the Company. The rights of the holders of Class A and Class B common stock are identical, except with respect to voting and conversion. Each share of Class A common stock is entitled to one vote per share and each share of Class B common stock is entitled to ten votes per share. Shares of Class B common stock may be converted into Class A common stock at any time at the option of the stockholder, and are automatically converted into Class A common stock upon sale or transfer, subject to certain limited exceptions.

Basic net loss per share attributable to common stockholders is computed using the weighted-average number of common shares outstanding during the period. Diluted net loss per share attributable to common stockholders is computed using the weighted-average number of common shares and, if dilutive, potential common shares outstanding during the period. The dilutive effect of these potential common shares is reflected in diluted earnings per share by application of the treasury stock method.

The following table sets forth the calculation of basic and diluted net loss per share attributable to common stockholders during the periods presented (in thousands, except share and per share data):

	_	Three Mon Septem		_	Nine Mon Septem	
		2018	2017		2018	2017
Net loss attributable to common stockholders	\$	(27,056)	\$ (23,453)	\$	(74,791)	\$ (44,789)
Weighted-average shares used to compute net loss per share attributable to common						
stockholders, basic and diluted		98,019,629	 92,156,768		96,359,437	 90,543,087
Net loss per share attributable to common stockholders, basic and diluted	\$	(0.28)	\$ (0.25)	\$	(0.78)	\$ (0.49)

The following outstanding shares of common stock equivalents were excluded from the calculation of the diluted net loss per share attributable to common stockholders because their effect would have been anti-dilutive:

	As of Septemb	As of September 30,		
	2018	2017		
Stock options issued and outstanding	8,567,152	11,380,189		
Nonvested restricted stock units issued and outstanding	7,368,918	4,384,898		
Class A common stock reserved for Twilio.org	635,014	680,397		
Class A common stock committed under 2016 ESPP	127,327	224,126		
Conversion spread*	233,799	_		
Unvested shares subject to repurchase	209	16,033		
Total	16,932,419	16,685,643		

*Since the Company expects to settle the principal amount of its outstanding convertible senior notes in cash and any excess in shares of the Company's Class A common stock, the Company uses the treasury stock method for calculating any potential dilutive effect of the conversion spread on diluted net income per share, if applicable. The conversion spread will have a dilutive impact on diluted net income per share of Class A common stock when the average market price of the Company's Class A common stock for a given period exceeds the conversion price of \$70.90 per share for the Notes.

15. Income Taxes

In October 2016, the FASB issued ASU 2016-16, "Intra-Entity Transfers Other Than Inventory", which requires entities to recognize the income tax consequences of an intra-entity transfer of an asset other than inventory when the transfer occurs. The Company adopted ASU 2016-16 in the first quarter 2018. Adoption of the ASU did not have any impact on the Company's financial statements.

The Company determines its income tax provision or benefit for interim periods using an estimate of its annual effective tax rate, adjusted for discrete items occurring in the quarter. The primary difference between the Company's effective tax rate and the federal statutory rate relates to the net operating losses in jurisdictions with a valuation allowance or a zero tax rate.

The Company recorded a provision for income taxes of \$0.1 million and \$0.4 million for the three and nine months ended September 30, 2018, respectively, and \$0.4 million and \$0.9 million for the three and nine months ended September 30, 2017, respectively. The provision for income taxes consists primarily of income taxes and withholding taxes in foreign jurisdictions in which the Company conducts business. The Company's U.S. operations have been in a loss position and the Company maintains a full valuation allowance against its U.S. deferred tax assets.

On December 22, 2017, the U.S. government enacted comprehensive tax legislation commonly referred to as the Tax Cuts and Jobs Act (the "Tax Act"). The Tax Act makes broad and complex changes to the U.S. tax code including, but not limited to, (1) reducing the U.S. federal corporate tax rate from 35 percent to 21 percent; (2) requiring companies to pay a one-time transition tax on certain unrepatriated earnings of foreign subsidiaries; (3) generally eliminating U.S. federal income taxes on dividends from foreign subsidiaries; (4) requiring a current inclusion in U.S. federal taxable income of certain earnings of controlled foreign corporations; (5) eliminating the corporate alternative minimum tax ("AMT") and changing how existing AMT credits can be realized; (6) creating the base erosion anti-abuse tax ("BEAT"), a new minimum tax; (7) creating a new limitation on deductible interest expense; and (8) changing rules related to uses and limitations of net operating loss carryforwards created in tax years beginning after December 31, 2017.

The SEC staff issued SAB 118, which provides guidance on accounting for the tax effects of the Tax Act. SAB 118 provides a measurement period that should not extend beyond one year from the Tax Act enactment date for companies to complete the accounting under ASC 740. The Company has not completed its accounting assessment for the effects of the Tax Act as of September 30, 2018; however, the final accounting assessment will occur no later than one year from the date the Tax Act was enacted. The Company has also considered and estimated a number of provisions of the Tax Act effective January 1, 2018 and, based on the initial assessment, the Company has determined that the Tax Act did not have a material effect on its consolidated financial statements for the three and nine months ended September 30, 2018.

16. Subsequent Events

On October 15, 2018, the Company and SendGrid, Inc. ("SendGrid") announced a definitive agreement (the "Merger Agreement") for the Company to acquire SendGrid.

Under the terms of the Merger Agreement, shareholders of SendGrid will receive 0.485 of a Company share of Class A common stock for each SendGrid share they hold. Based upon the SendGrid shares outstanding as of October 10, 2018, the Company would be required to issue approximately 26.3 million new shares, a value of approximately \$2.0 billion at the time of the announcement of the transaction.

The boards of directors of each of the Company and SendGrid have approved the transaction, which is also subject to approval by each of SendGrid's and the Company's stockholders, clearance under the Hart-Scott-Rodino Antitrust Improvements Act and other regulatory reviews, and other customary closing conditions.

Under the terms of the Merger Agreement, the Company will pay SendGrid \$120 million if the merger is not consummated under certain circumstances, including if the Merger Agreement is terminated relating to a Company change in board recommendation in connection with an alternative transaction. If the Merger Agreement is terminated relating to a change in board recommendation and/or alternative transaction by SendGrid, SendGrid will be required to pay the Company \$69 million.

The proposed transaction is expected to close in the first half of 2019. During the nine months ended September 30, 2018, the Company incurred \$1.0 million in expenses related to this transaction.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis of our financial condition and results of operations should be read in conjunction with our condensed consolidated financial statements and related notes appearing elsewhere in this Quarterly Report on Form 10-Q. In addition to historical financial information, the following discussion contains forward-looking statements that are based upon current plans, expectations and beliefs that involve risks and uncertainties. Our actual results may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those set forth under Part II, Item 1A, "Risk Factors" in this Quarterly Report on Form 10-Q. Our fiscal year ends on December 31.

Overview

We are the leader in the Cloud Communications Platform category. We enable developers to build, scale and operate real-time communications within their software applications via our simple-to-use Application Programming Interfaces ("API"). The power, flexibility, and reliability offered by our software building blocks empowers companies of virtually every shape and size to build world-class engagement into their customer experience.

Our platform consists of three layers: our Engagement Cloud, Programmable Communications Cloud and Super Network. Our Engagement Cloud software is a set of APIs that handles the higher level communication logic needed for nearly every type of customer engagement. These APIs are focused on the business challenges that a developer is looking to address, allowing our customers to more quickly and easily build better ways to engage with their customers throughout their journey. Our Programmable Communications Cloud software is a set of APIs that enables developers to embed voice, messaging and video capabilities into their applications. The Programmable Communications Cloud is designed to support almost all the fundamental ways humans communicate, unlocking innovators to address just about any communication market. The Super Network is our software layer that allows our customers' software to communicate with connected devices globally. It interconnects with communications networks around the world and continually analyzes data to optimize the quality and cost of communications that flow through our platform. The Super Network also contains a set of APIs that gives our customers access to more foundational components of our platform, like phone numbers.

As of September 30, 2018, our customers' applications that are embedded with our products could reach users via voice, messaging and video in nearly every country in the world, and our platform offered customers local telephone numbers in over 100 countries and text-to-speech functionality in 26 languages. We support our global business through 27 cloud data centers in nine regions around the world and have developed contractual relationships with network service providers globally.

Our business model is primarily focused on reaching and serving the needs of software developers, who we believe are becoming increasingly influential in technology decisions in a wide variety of companies. We call this approach our Business Model for Innovators, which empowers developers by reducing friction and upfront costs, encouraging experimentation, and enabling developers to grow as customers as their ideas succeed. We established and maintain our leadership position by engaging directly with, and cultivating, our developer community, which has led to the rapid adoption of our platform. We reach developers through community events and conferences, including our SIGNAL developer conferences, to demonstrate how every developer can create differentiated applications incorporating communications using our products.

Once developers are introduced to our platform, we provide them with a low friction trial experience. By accessing our easy-to-adopt APIs, extensive self-service documentation and customer support team, developers build our products into their applications and then test such applications through free trials. Once they have decided to use our products beyond the initial free trial period, customers provide their credit card information and only pay for the actual usage of our products. Historically, we have acquired the substantial majority of our customers through this self-service model. As customers expand their usage of our platform, our relationships with me often evolve to include business leaders within their organizations. Once our customers reach a certain spending level with us, we support them with account executives or customer success advocates within our sales organization to ensure their satisfaction and expand their usage of our products.

When potential customers do not have the available developer resources to build their own applications, we refer them to our network of Solution Partners, who embed our products in their solutions, such as software for contact centers, sales force automation and marketing automation that they sell to other businesses.

We supplement our self-service model with a sales effort aimed at engaging larger potential customers, strategic leads and existing customers through a direct sales approach. We augment this sales effort with the Twilio Enterprise Plan, which provides capabilities for advanced security, access management and granular administration, and is targeted at the needs of enterprise scale customers. Our sales organization works with technical and business leaders who are seeking to leverage software to the with differentiation. As we educate these leaders on the benefits of developing applications that incorporate our products to differentiate their business, they often consult with their developers regarding implementation. We believe that developers are often advocates for our products as a result of our developer-focused approach. Our sales organization includes sales development, inside sales, field sales, sales engineering and customer success personnel.

We generate the substantial majority of our revenue from customers based on their usage of our software products that they have incorporated into their applications. In addition, customers typically purchase one or more telephone numbers from us, for which we charge a monthly flat fee per number. Some customers also choose to purchase various levels of premium customer support for a monthly fee. Customers that register in our self-service model typically pay upfront via credit card and draw down their balance as they purchase or use our products. Most of our customers draw down their balance in the same month they pay up front and, as a result, our deferred revenue and customer deposits liability at any particular time is not a meaningful indicator of future revenue. As our customers' usage grows, some of our customers enter into contracts and are invoiced monthly in arrears. Many of these customer outracts have terms of 12 months and typically include some level of minimum revenue commitment revenue from customers with whom we have contracts has constituted a minority of our revenue in any period, and we expect this to continue in the future.

Our developer-focused products are delivered to customers and users through our Super Network, which uses software to optimize communications on our platform. We interconnect with communications networks globally to deliver our products, and therefore we have arrangements with network service providers in many regions in the world. Historically, a substantial majority of our cost of revenue has been network service provider fees. We continue to optimize our network service provider coverage and connectivity through continuous improvements in routing and sourcing in order to lower the usage expenses we incur for network service provider fees. As we benefit from our platform optimization efforts, we sometimes pass these savings on to customers in the form of lower usage prices on our products in an effort to drive increased usage and expand the reach and scale of our platform. In the near term, we intend to operate our business to expand the reach and scale of our platform, and to grow our revenue, rather than to maximize our gross margins.

We have achieved significant growth in recent periods. In the three months ended September 30, 2018 and 2017, our revenue was \$168.9 million and \$100.5 million, respectively. In the three months ended September 30, 2018 and 2017, our 10 largest Active Customer Accounts generated an aggregate of 18% and 17% of our revenue, respectively. For the three months ended September 30, 2018 and 2017, among our 10 largest Active Customer Accounts we had three and four Variable Customer Accounts, respectively, representing 8% of our revenue in each period. For the three months ended September 30, 2018 and 2017, our Base Revenue was \$154.3 million and \$20.0 million, respectively. We incurred a net loss of \$27.1 million and \$23.5 million, for the three months ended September 30, 2018 and 2017, respectively. See the section titled "— Key Business Metrics—Base Revenue" below for a discussion of Base Revenue.

Proposed Acquisition

On October 15, 2018, we entered into a definitive agreement to acquire SendGrid, Inc. ("SendGrid") pursuant to an Agreement and Plan of Merger and Reorganization (the "Merger Agreement"). SendGrid is a communication platform for transactional and marketing email. Under the terms and conditions of the Merger Agreement, at the closing of the proposed transaction each outstanding share of SendGrid common stock will be converted into the right to receive a 0.485 of a share of our Class A common stock. The transaction is expected to close in the first half of 2019. During the nine months ended September 30, 2018 we incurred \$1.0 million in expenses related to this proposed transaction. If consummated, the acquisition of SendGrid will have a significant impact on our liquidity, financial condition and results of operations. Unless otherwise noted, the following discussion and analysis of our results of operations and our liquidity and capital resources focuses on our existing operations exclusive of the proposed acquisition of SendGrid. Any forward-looking statements contained herein do not take into account the impact of such proposed acquisition. See Note 16 to our condensed consolidated financial statements for additional information regarding the proposed transaction with SendGrid.

Key Business Metrics

	Three Months Ended September 30,		
	2018		2017
Number of Active Customer Accounts (as of end date of period)	 61,153		46,489
Base Revenue (in thousands)	\$ 154,348	\$	91,965
Base Revenue Growth Rate	68%		43%
Dollar-Based Net Expansion Rate	145%		122%

Number of Active Customer Accounts. We believe that the number of our Active Customer Accounts is an important indicator of the growth of our business, the market acceptance of our platform and future revenue trends. We define an Active Customer Account at the end of any period as an individual account, as identified by a unique account identifier, for which we have

recognized at least \$5 of revenue in the last month of the period. We believe that the use of our platform by our customers at or above the \$5 per month threshold is a stronger indicator of potential future engagement than trial usage of our platform or usage at levels below \$5 per month. A single organization may constitute multiple unique Active Customer Accounts if it has multiple account identifiers, each of which is treated as a separate Active Customer Account.

In the three months ended September 30, 2018 and 2017, revenue from Active Customer Accounts represented over 99% of total revenue in each period.

Base Revenue. We monitor Base Revenue as one of the more reliable indicators of future revenue trends. Base Revenue consists of all revenue other than revenue from large Active Customer Accounts that have never entered into 12-month minimum revenue commitment contracts with us, which we refer to as Variable Customer Accounts. While almost all of our customer accounts exhibit some level of variability in the usage of our products, based on our experience, we believe that Variable Customer Accounts are more likely to have significant fluctuations in usage of our products from period to period, and therefore that revenue from Variable Customer Accounts are more likely to have significant fluctuations of such customers not to enter into contracts with us that contain minimum revenue commitments, even though they may spend significant amounts on the use of our products and they may be foregoing more favorable terms often available to customers that enter into committed contracts with us. This variability adversely affects our ability to rely upon revenue from Variable Customer Accounts when analyzing expected trends in future revenue.

For historical periods through March 31, 2016, we defined a Variable Customer Account as an Active Customer Account that (i) had never signed a minimum revenue commitment contract with us for a term of at least 12 months and (ii) had met or exceeded 1% of our revenue in any quarter in the periods presented through March 31, 2016. To allow for consistent period-to-period comparisons, in the event a customer account qualified as a Variable Customer Account as of March 31, 2016, or a previously Variable Customer Account ceased to be an Active Customer Account as of such date, we included such customer account as a Variable Customer Account in all periods presented. For reporting periods starting with the three months ended June 30, 2016, we define a Variable Customer Account in a customer account that (a) has been categorized as a Variable Customer Account in any prior quarter, as well as (b) any new customer account that (i) is with a customer that has never signed a minimum revenue commitment contract with us for a term of at least 12 months and (ii) meets or exceeds 1% of our revenue in a quarter. Once a customer account that (a) the be a Variable Customer Account in any period, it remains a Variable Customer Account in subsequent periods such customer enters into a minimum revenue commitment contract with us for a term of at least 12 months.

In each of the three months ended September 30, 2018 and 2017, we had six Variable Customer Accounts, which represented 9% of our total revenue.

Dollar-Based Net Expansion Rate. Our ability to drive growth and generate incremental revenue depends, in part, on our ability to maintain and grow our relationships with existing Active Customer Accounts and to increase their use of the platform. An important way in which we track our performance in this area is by measuring the Dollar-Based Net Expansion Rate for our Active Customer Accounts, other than our Variable Customer Accounts. Our Dollar-Based Net Expansion Rate decreases when such Active Customer Accounts increase usage of a product, extend usage of a product to new applications or adopt a new product. Our Dollar-Based Net Expansion Rate decreases when such Active Customer Accounts increase usage of a product or when we lower usage prices on a product. As our customers grow their businesses and extend the use of our platform, they sometimes create multiple customer accounts with us for operational or other reasons. As such, for reporting periods starting with the three months ended December 31, 2016, when we identify a significant customer organization (defined as a single customer organization generating more than 1% of our revenue in a quarterly reporting period) that has created a new Active Customer Account, this new Active Customer Account is included with, the original Active Customer Account for the purposes of calculating this metric. We believe measuring our Dollar-Based Net Expansion Rate on revenue generated from our Active Customer Accounts, other than our Variable Customer Accounts, provides a more meaningful indication of the performance of our efforts to increase revenue from existing customers.

Our Dollar-Based Net Expansion Rate compares the revenue from Active Customer Accounts, other than Variable Customer Accounts, in a quarter to the same quarter in the prior year. To calculate the Dollar-Based Net Expansion Rate, we first identify the cohort of Active Customer Accounts, other than Variable Customer Accounts, that were Active Customer Accounts in the same quarter of the prior year. The Dollar-Based Net Expansion Rate is the quotient obtained by dividing the revenue generated from that cohort in a quarter, by the revenue generated from that same cohort in the corresponding quarter in the prior year. When we calculate Dollar-Based Net Expansion Rate for periods longer than one quarter, we use the average of the applicable quarterly Dollar-Based Net Expansion Rates for each of the quarters in such period.

Key Components of Statements of Operations

Revenue. We derive our revenue primarily from usage-based fees earned from customers using the software products within our Engagement Cloud and Programmable Communications Cloud. These usagebased software products include offerings, such as Programmable Voice, Programmable Messaging and Programmable Video. Some examples of the usage-based fees for which we charge include minutes of call duration activity for our Programmable Voice products, number of text messages sent or received using our Programmable Messaging products and number of authentications for our Account Security products. In the three months ended September 30, 2018 and 2017, we generated 83% and 82% of our revenue, respectively, from usage-based fees. We also earn monthly flat fees from certain fee-based products, such as telephone numbers, short codes and customer support.

Customers typically pay upfront via credit card in monthly prepaid amounts and draw down their balances as they purchase or use our products. As customers grow their usage of our products they automatically receive tiered usage discounts. Our larger customers often enter into contracts, for at least 12 months that contain minimum revenue commitments, which may contain more favorable pricing. Customers on such contracts typically are invoiced monthly in arrears for products used.

Amounts that have been charged via credit card or invoiced are recorded in accounts receivable and in revenue, deferred revenue or customer deposits, depending on whether the revenue recognition criteria have been met. Given that our credit card prepayment amounts tend to be approximately equal to our credit card consumption amounts in each period, and that we do not have many invoiced customers on pre-payment contract terms, our deferred revenue at any particular time is not a meaningful indicator of future revenue.

We define U.S. revenue as revenue from customers with IP addresses at the time of registration in the United States, and we define international revenue as revenue from customers with IP addresses at the time of registration outside of the United States.

Cost of Revenue and Gross Margin. Cost of revenue consists primarily of fees paid to network service providers. Cost of revenue also includes cloud infrastructure fees, personnel costs, such as salaries and stock-based compensation for our customer support employees, and non-personnel costs, such as amortization of capitalized internal use software development costs. Our arrangements with network service providers require us to pay fees based on the volume of phone calls initiated or text messages sent, as well as the number of telephone numbers acquired by us to service our customers. Our arrangements with our cloud infrastructure provider require us to pay fees based on our server capacity consumption.

Our gross margin has been and will continue to be affected by a number of factors, including the timing and extent of our investments in our operations, increases or decreases in our network service provider and cloud infrastructure-related fees, the mix of U.S. usage compared to international usage, exchange rate fluctuations related to network service provider fees denominated in currencies other than the U.S. dollar, the mix of products that are less dependent on the traditional network service provider networks, the timing of amortization of capitalized software development costs, and the extent to which we periodically choose to pass on our cost savings from platform optimization efforts to our customers in the form of lower usage prices.

Operating Expenses. The most significant components of operating expenses are personnel costs, which consist of salaries, benefits, ad-hoc bonuses, stock-based compensation and compensation expenses related to stock repurchases from employees. We also incur other non-personnel costs related to our general overhead expenses. We expect that our operating costs will increase in absolute dollars.

Research and Development. Research and development expenses consist primarily of personnel costs, outsourced engineering services, cloud infrastructure fees for staging and development, amortization of capitalized internal use software development costs and an allocation of our general overhead expenses. We capitalize the portion of our software development costs that meets the criteria for capitalization.

We continue to focus our research and development efforts on adding new features and products, including new use cases, improving our platform and increasing the functionality of our existing products.

Sales and Marketing. Sales and marketing expenses consist primarily of personnel costs, including commissions for our sales employees. Sales and marketing expenses also include expenditures related to advertising, marketing, our brand awareness



activities and developer evangelism, costs related to our SIGNAL developer conferences, credit card processing fees, professional services fees and an allocation of our general overhead expenses.

We focus our sales and marketing efforts on generating awareness of our company, platform and products through our developer evangelist team and self-service model, creating sales leads and establishing and promoting our brand, both domestically and internationally. We plan to continue investing in sales and marketing by increasing our sales and marketing headcount, supplementing our self-service model with an enterprise sales approach, expanding our sales channels, driving our go-to-market strategies, building our brand awareness and sponsoring additional marketing events.

General and Administrative. General and administrative expenses consist primarily of personnel costs for our accounting, finance, legal, human resources and administrative support personnel and executives. General and administrative expenses also include costs related to business acquisitions, legal and other professional services fees, sales and other taxes, depreciation and an antization and an allocation of our general overhead expenses. We expect that we will incur costs associated with supporting the growth of our business and to meet the increased compliance requirements associated with both our international expansion and our transition to, and operation as, a public company.

Our general and administrative expenses include a significant amount of sales and other taxes to which we are subject based on the manner we sell and deliver our products. Prior to March 2017, we did not collect such taxes from our customers and recorded such taxes as general and administrative expenses. Effective March 2017, we began collecting these taxes from customers in certain jurisdictions and added more jurisdictions throughout 2017 and in the first quarter of 2018 we began to collect these taxes in one additional jurisdiction. We continue expanding the number of jurisdictions where we will be collecting these taxes in the future. We expect that these expenses will decline in future years as we continue collecting these taxes from our customers in more jurisdictions, which would reduce our rate of ongoing accrual.

Provision for Income Taxes. Our income tax provision or benefit for interim periods is determined using an estimate of our annual effective tax rate, adjusted for discrete items occurring in the quarter. The primary difference between our effective tax rate and the federal statutory rate relates to the net operating losses in jurisdictions with a valuation allowance or a zero tax rate.

On December 22, 2017, the U.S. government enacted comprehensive tax legislation commonly referred to as the Tax Cuts and Jobs Act (the "Tax Act"). The Tax Act makes broad and complex changes to the U.S. tax code including, but not limited to, (1) reducing the U.S. federal corporate tax rate from 35 percent to 21 percent; (2) requiring companies to pay a one-time transition tax on certain unrepatriated earnings of foreign subsidiaries; (3) generally eliminating U.S. federal income taxes on dividends from foreign subsidiaries; (4) requiring a current inclusion in U.S. federal taxable income of certain earnings of controlled foreign corporations; (5) eliminating the corporate alternative minimum tax ("AMT") and changing how existing AMT credits can be realized; (6) creating the base erosion anti-abuse tax ("BEAT"), a new minimum tax; (7) creating a new limitation on deductible interest expense; and (8) changing rules related to uses and limitations of net operating loss carryforwards created in tax years beginning after December 31, 2017.

We have not completed our accounting assessment for the effects of the Tax Act as of September 30, 2018; however, based on our initial assessment, we have determined that the Tax Act did not have a material effect on our condensed consolidated financial statements for the three months ended September 30, 2018.

Non-GAAP Financial Measures

We use the following non-GAAP financial information, collectively, to evaluate our ongoing operations and for internal planning and forecasting purposes. We believe that non-GAAP financial information, when taken collectively, may be helpful to investors because it provides consistency and comparability with past financial performance, facilitates period-to-period comparisons of results of operations, and assists in comparisons with other companies, many of which use similar non-GAAP financial information to supplement their GAAP results. Non-GAAP financial information is presented for supplemental information purposes only, and should not be considered a substitute for financial information presented in accordance with generally accepted accounting principles, and may be different from similarly-titled non-GAAP measures used by other companies. Whenever we use a non-GAAP financial measure, a reconciliation is provided to the most closely applicable financial measure stated in accordance with generally accepted accounting principles. Investors are encouraged to review the related GAAP financial measures and the reconciliation of these non-GAAP financial measures to their most directly comparable GAAP financial measures.

Non-GAAP Gross Profit and Non-GAAP Gross Margin. For the periods presented, we define non-GAAP gross profit and non-GAAP gross margin as GAAP gross profit and GAAP gross margin, respectively, adjusted to exclude stock-based compensation and amortization of acquired intangibles.

	 Three Months Ended September 30, 2018 (In thousands)		
Reconciliation:	(111 110)	isanusj	
Gross profit	\$ 91,864	\$	52,288
Non-GAAP adjustments:			
Stock-based compensation	284		180
Amortization of acquired intangibles	1,396		1,250
Non-GAAP gross profit	\$ 93,544	\$	53,718
Non-GAAP gross margin	 55%		53%

Non-GAAP Operating Expenses. For the periods presented, we define non-GAAP operating expenses (including categories of operating expenses) as GAAP operating expenses (and categories of operating expenses) adjusted to exclude, as applicable, stock-based compensation, amortization of acquired intangibles, acquisition-related expenses, legal settlements/accruals, charitable contributions and payroll taxes related to stock-based compensation.

		Three Months Ended September 30,		
	2018			2017
		(In thousands)		
Reconciliation:				
Operating expenses	\$	116,897	\$	76,319
Non-GAAP adjustments:				
Stock-based compensation		(22,457)		(14,008)
Amortization of acquired intangibles		(410)		(265)
Acquisition-related expenses		(1,554)		(35)
Legal settlements/accruals		(1,510)		
Charitable contributions		(175)		
Payroll taxes related to stock-based compensation		(1,498)		(595)
Non-GAAP operating expenses	\$	89,293	\$	61,416

(1)

Non-GAAP Income (Loss) from Operations and Non-GAAP Operating Margin. For the periods presented, we define non-GAAP income (loss) from operations and non-GAAP operating margin as GAAP loss from operations and GAAP operating margin, respectively, adjusted to exclude stock-based compensation, amortization of acquired intangibles, acquisition-related expenses, legal settlements/accruals, charitable contributions and payroll taxes related to stock-based compensation.

	Three Months Ended September 30,					
	2018 2017					
Reconciliation:		(In thou	sands)			
Loss from operations	\$	(25,033)	\$	(24,031)		
Non-GAAP adjustments:						
Stock-based compensation		22,741		14,188		
Amortization of acquired intangibles		1,806		1,515		
Acquisition-related expenses		1,554		35		
Legal settlements/accruals		1,510		—		
Charitable contributions		175		_		
Payroll taxes related to stock-based compensation		1,498		595		
Non-GAAP income (loss) from operations	\$	4,251	\$	(7,698)		
Non-GAAP operating margin		3%		(8)%		

Results of Operations

The following tables set forth our results of operations for the periods presented and as a percentage of our total revenue for those periods. The period-to-period comparison of our historical results are not necessarily indicative of the results that may be expected in the future.

	Three Months Ended September 30,					Nine Months Ended September 30,			
		2018		2017		2018		2017	
				(In tho	usands)				
Condensed Consolidated Statements of Operations Data:									
Revenue	\$	168,895	\$	100,542	\$	445,765	\$	283,784	
Cost of revenue ^{(1) (2)}		77,031		48,254		204,553		127,873	
Gross profit		91,864		52,288		241,212		155,911	
Operating expenses:									
Research and development ⁽¹⁾⁽²⁾		42,340		31,674		119,727		87,910	
Sales and marketing ⁽¹⁾⁽²⁾		45,949		25,778		116,520		73,047	
General and administrative ^{(1) (2)}		28,608		18,867		76,213		40,810	
Total operating expenses		116,897		76,319		312,460		201,767	
Loss from operations		(25,033)		(24,031)		(71,248)		(45,856)	
Other income (expenses), net		(1,939)		1,000		(3,172)		1,969	
Loss before provision for income taxes		(26,972)		(23,031)		(74,420)		(43,887)	
Provision for income taxes		(84)		(422)		(371)		(902)	
Net loss attributable to common stockholders	\$	(27,056)	\$	(23,453)	\$	(74,791)	\$	(44,789)	

Includes stock-based compensation expense as follows:

	Three Mo Septen	Nine Months Ended September 30,				
	2018	2017		2018		2017
		(In tho	isands)			
Cost of revenue	\$ 284	\$ 180	\$	772	\$	460
Research and development	10,879	6,493		28,500		16,687
Sales and marketing	5,246	2,603		14,154		6,961
General and administrative	6,332	4,912		17,861		11,865
Total	\$ 22,741	\$ 14,188	\$	61,287	\$	35,973
	36					

(2) Includes amortization of acquired intangibles as follows:

			nths Ended nber 30,	Nine Months Ended September 30,			
		2018		2017	2018		2017
	¢	1.207	¢		ousands)	2 710	2.420
Cost of revenue	\$	1,396	\$	1,250	\$	3,719 \$	
Research and development				25		22	101
Sales and marketing		390		220		816	539
General and administrative		20		20		60	64
Total	\$	1,806	\$	1,515	\$	4,617 \$	4,133
				Three Months Ended September 30,			Nine Months Ended September 30,
			2018		2017	2018	2017
onsolidated Statements of Operations, as a percentage of revenue:**							
evenue				100%	100%		00% 10
ost of revenue				46	48		46
ross profit				54	52		54 :
perating expenses:							
Research and development				25	32		27
Sales and marketing				27	26		26
General and administrative				17	19		17
Total operating expenses			-	69	76		70
oss from operations				(15)	(24)	(16) (1
ther income (expenses), net				(1)	1		(1)
Loss before provision for income taxes				(16)	(23)	(17) (1
ovision for income taxes				*	*		*
Net loss attributable to common stockholders			-	(16)%	(23)%	(17)%

* Less than 0.5% of revenue.

** Columns may not add up to 100% due to rounding.

Comparison of the Three Months Ended September 30, 2018 and 2017

Revenue

2018		2017		Change	
		(Dollars in	thousands)		
\$ 154,348	\$	91,965	\$	62,383	68%
14,547		8,577		5,970	70%
\$ 168,895	\$	100,542	\$	68,353	68%
\$	Septem 2018 \$ 154,348 14,547	September 30, 2018 \$ 154,348 \$ 14,547	2018 2017 \$ 154,348 \$ 91,965 14,547 8,577	September 30, 2018 2017 (Dollars in thousands) (Dollars in thousands) \$ 154,348 91,965 14,547 8,577	September 30, Change 2018 2017 Change (Dollars in thousands) (Dollars in thousands) (Dollars in thousands) \$ 154,348 \$ 91,965 \$ 62,383 14,547 8,577 5,970

In the three months ended September 30, 2018, Base Revenue increased by \$62.4 million, or 68%, compared to the same period last year, and represented 91% of total revenue in both the three months ended September 30, 2018 and 2017. This increase was primarily attributable to an increase in the usage of our products, particularly our Programmable Messaging products and Programmable Voice products, and the adoption of additional products by our existing customers. This increase was partially offset by pricing decreases that we have implemented over time in the form of lower usage prices, in an effort to increase the reach and scale of our platform. The changes in usage and price in the three months ended September 30, 2018 were reflected in our Dollar-Based Net Expansion Rate of 145%. The increase in usage was also attributable to a 32% increase in the number of Active Customer Accounts, from 46,489 as of September 30, 2017, to 61,153 as of September 30, 2018.

In the three months ended September 30, 2018, Variable Revenue increased by \$6.0 million, or 70%, compared to the same period last year, and represented 9% of total revenue in both the three months ended September 30, 2018 and 2017. This increase was primarily attributable to the increase in the usage of products by our existing Variable Customer Accounts.

U.S. revenue and international revenue represented \$125.7 million, or 74%, and \$43.2 million, or 26%, respectively, of total revenue in the three months ended September 30, 2018, compared to \$76.7 million, or 76%, and \$23.8 million, or 24%, respectively, of total revenue in the three months ended September 30, 2017. The increase in international revenue was attributable to the growth in usage of our products, particularly our Programmable Messaging products and Programmable Voice products, by our existing international Active Customer Accounts, and a 36% increase in the number of international Active Customer Accounts driven in part by our focus on expanding our sales to customers outside of the United States.

Cost of Revenue and Gross Margin

	Three Months Ende September 30,	ed			
	 2018	2017		Change	
		(Dollars i	n thousands)		
Cost of revenue	\$ 77,031 \$	48,254	\$	28,777	60%
Gross margin	54%	52%	, D		

In the three months ended September 30, 2018, cost of revenue increased by \$28.8 million, or 60%, compared to the same period last year. The increase in cost of revenue was primarily attributable to a \$24.9 million increase in network service providers' costs and a \$1.8 million increase in cloud infrastructure fees both to support the growth in usage of our products.

In the three months ended September 30, 2018, gross margin percentage increased primarily as a result of lower network service provider costs related to foreign currency fluctuations, operational improvements, and product mix, which was partially offset by an increasing mix of international product usage.

Operating Expenses

		nths Ended 1ber 30,	I			
	2018		2017		Change	
			(Dollars in	thousands)		
Research and development	\$ 42,340	\$	31,674	\$	10,666	34%
Sales and marketing	45,949		25,778		20,171	78%
General and administrative	28,608		18,867		9,741	52%
Total operating expenses	\$ 116,897	\$	76,319	\$	40,578	53%

In the three months ended September 30, 2018, research and development expenses increased by \$10.7 million, or 34%, compared to the same period last year. The increase was primarily attributable to a \$6.9 million increase in personnel costs, net of a \$0.9 million increase in capitalized software development costs, largely as a result of a 17% average increase in our research and development headcount, as we continued to focus on enhancing our existing products and introducing new products, as well as enhancing product management and other technical functions. The increase was also due to a \$0.9 million increase in personnel costs, encoded as \$0.4 million increase in outsourced engineering services and a \$0.3 million increase in cloud infrastructure fees to support the staging and development of our products.

In the three months ended September 30, 2018, sales and marketing expenses increased by \$20.2 million, or 78%, compared to the same period last year. The increase was primarily attributable to a \$12.5 million increase in personnel costs, largely as a result of a 50% average increase in sales and marketing headcount as we continued to expand our sales efforts in the United States and internationally, a \$2.1 million increase in advertising expenses, a \$2.1 million increase related to our brand awareness programs, conferences and events and a \$0.5 million increase in credit card processing fees due to increased volumes.

In the three months ended September 30, 2018, general and administrative expenses increased by \$9.7 million, or 52%, compared to the same period last year. The increase was primarily attributable to a \$3.4 million increase in personnel costs, largely as a result of a 32% average increase in general and administrative headcount to support the growth of our business domestically and internationally, a \$3.4 million increase in professional services fees primarily related to our operations as a public company and our on-going litigation matters, including legal settlements/accruals, a \$1.5 million increase in the professional fees specifically related to the business adescribed in Note 5 to our condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q and a \$0.4 million increase in depreciation and amortization.

Other Income (Expenses), net

	Three Mon Septem	ed			
	2018	2017		Change	
		(Dollars in	thousar	ıds)	
Other income (expenses), net	\$ (1,939)	\$ 1,000	\$	(2,939)	(294)%

In the three months ended September 30, 2018, net other income (expenses) decreased by \$2.9 million, or 294%, compared to the same period last year, primarily due to \$5.7 million amortization of debt discount and issuance costs related to our 0.25% convertible senior notes due 2023 (the "Notes") issued in May 2018. This decrease was partially offset by \$3.0 million of interest earned on our marketable securities portfolio.

Comparison of the Nine Months Ended September 30, 2018 and 2017

Revenue

	Nine Months End	led Septem	ıber 30,						
	 2018		2017		Change				
			(Dollars in	thousands)					
Base Revenue	\$ 406,859	\$	260,191	\$	146,668	56%			
Variable Revenue	38,906		23,593		15,313	65%			
Total revenue	\$ 445,765	\$	283,784	\$	161,981	57%			

In the nine months ended September 30, 2018, Base Revenue increased by \$146.7 million, or 56%, compared to the same period last year, and represented 91% and 92% of total revenue in the nine months ended September 30, 2018 and 2017, respectively. This increase was primarily attributable to an increase in the usage of our products, particularly our Programmable Messaging products and Programmable Voice products, and the adoption of additional products by our existing customers. This increase was partially offset by pricing decreases that we have implemented over time in the form of lower usage prices, in an effort to increase the reach and scale of our platform. The changes in usage and price in the nine months ended September 30, 2018 were reflected in our Dollar-Based Net Expansion Rate of 138%. The increase in usage was also attributable to a 32% increase in the number of Active Customer Accounts, from 46,489 as of September 30, 2017, to 61,153 as of September 30, 2018.

In the nine months ended September 30, 2018, Variable Revenue increased by \$15.3 million, or 65%, compared to the same period last year, and represented 9% and 8% of total revenue in the nine months ended September 30, 2018 and 2017, respectively. This increase was primarily attributable to the increase in the usage of products by our existing Variable Customer Accounts.

U.S. revenue and international revenue represented \$335.6 million, or 75%, and \$110.2 million, or 25%, respectively, of total revenue in the nine months ended September 30, 2018, compared to \$221.9 million, or 78%, and \$61.9 million, or 22%, respectively, of total revenue in the nine months ended September 30, 2017. The increase in international revenue was attributable to the growth in usage of our products, particularly our Programmable Messaging products and Programmable Voice products, by our existing international Active Customer Accounts, and a 36% increase in the number of international Active Customer Accounts driven in part by our focus on expanding our sales to customers outside of the United States.

Cost of Revenue and Gross Margin

	Nine Mon Septem				
	2018	2017		Change	
		(Dollars in	thousands		
Cost of revenue	\$ 204,553	\$ 127,873	\$	76,680	60%
Gross margin	54%	55%			

In the nine months ended September 30, 2018, cost of revenue increased by \$76.7 million, or 60%, compared to the same period last year. The increase in cost of revenue was primarily attributable to a \$67.7 million increase in network service providers' costs and a \$4.9 million increase in cloud infrastructure fees both to support the growth in usage of our products.

In the nine months ended September 30, 2018, gross margin percentage declined primarily as a result of higher network service provider costs related to foreign currency fluctuations, certain price adjustments that were made by us in 2017 as a result of the high volume growth of a large customer, an increasing mix of international product usage, and an increase in network service provider fees in certain geographies, which was partially offset by operational improvements and product mix.

Operating Expenses

	Nine Mo Septer	nths En mber 30				
	 2018		2017		Change	
			(Dollars in	thousands		
Research and development	\$ 119,727	\$	87,910	\$	31,817	36%
Sales and marketing	116,520		73,047		43,473	60%
General and administrative	76,213		40,810		35,403	87%
Total operating expenses	\$ 312,460	\$	201,767	\$	110,693	55%

In the nine months ended September 30, 2018, research and development expenses increased by \$31.8 million, or 36%, compared to the same period last year. The increase was primarily attributable to a \$22.3 million increase in personnel costs, net of a \$4.0 million increase in capitalized software development costs, largely as a result of an 25% average increase in our research and development headcount, as we continued to focus on enhancing our existing products and introducing new products, as well as enhancing product management and other technical functions. The increase was also due to a \$2.0 million increase in cloud infrastructure fees to support the staging and development of our products, a \$1.4 million increase in professional services fees, a \$0.9 million increase in outsourced engineering services and a \$1.4 million increase.

In the nine months ended September 30, 2018, sales and marketing expenses increased by \$43.5 million, or 60%, compared to the same period last year. The increase was primarily attributable to a \$32.4 million increase in personnel costs, largely as a result of a 54% average increase in sales and marketing headcount as we continued to expand our sales efforts in the United States and internationally, a \$2.5 million increase in the expenses related to brand awareness programs and a \$2.5 million increase in advertising expenses.

In the nine months ended September 30, 2018, general and administrative expenses increased by \$35.4 million, or 87%, compared to the same period last year. The increase was primarily attributable to the \$13.1 million release of tax liability upon certain obligation settlements and estimate revisions that occurred in the nine months ended September 30, 2017, and did not reoccur in the same period in 2018. The remaining increase was primarily attributable to a \$10.8 million increase in personnel costs, largely as a result of a 37% average increase in general and administrative headcount to support the growth of our business domestically and internationally and a \$6.2 million increase in professional services fees primarily related to our operations as a public company and our on-going litigation matters, including legal settlements/accruals, and a \$1.2 million increase in professional expenses specifically related to our business acquisitions as described in Note 5 to our condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q.

Other Income (Expenses), net

	Nine Mont Septem	d			
	 2018	2017		Change	
		(Dollars in	thousands	s)	
Other income (expenses), net	\$ (3,172)	\$ 1,969	\$	(5,141)	(261)%

In the nine months ended September 30, 2018, net other income (expenses) decreased by \$5.1 million, or 261%, compared to the same period last year, primarily due to \$8.4 million amortization of debt discount and issuance costs related to our 0.25%

convertible senior notes due 2023 (the "Notes") issued in May 2018. This decrease was partially offset by \$5.8 million of interest earned on our marketable securities portfolio.

Liquidity and Capital Resources

To date, our principal sources of liquidity have been (i) the net proceeds of \$155.5 million and \$64.4 million, net of underwriting discounts and offering expenses, from our initial public offering in June 2016 and our follow-on public offering in October 2016, respectively; (ii) the net proceeds we received through private sales of equity securities; (iii) the net proceeds of approximately \$537.0 million, after deducting initial purchaser discounts and debt issuance costs paid or payable by us, from issuance of the Notes, as described in Note 8 to our unaudited condensed consolidated financial statement included elsewhere in this Quarterly Report on Form 10-Q; and (iv) the payments received from customers using our products. From our inception through March 31, 2016, we completed several rounds of equity financing through the sale of our convertible preferred stock for total net proceeds of \$237.1 million.

We believe that our cash, cash equivalents and marketable securities balances, as well as the cash flows generated by our operations will be sufficient to satisfy our anticipated cash needs for working capital and capital expenditures for at least the next 12 months. However, our belief may prove to be incorrect, and we could utilize our available financial resources sooner than we currently expect. Our future capital requirements and the adequacy of available funds will depend on many factors, including those set forth in Part II, Item 1A, "Risk Factors." We may be required to seek additional equity or debt financing in order to meet these future capital requirements. In the event that additional financing is required from outside sources, we may not be able to raise it on terms acceptable to us, or at all. If we are unable to raise additional capital when desired, our business, results of operations and financial condition would be adversely affected.

On October 15, 2018, we, Topaz Merger Subsidiary, Inc. and SendGrid entered into the Merger Agreement, pursuant to which we agreed to acquire SendGrid, subject to the terms and conditions set forth therein. The Merger Agreement provides customary termination rights for both Twilio and SendGrid.

In addition, under the terms of the Merger Agreement, we will pay SendGrid \$120.0 million (the "Twilio Termination Fee") if the merger is not consummated under certain circumstances, including if the Merger Agreement is terminated relating to a change in our board recommendation in connection with an alternative transaction. If the Merger Agreement is terminated relating to a change in board recommendation and/or alternative transaction by SendGrid, SendGrid will be required to pay us \$69.0 million (the "SendGrid Termination Fee").

Further, if the Merger Agreement is terminated for our failure to obtain stockholder approval and the shares of Twilio Class B Common Stock subject to certain proxies fail to be voted in favor of the issuance of our Class A common stock in the merger, we will be required to pay the Twilio Termination Fee. Except as described in the previous sentence, in connection with the termination by either party due to such party's failure to obtain its requisite stockholder approval, such party shall reimburse the other party for all reasonable and documented expenses of the other party up to \$5.0 million as set forth in the Merger Agreement.

If consummated, the acquisition of SendGrid will have a significant impact on our liquidity, financial condition and results of operations.

Cash Flows

The following table summarizes our cash flows for the periods indicated (in thousands):

	Nine Months Ended September 30,			
		2018		2017 (As Adjusted)
Cash provided (used) by operating activities	\$	12,833	\$	(7,054)
Cash used in investing activities		(148,402)		(236,965)
Cash provided by financing activities		503,990		29,002
Effect of exchange rate changes on cash, cash equivalents and restricted cash		105		88
Net increase (decrease) in cash, cash equivalents and restricted cash	\$	368,526	\$	(214,929)

Cash Flows from Operating Activities

In the nine months ended September 30, 2018, cash provided by operating activities consisted primarily of our net loss of \$74.8 million adjusted for non-cash items, including \$61.3 million of stock-based compensation expense, \$18.2 million of depreciation and amortization expense, \$8.4 million amortization of the debt discount and issuance costs related to our Notes and \$4.6 million of cumulative changes in operating assets and liabilities. With respect to changes in operating assets and liabilities, accounts payable and other current liabilities increased \$41.4 million primarily due to increases in transaction volumes. Deferred revenue decreased \$3.4 million primarily due to a \$8.5 million reclassification of customer deposits caused by the implementation of the new accounting guidance for revenue, as described in Note 2 to our unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q. This decrease in deferred revenue was partially offset by increases due to transaction volumes and timing of customer prepayments. Accounts receivable and prepaid expenses increased \$46.6 million, which resulted primarily from the timing of cash receipts from certain of our larger customers, pre-payments for cloud infrastructure fees and certain operating expenses.

In the nine months ended September 30, 2017, cash used in operating activities consisted primarily of our net loss of \$44.8 million adjusted for non-cash items, including \$36.0 million of stock-based compensation expense, \$13.4 million of depreciation and amortization expense and \$12.0 million of cumulative changes in operating assets and liabilities. With respect to changes in operating assets and liabilities, accounts payable and other current liabilities remained unchanged and deferred revenue increased \$3.4 million due to increases in transaction volumes which were partially offset by the \$13.1 million release of tax liability upon certain obligation settlements and estimate revisions. Accounts receivable and prepaid expenses increased \$14.2 million, which resulted primarily from the timing of cash receipts from certain of our larger customers, pre-payments for cloud infrastructure fees and certain operating expenses.

Cash Flows from Investing Activities

In the nine months ended September 30, 2018, cash used in investing activities was \$148.4 million primarily consisting of \$100.0 million of purchases of marketable securities, net of maturities, \$29.7 million of cash paid to acquire other businesses as described in Note 5 to our condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q and \$15.3 million related to capitalized software development costs.

In the nine months ended September 30, 2017, cash used in investing activities was \$237.0 million, primarily consisting of \$193.2 million of purchases of marketable securities, net of maturities, a \$22.6 million payment for a business acquisition, net of cash acquired, \$12.3 million related to capitalized software development costs and a \$8.6 million increase in purchases of capital assets primarily related to the leasehold improvements under our new office lease.

Cash Flows from Financing Activities

In the nine months ended September 30, 2018, cash provided by financing activities was \$504.0 million primarily consisting of \$537.1 million net proceeds from our Notes, net of initial purchaser discounts and issuance costs paid in the period, \$22.6 million proceeds from stock options exercised by our employees and \$4.5 million proceeds from shares issued under our employee stock purchase plan. This was partially offset by a \$58.5 million payment for capped call transactions.

In the nine months ended September 30, 2017, cash provided by financing activities was \$29.0 million, primarily consisting

of \$22.5 million proceeds from stock options exercises by our employees and \$7.4 million proceeds from shares issued under our employee stock purchase plan.

Critical Accounting Policies and Estimates

Our condensed consolidated financial statements are prepared in accordance with generally accepted accounting principles in the United States of America. The preparation of these condensed consolidated financial statements requires us to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue, expenses and related disclosures. We evaluate our estimates and assumptions on an ongoing basis. Our estimates are based on historical experience and various other assumptions that we believe to be reasonable under the circumstances. Our actual results could differ from these estimates.

We believe that the assumptions and estimates associated with revenue recognition, capitalization of our internal-use software development costs and accruals and contingencies have the greatest potential impact on our condensed consolidated financial statements. Therefore, we consider these to be our critical accounting policies and estimates.

Effective January 1, 2018, we adopted Accounting Standards Codification ("ASC") 606, "Revenue from Contracts with Customers", which replaced the existing revenue recognition guidance, ASC 605, and outlines a single set of comprehensive principles for recognizing revenue under U.S. GAAP. Among other things, ASC 606 requires entities to assess the products or services promised in contracts with customers at contract inception to determine the appropriate unit at which to record revenue, which is referred to as a performance obligation. Revenue is recognized when control of the promised products or services is transferred to customers at an amount that reflects the consideration to which the entity expects to be entitled to in exchange for those products or services.

We adopted ASC 606 using the modified retrospective method with cumulative catch-up adjustment to the opening retained earnings as of January 1, 2018. Results for reporting periods beginning after December 31, 2017 are presented under ASC 606, while prior period amounts are not adjusted and continue to be reported in accordance with our historical accounting policies prior to adoption. In adopting the standard, we elected to apply the new guidance only to those contracts which were not completed as of the date of the adoption.

The impact of adopting the new standard on our consolidated financial statements was insignificant. We recorded a net cumulative catch-up adjustment to the beginning retained earnings as of January 1, 2018, of \$0.7 million.

The primary impact relates to the deferral of incremental commission costs of obtaining new contracts. Under ASC 605, we deferred only direct and incremental commission costs to obtain a contract and amortized those costs on a straight-line basis over the term of the related subscription contract. Under the new standard, we defer all incremental commission costs to obtain the contract and amortize these costs on a straight-line basis over the term of benefit of the underlying asset, which was determined to be five years.

The impact on our revenue recognition policies was insignificant. Prior to the adoption of ASC 606, we recognized the majority of our revenue according to the usage by our customers in the period in which that usage occurred. ASC 606 continuous transfer of control to the customer. The impact on the Company's balance sheet presentation includes presenting customer refundable prepayments as customer deposit liabilities, whereas under ASC 605 these were included in deferred revenues.

There was not a significant tax impact to our consolidated statements of operations and consolidated balance sheet relating to the adoption of the new standard as there is a full valuation allowance due to our history of continued losses.

Other than adoption of ASC 606, there have been no material changes to our critical accounting policies and estimates as compared to the critical accounting policies and estimates disclosed in our most recent Annual Report on Form 10-K.

Recently Issued Accounting Guidance

See Note 2 of the notes to our unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q for a summary of recently issued and not yet adopted accounting pronouncements.

Off-Balance Sheet Arrangements

We have not entered into any off-balance sheet arrangements and do not have any holdings in variable interest entities.

Contractual Obligations and Other Commitments

Our principal commitments consist of obligations under our operating leases for office space and contractual commitments to our cloud infrastructure and network service providers. In the three months ended September 30, 2018, we entered into several non-cancellable vendor agreements with a term from one to two years for a total purchase commitments of \$2.2 million. Except for these new commitment, there have been no material changes to our principle commitments described in our most recent Annual Report on Form 10-K.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

We are exposed to certain market risks in the ordinary course of our business. These risks primarily include interest rate sensitivities as follows:

Interest Rate Risk

We had cash and cash equivalents of \$469.1 million and marketable securities of \$276.2 million as of September 30, 2018. Cash and cash equivalents consist of bank deposits and money market funds. Marketable securities consist of U.S. treasury securities, high credit quality corporate debt securities and commercial paper. The cash and cash equivalents and marketable securities are held for working capital purposes. Such interest-earning instruments carry a degree of interest rate risk. To date, fluctuations in interest income have not been significant. The primary objective of our investment activities is to preserve principal while maximizing income without significantly increasing risk. We do not enter into investments for trading or speculative purposes and have not used any derivative financial instruments to manage our interest rate risk exposure. Due to the short-term nature of our investments, we have not been exposed to, nor do we anticipate being exposed to, material risks due to changes in interest rates during any of the periods presented would not have had a material impact on our condensed consolidated financial statements.

In May 2018, we issued \$550.0 million aggregate principal amount of Notes. The fair market value of the Notes is affected by our stock price. The fair value of the Notes will generally increase as our common stock price increases and will generally decrease as our common stock price declines in value. In addition, the fair market value of the Notes are exposed to interest rate risk. Generally, the fair market value of our fixed interest rate Notes will increase as interest rates fall and decrease as interest rates rise. Additionally, on our balance sheet we carry the Notes at face value less unamortized discount and debt issuance cost and we present the fair value for required disclosure purposes only.

Currency Exchange Risks

The functional currency of our foreign subsidiaries is the U.S. dollar and the Euro. Therefore, we are exposed to foreign exchange rate fluctuations as we convert the financial statements of our foreign subsidiaries into U.S. dollars. The local currencies of our foreign subsidiaries are the British pound, the Euro, the Colombian peso, the Singapore dollar, the Hong Kong dollar, the Swedish Krona and the Czech Koruna. Our subsidiaries remeasure monetary assets and liabilities at period-end exchange rates, while non-monetary items are remeasured at historical rates. Revenue and expense accounts are remeasured at the average exchange rate in effect during the year. If there is a change in foreign currency exchange rates, the conversion of our foreign subsidiaries' financial statements into U.S. dollars would result in a realized gain or loss which is recorded in our condensed consolidated statements of operations. We do not currently engage in any hedging activity to reduce our potential exposure to currency fluctuations, although we may choose to do so in the future. A hypothetical 10% change in foreign exchange rates during any of the periods presented would not have had a material impact on our condensed consolidated financial statements.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Our management, with the participation of our principal executive officer and principal financial officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")), as of the end of the period covered by this Quarterly Report on Form 10-Q. Based on such evaluation, our principal executive officer and principal financial officer have concluded that as of September 30, 2018, our disclosure controls and procedures were not effective as a result of the material weakness in our internal control over financial reporting related to accounting for capitalized software development costs as described in Part II, Item 9A, "*Controls and Procedures*", of our most recently filed Annual Report on Form 10-K.

Changes in Internal Control

We are taking actions to remediate the material weakness relating to our internal control over financial reporting, as described in Part II, Item 9A, "Controls and Procedures", of our most recently filed Annual Report on Form 10-K. There were no changes in our internal control over financial reporting in connection with the evaluation required by Rules 13a-15 (d) and 15d-15 (d) of the Exchange Act that occurred during the quarter ended September 30, 2018, that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

Inherent Limitations on Effectiveness of Controls

Our management, including our Chief Executive Officer and Chief Financial Officer, does not expect that our disclosure controls and procedures or our internal control over financial reporting will prevent or detect all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within the Company have been detected. The design of any system of controls also is based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Over time, controls may become inadequate because of the adoges in conditions, or the degree of compliance with the policies or procedures may deteriorate. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings

On April 30, 2015, Telesign Corporation ("Telesign"), filed a lawsuit against us in the United States District Court, Central District of California ("Telesign II"). Telesign alleges that we are infringing three U.S. patents that it holds: U.S. Patent No. 8,462,920 ("920"), U.S. Patent No. 8,687,038 ("038") and U.S. Patent No. 7,945,034 ("034"). The patent infringement allegations in the lawsuit relate to our Account Security products, our two-factor authentication use case and an API tool to find information about a phone number.

On March 28, 2016, Telesign filed a second lawsuit against us in the United States District Court, Central District of California ("Telesign II"), alleging infringement of U.S. Patent No. 9,300,792 ("792") held by Telesign. The '792 patent is in the same patent family as the '920 and '038 patents asserted in Telesign I. On March 8, 2017, in response to a petition by the Company, the U.S. Patent and Trademark Officer ("PTO") issued an order instituting the *inter partes* review for the '792 patent. On March 6, 2018, the PTO found all claims challenged by Twilio in the *inter partes* review unpatentable. On March 15, 2017, Twilio filed a motion to consolidate and stay related cases pending the conclusion of the '792 patent *inter partes* review, which the court granted. The Central District of California court lifted the stay on April 13, 2018. The court transferred the cases to the United States District Court, Northern District of California. With respect to each of the patents asserted in the new-consolidated Telesign I and Telesign II cases ("Telesign I/II"), the consolidated complaint seeks, among other things, to enjoin us from allegedly infringing the patents, along with damages for lost profits. On August 14, 2018, we filed a motion seeking a judgment that all asserted claims are invalid under 35 U.S.C. 101. On October 19, 2018, the court granted our motion and entered judgment in our favor. On November 8, 2018, Telesign filed a notice of appeal. Based on final judgment being entered by the district court in our favor, we do not believe a loss is reasonably possible or estimable.

On December 1, 2016, we filed a patent infringement lawsuit against Telesign in the United States District Court, Northern District of California ("Telesign III"), alleging indirect infringement of United States Patent No. 8,306,021 ("021"), United States Patent No. 8,837,465 ("465"), United States Patent No. 8,755,376 ("376"), United States Patent No. 8,736,051 ("051"), United States Patent No. 8,737,962 ("962"), United States Patent No. 9,270,833 ("833"), and United States Patent No. 9,226,217 ("217"). Telesign filed a motion to dismiss the complaint on January 25, 2017. In two orders, issued on March 31, 2017 and April 17, 2017, the court granted Telesign's motion to dismiss with respect to the '962, '833, '051 and '217 patents, but denied Telesign's motion to dismiss as to the '021, '465 and '376 patents. On August 23, 2017, Telesign petitioned the U.S. Patent and Trademark Office ("U.S. PTO") for *inter partes* review of the '021, '465, and '376 patents. On March 9, 2018, the PTO denied Telesign's petitions for *inter partes* review of the '465 and '376 patents. The Company is seeking a judgment of infringement, a judgment of willful infringement, monetary and injunctive relief, enhanced damages, and an award of costs and expenses against Telesign.

On February 18, 2016, a putative class action complaint was filed in the Alameda County Superior Court in California, entitled Angela Flowers v. Twilio Inc. The complaint alleges that our products permit the interception, recording and disclosure of communications at a customer's request and are in violation of the California Invasion of Privacy Act. The complaint seeks injunctive relief as well as monetary damages. On May 27, 2016, we filed a demurrer to the complaint. On August 2, 2016, the court issued an order denying the demurrer in part and granting it in part, with leave to amend by August 18, 2016 to address any claims under California's Unfair Competition Law. The plaintiff opted not to amend the complaint. Following a period of discovery, the plaintiff filed a motion for class certification on September 20, 2017. On January 2, 2018, the court issued an order granting in part and denying in part the plaintiff's class certification motion. The court certified two classes of individuals who, during specified time periods, allegedly sent or received certain communications involving the accounts of three of our customers that were recorded. The parties held a mediation on August 29, 2018. Following further discussions in coordination with the mediator, on September 27, 2018, the parties reached an agreement in principle to settle the case. The parties are preparing a long-form settlement agreement, and a preliminary class settlement approval hearing has been scheduled for January 15, 2019. We currently estimate our potential liability in the Flowers matter to be \$1.5 million and we reserved this amount in our condensed consolidated balance sheet as of September 30, 2018, presented elsewhere in this Quarterly Report on Form 10-Q.

We intend to vigorously defend ourselves against these lawsuits and believe we have meritorious defenses to each matter in which we are a defendant. However, litigation is inherently uncertain, and any judgment or injunctive relief entered against us or any adverse settlement could negatively affect our business, results of operations and financial condition.

In addition to the litigation discussed above, from time to time, we may be subject to legal actions and claims in the ordinary course of business. We have received, and may in the future continue to receive, claims from third parties asserting, among other



things, infringement of their intellectual property rights. Future litigation may be necessary to defend ourselves, our partners and our customers by determining the scope, enforceability and validity of third-party proprietary rights, or to establish our proprietary rights. The results of any current or future litigation cannot be predicted with certainty, and regardless of the outcome, litigation can have an adverse impact on us because of defense and settlement costs, diversion of management resources, and other factors.

Item 1A. Risk Factors

A description of the risks and uncertainties associated with our business is set forth below. You should carefully consider the risks and uncertainties described below, together with all of the other information in this Quarterly Report on Form 10-Q, including Part I, Item 2, "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our condensed consolidated financial statements and related notes appearing elsewhere in this Quarterly Report on Form 10-Q. The risks and uncertainties described below may not be the only ones we face. If any of the risks actually occur, our business, financial condition, results of operations and prospects could be materially and adversely affected. In that event, the market price of our Class A common stock could decline.

Risks Related to Our Business and Our Industry

The market for our products and platform is new and unproven, may decline or experience limited growth and is dependent in part on developers continuing to adopt our platform and use our products.

We were founded in 2008, and we have been developing and providing a cloud-based platform that enables developers and organizations to integrate voice, messaging and video communications capabilities into their software applications. This market is relatively new and unproven and is subject to a number of risks and uncertainties. We believe that our revenue currently constitutes a significant portion of the total revenue in this market, and therefore, we believe that our future success will depend in large part on the growth, if any, of this market. The utilization of APIs by developers and organizations to build communications functionality into their applications is still relatively new, and developers and organizations may not recognize the need for, or benefits of, our products and platform. Moreover, if they do not recognize the need for and benefits of our products and platform, they may decide to adopt alternative products and services to satisfy some portion of their business needs. In order to grow our business and extend our market position, we intend to focus on educating developers and other potential customers about the benefits of our products and platform. Our ability to expand the market that our products and platform address depends upon a number of factors, including the cost, performance and perceived value associated with such products and platform. The market for our products and platform could fail to grow significantly or there could be a reduction in demand for our products as a result of a lack of developer acceptance, technological conditions and other causes. If our market does not experience significant growth or demand for our products decreases, then our business, results of operations and financial condition could be adversely affected.

We have a history of losses and we are uncertain about our future profitability.

We have incurred net losses in each year since our inception, including net losses of \$74.8 million, \$63.7 million and \$41.3 million in the nine months ended September 30, 2018 and in the years 2017 and 2016, respectively. We had an accumulated deficit of \$324.5 million as of September 30, 2018. We expect to continue to expend substantial financial and other resources on, among other things:

- · investments in our engineering team, the development of new products, features and functionality and enhancements to our platform;
- sales and marketing, including the continued expansion of our direct sales organization and marketing programs, especially for enterprises and for organizations outside of the United States, and expanding
 our programs directed at increasing our brand awareness among current and new developers;
- · expansion of our operations and infrastructure, both domestically and internationally; and
- · general administration, including legal, accounting and other expenses related to being a public company.

These investments may not result in increased revenue or growth of our business. We also expect that our revenue growth rate will decline over time. Accordingly, we may not be able to generate sufficient revenue to offset our expected cost increases and achieve

and sustain profitability. If we fail to achieve and sustain profitability, then our business, results of operations and financial condition would be adversely affected.

We have experienced rapid growth and expect our growth to continue, and if we fail to effectively manage our growth, then our business, results of operations and financial condition could be adversely affected.

We have experienced substantial growth in our business since inception. For example, our headcount has grown from 955 employees on September 30, 2017 to 1,275 employees on September 30, 2018. In addition, we are rapidly expanding our international operations. Our international headcount grew from 202 employees as of September 30, 2017 to 310 employees as of September 30, 2018. We expect to continue to expand our international operations in the future. We have also experienced significant growth in the number of customers, usage and amount of data that our platform and associated infrastructure support. This growth has placed and may continue to place significant demands on our corporate culture, operational infrastructure and management.

We believe that our corporate culture has been a critical component of our success. We have invested substantial time and resources in building our team and nurturing our culture. As we expand our business and mature as a public company, we may find it difficult to maintain our corporate culture while managing this growth. Any failure to manage our anticipated growth and organizational changes in a manner that preserves the key aspects of our culture could hurt our chance for future success, including our ability to recruit and retain personnel, and effectively focus on and pursue our corporate objectives. This, in turn, could adversely affect our business, results of operations and financial condition.

In addition, in order to successfully manage our rapid growth, our organizational structure has become more complex. In order to manage these increasing complexities, we will need to continue to scale and adapt our operational, financial and management controls, as well as our reporting systems and procedures. The expansion of our systems and infrastructure will require us to commit substantial financial, operational and management resources before our revenue increases and without any assurances that our revenue will increase.

Finally, continued growth could strain our ability to maintain reliable service levels for our customers. If we fail to achieve the necessary level of efficiency in our organization as we grow, then our business, results of operations and financial condition could be adversely affected.

Our quarterly results may fluctuate, and if we fail to meet securities analysts' and investors' expectations, then the trading price of our Class A common stock and the value of your investment could decline substantially.

Our results of operations, including the levels of our revenue, cost of revenue, gross margin and operating expenses, have fluctuated from quarter to quarter in the past and may continue to vary significantly in the future. These fluctuations are a result of a variety of factors, many of which are outside of our control, may be difficult to predict and may or may not fully reflect the underlying performance of our business. If our quarterly results of operations fall below the expectations of investors or securities analysts, then the trading price of our Class A common stock could decline substantially. Some of the important factors that may cause our results of operations to fluctuate from quarter to quarter to quarter include:

- our ability to retain and increase revenue from existing customers and attract new customers;
- · fluctuations in the amount of revenue from our Variable Customer Accounts and our larger Base Customer Accounts;
- · our ability to attract and retain enterprises and international organizations as customers;
- · our ability to introduce new products and enhance existing products;
- competition and the actions of our competitors, including pricing changes and the introduction of new products, services and geographies;
- the number of new employees;
- changes in network service provider fees that we pay in connection with the delivery of communications on our platform;
- · changes in cloud infrastructure fees that we pay in connection with the operation of our platform;
- changes in our pricing as a result of our optimization efforts or otherwise;

- · reductions in pricing as a result of negotiations with our larger customers;
- · the rate of expansion and productivity of our sales force, including our enterprise sales force, which has been a focus of our recent expansion efforts;
- · change in the mix of products that our customers use;
- change in the revenue mix of U.S. and international products;
- · changes in laws, regulations or regulatory enforcement, in the United States or internationally, that impact our ability to market, sell or deliver our products;
- the amount and timing of operating costs and capital expenditures related to the operations and expansion of our business, including investments in our international expansion,
- significant security breaches of, technical difficulties with, or interruptions to, the delivery and use of our products on our platform;
- the timing of customer payments and any difficulty in collecting accounts receivable from customers;
- general economic conditions that may adversely affect a prospective customer's ability or willingness to adopt our products, delay a prospective customer's adoption decision, reduce the revenue that we
 generate from the use of our products or affect customer retention;
- changes in foreign currency exchange rates;
- · extraordinary expenses such as litigation or other dispute-related settlement payments;
- sales tax and other tax determinations by authorities in the jurisdictions in which we conduct business;
- · the impact of new accounting pronouncements;
- · expenses in connection with mergers, acquisitions or other strategic transactions; and
- fluctuations in stock-based compensation expense.

The occurrence of one or more of the foregoing and other factors may cause our results of operations to vary significantly. As such, we believe that quarter-to-quarter comparisons of our results of operations may not be meaningful and should not be relied upon as an indication of future performance. In addition, a significant percentage of our operating expenses is fixed in nature and is based on forecasted revenue trends. Accordingly, in the event of a revenue shortfall, we may not be able to mitigate the negative impact on our income (loss) and margins in the short term. If we fail to meet or exceed the expectations of investors or securities analysts, then the trading price of our Class A common stock could fall substantially, and we could face costly lawsuits, including securities class action suits.

Additionally, certain large scale events, such as major elections and sporting events, can significantly impact usage levels on our platform, which could cause fluctuations in our results of operations. We expect that significantly increased usage of all communications platforms, including ours, during certain seasonal and one-time events could impact delivery and quality of our products during those events. We also experienced increased expenses in the second quarter of 2017 due to our developer conference, SIGNAL, which we hosted in the fourth quarter of 2018 and plan to host annually. Such annual and one-time events may cause fluctuations in our results of operations and may impact both our revenue and operating expenses.

If we are not able to maintain and enhance our brand and increase market awareness of our company and products, then our business, results of operations and financial condition may be adversely affected.

We believe that maintaining and enhancing the "Twilio" brand identity and increasing market awareness of our company and products, particularly among developers, is critical to achieving widespread acceptance of our platform, to strengthen our relationships with our existing customers and to our ability to attract new customers. The successful promotion of our brand will depend largely on

our continued marketing efforts, our ability to continue to offer high quality products, our ability to be thought leaders in the cloud communications market and our ability to successfully differentiate our products and platform from competing products and services. Our brand promotion and thought leadership activities may not be successful or yield increased revenue. In addition, independent industry analysts often provide reviews of our products and services, which may significantly influence the perception of our products in the marketplace. If these reviews are negative or not as strong as reviews of our competings' products and services, then our brand may be harmed.

From time to time, our customers have complained about our products, such as complaints about our pricing and customer support. If we do not handle customer complaints effectively, then our brand and reputation may suffer, our customers may lose confidence in us and they may reduce or cease their use of our products. In addition, many of our customers post and discuss on social media about Internet-based products and services, including our products and platform. Our success depends, in part, on our ability to generate positive customer feedback and minimize negative feedback on social media channels where existing and potential customers seek and share information. If actions we take or changes we make to our products or platform upset these customers, then their online commentary could negatively affect our brand and reputation. Complaints or negative publicity about us, our products or our platform could materially and adversely impact our ability to attract and retain customers, our business, results of operations and financial condition.

The promotion of our brand also requires us to make substantial expenditures, and we anticipate that these expenditures will increase as our market becomes more competitive and as we expand into new markets. To the extent that these activities increase revenue, this revenue still may not be enough to offset the increased expenses we incur. If we do not successfully maintain and enhance our brand, then our business may not grow, we may see our pricing power reduced relative to competitors and we may lose customers, all of which would adversely affect our business, results of operations and financial condition.

Our business depends on customers increasing their use of our products, and any loss of customers or decline in their use of our products could materially and adversely affect our business, results of operations and financial condition.

Our ability to grow and generate incremental revenue depends, in part, on our ability to maintain and grow our relationships with existing customers and to have them increase their usage of our products, then our revenue may decline and our results of operations may be harmed. For example, Uber, our largest Base Customer, decreased its usage of our products throughout 2017, which has and may result in decreased revenues from this customers recent historical periods. Customers are charged based on the usage of our products. Most of our customers do not have long-term contractual financial committements to us and, therefore, most of our customers may reduce or cease their use of our products at any time without penalty or termination charges. Customers may refuce or reduce their use of our products for any number of reasons, including if they are not satisfied with our products, the value proposition of our products or our ability to meet their needs and expectations. We cannot accurately predict customers" usage levels and the loss of customers are not satisfied with our products, the value proposition of our products or our ability to meet their needs and expectations. If a significant number of customers customer using, or reduce their usage of our products, the value proposition of our products and marketing expenditures could adversely affect our business, results of operations and financial condition are significant number of customers customers do and expectations. If a significant number of customers customer their usage of our products, the wave proposition of our products and marketing expenditures could adversely affect our business, results of operations and financial condition.

If we are unable to attract new customers in a cost-effective manner, then our business, results of operations and financial condition would be adversely affected.

In order to grow our business, we must continue to attract new customers in a cost-effective manner. We use a variety of marketing channels to promote our products and platform, such as developer events and developer evangelism, as well as search engine marketing and optimization. We periodically adjust the mix of our other marketing programs such as regional customer events, email campaigns, billboard advertising and public relations initiatives. If the costs of the marketing channels we use increase dramatically, then we may choose to use alternative and less expensive channels, which may not be as effective as the channels we currently use. As we add to or change the mix of our marketing strategies, we may need to expand into more expensive channels than those we are currently in, which could adversely affect our business, results of operations and financial condition. We will incur marketing expenses before we are able to recognize any revenue that the marketing initiatives may generate, and these expenses may not result in increased revenue or brand awareness. We have made in the past, and may make in the future, significant expenditures and investments in new marketing campaigns, and we cannot guarantee that any such investments will lead to the cost-effective acquisition of additional customers. If we are unable to maintain effective marketing programs, then our ability to attract new customers could be materially and adversely affected, our advertising and marketing expenses could increase substantially and our results of operations may suffer.



If we do not develop enhancements to our products and introduce new products that achieve market acceptance, our business, results of operations and financial condition could be adversely affected.

Our ability to attract new customers and increase revenue from existing customers depends in part on our ability to enhance and improve our existing products, increase adoption and usage of our products and introduce new products. The success of any enhancements or new products depends on several factors, including timely completion, adequate quality testing, actual performance quality, market-accepted pricing levels and overall market acceptance. Enhancements and new products that we develop may not be introduced in a timely or cost-effective manner, may contain errors or defects, may have interoperability difficulties with our platform or other products or may not achieve the broad market acceptance necessary to generate significant revenue. Furthermore, our ability to increase for our products depends, in part, on the development of new use cases for our products, which is typically driven by our developer community and may be outside of our control. We also have invested, and may continue to invest, in the acquisition of complementary businesses, technologies, services, products and other assets that expand the products that we can offer our customers. We may make these investments without being certain that they will result in products or enhancements that will be accepted by existing or prospective customers. Our ability to generate usage of additional products by our customers may also require increasingly sophisticated and more costly sales efforts and result in a longer sales cycle. If we are unable to successfully enhance our existing products to one evolving customer requirements, increase adoption and usage of our products, develop new products, or if our efforts to increase the usage of our products, evelop new products, or if our efforts to increase the usage of our products develop new products.

If we are unable to increase adoption of our products by enterprises, our business, results of operations and financial condition may be adversely affected.

Historically, we have relied on the adoption of our products by software developers through our self-service model for a significant majority of our revenue, and we currently generate only a small portion of our revenue from enterprise customers. Our ability to increase our customer base, especially among enterprises, and achieve broader market acceptance of our products will depend, in part, on our ability to effectively organize, focus and train our sales and marketing personnel. We have limited experience selling to enterprises and only recently established an enterprise-focused sales force.

Our ability to convince enterprises to adopt our products will depend, in part, on our ability to attract and retain sales personnel with experience selling to enterprises. We believe that there is significant competition for experienced sales professionals with the skills and technical knowledge that we require. Our ability to achieve significant revenue growth in the future will depend, in part, on our ability to recruit, train and retain a sufficient number of experienced sales professionals, particularly those with experience selling to enterprises. In addition, even if we are successful in hiring qualified sales personnel, new hires require significant training and experience before they achieve full productivity, particularly for sales efforts targeted at enterprises and new territories. Our ability to unable to hire or retain sufficient numbers of qualified individuals in the future in the markets where we do business. Because we do not have a long history of targeting our sales efforts at enterprises, we cannot predict whether, or to what extent, our sales will increase as we organize and train our sales force or how long it will take for sales personnel to become productive.

As we seek to increase the adoption of our products by enterprises, we expect to incur higher costs and longer sales cycles. In this market segment, the decision to adopt our products may require the approval of multiple technical and business decision makers, including security, compliance, procurement, operations and IT. In addition, while enterprise customers may quickly deploy our products on a limited basis, before they will commit to deploying our products at scale, they often require extensive education about our products and significant customer support time, engage in protracted pricing negotiations and seek to secure readily available development resources. In addition, sales cycles for enterprises are inherently more complex and less predictable than the sales through our self-service model, and some enterprise customers may not use our products enough to generate revenue that justifies the cost to obtain such customers. In addition, these complex and resources intensive sales efforts could place additional strain on our product and engineering resources. Further, enterprises, including some of our customers, may chevelop their own solutions that do not include our products. They also may deductions in pricing as their usage of our products increases, which could have an adverse impact on our gross margin. As a result of our limited experience selling and marketing to enterprises, our efforts to sell to these potential customers may not be successful. If we are unable to increase the revenue that we derive from enterprises, then our business, results of operations and financial condition may be adversely affected.

If we are unable to expand our relationships with existing Solution Partner customers and add new Solution Partner customers, our business, results of operations and financial condition could be adversely affected.

We believe that the continued growth of our business depends in part upon developing and expanding strategic relationships with Solution Partner customers. Solution Partner customers embed our software products in their solutions, such as software applications for contact centers and sales force and marketing automation, and then sell such solutions to other businesses. When potential customers do not have the available developer resources to build their own applications, we refer them to our network of Solution Partner customers.

As part of our growth strategy, we intend to expand our relationships with existing Solution Partner customers and add new Solution Partner customers. If we fail to expand our relationships with existing Solution Partner customers or establish relationships with new Solution Partner customers in a timely and cost-effective manner, or at all, then our business, results of operations and financial condition could be adversely affected. Additionally, even if we are successful at building these relationships but there are problems or issues with integrating our products into the solutions of these customers, our reputation and ability to grow our business may be harmed.

We rely upon Amazon Web Services to operate our platform, and any disruption of or interference with our use of Amazon Web Services would adversely affect our business, results of operations and financial condition.

We outsource substantially all of our cloud infrastructure to Amazon Web Services ("AWS"), which hosts our products and platform. Our customers need to be able to access our platform that we access, and we are, therefore, vulnerable to service interruptions at AWS. We have experienced, and expect that in the future we may experience interruptions, delays and outages in service and availability due to a variety of factors, including infrastructure changes, human or software errors, website hosting disruptions and capacity constraints. Capacity constraints could be due to a number of potential causes, including technical failures, natural disaters, fraud or security attacks. For instance, in September 2015, AWS suffered a significant outage that had a widespread impact on the ability of our customers to use several of our products. In addition, if our security, or that of AWS, is compromised, or our products or platform are unavailable or our users are unable to use or causes of these performance problems within a period of time acceptable to our customers. It may become increasingly difficult to maintain and improve our platform performance, especially during peak usage times, as our products become more complex and the usage of our products increases. To the extent that we do not effectively address capacity constraints, either through AWS or alternative providers of cloud infrastructure, our business, results of operations and financial condition may be adversely affected. In addition, any changes in service levels from AWS may adversely affect our ability to meet our customers.

The substantial majority of the services we use from AWS are for cloud-based server capacity and, to a lesser extent, storage and other optimization offerings. AWS enables us to order and reserve server capacity in varying amounts and sizes distributed across multiple regions. We access AWS infrastructure through standard IP connectivity. AWS provides us with computing and storage capacity pursuant to an agreement that continues until terminated by either party. AWS may terminate the agreement by providing 30 days prior written notice, and it may in some cases terminate the agreement immediately for cause upon notice. Although we expect that we could receive similar services from other third parties, if any of our arrangements with AWS are terminated, we could experience interruptions on our platform and in our ability to make our products available to customers, as well as delays and additional expenses in arranging alternative cloud infrastructure services.

Any of the above circumstances or events may harm our reputation, cause customers to stop using our products, impair our ability to increase revenue from existing customers, impair our ability to grow our customer base, subject us to financial penalties and liabilities under our service level agreements and otherwise harm our business, results of operations and financial condition.

To deliver our products, we rely on network service providers for our network service.

We currently interconnect with network service providers around the world to enable the use by our customers of our products over their networks. We expect that we will continue to rely heavily on network service providers for these services going forward. Our reliance on network service providers has reduced our operating flexibility, ability to make timely service changes and control quality of service. In addition, the fees that we are charged by network service providers may change daily or weekly, while we do not typically change our customers' pricing as rapidly.

At times, network service providers have instituted additional fees due to regulatory, competitive or other industry related changes that increase our network costs. For example, starting in February 2019, we are expecting one of the major U.S. cellular carriers to introduce a new service offering for A2P, or Application to Person, SMS messages that will add a new fee for A2P SMS messages delivered to its subscribers. While we have historically responded to these types of fee increases through a combination of further negotiating efforts with our network service providers, absorbing the increased costs or changing our prices to customers, there is no guarantee that we will continue to be able to do so in the future without a material negative impact to our business. In the case of this new A2P SMS fee anticipated in early 2019, we plan to pass these fees on to our customers and do not expect this to impact the gross profit dollars per message we receive when customers send SMS messages to this carrier's subscribers. However, mathematically this would still have a negative impact on our gross margins. Additionally, our ability to respond to any new fees may be constrained if all network providers in a particular market impose equivalent fee structures, if the magnitude of the fees is disproportionately large when compared to the underlying prices paid by our customers, or if the market conditions limit our ability to increase the price we charge our customers.

Furthermore, many of these network service providers do not have long-term committed contracts with us and may terminate their agreements with us without notice or restriction. If a significant portion of our network service providers stop providing us with access to their infrastructure, fail to provide these services to us on a cost-effective basis, cease operations, or otherwise terminate these services, the delay caused by qualifying and switching to other network service providers ould be time consuming and costly and could adversely affect our business, results of operations and financial condition. Further, if problems occur with our network service providers, it may cause errors or poor quality communications with our products, and we could encounter difficulty identifying the source of the problem. The occurrence of errors or poor quality communications on our products, whether caused by our platform or a network service provider, may result in the loss of our existing customers or the delay of adoption of our products by potential customers and may adversely affect our business, results of operations and financial condition.

Our future success depends in part on our ability to drive the adoption of our products by international customers.

In the nine months ended September 30, 2018, and in the years 2017 and 2016, we derived 25%, 23% and 16% of our revenue, respectively, from customer accounts located outside the United States. The future success of our business will depend, in part, on our ability to expand our customer base worldwide. While we have been rapidly expanding our sales efforts internationally, our experience in selling our products outside of the United States is limited. Furthermore, our developer-first business model may not be successful or have the same traction outside the United States. As a result, our investment in marketing our products to these potential customers, then our business, results of operations and financial condition may be adversely affected.

We are in the process of expanding our international operations, which exposes us to significant risks.

We are continuing to expand our international operations to increase our revenue from customers outside of the United States as part of our growth strategy. Between September 30, 2017 and September 30, 2018, our international headcount grew from 202 employees to 310 employees. We expect to open additional foreign offices and hire employees to work at these offices in order to reach new customers and gain access to additional technical talent. Operating in international markets requires significant resources and management attention and will subject us to regulatory, economic and political risks in addition to those we already face in the United States. Because of our limited experience with international operations or with developing and managing sales in international markets, our international expansion efforts may not be successful.

In addition, we will face risks in doing business internationally that could adversely affect our business, including:

- exposure to political developments in the United Kingdom ("U.K."), including the planned departure of the U.K. from the European Union (EU) in March 2019, which has created an uncertain political and economic environment, instability for businesses and volatility in global financial markets;
- the difficulty of managing and staffing international operations and the increased operations, travel, infrastructure and legal compliance costs associated with numerous international locations;
- · our ability to effectively price our products in competitive international markets;
- · new and different sources of competition;
- our ability to comply with the General Data Protection Regulation ("GDPR"), which went into effect on May 25, 2018, and the California Consumer Privacy Act, which will be effective as of January 1, 2020;
- · potentially greater difficulty collecting accounts receivable and longer payment cycles;
- · higher or more variable network service provider fees outside of the United States;
- · the need to adapt and localize our products for specific countries;
- · the need to offer customer support in various languages;
- · difficulties in understanding and complying with local laws, regulations and customs in foreign jurisdictions;

- difficulties with differing technical and environmental standards, data privacy and telecommunications regulations and certification requirements outside the United States, which could prevent customers
 from deploying our products or limit their usage;
- export controls and economic sanctions administered by the Department of Commerce Bureau of Industry and Security and the Treasury Department's Office of Foreign Assets Control;

compliance with various anti-bribery and anti-corruption laws such as the Foreign Corrupt Practices Act and United Kingdom Bribery Act of 2010;

- tariffs and other non-tariff barriers, such as quotas and local content rules;
- · more limited protection for intellectual property rights in some countries;
- adverse tax consequences;
- fluctuations in currency exchange rates, which could increase the price of our products outside of the United States, increase the expenses of our international operations and expose us to foreign currency
 exchange rate risk;
- · currency control regulations, which might restrict or prohibit our conversion of other currencies into U.S. dollars;
- · restrictions on the transfer of funds;
- · deterioration of political relations between the United States and other countries; and
- · political or social unrest or economic instability in a specific country or region in which we operate, which could have an adverse impact on our operations in that location.

Also, due to costs from our international expansion efforts and network service provider fees outside of the United States, which generally are higher than domestic rates, our gross margin for international customers is typically lower than our gross margin for domestic customers. As a result, our gross margin may be impacted and fluctuate as we expand our operations and customer base worldwide.

Our failure to manage any of these risks successfully could harm our international operations, and adversely affect our business, results of operations and financial condition.

We currently generate significant revenue from our largest customers, and the loss or decline in revenue from any of these customers could harm our business, results of operations and financial condition.

In the nine months ended September 30, 2018, and in the years 2017 and 2016, our 10 largest Active Customer Accounts generated an aggregate of 18%, 19% and 30% of our revenue, respectively. In addition, a significant portion of our revenue comes from two customers, one of which is a Variable Customer Account.

In the nine months ended September 30, 2018, and in the years 2017 and 2016, WhatsApp, a Variable Customer, accounted for 7%, 6% and 9% of our revenue, respectively. WhatsApp uses our Programmable Voice products and Programmable Messaging products in its applications to verify new and existing users on its service. Our Variable Customer Accounts, including WhatsApp, do not have long-term contracts with us and may reduce or fully terminate their usage of our products at any time without notice, penalty or termination charges. In addition, the usage of our products by WhatsApp and other Variable Customer Accounts may change significantly between periods.

In the nine months ended September 30, 2018, and in the years 2017 and 2016, a second customer, Uber, a Base Customer, accounted for 4%, 8% and 14% of our revenue, respectively. Uber uses our Programmable Messaging products and Programmable Voice products. Uber, or any other one of our Base Customers, could significantly reduce their usage of our products without notice or penalty. Uber decreased its usage of our products throughout 2017 and may continue to do so in the future.

In the event that our large customers do not continue to use our products, use fewer of our products, or use our products in a more limited capacity, or not at all, our business, results of operations and financial condition could be adversely affected.

The market in which we participate is intensely competitive, and if we do not compete effectively, our business, results of operations and financial condition could be harmed.

The market for cloud communications is rapidly evolving, significantly fragmented and highly competitive, with relatively low barriers to entry in some segments. The principal competitive factors in our market include completeness of offering, credibility with developers, global reach, ease of integration and programmability, product features, platform scalability, reliability, security and performance, brand awareness and reputation, the strength of sales and marketing efforts, customer support, as well as the cost of deploying and using our products. Our competitors fall into four primary categories:

- · legacy on-premise vendors, such as Avaya and Cisco;
- · regional network service providers that offer limited developer functionality on top of their own physical infrastructure;
- · smaller software companies that compete with portions of our product line; and
- Software-as-a-service ("SaaS") companies that offer prepackaged applications for a narrow set of use cases.

Some of our competitors and potential competitors are larger and have greater name recognition, longer operating histories, more established customer relationships, larger budgets and significantly greater resources than we do. In addition, they have the operating flexibility to bundle competing products and services at little or no perceived incremental cost, including offering them at a lower price as part of a larger sales transaction. As a result, our competitors may be able to respond more quickly and effectively than we can to new or changing opportunities, technologies, standards or customer requirements. In addition, some competitors may offer products and services that address one or a limited number of functions at lower prices, with greater depth than our products or in different geographies. Our current and potential competitors may develop and market new products and services with comparable functionality to our products, and this could lead to us having to decrease prices in order to remain competitive. Customers utilize our products in many ways and use varying levels of functionality that our products offer or are capable of supporting or enabling within their applications. Customers that use many of the features of our products or use our products to support or enable core functionality for their applications may have difficulty or find it impractical to replace our products with a competitor's products provide, which may limit of functionality may be able to more easily replace our products with competitive offerings. Our customers also may choose to build some of the functionality or provide, which may limit or eliminate their demand for our products.

With the introduction of new products and services and new market entrants, we expect competition to intensify in the future. In addition, some of our customers may choose to use our products and our competitors' products at the same time. Further, customers and consumers may choose to adopt other forms of electronic communications or alternative communication platforms.

Moreover, as we expand the scope of our products, we may face additional competition. If one or more of our competitors were to merge or partner with another of our competitors, the change in the competitive landscape could also adversely affect our ability to compete effectively. In addition, some of our competitors have lower list prices than us, which may be attractive to certain customers even if those products have different or lesser functionality. If we are unable to maintain our current pricing due to competitive pressures, our margins will be reduced and our business, results of operations and financial condition would be adversely affected. In addition, pricing pressures and increased competition generally could result in reduced revenue, reduced margins, increased losses or the failure of our products to achieve or maintain widespread market acceptance, any of which could harm our business, results of operations and financial condition.

We have a limited operating history, which makes it difficult to evaluate our current business and future prospects and increases the risk of your investment.

We were founded and launched our first product in 2008. As a result of our limited operating history, our ability to forecast our future results of operations is limited and subject to a number of uncertainties, including our ability to plan for future growth. Our historical revenue growth should not be considered indicative of our future performance. We have encountered and will encounter risks and uncertainties frequently experienced by growing companies in rapidly changing industries, such as:

- · market acceptance of our products and platform;
- adding new customers, particularly enterprises;
- retention of customers;
- · the successful expansion of our business, particularly in markets outside of the United States;
- competition;



- · our ability to control costs, particularly our operating expenses;
- network outages or security breaches and any associated expenses;
- · foreign currency exchange rate fluctuations;
- · executing acquisitions and integrating the acquired businesses, technologies, services, products and other assets; and
- general economic and political conditions.
- If we do not address these risks successfully, our business, results of operations and financial condition could be adversely affected.

We have limited experience with respect to determining the optimal prices for our products.

We charge our customers based on their use of our products. We expect that we may need to change our pricing from time to time. In the past we have sometimes reduced our prices either for individual customers in connection with long-term agreements or for a particular product. One of the challenges to our pricing is that the fees that we pay to network service providers over whose networks we transmit communications can vary daily or weekly and are affected by volume and other factors that may be outside of our control and difficult to predict. This can result in us incurring increased costs that we may be unable or unwilling to pass through to our customers, which could adversely impact our business, results of operations and financial condition.

Further, as competitors introduce new products or services that compete with ours or reduce their prices, we may be unable to attract new customers or retain existing customers based on our historical pricing. As we expand internationally, we also must determine the appropriate price to enable us to compete effectively internationally. Moreover, enterprises, which are a primary focus for our direct sales efforts, may demand substantial price concessions. In addition, if the mix of products sold changes, including for a shift to IP-based products, then we may need to, or choose to, revise our pricing. As a result, in the future we may be required or choose to reduce our prices or change our pricing model, which could adversely affect our business, results of operations and financial condition.

We typically provide monthly uptime service level commitments of up to 99.95% under our agreements with customers. If we fail to meet these contractual commitments, then our business, results of operations and financial condition could be adversely affected.

Our agreements with customers typically provide for service level commitments. If we suffer extended periods of downtime for our products or platform and we are unable to meet these commitments, then we are contractually obligated to provide a service credit, which is typically 10% of the customer's amounts due for the month in question. In addition, the performance and availability of AWS, which provides our cloud infrastructures is outside of our control and, therefore, we are not in full control of whether we meet our service level commitments. As a result, our business, results of operations and financial condition could be adversely affected if we suffer unscheduled downtime that exceeds the service level commitments we have made to our customers. Any extended service outages coultaged outgative outgates coultaged and financial condition could be adversely affect our business and reputation.

Breaches of our networks or systems, or those of AWS or our service providers, could degrade our ability to conduct our business, compromise the integrity of our products, platform and data, result in significant data losses and the theft of our intellectual property, damage our reputation, expose us to liability to third parties and require us to incur significant additional costs to maintain the security of our networks and data.

We depend upon our IT systems to conduct virtually all of our business operations, ranging from our internal operations and research and development activities to our marketing and sales efforts and communications with our customers and business partners. Individuals or entities may attempt to penetrate our network security, or that of our platform, and to cause harm to our business operations, including by misappropriating our proprietary information or that of our customers, employees and business partners or to cause interruptions of our products and platform. Because the techniques used by such individuals or entities to access, disrupt or sabotage devices, systems and networks change frequently and may not be recognized until launched against a target, we may be unable to anticipate these techniques, and we may not become aware in a timely manner of such a security breach, which could exacerbate any damage we experience. Additionally, we depend upon our employees and contractors to appropriately handle confidential and sensitive data, including customer data, and to deploy our IT resources in a safe and secure manner that does not expose our network systems to security breaches or the loss of data. Any data security incidents, including internal malfeasance by our employees, unauthorized access or usage, virus or similar breach or disruption of us or our service providers, such as AWS or service providers, could result in loss of confidential information, damage to our reputation, loss of customers, litigation, regulatory investigations, fines, penalties and other liabilities. Accordingly, if our cybersecurity measures or those of AWS or our service providers, fail to protect against unauthorized access, attacks (which may include sophisticated cyberattacks), compromise or the mishandling of data by our employees and contractors, then our reputation, business, results of operations and financial condition could be adversely affected.

Defects or errors in our products could diminish demand for our products, harm our business and results of operations and subject us to liability.

Our customers use our products for important aspects of their businesses, and any errors, defects or disruptions to our products and any other performance problems with our products could damage our customers' businesses and, in turn, hurt our brand and reputation. We provide regular updates to our products, which have in the past contained, and may in the future contain, undetected errors, failures, vulnerabilities and bugs when first introduced or released. Real or perceived errors, failures or bugs in our products could result in negative publicity, loss of or delay in market acceptance of our platform, loss of competitive position, lower customer retention or claims by customers for losses sustained by them. In such an event, we may be required, or may choose, for customer relations or other reasons, to expend additional resources in order to help correct the problem. In addition, we may not carry insurance sufficient to compensate us for any losses that may result from claims arising from defects or disruptions in our products. As a result, our reputation and our brand could be harmed, and our business, results of operations and financial condition may be adversely affected.

If we fail to adapt and respond effectively to rapidly changing technology, evolving industry standards, changing regulations, and changing customer needs, requirements or preferences, our products may become less competitive.

The market for communications in general, and cloud communications in particular, is subject to rapid technological change, evolving industry standards, changing regulations, as well as changing customer needs, requirements and preferences. The success of our business will depend, in part, on our ability to adapt and respond effectively to these changes on a timely basis. If we are unable to develop new products that satisfy our customers and provide enhancements and new features for our existing products that keep pace with rapid technological and industry change, our business, results of operations and financial condition could be adversely affected. If new technologies emerge that are able to deliver competitive products and services at lower prices, more efficiently, more conveniently or more securely, such technologies could adversely impact our ability to compete effectively.

Our platform must integrate with a variety of network, hardware, mobile and software platforms and technologies, and we need to continuously modify and enhance our products and platform to adapt to changes and innovation in these technologies. If customers adopt new software platforms or infrastructure, we may be required to develop new versions of our products to work with those new platforms or infrastructure. This development effort may require significant resources, which would adversely affect our business, results of operations and financial condition. Any failure of our products and platform to operate effectively with evolving or new platforms and technologies could reduce the demand for our products. If we are unable to respond to these changes in a cost-effective manner, our products may become less marketable and less competitive or obsolete, and our business, results of operations and financial condition could be adversely affected.

Our reliance on SaaS technologies from third parties may adversely affect our business, results of operations and financial condition.

We rely heavily on hosted SaaS technologies from third parties in order to operate critical internal functions of our business, including enterprise resource planning, customer support and customer relations management services. If these services become unavailable due to extended outages or interruptions, or because they are no longer available on commercially reasonable terms or prices, our expenses could increase. As a result, our ability to manage our operations could be interrupted and our processes for managing our sales process and supporting our customers could be impaired until equivalent services, if available, are identified, obtained and implemented, all of which could adversely affect our business, results of operations and financial condition.

If we are unable to develop and maintain successful relationships with independent software vendors and system integrators, our business, results of operations and financial condition could be adversely affected.

We believe that continued growth of our business depends in part upon identifying, developing and maintaining strategic relationships with independent software vendor ("ISV") development platforms and system integrators. As part of our growth strategy, we plan to further develop product partnerships with ISV development platforms to embed our products as additional distribution channels and also intend to further develop partnerships and specific solution areas with systems integrators. If we fail to establish these relationships in a timely and cost-effective manner, or at all, then our business, results of operations and financial condition could be adversely affected. Additionally, even if we are successful at developing these relationships but there are problems or issues with the integrations or enterprises are not willing to purchase through ISV development platforms, our reputation and ability to grow our business may be adversely affected.



Any failure to offer high quality customer support may adversely affect our relationships with our customers and prospective customers, and adversely affect our business, results of operations and financial condition.

Many of our customers depend on our customer support team to assist them in deploying our products effectively to help them to resolve post-deployment issues quickly and to provide ongoing support. If we do not devote sufficient resources or are otherwise unsuccessful in assisting our customers effectively, it could adversely affect our ability to retain existing customers and could prevent prospective customers from adopting our products. We may be unable to respond quickly enough to accommodate short-term increases in demand for customer support. We also may be unable to modify the nature, scope and delivery of our customer support to compete with changes in the support services provided by our competitors. Increased demand for customer support, without corresponding revenue, could increase costs and adversely affect our business, results of operations and financial condition. Our sales are highly dependent on our business reputation and on positive recommendations from developers. Any failure to maintain high quality customer support, could adversely affect our reputation, business, results of operations and financial condition.

We have been sued, and may, in the future, be sued by third parties for alleged infringement of their proprietary rights, which could adversely affect our business, results of operations and financial condition.

There is considerable patent and other intellectual property development activity in our industry. Our future success depends, in part, on not infringing the intellectual property rights of others. Our competitors or other third parties have claimed and may, in the future, claim that we are infringing upon their intellectual property rights, and we may be found to be infringing upon such rights. For example, on April 30, 2015, Telesign Corporation ("Telesign"), filed a lawsuit against us in the United States District Court, Central District of California ("Telesign I"). Telesign alleges that we are infringing three U.S. patents that it holds: U.S. Patent No. 8,687,038 ("038") and U.S. Patent No. 7,945,034 ("034"). The patent infringement allegations in the lawsuit relate to our Account Security products, our two-factor authentication use case and an API tool to find information about a phone number. Subsequently, on March 28, 2016, Telesign filed a second lawsuit against us in the United States District Court, Central District of California ("Telesign II"), alleging infringement of U.S. Patent No. 9,300,792 ("792") held by Telesign. The '792 patent is in the same patent family as the '920 and '038 patents asserted in Telesign I, and the infringement allegations in Telesign II relate to our Account Security products and our two-factor authentication use case. On October 19, 2018, the court entered judgment in our favor on all asserted claims. On November 8, 2018, Telesign filed a notice of appeal. See the section titled "Item 1. Legal Proceedings." We intend to vigorously defend ourselves against such lawsuits and believe we have meritorious defenses to matters in which we are a defendant. However, litigation is inherently uncertain, and any judgment or injunctive relief entered against us or any adverse settlement could negatively affect our business, results of operations and other interim developments related to the litigation. If securities analysts or investors regard these announcements as negative, the tr

In the future, we may receive claims from third parties, including our competitors, that our products or platform and underlying technology infringe or violate a third party's intellectual property rights, and we may be found to be infringing upon such rights. We may be unaware of the intellectual property rights of others that may cover some or all of our technology. Any claims or litigation could cause us to incur significant expenses and, if successfully asserted against us, could require that we pay substantial damages or ongoing royalty payments, prevent us from offering our products, or require that we comply with other unfavorable terms. We may also be obligated to indemnify our customers or business partners in connection with any such litigation and to obtain licenses or modify our products or platform, which could further exhaust our resources. Even if we were to prevail in the event of claims or litigation us, any claim or litigation regarding intellectual property could be costly and time-consuming and divert the attention of our management and other employees from our business. Patent infringement, trademark infringement, trade secret misappropriation and other intellectual property claims and proceedings brought against us, whether successful or not, could harm our brand, business, results of operations and financial condition.

Indemnity provisions in various agreements potentially expose us to substantial liability for intellectual property infringement and other losses.

Our agreements with customers and other third parties typically include indemnification or other provisions under which we agree to indemnify or otherwise be liable to them for losses suffered or incurred as a result of claims of intellectual property infringement, damages caused by us to property or persons or other liabilities relating to or arising from our products or platform or other acts or omissions. The term of these contractual provisions often survives termination or expiration of the applicable agreement. Large indemnity payments or damage claims from contractual breach could harm our business, results of operations and financial condition. Although typically we contractually limit our liability with respect to such obligations, we may still incur substantial liability related to them. Any dispute with a customer with respect to such obligations could have adverse effects on our relationship with that customer and other current and prospective customers, demand for our products and adversely affect our business, results of operations and financial condition.

We could incur substantial costs in protecting or defending our intellectual property rights, and any failure to protect our intellectual property could adversely affect our business, results of operations and financial condition.

Our success depends, in part, on our ability to protect our brand and the proprietary methods and technologies that we develop under patent and other intellectual property laws of the United States and foreign jurisdictions so that we can prevent others from using our inventions and proprietary information. As of December 31, 2017, in the United States, we had been issued 77 patents, which expire between 2029 and 2036, and had 42 patent applications pending for examination and three pending provisional applications. As of such date, we also had seven issued patents and seven patent applications pending for examination in foreign jurisdictions, all of which are related to U.S. patents and patent applications. There can be no assurance that additional patents will be issued or that any patents that have been issued or that may be issued in the future will provide significant protection for our intellectual property. As of December 31, 2017, we had 14 registered trademarks in the United States and 61 registered trademarks in foreign jurisdictions. If we fail to protect our technology and our business, results of operations and financial condition may be adversely affected.

There can be no assurance that the particular forms of intellectual property protection that we seek, including business decisions about when to file patent applications and trademark applications, will be adequate to protect our business. We could be required to spend significant resources to monitor and protect our intellectual property rights. Litigation may be necessary in the future to enforce our intellectual property rights, determine the validity and scope of our proprietary rights or those of others, or defend against claims of infringement or invalidity. Such litigation could be costly, time-consuming and distracting to management, result in a diversion of significant resources, the narrowing or invalidation of portions of our intellectual property and have an adverse effect on our business, results of operations and financial condition. Our efforts to enforce our intellectual property rights or unterleains and countersuits attacking the validity of our intellectual property rights or a diversion of portions of our patents, counterclaims and countersuits attacking the validity of our intellectual property rights or alleging that we infringe the counterclaimant's own intellectual property. Any of our patents, copyrights, trademarks or other intellectual property rights could be challenged by others or invalidated through administrative process or litigation.

We also rely, in part, on confidentiality agreements with our business partners, employees, consultants, advisors, customers and others in our efforts to protect our proprietary technology, processes and methods. These agreements may not effectively prevent disclosure of our confidential information, and it may be possible for unauthorized parties to copy our software or other proprietary technology or information, or to develop similar software independently without our having an adequate remedy for unauthorized use or disclosure of our confidential information. In addition, others may independently discover our trade secrets and proprietary information, and in these cases we would not be able to assert any trade secret rights against those parties. Costly and time-consuming litigation could be necessary to enforce and determine the scope of our proprietary rights, and failure to obtain or maintain trade secret protection could adversely affect our competitive business position.

In addition, the laws of some countries do not protect intellectual property and other proprietary rights to the same extent as the laws of the United States. To the extent we expand our international activities, our exposure to unauthorized copying, transfer and use of our proprietary technology or information may increase.

We cannot be certain that our means of protecting our intellectual property and proprietary rights will be adequate or that our competitors will not independently develop similar technology. If we fail to meaningfully protect our intellectual property and proprietary rights, our business, results of operations and financial condition could be adversely affected.

Our use of open source software could negatively affect our ability to sell our products and subject us to possible litigation.

Our products and platform incorporate open source software, and we expect to continue to incorporate open source software in our products and platform in the future. Few of the licenses applicable to open source software have been interpreted by courts, and there is a risk that these licenses could be construed in a manner that could impose unanticipated conditions or restrictions on our ability to commercialize our products and platform. Moreover, although we have implemented policies to regulate the use and incorporation of open source software into our products and platform, we cannot be certain that we have not incorporate open source software into our products the platform in a manner that is inconsistent with such policies. If we fail to comply with open source licenses, we may be subject to certain requirements, including requirements that we offer our products that incorporate the open source software for no cost, that we make available source code for modifications or derivative works we create based upon, incorporating or using the open source software and that we license such modifications or derivative works under the terms of applicable open source licenses. If an author or other third party that distributes such open source software were to allege that we had not complied with the conditions of one or more of these licenses, we could be required to incur significant legal expenses defending against such allegations and could be subject to significant damages, enjoined from generating revenue from customers using products that contained the open source software and platform and to re-engineering cannot be accomplished on a timely basis. Any of the orgening could require to to devote additional research and development resources to re-engineer our products or platform or discontinue offering our products to customers in the event re-engineering cannot be accomplished on a timely basis, results of operations and flatform.

We may acquire or invest in companies, which may divert our management's attention and result in debt or dilution to our stockholders. We may be unable to integrate acquired businesses and technologies successfully or achieve the expected benefits of such acquisitions.

We may evaluate and consider potential strategic transactions, including acquisitions of, or investments in, businesses, technologies, services, products and other assets in the future. For example, we have entered into the proposed acquisition of SendGrid. If that transaction closes, our stockholders will incur substantial dilution. For further risks related to the proposed acquisition of SendGrid, please see below under "-Risks Related to the Acquisition of SendGrid." We also may enter into relationships with other businesses to expand our products and platform, which could involve preferred or exclusive licenses, additional channels of distribution, discount pricing or investments in other companies.

Any acquisition, investment or business relationship may result in unforeseen operating difficulties and expenditures. In particular, we may encounter difficulties assimilating or integrating the businesses, technologies, products, personnel or operations of the acquired companies, particularly if the key personnel of the acquired company choose not to work for us, their products or services are not easily adapted to work with our platform, or we have difficultiretation that would otherwise be available for development of our existing business. Moreover, the anticipated benefits of any acquisition, investment or business relationship may not be realized or we may be exposed to unknown risks or liabilities.

Negotiating these transactions can be time consuming, difficult and expensive, and our ability to complete these transactions may often be subject to approvals that are beyond our control. Consequently, these transactions, even if announced, may not be completed. For one or more of those transactions, we may:

- issue additional equity securities that would dilute our existing stockholders;
- · use cash that we may need in the future to operate our business;
- · incur large charges or substantial liabilities;
- · incur debt on terms unfavorable to us or that we are unable to repay;
- encounter difficulties retaining key employees of the acquired company or integrating diverse software codes or business cultures; or
- · become subject to adverse tax consequences, substantial depreciation, or deferred compensation charges.
- The occurrence of any of these foregoing could adversely affect our business, results of operations and financial condition.

We depend largely on the continued services of our senior management and other key employees, the loss of any of whom could adversely affect our business, results of operations and financial condition.

Our future performance depends on the continued services and contributions of our senior management and other key employees to execute on our business plan, to develop our products and platform, to deliver our products to customers, to attract and retain customers and to identify and pursue opportunities. The loss of services of senior management or other key employees could significantly delay or prevent the achievement of our development and strategic objectives. In particular, we depend to a considerable degree on the vision, skills, experience and effort of our co-founder and Chief Executive Officer, Jeff Lawson. None of our executive officers or other senior management personnel is bound by a written employment argreement and any of them may terminate employment with us at any time with no advance notice. On October 24, 2018, we announced that our Chief Financial Officer, Lee Kirkpatrick, will be resigning as our Chief Financial Officer on or about November 12, 2018 and Khozema Shipchandler will be succeeding him as Chief Financial Officer at that time. Though Mr. Kirkpatrick will remain an employee for a transition period, we could experience a delay or disruption in the achievement of our business objectives during the period our new Chief Financial Officer gets up to speed on our business objectives. The loss of the services of other of our senior management personnel would likely involve significant time and costs, and such loss could significantly delay or prevent the achievement of our business objectives. The loss of the services of other of our senior management or other key employees for any reason could adversely affect our business, results of operations and financial condition.

If we are unable to hire, retain and motivate qualified personnel, our business will suffer.

Our future success depends, in part, on our ability to continue to attract and retain highly skilled personnel. We believe that there is, and will continue to be, intense competition for highly skilled management, technical, sales and other personnel with experience in our industry in the San Francisco Bay Area, where our headquarters are located, and in other locations where we maintain offices. We must provide competitive compensation packages and a high quality work environment to hire, retain and motivate employees. If we are unable to retain and motivate our existing employees and attract qualified personnel to fill key positions, we may be unable to manage our business effectively, including the development, marketing and sale of our products, which could adversely affect our business, results of operations and financial condition. To the extent we hire personnel from competitors, we also may be subject to allegations that they have been improperly solicited or divulged proprietary or other confidential information.

Volatility in, or lack of performance of, our stock price may also affect our ability to attract and retain key personnel. Many of our key personnel are, or will soon be, vested in a substantial number of shares of Class A common stock or stock options. Employees may be more likely to terminate their employment with us if the shares they own or the shares underlying their vested options have significantly appreciated in value relative to the original purchase prices of the shares or the exercise prices of the options, or, conversely, if the exercise of the options that they hold are significantly above the trading price of our Class A common stock. If we are unable to retain our employees, our business, results of operations and financial condition could be adversely affected.

Our products and platform and our business are subject to a variety of U.S. and international laws and regulations, including those regarding privacy, data protection and information security, and our customers may be subject to regulations related to the handling and transfer of certain types of sensitive and confidential information. Any failure of our products to comply with or enable our customers and channel partners to comply with applicable laws and regulations would harm our business, results of operations and financial condition.

We and our customers that use our products may be subject to privacy- and data protection-related laws and regulations that impose obligations in connection with the collection, processing and use of personal data, financial data, health or other similar data. The U.S. federal and various state and foreign governments have adopted or proposed limitations on, or requirements regarding, the collection, distribution, use, security and storage of personally identifiable information of individuals. The U.S. Federal Trade Commission and numerous state attorneys general are applying federal and state consumer protection laws to impose standards on the online collection, use and dissemination of data, and to the security measures applied to such data.

Similarly, many foreign countries and governmental bodies, including the European Union ("EU") member states, have laws and regulations concerning the collection and use of personally identifiable information obtained from individuals located in the EU or by businesses operating within their jurisdiction, which are often more restrictive than those in the United States. Laws and regulations in these jurisdictions apply broadly to the collection, use, storage, disclosure and security of personally identifiable information that identifies or may be used to identify an individual, such as names, telephone numbers, email addresses and, in some jurisdictions, IP addresses and other online identifiers.

For example, in April 2016 the EU adopted the General Data Protection Regulation ("GDPR"), which took full effect on May 25, 2018. The GDPR enhances data protection obligations for businesses and requires service providers (data processors) processing personal data on behalf of customers to cooperate with European data protection authorities, implement security measures and keep records of personal data processing activities. Noncompliance with the GDPR can trigger fines equal to or greater of £20 million or 4% of global annual revenues. Given the breadth and depth of changes in data protection obligations, preparing to meet the requirements of GDPR before its effective date has required significant time and resources, including a review of our technology and systems currently in use against the requirements of GDPR. There are also additional EU laws and regulations (and member states implementations thereof) which govern the protection of consumers and of electronic communications. If our efforts to comply with GDPR or other applicable EU laws and regulations are not successful, we may be subject to penalties and fines that would adversely impact our business and results of operations, and our ability to conduct business in the EU could be significantly impaired.

We have in the past relied on the EU-U.S and the Swiss-U.S. Privacy Shield frameworks approved by the European Commission in July 2016 and the Swiss Government in January 2017, respectively, which were designed to allow U.S. corporations to self-certify to the U.S. Department of Commerce and publicly commit to comply with the Privacy Shield requirements to freely import personal data from the EU and Switzerland. However, ongoing legal challenges to these frameworks has resulted in some uncertainty as to their validity. While our Binding Corporate Rules, approved in May 2018, now serve as our primary mechanism to legitimize data transfers from the European Economic Area, we may nonetheless experience hesitancy, reluctance, or refusal by European or multinational customers to continue to use our services due to the potential risk exposure to such customers as a result of a European Union Court of Justice ruling negatively impacting the Privacy Shield frameworks. We and our customers are at risk of enforcement actions taken by an EU data protection authority until such point in time that we are able to ensure that all data transfers from the European Economic Area are legitimized. In addition, as the United Kingdom transitions out of the EU, we may encounter additional complexity with respect to data privacy and data transfers from the U.K.

Furthermore, outside of the EU, we continue to see increased regulation of data privacy and security, including the adoption of more stringent subject matter specific state laws in the United States. For example, on June 28, 2018, California enacted the California Consumer Privacy Act (CCPA), which takes effect on January 1, 2020. The CCPA gives California residents expanded rights to access and delete their personal information, opt out of certain personal information sharing, and receive detailed information about how their personal information is used. The CCPA provides for civil penalties for violations, as well as a private right of action for data breaches that is expected to increase data breach litigation. The CCPA may increase our compliance costs and potential liability. Some observers have noted that the CCPA could mark the beginning of a trend toward more stringent state privacy legislation in the U.S., which could increase our potential liability and adversely affect our business.

As well, we continue to see jurisdictions imposing data localization laws, which require personal information, or certain subcategories of personal information to be stored in the jurisdiction of origin. These regulations may inhibit our ability to expand into those markets or prohibit us from continuing to offer services in those markets without significant additional costs

The uncertainty and changes in the requirements of multiple jurisdictions may increase the cost of compliance, delay or reduce demand for our services, restrict our ability to offer services in certain locations, impact our customers' ability to deploy our solutions in certain jurisdictions, or subject us to sanctions, by national data protection regulators, all of which could harm our business, financial condition and results of operations.

Additionally, although we endeavor to have our products and platform comply with applicable laws and regulations, these and other obligations may be modified, they may be interpreted and applied in an inconsistent manner from one jurisdiction to another, and they may conflict with one another, other regulatory requirements, contractual commitments or our internal practices

We also may be bound by contractual obligations relating to our collection, use and disclosure of personal, financial and other data or may find it necessary or desirable to join industry or other self-regulatory bodies or other privacy- or data protection-related organizations that require compliance with their rules pertaining to privacy and data protection.

We expect that there will continue to be new proposed laws, rules of self-regulatory bodies, regulations and industry standards concerning privacy, data protection and information security in the United States, the European Union and other jurisdictions, and we cannot yet determine the impact such future laws, rules, regulations and standards may have on our business. Moreover, existing U.S. federal and various state and foreign privacy- and data protection-related laws and regulations are evolving and subject to potentially differing interpretations, and various legislative and regulatory bodies may expand current or enact new laws and regulations regarding privacy- and data protection-related matters. Because global laws, regulations and industry standards concerning privacy and data security have continued to develop and evolve rapidly, it is possible that we or our products or platform may not have been, compliant with each such applicable law, regulation and industry standard and compliance with such new laws or to changes to existing laws may impact our business. These developments could adversely affect our business, results of operations and financial condition.

Any failure or perceived failure by us, our products or our platform to comply with new or existing U.S., EU or other foreign privacy or data security laws, regulations, policies, industry standards or legal obligations, or any security incident that results in the unauthorized access to, or acquisition, release or transfer of, personally identifiable information or other customer data may result in governmental investigations, inquiries, enforcement actions and prosecutions, private litigation, fines and penalties, adverse publicity or potential loss of business. For example, on February 18, 2016, a putative class action complaint was filed in the Alameda County Superior Court in California. The complaint alleged that our products permit the interception, recording and disclosure of communications at a customer's request and in violation of the California Invasion of Privacy Act. This litigation, or any other such actions in the future and related penalties could divert management's attention and resources, adversely affect our brand, business, results of operations and financial condition.

Changes in laws and regulations related to the Internet or changes in the Internet infrastructure itself may diminish the demand for our products, and could adversely affect our business, results of operations and financial condition.

The future success of our business depends upon the continued use of the Internet as a primary medium for commerce, communications and business applications. Federal, state or foreign government bodies or agencies have in the past adopted, and may in the future adopt, laws or regulations affecting the use of the Internet as a commercial medium. Changes in these laws or regulations could require us to modify our products and platform in order to comply with these changes. In addition, government agencies or private organizations have imposed and may impose additional taxes, fees or other charges for accessing the Internet or commerce conducted via the Internet. These laws or charges could limit the growth of Internet-related commerce or communications generally, or result in reductions in the demand for Internet-based products and sour products and platform. In addition, the use of

the Internet as a business tool could be adversely affected due to delays in the development or adoption of new standards and protocols to handle increased demands of Internet activity, security, reliability, cost, easeof-use, accessibility and quality of service. The performance of the Internet and its acceptance as a business tool has been adversely affected by "viruses", "worms", and similar malicious programs. If the use of the Internet is reduced as a result of these or other issues, then demand for our products could decline, which could adversely affect our business, results of operations and financial condition.

Certain of our products are subject to telecommunications-related regulations, and future legislative or regulatory actions could adversely affect our business, results of operations and financial condition.

As a provider of communications products, we are subject to existing or potential Federal Communications Commission ("FCC") regulations relating to privacy, Telecommunications Relay Service fund contributions and other requirements. FCC classification of our Internet voice communications products as telecommunications services could result in additional federal and state regulatory obligations. If we do not comply with FCC rules and regulations, we could be subject to FCC enforcement actions, fines, loss of licenses and possibly restrictions on our ability to operate or offer certain of our products. Any enforcement action by the FCC, which may be a public process, would hurt our reputation in the industry, possibly impair our ability to sell our products to customers and could adversely affect our business, results of operations and financial condition.

Our products are subject to a number of FCC regulations and laws that are administered by the FCC. Among others, we must comply (in whole or in part) with:

- the Communications Act of 1934, as amended, which regulates communications services and the provision of such services;
- the Telephone Consumer Protection Act, which limits the use of automatic dialing systems, artificial or prerecorded voice messages, SMS text messages and fax machines;
- · the Communications Assistance for Law Enforcement Act ("CALEA"), which requires covered entities to assist law enforcement in undertaking electronic surveillance;
- · requirements to safeguard the privacy of certain customer information;
- · payment of annual FCC regulatory fees based on our interstate and international revenues;
- · rules pertaining to access to our services by people with disabilities and contributions to the Telecommunications Relay Services fund; and
- · FCC rules regarding the use of customer proprietary network information.

If we do not comply with any current or future rules or regulations that apply to our business, we could be subject to substantial fines and penalties, and we may have to restructure our offerings, exit certain markets or raise the price of our products. In addition, any uncertainty regarding whether particular regulations apply to our business, and how they apply, could increase our costs or limit our ability to grow. Any of the foregoing could adversely affect our business, results of operations and financial condition.

As we continue to expand internationally, we have become subject to telecommunications laws and regulations in the foreign countries where we offer our products. Internationally, we currently offer our products in over 180 countries.

Our international operations are subject to country-specific governmental regulation and related actions that have increased and may continue to increase our costs or impact our products and platform or prevent us from offering or providing our products in certain countries. Certain of our products may be used by customers located in countries where voice and other forms of IP communications may be illegal or require special licensing or in countries on a U.S. embargo list. Even where our products are reportedly illegal or become illegal or where users are located in an embargoed country, users in those countries may be able to continue to use our products in tosse countries on the illegality or embargo. We may be subject to penalties or governmental action if consumers continue to use our products in countries where it is illegal to do so, and any such penalties or governmental action may be costly and may harm our business and damage our brand and reputation. We may be required to incur additional expenses to meet applicable international regulatory requirements or be required to discontinue those services if required by law or if we cannot or will not meet those requirements.

If we are unable to effectively process local number and toll-free number portability provisioning in a timely manner or to obtain or retain direct inward dialing numbers and local or toll-free numbers, our business and results of operations may be adversely affected.

We support local number and toll-free number portability, which allows our customers to transfer their existing phone numbers to us and thereby retain their existing phone numbers when subscribing to our voice products. Transferring existing numbers is a manual process that can take up to 15 business days or longer to complete. A new customer of our voice products must maintain both our voice product and the customer's existing phone service during the number transferring process. Any delay that we experience in transferring these numbers typically results from the fact that we depend on network service providers to transfer these numbers, a process that we do not control, and these network service providers may refuse or substantially delay the transfer of these numbers to us. Local number portability is considered an important feature by many potential customers, and if we fail to reduce any related delays, then we may experience increased difficulty in acquiring new customers.

In addition, our future success depends in part on our ability to procure large quantities of local and toll-free direct inward dialing numbers ("DIDs"), in the United States and foreign countries at a reasonable cost and without restrictions. Our ability to procure, distribute and retain DIDs depends on factors outside of our control, such as applicable regulations, the practices of network service providers that provide DIDs, such as offering DIDs with conditional minimum volume call level requirements, the cost of these DIDs and the level of overall competitive demand for new DIDs. Due to their limited availability, there are certain popular area code prefixes that we generally cannot obtain. Our inability to acquire or retain DIDs for our operations would make our voice and messaging products less attractive to potential customers in the affected local geographic areas. In addition, future growth in our customer base, together with growth in the customer bases of other providers of cloud communications, has increased, which increases our dependence on needing sufficiently large quantities of DIDs. It may become increasingly difficult to source larger quantities of DIDs as we scale and we may need to pay higher costs for DIDs, and DIDs may become subject to more stringent usage conditions. Any of the foregoing could adversely affect our business, results of operations and financial condition.

We face a risk of litigation resulting from customer misuse of our software to send unauthorized text messages in violation of the Telephone Consumer Protection Act.

The actual or perceived improper sending of text messages may subject us to potential risks, including liabilities or claims relating to consumer protection laws. For example, the Telephone Consumer Protection Act of 1991 restricts telemarketing and the use of automatic SMS text messages without proper consent. This has resulted in civil claims against our company and requests for information through thirdparty subpoenas. The scope and interpretation of the laws that are or may be applicable to the delivery of text messages are continuously evolving and developing. If we do not comply with these laws or regulations or if we become liable under these laws or regulations due to the failure of our customers to comply with these laws by obtaining proper consent, we could face direct liability.

We may be subject to governmental export controls and economic sanctions regulations that could impair our ability to compete in international markets due to licensing requirements and could subject us to liability if we are not in compliance with applicable laws.

Certain of our products and services may be subject to export control and economic sanctions regulations, including the U.S. Export Administration Regulations, U.S. Customs regulations and various economic and trade sanctions regulations administered by the U.S. Treasury Department's Office of Foreign Assets Controls. Exports of our products and the provision of our services must be made in compliance with these laws and regulations. Although we take precautions to prevent our products from being provided in violation of such laws, we are aware of previous exports of certain of our products to a small number of persons and organizations that are the subject of U.S. sanctions or located in countries or regions subject to U.S. sanctions. If we fail to comply with these laws and regulations, we and certain of our employees could be subject to substantial civil or criminal penalties, including: the possible loss of export privileges; fines, which may be imposed on us and responsible employees or managers; and, in extreme cases, the incarceration of responsible employees or managers. Obtaining the necessary authorizations, including any required license, for a particular deployment may be time-consuming, is not guaranteed and may result in the delay or loss of sales opportunities. In addition, changes in our products or services, or changes in applicable export or economic sanctions regulations may create delays in the introduction and deployment of our products and services in international markets, or, in some cases, prevent the export of our products or provision of our services to certain countries or end users. Any change in export or economic sanctions regulations, sould also result in decreased use of our products and services, or in our decreased ability to export our products or provide our services could adversely affect our business, results of operations and to prestions. Any decreased use of our products and services or in our ability to export our products or provide our services could adversely affect

Further, we incorporate encryption technology into certain of our products. Various countries regulate the import of certain encryption technology, including through import permitting and licensing requirements, and have enacted laws that could limit our customers' ability to import our products into those countries. Encryption products and the underlying technology may also be subject to export control restrictions. Governmental regulation of encryption technology and regulation of exports of encryption products, or our failure to obtain required approval for our products, when applicable, could harm our international sales and adversely affect our revenue. Compliance with applicable regulatory requirements regarding the export of our products and provision of our services, including with respect to new releases of our products and services in international markets, prevent our customers with international operations from deploying our products and using our services throughout their globally-distributed systems or, in some cases, prevent the export of our provision of our services to some countries altogether.

We may have additional tax liabilities, which could harm our business, results of operations and financial condition.

Significant judgments and estimates are required in determining our provision for income taxes and other tax liabilities. Our tax expense may be impacted, for example, if tax laws change or are clarified to our detriment or if tax authorities successfully challenge the tax positions that we take, such as, for example, positions relating to the arms-length pricing standards for our intercompany transactions and our state sales and use tax positions. In determining the adequacy of income taxes, we assess the likelihood of adverse outcomes that could result if our tax positions were challenged by the Internal Revenue Service ("IRS"), and other tax authorities. Should the IRS or other tax authorities assess additional taxes as a result of examinations, we may be required to record charges to operations that could adversely affect our results of operations and financial condition. We are currently in discussions with certain states regarding prior state sales taxes that we may owe. We have reserved \$22.7 million on our September 30, 2018 balance sheet for these tax payments. The actual exposure could differ materially from our current estimates, and if the actual payments we make to these and other states exceed the accrual in our balance sheet, our results of operations would be harmed.

We could be subject to liability for historical and future sales, use and similar taxes, which could adversely affect our results of operations.

We conduct operations in many tax jurisdictions throughout the United States. In many of these jurisdictions, non-income-based taxes, such as sales and use and telecommunications taxes, are assessed on our operations. We are subject to indirect taxes, and may be subject to certain other taxes, in some of these jurisdictions. Historically, we have not billed or collected these taxes and, in accordance with generally accepted accounting principles in the United States ("U.S. GAAP"), we have recorded a provision for our tax exposure in these jurisdictions when it is both probable that a liability has been incurred and the amount of the exposure can be reasonably estimated. These estimates include several key assumptions, including, but not limited to, the taxability of our products, the jurisdictions in which we believe we have nexus, and the sourcing of revenues to those jurisdictions. In the event these jurisdictions challenge our assumptions and analysis, our actual exposure could differ materially from our current estimates.

We may be subject to scrutiny from state tax authorities in various jurisdictions and may have additional exposure related to our historical operations. Furthermore, certain jurisdictions in which we do not collect such taxes may assert that such taxes are applicable, which could result in tax assessments, penalties and interest, and we may be required to collect such taxes in the future. Such tax assessments, penalties and interest or future requirements may adversely affect our business, results of operations and financial condition.

Effective March 2017, we began collecting telecommunications-based taxes from our customers in certain jurisdictions. Since then, we have added more jurisdictions where we collect these taxes and we expect to continue expanding the number of jurisdictions in which we will collect these taxes in the future. Some customers may question the incremental tax charges and some may seek to negotiate lower pricing from us, which could adversely affect our business, results of operations and financial condition.

Our global operations and structure subject us to potentially adverse tax consequences.

We generally conduct our global operations through subsidiaries and report our taxable income in various jurisdictions worldwide based upon our business operations in those jurisdictions. In particular, our intercompany relationships are subject to complex transfer pricing regulations administered by taxing authorities in various jurisdictions. Also, our tax expense could be affected depending on the applicability of withholding and other taxes (including withholding and indirect taxes on software licenses and related intercompany transactions) under the tax laws of certain jurisdictions in which we have business operations. The relevant revenue and taxing authorities may disagree with positions we have taken generally, or our determinations as to the value of assets sold or acquired or income and expenses attributable to specific jurisdictions. If such a disagreement were to occur, and our position were not sustained, we could be required to pay additional taxes, interest and penalties, which could result in one-time tax charges, higher effective tax rates, reduced eash flows and lower overall profitability of our operations.

Certain government agencies in jurisdictions where we and our affiliates do business have had an extended focus on issues related to the taxation of multinational companies. In addition, the Organization for Economic Co-operation and Development is conducting a project focused on base erosion and profit shifting in international structures, which seeks to establish certain international standards for taxing the worldwide income of multinational companies. As a result of these developments, the tax laws of certain countries in which we and our affiliates do business could change on a prospective or retroactive basis, and any such changes could increase our liabilities for taxes, interest and penalties, and therefore could harm our business, cash flows, results of operations and financial position.

Changes in the U.S. taxation of international business activities or the adoption of other tax reform policies could materially impact our business, results of operations and financial condition.

Changes to U.S. tax laws that may be enacted in the future could impact the tax treatment of our foreign earnings. Due to the expansion of our international business activities, any changes in the U.S. taxation of such activities may increase our worldwide effective tax rate and adversely affect our business, results of operations and financial condition.

If we experience excessive credit card or fraudulent activity, we could incur substantial costs.

Most of our customers authorize us to bill their credit card accounts directly for service fees that we charge. If people pay for our services with stolen credit cards, we could incur substantial third-party vendor costs for which we may not be reimbursed. Further, our customers provide us with credit card billing information online, and we do not review the physical credit cards used in these transactions, which increases our risk of exposure to fraudulent activity. We also incur charges, which we refer to as chargebacks, from the credit card companies from claims that the customer did not authorize the credit card transaction to purchase our services. If the number of unauthorized credit card transactions becomes excessive, we could be assessed substantial fines for excess chargebacks and we could lose the right to accept credit cards for payment.

Our products may also be subject to fraudulent usage, including but not limited to revenue share fraud, domestic traffic pumping, subscription fraud, premium text message scams and other fraudulent schemes. Although our customers are required to set passwords or personal identification numbers to protect their accounts, third parties have in the past been, and may in the future be, able to access and use their accounts through fraudulent means. Furthermore, spannmers attempt to use our products to send targeted and untargeted span messages. We cannot be certain that our efforts to defeat spanning attacks will be successful in eliminating all span messages from being sent using our platform. In addition, a cybersecurity breach of our customers' systems could result in exposure of their authentication credentials, unauthorized access to their accounts or fraudulent calls on their accounts, any of which could adversely affect our business, results of operations and financial condition.

Unfavorable conditions in our industry or the global economy or reductions in spending on information technology and communications could adversely affect our business, results of operations and financial condition.

Our results of operations may vary based on the impact of changes in our industry or the global economy on our customers. Our results of operations depend in part on demand for information technology and cloud communications. In addition, our revenue is dependent on the usage of our products, which in turn is influenced by the scale of business that our customers are conducting. To the extent that weak economic conditions result in a reduced volume of business for, and communications by, our customers and prospective customers, demand for, and use of, our products may decline. Furthermore, weak economic conditions may make it more difficult to collect on outstanding accounts receivable. Additionally, historically, we have generated the substantial majority of our revenue from small and medium-sized businesses, and we expect this to continue for the foreseeable future. Small and medium-sized business may be affected by economic downturns to a greater extent than enterprises, and typically have more limited financial resources, including adversely affect our business, results of operations and financial condition.

We may require additional capital to support our business, and this capital might not be available on acceptable terms, if at all.

We intend to continue to make investments to support our business and may require additional funds. In particular, we may seek additional funds to develop new products and enhance our platform and existing products, expand our operations, including our sales and marketing organizations and our presence outside of the United States, improve our infrastructure or acquire complementary businesses, technologies, services, products and other assets. In addition, we may use a portion of our cash to satisfy tax withholding and remittance obligations related to outstanding restricted stock units. Accordingly, we may need to engage in equity or debt financings to secure additional funds. If we raise additional funds through future issuances of equity or convertible debt securities, our stockholders could suffer significant dilution, and any new equity securities we issue could have rights, preferences and privileges

superior to those of holders of our Class A and Class B common stock. Any debt financing that we may secure in the future could involve restrictive covenants relating to our capital raising activities and other financial and operational matters, which may make it more difficult for us to obtain additional capital and to pursue business opportunities. We may not be able to obtain additional financing on terms favorable to us, if at all. If we are unable to obtain additional financing on terms satisfactory to us when we require it, our ability to continue to support our business growth, scale our infrastructure, develop product enhancements and to respond to business challenges could be significantly impaired, and our business, results of operations and financial condition may be adversely affected.

We face exposure to foreign currency exchange rate fluctuations, and such fluctuations could adversely affect our business, results of operations and financial condition.

As our international operations expand, our exposure to the effects of fluctuations in currency exchange rates grows. While we have primarily transacted with customers and business partners in U.S. dollars, we have transacted with customers in Japan in Japanese Yen and in Europe in Euros and Swedish Kronas. We expect to significantly expand the number of transactions with customers that are denominated in foreign currencies in the future as we continue to expand our business internationally. We also incur expenses for some of our network service provider costs outside of the United States in local currencies and for employee compensation and other operating expenses at our non-U.S. locations in the local currency for such locations. Fluctuations in the exchange rates between the U.S. dollar and other currencies could result in an increase to the U.S. dollar expenses.

In addition, our international subsidiaries maintain net assets that are denominated in currencies ofter than the functional operating currencies of these entities. As we continue to expand our international operations, we become more exposed to the effects of fluctuations in currency exchange rates. Accordingly, changes in the value of foreign currencies relative to the U.S. dollar can affect our results of operations due to transactional and translational remeasurements. As a result of such foreign currency exchange rate fluctuations, it could be more difficult to detect underlying trends in our business and results of operations. In addition, to the extent that fluctuations in currency exchange rates cause our results of operations to differ from our expectations or the expectations of our investors and securities analysts who follow our stock, the trading price of our Class A common stock could be adversely affected.

We do not currently maintain a program to hedge transactional exposures in foreign currencies. However, in the future, we may use derivative instruments, such as foreign currency forward and option contracts, to hedge certain exposures to fluctuations in foreign currency exchange rates. The use of such hedging activities may not offset any or more than a portion of the adverse financial effects of unfavorable movements in foreign exchange rates over the limited time the hedges are in place. Moreover, the use of hedging instruments may introduce additional risks if we are unable to structure effective hedges with such instruments.

Our ability to use our net operating losses to offset future taxable income may be subject to certain limitations.

As of December 31, 2017, we had federal and state net operating loss carryforwards ("NOLs"), of \$229.3 million and \$159.6 million, respectively, due to prior period losses. In general, under Section 382 of the Internal Revenue Code of 1986, as amended (the "Code"), a corporation that undergoes an "ownership change" (generally defined as a greater than 50-percentage-point cumulative change (by value) in the equity ownership of certain stockholders over a rolling three-year period) is subject to limitations on its ability to utilize its pre-change NOLs to offset post-change taxable income. Our existing NOLs may be subject to limitations arising from previous ownership changes, and if we undergo an ownership change in the future, our ability to utilize NOLs could be further limited by Section 382 of the Code. Future changes in our stock ownership, some of which may be outside of our control, could result in an ownership change under Section 382 of the Code.

On December 22, 2017, the U.S. government enacted new tax legislation commonly referred to as the Tax Cuts and Jobs Act (the "Tax Act"). The Tax Act makes broad and complex changes to the U.S. tax code including changes to the uses and limitations of net operating losses. For example, while the Tax Act allows for federal net operating losses incurred in tax years beginning after December 31, 2017 to be carried forward indefinitely, the Tax Act also imposes an 80% limitation on the use of net operating losses that are generated in tax years beginning after December 31, 2017. However, net operating losses generated prior to December 31, 2017 will still have a 20-year carryforward period, but are not subject to the 80% limitation. Furthermore, our ability to utilize our net operating losses is conditioned upon our maintaining profitability in the future and generating U.S. federal taxable income. Since we do not know whether or when we will generate the U.S. federal taxable income necessary to utilize our remaining net operating losses, these net operating loss carryforwards generated prior to December 31, 2017 could expire unused.



If our estimates or judgments relating to our critical accounting policies prove to be incorrect, our results of operations could be adversely affected.

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the amounts reported in the consolidated financial statements and accompanying notes. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, as provided Part I, Item 2, "Management's Discussion and Analysis of Financial Condition and Results of Operations." The results of these estimates form the basis for making judgments about the carrying values of assets, liabilities and equity, and the amount of revenue and expenses that are not readily apparent from other sources. Significant assumptions and estimates used in preparing our consolidated financial statements include those related to revenue recognition, capitalization of our internal-use software development costs and accruals and contingencies. Our results of operations to fall below the expectations to fall below the expectations to fall below the expectations of securities analysts and investors, resulting in a decline in the trading price of our Class A common stock.

Changes in financial accounting standards or practices may cause adverse, unexpected financial reporting fluctuations and affect our results of operations.

A change in accounting standards or practices may have a significant effect on our results of operations and may even affect our reporting of transactions completed before the change is effective. New accounting pronouncements and varying interpretations of accounting pronouncements have occurred and may occur in the future. Changes to existing rules or the questioning of current practices may adversely affect our reported financial results or the way we conduct our business.

For example, a new accounting guidance, Accounting Standards Codification ("ASC") 606, "*Revenue from Contracts with Customers*", became effective on January 1, 2018. We adopted ASC 606 on January 1, 2018, and it resulted in an adjustment of \$0.7 million to our opening retained earnings as of January 1, 2018. Although this accounting guidance did not have a material impact on our consolidated financial statements, there may be other standards that can become effective in the future that may have a material impact on our consolidated financial statements, such as Accounting Standards Update ("ASU") 2016-12, "*Leases*", which will become effective on January 1, 2019, and will result in a significant gross up of our assets and liabilities.

We have identified a material weakness in our internal controls related to the tracking of qualifying internal use software development costs eligible for capitalization; our failure to remediate the identified deficiency may cause us not to be able to accurately or timely report our financial condition or results of operations. If one or more of our internal controls over financial reporting are not effective, it could adversely affect investor confidence in us and our reputation, business or stock price.

In reviewing the accounting for our software development activities, our management has concluded that our internal controls did not effectively track and categorize software development costs between period expenses and capitalization as a fixed asset in accordance with U.S. GAAP. As described under Part II, Item 9A, "Controls and Procedures," of our most recently filed Annual Report on Form 10-K, our management has concluded that the identified deficiencies constitute a material weakness in our internal control over financial reporting. As of September 30, 2018, we have not yet remediated this material weakness. Notwithstanding the foregoing, our management has concluded that the condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q fairly present, in all material respects, our financial position, results of operations and cash flows for the periods presented in this Quarterly Report on Form 10-Q in conformity with GAAP.

A material weakness is a deficiency, or combination of deficiencies, in internal control over financial reporting such that there is a reasonable possibility that a material misstatement of our annual or interim consolidated financial statements will not be prevented or detected on a timely basis. Although we plan to remediate the identified deficiencies as quickly as possible, we cannot, at this time, estimate when such remediation may occur, and our initiatives may not prove successful.

We cannot guarantee that we will not identify additional deficiencies in our internal control over financial reporting in the future. If we are unable to remediate the deficiencies or identify additional deficiencies in the future, our ability to record, process and report financial information accurately, and to prepare financial statements within the time periods specified by the rules and forms of the SEC, could be adversely affected. The occurrence of or failure to remediate a material weakness may adversely affect our reputation and business and the market price of our common stock and any other securities we may issue.

If our goodwill or intangible assets become impaired, we may be required to record a significant charge to earnings.

We review our intangible assets for impairment when events or changes in circumstances indicate the carrying value may not be recoverable. Goodwill is required to be tested for impairment at least annually. As of September 30, 2018, we carried a net \$64.4 million of goodwill and intangible assets related to acquired businesses. An adverse change in market conditions, particularly if such change has the effect of changing one of our critical assumptions or estimates, could result in a change to the estimation of fair value that could result in an impairment charge to our goodwill or intangible assets. Any such charges may adversely affect our results of operations.

Our business is subject to the risks of earthquakes, fire, floods and other natural catastrophic events, and to interruption by man-made problems such as power disruptions, computer viruses, data security breaches or terrorism.

Our corporate headquarters are located in the San Francisco Bay Area, a region known for seismic activity. A significant natural disaster, such as an earthquake, fire or flood, occurring at our headquarters, at one of our other facilities or where a business partner is located could adversely affect our business, results of operations and financial condition. Further, if a natural disaster or man-made problem were to affect our service providers, this could adversely affect the ability of our customers to use our products and platform. In addition, natural disasters and acts of terrorism could cause disruptions in our or our customers' businesses, national economies or the world economy as a whole. We also rely on our network and third-party infrastructure and enterprise applications and internal technology systems for our engineering, sales and marketing, and operations activities. Although we maintain internulent and disaster response plans, in the event of a major disruption caused by a natural disaster or man-made problem, we may be unable to continue our operations and marketing in our development activities, lengthy interruptions in service, breaches of data security and loss of critical data, any of which could adversely affect our business, results of operations and financial condition.

In addition, computer malware, viruses and computer hacking, fraudulent use attempts and phishing attacks have become more prevalent in our industry, have occurred on our platform in the past and may occur on our platform in the future. Though it is difficult to determine what, if any, harm may directly result from any specific interruption or attack, any failure to maintain performance, reliability, security, integrity and availability of our products and technical infrastructure to the satisfaction of our users may harm our reputation and our ability to retain existing users and attract new users.

Risks Related to Ownership of Our Class A Common Stock

The trading price of our Class A common stock has been volatile and may continue to be volatile, and you could lose all or part of your investment.

Prior to our initial public offering in June 2016, there was no public market for shares of our Class A common stock. On June 23, 2016, we sold shares of our Class A common stock to the public at \$15.00 per share. From June 23, 2016, the date that our Class A common stock started trading on the New York Stock Exchange, through October 30, 2018, the trading price of our Class A common stock has ranged from \$22.80 per share to \$88.88 per share. The trading price of our Class A common stock may continue to fluctuate significantly in response to numerous factors, many of which are beyond our control, including:

- \cdot price and volume fluctuations in the overall stock market from time to time;
- volatility in the trading prices and trading volumes of technology stocks;
- changes in operating performance and stock market valuations of other technology companies generally, or those in our industry in particular;
- · sales of shares of our Class A common stock by us or our stockholders;
- failure of securities analysts to maintain coverage of us, changes in financial estimates by securities analysts who follow our company, or our failure to meet these estimates or the expectations of investors;
- the financial projections we may provide to the public, any changes in those projections or our failure to meet those projections;
- · announcements by us or our competitors of new products or services;
- · the public's reaction to our press releases, other public announcements and filings with the SEC;
- · rumors and market speculation involving us or other companies in our industry;
- · actual or anticipated changes in our results of operations or fluctuations in our results of operations;
- · actual or anticipated developments in our business, our competitors' businesses or the competitive landscape generally;

- · litigation involving us, our industry or both, or investigations by regulators into our operations or those of our competitors;
- · developments or disputes concerning our intellectual property or other proprietary rights;
- · announced or completed acquisitions of businesses, products, services or technologies by us or our competitors;
- new laws or regulations or new interpretations of existing laws or regulations applicable to our business;
- · changes in accounting standards, policies, guidelines, interpretations or principles;
- · any significant change in our management; and
- general economic conditions and slow or negative growth of our markets.

In addition, in the past, following periods of volatility in the overall market and the market price of a particular company's securities, securities class action litigation has often been instituted against these companies. This litigation, if instituted against us, could result in substantial costs and a diversion of our management's attention and resources.

Substantial future sales of shares of our Class A common stock could cause the market price of our Class A common stock to decline.

The market price of our Class A common stock could decline as a result of substantial sales of our Class A common stock, particularly sales by our directors, executive officers and significant stockholders, or the perception in the market that holders of a large number of shares intend to sell their shares.

Additionally, the shares of Class A common stock subject to outstanding options and restricted stock unit awards under our equity incentive plans and the shares reserved for future issuance under our equity incentive plans will become eligible for sale in the public market upon issuance, subject to applicable insider trading policies. Certain holders of our Class A common stock have rights, subject to some conditions, to require us to file registration statements covering their shares or to include their shares in registration statements that we may file for our stockholders or ourselves.

The dual class structure of our common stock has the effect of concentrating voting control with those stockholders who held our capital stock prior to the completion of our initial public offering, including our directors, executive officers and their respective affiliates. This limits or precludes your ability to influence corporate matters, including the election of directors, amendments of our organizational documents and any merger, consolidation, sale of all or substantially all of our assets, or other major corporate transaction requiring stockholder approval.

Our Class B common stock has 10 votes per share, and our Class A common stock has one vote per share. As of September 30, 2018, our directors, executive officers and their respective affiliates, held in the aggregate 48.73% of the voting power of our capital stock. Because of the 10-to-one voting ratio between our Class B common stock and Class A common stock, the holders of our Class B common stock collectively will continue to control a majority of the combined voting power of our common stock and therefore be able to control all matters submitted to our stockholders for approval until the earlier of (i) June 28, 2023, or (ii) the date the holders of two-thirds of our Class B common stock elect to convert the Class B common stock to Class A common stock. This concentrated control limits or precludes your ability to influence corporate matters for the foreseeable future, including the election of directors, amendments of our organizational documents, and any merger, consolidation, sale of all or substantially all of our assets, or other major corporate transaction requiring stockholder approval. In addition, this may prevent or discourage unsolicited acquisition proposals or offers for our capital stock that you may feel are in your best interest as one of our stockholders.

Future transfers by holders of Class B common stock will generally result in those shares converting to Class A common stock, subject to limited exceptions, such as certain transfers effected for estate planning purposes. The conversion of Class B common stock to Class A common stock will have the effect, over time, of increasing the relative voting power of those holders of Class B common stock who retain their shares in the long term.

If securities or industry analysts cease publishing research or reports about us, our business or our market, or if they change their recommendations regarding our Class A common stock adversely, the trading price of our Class A common stock and trading volume could decline.

The trading market for our Class A common stock is influenced by the research and reports that securities or industry analysts may publish about us, our business, our market or our competitors. If any of the analysts who cover us change their recommendation regarding our Class A common stock adversely, or provide more favorable relative recommendations about our competitors, the trading price of our Class A common stock adversely, or provide more favorable relative recommendations about our competitors, the trading price of our Class A common stock adversely or four company or fail to regularly publish reports on us, we could lose visibility in the financial markets, which in turn could cause the trading price of our Class A common stock or deline.

Anti-takeover provisions contained in our amended and restated certificate of incorporation and amended and restated bylaws, as well as provisions of Delaware law, could impair a takeover attempt.

Our amended and restated certificate of incorporation, amended and restated bylaws and Delaware law contain provisions which could have the effect of rendering more difficult, delaying, or preventing an acquisition deemed undesirable by our board of directors. Among other things, our amended and restated certificate of incorporation and amended and restated bylaws include provisions:

- authorizing "blank check" preferred stock, which could be issued by our board of directors without stockholder approval and may contain voting, liquidation, dividend and other rights superior to our Class A and Class B common stock;
- · limiting the liability of, and providing indemnification to, our directors and officers;
- · limiting the ability of our stockholders to call and bring business before special meetings;
- providing for a dual class common stock structure in which holders of our Class B common stock have the ability to control the outcome of matters requiring stockholder approval, even if they own
 significantly less than a majority of the outstanding shares of our Class A and Class B common stock, including the election of directors and significant corporate transactions, such as a merger or other
 sale of our company or its assets;
- · providing that our board of directors is classified into three classes of directors with staggered three-year terms;
- · prohibit stockholder action by written consent, which requires all stockholder actions to be taken at a meeting of our stockholders;
- · requiring advance notice of stockholder proposals for business to be conducted at meetings of our stockholders and for nominations of candidates for election to our board of directors; and
- · controlling the procedures for the conduct and scheduling of board of directors and stockholder meetings.

These provisions, alone or together, could delay or prevent hostile takeovers and changes in control or changes in our management.

As a Delaware corporation, we are also subject to provisions of Delaware law, including Section 203 of the Delaware General Corporation law, which prevents certain stockholders holding more than 15% of our outstanding common stock from engaging in certain business combinations without approval of the holders of at least two-thirds of our outstanding common stock not held by such 15% or greater stockholder.

Any provision of our amended and restated certificate of incorporation, amended and restated bylaws or Delaware law that has the effect of delaying, preventing or deterring a change in control could limit the opportunity for our stockholders to receive a premium for their shares of our common stock and could also affect the price that some investors are willing to pay for our Class A common stock.



We do not expect to declare any dividends in the foreseeable future.

We do not anticipate declaring any cash dividends to holders of our common stock in the foreseeable future. Consequently, investors may need to rely on sales of their Class A common stock after price appreciation, which may never occur, as the only way to realize any future gains on their investment. Investors seeking cash dividends should not purchase our Class A common stock.

Risks Related to the Outstanding Notes

Servicing our future debt may require a significant amount of cash, and we may not have sufficient cash flow from our business to pay our indebtedness.

Our ability to make scheduled payments of the principal of, to pay interest on or to refinance our indebtedness, including the Notes, depends on our future performance, which is subject to economic, financial, competitive and other factors beyond our control. Our business may not generate cash flow from operations in the future sufficient to service our debt and make necessary capital expenditures. If we are unable to generate such cash flow, we may be required to adopt one or more alternatives, such as selling assets, restructuring debt or obtaining additional debt financing or equity capital on terms that may be onerous or highly dilutive. Our ability to refinance any future indebtedness will depend on the capital markets and our financial condition at such time. We may not be able to engage in any of these activities or engage in these activities on desirable terms, which could result in a default on our debt obligations. In addition, any of our future debt agreements may contain restrictive covenants that may prohibit us from adopting any of these alternatives. Our failure to comply with these covenants could result in an event of default which, if not cured or waived, could result in the acceleration of our debt.

We may not have the ability to raise the funds necessary for cash settlement upon conversion of the Notes or to repurchase the Notes for cash upon a fundamental change, and our future debt may contain limitations on our ability to pay cash upon conversion of the Notes or to repurchase the Notes.

Subject to limited exceptions, holders of the Notes have the right to require us to repurchase their Notes upon the occurrence of a fundamental change at a fundamental change repurchase price equal to 100% of the principal amount of the Notes to be repurchased, plus accrued and unpaid interest, if any, to, but excluding, the fundamental change repurchase date. In addition, upon conversion of the Notes, unless we elect to deliver solely shares of our Class A common stock to settle such conversion (other than paying cash in lieu of delivering any fractional share), we will be required to make cash payments in respect of the Notes being converted. However, we may not have enough available cash or be able to obtain financing at the time we are required to make repurchases of Notes surrendered therefor or pay any cash amounts due upon conversion. In addition, our ability to repurchase the Notes or to pay cash upon conversions of the Notes or to pay cash payments governing our future indebtedness. Our failure to repurchase Notes at a time when the repurchase is required by the indenture governing the Notes or to pay any cash payable on future conversions of the Notes are equired by such indenture would constitute a default under such indenture. A default under the indenture or the fundamental change itself could also lead to a default under agreements governing our future indebtedness. If the repayment of the related indebtedness were to be accelerated after any applicable notice or grace periods, we may not have sufficient funds to repay the indebtedness and repurchase the Notes or make cash payments upon conversions thereof.

The conditional conversion feature of the Notes, if triggered, may adversely affect our financial condition and operating results.

In the event the conditional conversion feature of the Notes is triggered, holders of Notes will be entitled to convert the Notes at any time during specified periods at their option. If one or more holders elect to convert their Notes, unless we elect to satisfy our conversion obligation by delivering solely shares of our Class A common stock (other than paying cash in lieu of delivering any fractional share), we would be required to settle a portion or all of our conversion obligation through the payment of cash, which could adversely affect our liquidity. In addition, even if holders do not elect to convert their Notes, we could be required under applicable accounting rules to reclassify all or a portion of the outstanding principal of the Notes as a current rather than long-term liability, which would result in a material reduction of our net working capital.

The accounting method for convertible debt securities that may be settled in cash, such as the Notes, could have a material effect on our reported financial results.

Under Financial Accounting Standards Board Accounting Standards Codification 470-20, Debt with Conversion and Other Options, which we refer to as ASC 470-20, an entity must separately account for the liability and equity components of convertible debt instruments (such as the Notes) that may be settled entirely or partially in cash upon conversion in a manner that reflects the issuer's economic interest cost. ASC 470-20 requires the value of the conversion option of the Notes, representing the equity component, to be recorded as additional paid-in capital within stockholders' equity in our consolidated balance sheet and as a

discount to the debt component of the Notes, which reduces their initial debt carrying value reflected as a liability on our balance sheets. The carrying value of the debt component of the Notes, net of the discount recorded, will be accreted up to the principal amount of the Notes from the issuance date until maturity, which will result in non-cash charges to interest expense in our consolidated statement of operations. Accordingly, we will report lower net income or higher net loss in our financial results because ASC 470-20 requires interest to include both the current period's accretion of the debt discount and the instrument's coupon interest, which could adversely affect our reported or future financial results, the trading price of our Class A common stock and the trading price of the Notes.

In addition, under certain circumstances, convertible debt instruments (such as the Notes) that may be settled entirely or partly in cash are currently accounted for utilizing the treasury stock method, the effect of which is that the shares issuable upon conversion of the Notes are not included in the calculation of diluted earnings per share except to the extent that the conversion value of the Notes exceeds their principal amount. Under the treasury stock method, for diluted earnings per share purposes, the transaction is accounted for as if the number of shares of Class A common stock that would be necessary to settle such excess, if we elected to settle such excess in shares, are issued. We cannot be sure that the accounting standards in the future will continue to permit the use of the treasury stock method. If we are unable to use the treasury stock method in accounting for the shares issuable upon conversion of the Notes, then our diluted earnings per share would be adversely affected in periods when we report net income.

The capped call transactions may affect the value of the Notes and our Class A common stock.

In connection with the pricing of the Notes, we entered into privately negotiated capped call transactions with the option counterparties. The capped call transactions are expected generally to reduce the potential dilution to our Class A common stock upon any conversion of the Notes and/or offset any potential cash payments we are required to make in excess of the principal amount of converted Notes, as the case may be, with such reduction and/or offset subject to a cap.

In connection with establishing their initial hedges of the capped call transactions, the option counterparties or their respective affiliates entered into various derivative transactions with respect to our Class A common stock and/or purchased shares of our Class A common stock concurrently with or shortly after the pricing of the Notes.

In addition, the option counterparties and/or their respective affiliates may modify their hedge positions by entering into or unwinding various derivatives with respect to our Class A common stock and/or purchasing or selling our Class A common stock or other securities of ours in secondary market transactions at any time prior to the maturity of the Notes (and are likely to do so during any observation period related to a conversion of Notes). This activity could cause or avoid an increase or a decrease in the market price of our Class A common stock.

We do not make any representation or prediction as to the direction or magnitude of any potential effect that the transactions described above may have on the price of the Notes or our Class A common stock. In addition, we do not make any representation that the option counterparties will engage in these transactions or that these transactions, once commenced, will not be discontinued without notice.

We are subject to counterparty risk with respect to the capped call transactions.

The option counterparties are financial institutions, and we will be subject to the risk that any or all of them might default under the capped call transactions. Our exposure to the credit risk of the option counterparties will not be secured by any collateral. Past global economic conditions have resulted in the actual or perceived failure or financial difficulties of many financial institutions. If an option counterparty becomes subject to insolvency proceedings, we will become an unsecured creditor in those proceedings with a claim equal to our exposure at that time under the capped call transactions with such option counterparty. Our exposure will depend on many factors but, generally, an increase in our exposure will be correlated to an increase in the market price and in the volatility of our Class A common stock. In addition, upon a default by an option counterparty, we may suffer adverse tax consequences and more dilution than we currently anticipate with respect to our Class A common stock. We can provide no assurances as to the financial stability of viability of the option counterparties.

Risks Related to the Acquisition of SendGrid

The proposed acquisition of SendGrid, Inc. may cause a disruption in our business.

The merger agreement related to our acquisition (the "Acquisition") of SendGrid, Inc. ("SendGrid") restricts us from taking certain specified actions without SendGrid's consent until the Acquisition is completed or the merger agreement is terminated,

Table of Contents

including making certain acquisitions. These restrictions may affect our ability to execute our business strategies and attain our financial and other goals and may impact our financial condition, results of operations and cash flows.

The proposed Acquisition could cause disruptions to our business or business relationships, which could have an adverse impact on results of operations. Parties with which we have business relationships may experience uncertainty as to the future of such relationships and may delay or defer certain business decisions, seek alternative relationships with third parties or seek to alter their present business relationships with us. Parties with whom we otherwise may have sought to establish business relationships may seek alternative relationships with third parties.

The pursuit of the Acquisition, the preparation for the integration and the integration of SendGrid may place a significant burden on our management and internal resources. The diversion of management's attention away from day-to-day business concerns and any difficulties encountered in the transition and integration process could adversely affect our financial results.

We have incurred and expect to continue to incur significant costs, expenses and fees for professional services and other transaction costs in connection with the Acquisition. We may also incur unanticipated costs in the integration of SendGrid's business with our business. The substantial majority of these costs will be non-recurring expenses relating to the Acquisition, and many of these costs are payable regardless of whether or not the Acquisition is consummated. We also could be subject to litigation related to the proposed Acquisition, which could result in significant costs and expenses.

Failure to complete the Acquisition in a timely manner or at all could negatively impact the market price of our Class A common stock, as well as our future business and our financial condition, results of operations and cash flows.

We currently anticipate the Acquisition will be completed within the first half of 2019, but we cannot be certain when or if the conditions for the Acquisition will be satisfied or (if permissible under applicable law) waived. The Acquisition cannot be completed until the conditions to closing are satisfied or (if permissible under applicable law) waived, including (i) the adoption of the merger agreement by the requisite vote of SendGrid's stockholders and the approval of the stock issuance by the requisite vote of the Company's stockholders, (ii) receipt of certain required governmental approvals and consents, (iii) the effectiveness of the registration statement with respect to the issuance of the Company Class A common stock in connection with the Acquisition, (iv) authorization of the Class A common stock for listing on New York Stock Exchange, (v) receipt of certain tax opinions by us and SendGrid, including a tax opinion regarding the tax-free treatment of the Acquisition to the SendGrid stockholders for U.S. federal income tax purposes, and (vi) the accuracy of the representations and warranties made by us or SendGrid, as applicable, in the merger agreement. Our obligation to complete the Acquisition is also subject to, among other conditions, the absence of regulatory authorities requiring certain actions on our part.

The satisfaction of the required conditions could delay the completion of the Acquisition for a significant period of time or prevent it from occurring. Further, there can be no assurance that the conditions to the closing of the Acquisition will be satisfied or waived or that the Acquisition will be completed.

If the Acquisition is not completed in a timely manner or at all, our ongoing business may be adversely affected as follows:

- we may experience negative reactions from the financial markets, and our stock price could decline to the extent that the current market price reflects an assumption that the transaction will be completed;
- · we may experience negative reactions from employees, customers, suppliers or other third parties;
- management's focus may have been diverted from pursuing other opportunities that could have been beneficial to us; and
- · our costs of pursuing the Acquisition may be higher than anticipated.

In addition to the above risks, we may be required, under certain circumstances, to pay SendGrid a termination fee equal to \$120 million. If the Acquisition is not consummated, there can be no assurance that these risks will not materialize and will not materially adversely affect our stock price, business, financial conditions, results of operations or cash flows.



Table of Contents

Although we expect that the Acquisition will result in synergies and other benefits to us, we may not realize those benefits because of difficulties related to integration, the achievement of synergies, and other challenges.

We and SendGrid have operated and, until completion of the Acquisition, will continue to operate, independently, and there can be no assurances that our businesses can be combined in a manner that allows for the achievement of substantial benefits. Any integration process may require significant time and resources, and we may not be able to manage the process successfully as our ability to acquire and integrate larger or more complex companies, products, or technology in a successful manner is unproven. If we are not able to successfully integrate SendGrid's businesses with ours or pursue our customer and product strategy successfully, the anticipated benefits of the Acquisition may not be realized fully or may take longer than expected to be realized. Further, it is possible that there could be a loss of our and/or SendGrid's key employees and customers, disruption of either company's or both companies' ongoing businesses or unexpected issues, higher than expected costs and an overall post-completion process that takes longer than originally anticipated. Specifically, the following issues, among others, must be addressed in combining SendGrid's operations with ours in order to realize the anticipated benefits of the Acquisition so the combined company performs as the parties hope:

· combining the companies' corporate functions;

- combining SendGrid's business with our business in a manner that permits us to achieve the synergies anticipated to result from the Acquisition, the failure of which would result in the anticipated benefits of the Acquisition not being realized in the time frame currently anticipated or at all;
- maintaining existing agreements with customers, distributors, providers, talent and vendors and avoiding delays in entering into new agreements with prospective customers, distributors, providers, talent and vendors;
- · determining whether and how to address possible differences in corporate cultures and management philosophies;
- · integrating the companies' administrative and information technology infrastructure;
- · developing products and technology that allow value to be unlocked in the future;
- · evaluating and forecasting the financial impact of the acquisition transaction, including accounting charges; and
- · effecting potential actions that may be required in connection with obtaining regulatory approvals.

In addition, at times the attention of certain members of our management and resources may be focused on completion of the Acquisition and integration planning of the businesses of the two companies and diverted from day-to-day business operations, which may disrupt our ongoing business and the business of the combined company.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Unregistered Sales of Equity Securities

None

Item 3. Defaults Upon Senior Securities

Not applicable

Item 4. Mine Safety Disclosures

Not applicable

Item 5. Other Information

Not applicable

Item 6. Exhibits

The documents listed in the Exhibit Index of this Quarterly Report on Form 10-Q are incorporated by reference or are filed with this Quarterly Report on Form 10-Q, in each case as indicated therein.

EXHIBIT INDEX

Exhibit Number	Description	Filing Date
10.1	Sublease dated as of August 30, 2018, by and between Salesforce.com, Inc. and Twilio Inc.	Filed herewith
10.2	Consent to Sublease Agreement dated as of September 25, 2018, by and among Hudson Rincon Center, LLC, Salesforce.com Inc, and Twilio Inc.	Filed herewith
31.1	Certification of the Chief Executive Officer pursuant to Exchange Act Rules 13a-14(a) and 15d-14(a), as adopted pursuant to Section 302 of the	Filed herewith
	Sarbanes-Oxley Act of 2002	
31.2	Certification of the Chief Financial Officer pursuant to Exchange Act Rules 13a-14(a) and 15d-14(a), as adopted pursuant to Section 302 of the	Filed herewith
	Sarbanes-Oxley, Act of 2002	
32.1*	Certification of the Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the	Furnished herewith
	Sarbanes-Oxley Act of 2002	
101.INS	XBRL Instance Document.	Filed herewith
101.SCH	XBRL Taxonomy Extension Schema Document.	Filed herewith
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document.	Filed herewith
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document.	Filed herewith
101.LAB	XBRL Taxonomy Extension Label Linkbase Document.	Filed herewith
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document.	Filed herewith

* The certifications furnished in Exhibit 32.1 hereto are deemed to accompany this Quarterly Report on Form 10-Q and will not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, except to the extent that the registrant specifically incorporates it by reference.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

 Twilio Inc.

 Date: November 8, 2018

 Date: November 8, 2018

 Date: November 8, 2018

 /s/ LEE KIRKPATRICK

 Lee Kirkpatrick

 Chief Financial Officer (Principal Accounting and Financial Officer)

 78

SUBLEASE

THIS SUBLEASE (this "Sublease") is made as of August 30, 2018 (the "Effective Date") by and between SALESFORCE.COM, INC., a Delaware corporation ("Sublandlord"), and TWILIO INC., a Delaware corporation ("Sublease").

RECITALS

A. Hudson Rincon Center, LLC, a Delaware limited liability company ("Master Landlord"), and Sublandlord are parties to that certain Office Lease, dated as of December 28, 2012 (the "Original Master Lease"), as amended by that certain First Amendment to Office Lease, dated as of April 20, 2016 (the "First Amendment" and, together with the Original Master Lease and as the same may be further amended from time to time in accordance herewith, the "Master Lease") whereby Master Landlord has leased to Sublandlord approximately 259,416 rentable square feet of space (the "Premises"), comprised of a portion of the first (1s) floor (the "First Floor") and the entirety of the second (2nd), third (3nd), fourth (4th) and fifth (5th) floors (the "Second Floor", the "Fourth Floor" and the "Fifth Floor", respectively, and each of the foregoing and the First Floor") of the office building located at 101 Spars Street, San Francisco, California 94105, commonly known as One Rincon (the "Building"). Each Floor is more particularly described in the Master Lease. A redacted copy of the Master Lease is attached hereto as <u>Exhibit A</u>. Capitalized terms used in this Sublease but not defined herein shall have the meanings given to them in the Master Lease.

B. Subtenant desires to sublease from Sublandlord, and Sublandlord desires to sublease to Subtenant, each Premises Component, all on the terms, covenants and conditions set forth in this Sublease.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing Recitals and the mutual covenants and promises of the parties hereinafter set forth, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Sublandlord and Subtenant agree as follows:

1. SUBLEASE.

1.1 Sublease. Upon and subject to the terms, covenants and conditions hereinafter set forth, effective as of the Commencement Date (as defined below), Sublandlord hereby leases to Subtenant, and Subtenant hereby leases from Sublandlord, the Premises.

1.2 Agreed Area of Premises. The parties hereby stipulate that the Premises contain 259,416 rentable square feet, and that such square footage amount is not subject to adjustment or remeasurement by Subtenant or Sublandlord at any time during the Term (as defined below).

2. TERM; POSSESSION.

2.1 Term. Subject to Section 3 below, the term of this Sublease with respect to each Premises Component shall commence on: (a) with respect to the First Floor, the Second Floor and the Third Floor (collectively, the "First Increment"), the date that is the later of December 1, 2018 and the date of Master Landlord's Consent (as defined below) (the "First Increment Commencement Date"), and (b) with respect to the Fourth Floor and the Fifth Floor (collectively, the "Second Increment"),

April 1, 2020 (as the same may be extended as provided hereunder, the "Second Increment Commencement Date"; each such date, the applicable "Commencement Date") and shall expire on (i) with respect to the First Floor and the Third Floor, June 30, 2028, (ii) with respect to the Second Floor, March 31, 2025, and (iii) with respect to the Fourth Floor and Fifth Floor, December 31, 2026 (each such date, the applicable "Expiration Date"), unless sooner terminated pursuant to any provision herein (each, the "Term"). The following reflects the anticipated schedule for certain dates of this Sublease as of the Effective Date and is included for the convenience of the parties only; in the event of a conflict between the following and the terms of this Sublease, the terms of this Sublease shall prevail.

First Increment or Second		Rentable Square		Rent	
Increment	Floor	Footage	Commencement Date	Abatement	Expiration Date
First Increment	1 st Floor	18,062	December 1, 2018	2 months	June 30, 2028
First Increment	2 nd Floor	83,016	December 1, 2018	2 months	March 31, 2025
First Increment	3 rd Floor	74,966	December 1, 2018	2 months	June 30, 2028
Second Increment	4 th Floor	41,758	April 1, 2020	None	December 31, 2026
Second Increment	5 th Floor	41,614	April 1, 2020	None	December 31, 2026

2.2 Sublandlord's Election to Delay Delivery. Notwithstanding the foregoing, Sublandlord may in its sole and absolute discretion delay the Commencement Date applicable to the Fourth Floor and/or the Fifth Floor (in each case, in its entirety) by up to two (2) months in full month increments, exercisable by notice (the "Delay Notice") thereof to Subtenant delivered at any time prior to the date that is one (1) year prior to the Second Increment Commencement Date (and, for the avoidance of doubt, the Commencement Date with respect to any such Premises Component shall be the Commencement Date as so delayed). The Delay Notice shall state the Premises Component for which there will be a delay and the new Commencement Date for each such Premises Component. For the avoidance of doubt, the "Premises" hereunder shall refer to each Premises Component for which the Commencement Date has not occurred as of the date of determination. In the event that delivery of possession of the First Increment Date or delivery of the Second Increment is delayed past the Second Increment Date, then, except as set forth in <u>Section 2.3</u> below, this Sublease shall not be void or voidable (or terminable by Subtenant), the Term shall not be extended, and Sublandlord shall not be liable to Subtenant for any loss or damage resulting from such delay or from the failure of the delivery of possession of the Premises to occur on any particular date.

2.3 Delay in Delivery. Notwithstanding anything to the contrary set forth in this Sublease, if Sublandlord has not tendered possession of a Premises Component to Subtenant on or before the Commencement Date therefor, then for the period commencing on the date following such Commencement Date and ending on the date immediately preceding the date on which Sublandlord

2
2

tenders possession of such Premises Component to Subtenant, Subtenant shall be entitled to a credit against Base Rent for such Premises Component in the amount of (x) one (1) day's Base Rent for such Premises Component for each day of such period and (y) except to the extent that such delay is caused, contributed to or exacerbated by Force Majeure, for each day of such period after the first thirty (30) days, an additional one (1) day's Base Rent for such Premises Component for each day of such period after the first thirty (30) days. An additional one (1) day's Base Rent for such Premises Component for each day of such period after the first thirty (30) days. Moreover, if Sublandlord has not tendered possession of a Premises Component to Subtenant by the date that is ninety (90) days following the Commencement Date therefor (each, a "Trigger Date"), then Subtenant shall have the right to terminate this Subleas with respect to such Premises Component by written notice thereof delivered to Sublandlord at any time within ten (10) Business Days after the applicable Trigger Date and prior to the date on which such Premises Component is delivered to Subtenant.

2.4 <u>Rent Commencement Date</u>. Subject to Section 4.2.1 below, Subtenant shall commence paying Rent (as defined below) for each Premises Component on the applicable Commencement Date.

3. MASTER LANDLORD'S CONSENT.

This Sublease is not and shall not be effective unless and until Master Landlord provides its consent to this Sublease under **Section 14.1** [Transfers] of the Master Lease and this <u>Section 3</u> ("**Master Landlord's Consent**"). If Master Landlord's Consent has not been received within sixty (60) days after the Effective Date despite Sublandlord's commercially reasonable efforts to obtain same, Sublandlord or Subtenant may terminate this Sublease by written notice thereof delivered to the other party at any time prior to receipt of Master Landlord's Consent and in such event neither party shall have any obligations to the other party under this Sublease, except that Sublandlord shall request Master Landlord's consent to this Sublease no later than the date that is two (2) Business Days following the full execution and delivery of this Sublease by Subtenant and Sublandlord. Sublandlord shall reaceuting the Master Landlord's Consent as required. Subtenant shall promptly (and in any event prior to the date due under the Master Lease) pay all administrative fees, costs and expenses (including those incurred prior to the Effective Date) charged by Master Landlord in connection with Master Landlord's Consent that could constitut an amendment to or modification or waiver of any reprivation of the Master Lease. Subtenant shall not be reasonable for Subtenant to fail to execute Master Landlord's Consent on such basis. Without limiting the generality of the foregoing, Subtenant shall not request of Master Landlord's Consent to Alterations of the Master Lease; (ii) Subtenant shall have the right to use the Storage Space pursuant to Section 29.47 [Storage] of the Master Lease; (ii) Subtenant shall have the right to use the Storage Space pursuant to Section 29.47 [Storage] of the Master Lease; (ii) Subtenant shall have the right to use the Storage Space pursuant to Section 29.47 [Storage] of the Master Lease; (ii) Subtenant shall have the right to subtandord pursuant to Section 71.(y)).

^{4.} BASE RENT.

^{4.1} Base Rent. With respect to each Premises Component, commencing on the Commencement Date therefor and continuing through the Expiration Date therefor, Subtenant agrees to

pay Sublandlord as base rent ("**Base Rent**") for such Premises Component the sum of Eighty Four and No/100 Dollars (\$84.00) per rentable square foot therein per annum. The Base Rent shall increase by three percent (3.0%) of the prior amount per rentable square foot on the first (1st) anniversary of the Commencement Date of the first Premises Component delivered to Subtenant and each anniversary thereafter (and, for the avoidance of doubt, the Base Rent for each Premises Component, regardless of whether part of the First Increment or the Second Increment or the Commencement Date or Expiration Date therefor, shall at all times be the same, as increased pursuant to the foregoing, as of the applicable Commencement Date). The parties acknowledge that the Base Rent payable under this Sublease is being paid by Subtenant for the use of the following by Subtenant throughout the Term: (i) the Premises; (ii) as an economic concession to Subtenant, Sublandlord's improvements in the applicable Premises of the Commencement Date therefor; and (iii) as an economic concession to Subtenant, sublandlord's improvements of the Commencement Date therefor.

4.2 Base Rent Abatement and Credit.

4.2.1 <u>Abatement Period</u>. Notwithstanding any provision to the contrary in this Sublease, so long as Subtenant is not then in default under this Sublease beyond any applicable notice and cure periods, Subtenant shall be entitled to abatement of Base Rent for the first two (2) months of the Term of each Premises Component within the First Increment (the "Abatement Period"). If Subtenant is in default under this Sublease during the Abatement Period beyond any applicable notice and cure periods, Subtenant's right to abatement in accordance with the foregoing shall be suspended until such default is cured, and following such cure Subtenant shall then be entitled to the benefit of such abatement for any then-remaining Abatement Period (but not for any abatement during the period of such default). Beginning immediately following the expiration of the Abatement Period and continuing thereafter throughout the balance of the Term, Subtenant shall pay Sublandlord Base Rent in the amounts and at the times set forth herein (i.e., without abatement). By way of example, if the Commencement Date for the First Increment (in its entirety) is December 1, 2018 and for the Second Increment (in its entirety) is April 1, 2020 and Subtenant is not in default under this Sublease beyond any applicable notice and cure periods during the Abatement Period, then Base Rent for the First Increment would abate in accordance with the foregoing for two (2) months and would be first due on February 1, 2019 at the rental rate of \$84.00 per square foot of the First Increment period from February 1, 2019 through November 30, 2019 and, thereafter, Base Rent would increase in accordance with the following schedule:

Sublease Months	Square Footage Then Under Sublease	Monthly Base Rental Rate Per Rentable Square Foot	Monthly Rent
12/01/2018 - 11/30/2019*	176,044	\$ 7.00	\$ 1,232,308.00
12/01/2019 - 3/31/2020*	176,044	\$ 7.210	\$ 1,269,277.24
4/01/2020 - 11/30/2020	259,416	\$ 7.210	\$ 1,870,389.36
12/01/2020 - 11/30/2021*	259,416	\$ 7.426	\$ 1,926,501.04
12/01/2021 - 11/30/2022*	259,416	\$ 7.649	\$ 1,984,296.07
12/01/2022 - 11/30/2023*	259,416	\$ 7.879	\$ 2,043,824.95
12/01/2023 - 11/30/2024*	259,416	\$ 8.115	\$ 2,105,139.70
12/01/2024 - 3/31/20251*	259,416	\$ 8.358	\$ 2,168,293.89
4/01/2025 - 11/30/2025*	176,400	\$ 8.358	\$ 1,474,415.78

12/01/2025 - 11/30/2026*	176,400	\$ 8.609	\$ 1,518,648.25
12/01/2026 - 12/31/2026 ² *	176,400	\$ 8.867	\$ 1,564,207.70
1/01/2027 - 11/30/2027*	93,028	\$ 8.867	\$ 824,915.61
12/01/2027 - 6/30/2028 ³	93,028	\$ 9.133	\$ 849,663.08

* Each December 1 date above is subject to day for day delay if the First Increment is not delivered on December 1, 2018. Subject to adjustment if the entirety of the First Increment is not delivered to Subtenant on December 1, 2018.

** Base Rent is subject to abatement as provided in this Section 4.2.1.

*** Subject to adjustment if the entirety of the Second Increment is not delivered to Subtenant on April 1, 2020.

¹ Expiration of 2nd Floor of First Increment

² Expiration of Second Increment

 3 Expiration of 1^{st} Floor and 3^{rd} Floor of First Increment

The foregoing example is included for the convenience of the parties only; in the event of a conflict between such example and the terms of this Sublease, the terms of this Sublease shall prevail.

4.2.2 Base Rent Credit. In addition to the abatement provided in Section 4.2.1, Subtenant shall receive a one-time credit of Two Hundred Sixty Thousand Dollars (\$260,000) applied against the first (1st) installment of Base Rent due under this Sublease (for the avoidance of doubt, following the expiration of the initial Abatement Period).

4.3 <u>Process for Payment</u>. Each monthly installment of Base Rent shall be payable in advance on the first (1st) day of each calendar month during the Term, except that Base Rent for the First Increment for the first (1st) full month for which Base Rent is due therefor (i.e., the first full month following the Abatement Period) shall be paid upon Subtenant's receipt of Master Landlord's Consent and will be credited against the first (1st) installment(s) of Base Rent due under this Sublease (which, for the avoidance of doubt, will, after application of the \$260,000 credit under <u>Section 4.2.2</u>, be in the aggregate amount of \$972,308). If the Term commences on a day other than the first (1st) day of a calendar month or ends on a day other than the last day of a calendar month, then the Base Rent for the month in which this Sublease commences or ends shall be prorated (and paid at the beginning of each such month) by the number of days this Sublease is in effect during such month based upon a thirty (30) day month, and such partial month's installment shall be paid to later than the commencement of such month. In addition to the Base Rent, Subtenant agrees to pay Additional Rent (as defined in <u>Section 5</u>) as and when the same is due. All "**Rent**" (which shall include Base Rent, Additional Rent and other sums due to Sublandlord) shall be paid to Sublandlord, without prior demand and without any deduction, offset, counterclaim or abatement, in lawful money of the United States of America, via electronic funds transfer to:

Bank Name: SWIFT: ABA Routing Number: Account Name: Account Number:

or to such other person or at such other place as Sublandlord may from time to time designate in writing. Subtenant's covenant to pay Rent shall be independent of every other covenant in this Sublease.

4.4 Late Payments. If Subtenant fails to pay any installment of Rent, then interest shall accrue on such installment at the Interest Rate from the date due until paid and Subtenant shall pay to Sublandlord late charges in accordance with Article 25 [Late Charges] of the Master Lease. Subtenant acknowledges that late payments will cause Sublandlord to incur costs not contemplated by this Sublease, the exact amount of which costs are extremely difficult and impracticable to calculate. The parties hereto agree that the late charge described above represents a fair and reasonable estimate of the extra costs incurred by Sublandlord as a result of such late payment.

4.5 <u>No Waiver</u>. No payment by Subtenant or receipt and acceptance by Sublandlord of a lesser amount than the Rent shall be deemed to be other than part payment of the full amount then due and payable; nor shall any endorsement or statement on any check or any letter accompanying any check, payment of Rent or other payment, be deemed an accord and satisfaction; and Sublandlord may accept, but is not obligated to accept, such part payment without prejudice to Sublandlord's right to recover the balance due and payable or to pursue any other remedy provided in this Sublease or by law. If Sublandlord shall at any time or times accept Rent after it becomes due and payable, such acceptance shall not excuse a subsequent delay or constitute a waiver of Sublandlord's rights hereunder.

5. ADDITIONAL RENT.

5.1 <u>Building Office Pool Direct Expenses</u>. In addition to Base Rent, for each year subsequent to the calendar year 2019 (the "**Sublease Base Year**") during the Term, Subtenant shall pay to Sublandlord, as Additional Rent, Subtenant's Share of Tenant's Share of the annual Building Office Pool Direct Expenses payable by Sublandlord under the Master Lease that are in excess of the amount of Building Office Pool Direct Expenses applicable to the Sublease Base Year, provided that in no event shall any decrease in Direct Expenses for any Expense Year below Direct Expenses for the Sublease Base Year entitle Subtenant to any decrease in Base Rent or any credit against sums due under this Sublease. Said sums (including any estimates of such sums) shall be paid to Sublandlord at the times required pursuant to the terms and conditions of the Master Lease, provided that the first monthly installment of Tenant's Share of any Extimated Excess shall first be due and payable for the calendar month occurring immediately following the expiration of the Base Year. For purposes of the foregoing, "Subtenant's Share" shall be the percentage of the square footage of each Premises Component that is then part of the Premises divided by the aggregate square footage of all Premises Components. The Subtenant's Share of each Premises Component is as follows:

Premises Component	Square Footage of Premises Component	Subtenant's Share
1 st Floor	18,062	6.96%
2 nd Floor	83,016	32%
3 rd Floor	74,966	28.90%
4 th Floor	41,758	16.10%
5 th Floor	41,614	16.04%

5.2 Other Expenses

5.2.1 <u>All Other Expenses</u>. Subtenant shall be responsible to pay all other sums (excluding Base Rent as defined in the Master Lease) that Sublandlord is obligated to pay under the Master Lease with respect to the Premises (but not other sums which result from a default by Sublandlord

under the Master Lease unless caused in whole or in part by the acts or omissions of Subtenant) and shall also be responsible to pay for any additional charges and expenses imposed by Master Landlord pursuant to the terms of the Master Lease with respect to the Premises or this Sublease (including in connection with parking passes) or related exclusively to Subtenant's use and occupancy of the Premises during the Term, said sums shall be paid to Sublandlord at the times required pursuant to the terms and conditions of the Master Lease.

5.2.2 <u>Payment of Additional Rent</u>. Commencing with calendar year 2020, Sublandlord shall provide Subtenant with written notice of Sublandlord's estimate of the amount of Building Office Pool Direct Expenses per month payable pursuant to this <u>Section 5.2</u> for each calendar year during the Term following the Sublease Base Year promptly following the Sublandlord's receipt of Master Landlord's estimate of the Building Office Pool Direct Expenses payable under the Master Lease ("**Master Landlord's Statement**") in accordance with <u>Section 5.3</u> below. Thereafter, Subtenant's Share of the Building Office Pool Direct Expenses payable pursuant to this <u>Section 5.2</u> shall be determined and adjusted in accordance with the provisions of <u>Section 5.2.3</u> below.

5.2.3 <u>Procedure</u>. The determination and adjustment of Building Office Pool Direct Expenses payable hereunder shall be made in accordance with the following procedures:

(1) <u>Delivery of Estimate; Payment</u>. As soon as is practicable after Sublandlord's receipt of Master Landlord's Statement, Sublandlord shall give Subtenant written notice of its estimate of Subtenant's Share of the Building Office Pool Direct Expenses payable under <u>Section 5.2</u> for the ensuing calendar year, which estimate shall be prepared based on Master Landlord's Statement (as Master Landlord's Statement may change from time to time) (the "**Estimated OpEx Statement**"), together with a copy of Master Landlord's Statement. On or before the first day of each month during each calendar year during the Term following the Sublease Base Year, Subtenant shall pay to Sublandlord as Additional Rent Subtenant's Share of one-twelfth (1/12th) of such estimated amount.

(2) Sublandlord's Failure to Deliver Estimate. In the event the Estimated OpEx Statement is not given on or before December of the calendar year preceding the calendar year for which the Estimated OpEx Statement is applicable, then until the calendar month after the Estimated OpEx Statement is delivered by Sublandlord, Subtenant shall continue to pay to Sublandlord monthly, during the ensuing calendar year, estimated payments equal to the amounts payable hereunder during the calendar year just ended. Upon receipt of any Estimated OpEx Statement delivered after the month of December, Subtenant shall (i) commence as of the immediately following calendar month, and continue for the remainder of such calendar year, to pay to Sublandlord within thirty (30) days of the receipt of such updated Estimated OpEx Statement an amount equal to the difference of such monthly installment multiplied by the number of full and partial calendar months of such year preceding the delivery of such updated Estimated OpEx Statement. In each case, Subtenant shall have no less than thirty (30) days to pay such newly estimated amount following receipt of an updated Estimated OpEx Statement.

(3) <u>Year End Reconciliation</u>. Within thirty (30) days following Sublandlord's receipt of the final Statement delivered by Master Landlord to Sublandlord with respect to each calendar year following the Sublease Base Year in accordance with **Article 4** [Additional Rent] of the Master Lease, Sublandlord shall deliver to Subtenant a statement of the adjustment to be made for the calendar year just ended, together with a copy of any corresponding statement received by Sublandlord from Master Landlord (the "**Annual Operating Expense Statement**"). If on the basis of the Annual Operating Expense Statement Subtenant owes an amount that is less than the estimated payments actually

made by Subtenant for the calendar year just ended, Sublandlord shall credit such excess to the next payments of Rent coming due or, if the final Expiration Date hereunder is less than thirty (30) days from the date of such determination, promptly refund such excess to Subtenant. If on the basis of the Annual Operating Expense Statement Subtenant owes an amount that is more than the estimated payments for the calendar year just ended previously made by Subtenant, Subtenant shall pay the deficiency to Sublandlord within thirty (30) days after delivery of the Annual Operating Expense Statement from Sublandlord to Subtenant.

Audit. Following Sublandlord's delivery to Subtenant of an Annual Operating Expense Statement, if Subtenant reasonably disputes any amount set forth in the Master Landlord's Statement upon which Sublandlord's Annual Operating Expense Statement is based, and provided that: (A) Sublandlord has not previously exercised its right under the Master Lease to inspect Master Landlord's books and records with respect to Master Landlord's Statement (and such right is otherwise available under the Master Lease); and (B) Subtenant notifies Sublandlord, within one hundred and eighty (180) days following Sublandlord's delivery of Sublandlord's Annual Operating Expense Statement to Subtenant (and in any event prior to the date that an inspection is permitted under Section 4.6 [Landlord's Books and Records] of the Master Lease), of Subtenant's desire to cause Sublandlord to exercise such inspection right, then Sublandlord will notify Master Landlord pursuant to the provisions of Section 4.6 [Landlord's Books and Records] of the Master Lease that Sublandlord desires to cause Landlord's records with respect to the Annual Operating Expense Statement to be inspected (with respect to such disputed amounts only). The independent, certified public accounting firm for such inspection shall be designated by Subtenant (and approved by Sublandlord) and shall meet the requirements of Section 4.6 [Landlord's Books and Records] of the Master Lease. Sublandlord will promptly provide Subtenant with any report prepared by such accounting firm. All costs associated with any such inspection (including any cost or charges imposed by Master Landlord) shall be solely borne by Subtenant as Additional Rent hereunder unless as the result of such audit, either: (x) Sublandlord is reimbursed for such costs by Master Landlord pursuant to Section 4.6 [Landlord's Books and Records] of the Master Lease, or (y) Sublandlord also receives the benefit under the Master Lease of any cost savings revealed by such audit, in which case Sublandlord shall share equally in the cost of such audit (provided that in no event shall Sublandlord be required to pay any costs of such audit to the extent not reimbursed or otherwise benefitted from as provided in (x) and (y) above). If, on the basis of any such inspection or on the basis of an inspection initiated solely by Sublandlord, Sublandlord receives a credit of Building Office Pool Direct Expenses previously paid by Sublandlord to Master Landlord that is attributable to an overpayment by Sublandlord of Building Office Pool Direct Expenses, then any such credit shall be applied (i) first, to Sublandlord, to the extent of any costs associated with such inspection that have been incurred by Sublandlord and that have not to date been reimbursed to Sublandlord by Subtenant, (ii) second, to Subtenant, to the extent of the costs incurred with respect to such inspection that Subtenant has previously paid (either directly or by reimbursing Sublandlord) and (iii) third, as between Sublandlord and Subtenant as may be equitably necessary to adjust for any overpayment of Building Office Pool Direct Expenses by either Sublandlord or Subtenant, in any case after deducting costs due to Sublandlord under this Section 5.2.3 (including, if pursuant to an inspection initiated solely by Sublandlord, the costs thereof shall be deducted from any amount that would otherwise be applied for the benefit of Subtenant). For avoidance of doubt, if, as of the date that Subtenant notifies Sublandlord of Subtenant's desire to require Sublandlord to review Master Landlord's books and records, Sublandlord has already notified Landlord of Sublandlord's exercise of the inspection right described in Section 4.6 [Landlord's Books and Records] of the Master Lease, then Subtenant shall have no independent right to require any such inspection of Master Landlord's books and records, but Sublandlord agrees to: (A) promptly provide Subtenant with any report prepared by Sublandlord's accounting firm or accountant and, (B) equitably allocate to Subtenant any credit of Building Office Pool Direct Expenses attributable to an overpayment by Sublandlord of Building Office Pool Direct Expenses to the next payments of Rent and Building Office Pool Direct Expenses coming due, following Sublandlord's recovery of all costs associated with

such inspection, as may then be equitable given any corresponding overpayment of Building Office Pool Direct Expenses by Subtenant and in such event, Subtenant shall have no obligation to reimburse Sublandlord for the costs of the inspection initiated solely by Sublandlord. If Sublandlord receives a credit of Building Office Pool Direct Expenses at any time that relates to the Premises during the Term, whether pursuant to an inspection conducted pursuant to Section 4.6 [Landlord's Books and Records] of the Master Lease or otherwise, Subtenant shall be entitled to its proportionate share of such credit (subject to this Section 5.2.3).

5.3 <u>Landlord Statements</u>. Within a commercially reasonable period of time following Sublandlord's actual receipt from Master Landlord of a Statement or Estimate Statement under the Master Lease, Sublandlord or its third party consultants or other agents shall revise such Statement or Estimate Statement, as applicable, to account for the difference between the Sublease Base Year hereunder and the Base Year as defined in **Section 4.2.1** of the Master Lease (which difference the parties agree constitutes an economic concession to Subtenant) and deliver such revised version to Subtenant.

5.4 <u>Gross Receipts and Rental Tax</u>. Subtenant shall be liable to Sublandlord for any gross receipts and rental tax payable by Sublandlord to the City and County of San Francisco, California based upon the Rent payable hereunder.

5.5 <u>Additional Rent</u>. All sums payable pursuant to this <u>Section 5</u> and all other costs and expenses that Subtenant assumes or agrees to pay pursuant to this Sublease (other than Base Rent) shall be considered "Additional Rent" payable under this Sublease, and Sublandlord shall have all rights and remedies available hereunder for the failure to pay such Additional Rent.

6. SECURITY DEPOSIT; LETTER OF CREDIT.

6.1 Security Deposit. Subtenant shall, within ten (10) Business Days following Subtenant's execution and delivery of this Sublease, deposit a security deposit (the "Security Deposit") in the amount of Fourteen Million Six Hundred Sixty Seven Thousand Three Hundred Sixty Dollars (\$14,667,360) as security for Subtenant's faithful performance of Subtenant's obligations under this Sublease. If Subtenant is not in default of this Sublease as of the fourth (4th) anniversary of the earliest Commencement Date for the First Increment (or, if Subtenant is in default on such date, on the date after such anniversary that such default has been cured), the Security Deposit shall be reduced by fifty percent (50%) (i.e., from Fourteen Million Six Hundred Sixty Seven Thousand Three Hundred Sixty Dollars (\$1,3680) (the "Burn Down Right"). If Subtenant fails to pay Rent or otherwise defaults under this Sublease, Sublandlord may use the Security Deposit for the payment of any amount due Sublandlord the amount so used or applied so as to restore the Security Deposit to the amount set forth in this <u>Section 6.1</u>. Sublenant shall not demand pay Sublandlord. No part of the Security Deposit shall be considered to be held in trust, to bear interest, or to be prepayment for any monies to be paid by Subtenant under this Sublease. In the event of an assignment by Sublandlord of is interest under the Master Lease, Sublandlor shall have the right to transfer the Security Deposit to Subtenant agrees to look to such assignee solely for the return of the Security Deposit to any sublandlord. Subtenant the monies deposited shall have the right to every transfer or assignment made of the Security Deposit to subtenant agrees to look to such assignment by Sublandlord of subtenant the provisions hereof shall apply to every transfer or assignment made of the Security Deposit to a new sublandlord. Subtenant there the monies deposited herein as security Deposit to a new sublandlord. Subtenant agrees to look to such assignment to subtenant the rot to a s

neither Sublandlord nor its successors or assigns shall be bound by any such assignment, encumbrance, attempted assignment or attempted encumbrance.

6.2 Letter of Credit.

Notwithstanding Section 6.1 above, Subtenant shall, within ten (10) Business Days following Subtenant's execution and delivery of this Sublease, deliver to Sublandlord in lieu of cash 6.2.1 security deposits required pursuant to Section 6.1 an unconditional, clean, irrevocable letter of credit (the "Letter of Credit"), subject to the following terms and conditions. If Subtenant fails to deliver the Letter of Credit to Sublandlord within ten (10) Business Days following Subtenant's execution and delivery of this Sublease, Sublandlord may terminate this Sublease by written notice to Subtenant and in such event neither party shall have any obligations to the other party under this Sublease. Delivery of such Letter of Credit shall satisfy the Security Deposit required in Section 6.1. If Subtenant is entitled to the Burn Down Right as described in Section 6.1 above, then Subtenant may reduce the Letter of Credit to be in an aggregate amount equal to the Security Deposit resulting therefrom by issuance of an amendment to the existing Letter of Credit or replacement of the existing Letter of Credit with a new Letter of Credit meeting the requirements of this Section 6.2 (and, for the avoidance of doubt, any such replacement or amended letter of credit shall be deemed part of the Letter of Credit hereunder) and Sublandlord shall promptly following request of Subtenant complete all reasonable documentation reasonably required by the issuing bank to reduce the amount of the Letter of Credit in accordance with the foregoing. Subtenant's failure to deliver the Letter of Credit as provided above shall, notwithstanding anything in this Sublease or the Master Lease to the contrary, constitute a default by Subtenant hereunder for which there shall be no notice or grace or cure period applicable thereto and Subtenant shall not be entitled to possession hereunder until it has delivered the Letter of Credit as provided above. Subtenant shall pay all expenses, points and/or fees incurred by Subtenant in obtaining the Letter of Credit. Such Letter of Credit shall be: (i) in the form attached to and hereby made a part of this Sublease as Exhibit C; (ii) at all times in the amount of the Security Deposit and shall permit multiple draws without a corresponding reduction in the amount of the Letter of Credit required to be maintained by Subtenant hereunder; (iii) made payable to Sublandlord and expressly transferable and assignable; (iv) payable in the City of San Francisco, California or Santa Clara County, California and "callable" at sight upon presentment of a simple sight draft or certificate (including via delivery by overnight courier) stating only that Sublandlord is entitled to draw on the Letter of Credit pursuant to this Sublease; (v) for a term not less than one year; and (vi) contain an "evergreen" provision that provides that the Letter of Credit will be automatically renewed on an annual basis unless the issuer delivers at least thirty (30) days' prior written notice of cancellation to Sublandlord and Subtenant, such that the Letter of Credit remains in existence through the date occurring ninety (90) days after the expiration of the Term. At least thirty (30) days prior to the then current expiration date of the Letter of Credit, Subtenant shall cause the Letter of Credit to be renewed (or automatically and unconditionally extended), or replaced, from time to time for a period of one year with the final expiration date of the Letter of Credit being the ninetieth (90th) day after the final Expiration Date hereunder. The final Expiration Date hereunder is June 30, 2028, and thus, the ninetieth (90th) day after the final Expiration Date of the Term is September 28, 2028

6.2.2 Each Letter of Credit shall be issued by a federally chartered commercial bank that has a credit rating with respect to certificates of deposit, short term deposits or commercial paper of at least A3 (or equivalent) by Moody's Investment Service, Inc., or at least A4 (or equivalent) by Standard & Poor's, a division of McGraw-Hill, and shall be otherwise acceptable to Sublandlord in its sole and absolute discretion. If the issuer's credit rating is reduced below A3 (or equivalent) by Moody's Investment Service, Inc. or below AA (or equivalent) by Standard & Poor's, a division of McGraw-Hill, or if the financial condition of such issuer changes in any other materially adverse way as determined in Sublandlord's sole and absolute discretion, then Subtenant shall obtain from a different issuer a substitute Letter of Credit that complies in all respects with the requirements of <u>Section 6.2.1</u>, and Subtenant's

failure to obtain such substitute Letter of Credit within ten (10) Business Days following its receipt of Sublandlord's demand therefor (and Subtenant's failure to do so shall, notwithstanding anything in this Sublease or the Master Lease to the contrary, constitute a default by Subtenant hereunder for which there shall be no notice or grace or cure period applicable thereto) and shall entitle Sublandlord to immediately draw upon the then existing Letter of Credit in whole or in part, without notice to Subtenant and to hold the proceeds thereof as the Security Deposit. Sublandlord hereby approves of Silicon Valley Bank as the initial issuer of the Letter of Credit is insolvent or placed into receivership or conservatorship by the Federal Deposit Insurance Corporation, or any successor or similar entity, or if a trustee, receiver or liquidator is appointed for the issuer, then Subtenant shall obtain from a different issuer a substitute Letter of Credit within ten (10) Business Days following its receipt of Sublandlord's demand therefor (and Subtenant's failure to do so shall, notwithstanding anything in this Sublease or the Master Lease to the contrary, constitute a default of Subtenant for which there shall be no notice or grace or cure period applicable thereto) and shall entitle Sublandlord to immediately draw upon the then existing Letter of Credit in whole or in part, without notice to Subtenant and to hold the proceeds thereof as the Security Deposit.

6.2.3 Sublandlord shall have the right to assign its interest in any Letter of Credit delivered to it by Subtenant pursuant to this Section 6.2 to any assignee of Sublandlord so long as such assignee agrees to assume the obligations of Sublandlord under this Sublease and the obligations of Sublandlord, as tenant, under the Master Lease and Sublandlord shall give notice of any such assignment to Subtenant. If requested by any such assignee, Subtenant shall obtain an amendment to such Letter of Credit that names such assignee as the beneficiary thereof in lieu of Sublandlord. Any failure or refusal of the issuer to honor the Letter of Credit shall be at Subtenant's sole risk and shall not relieve Subtenant of its obligations hereunder with respect to the Security Deposit.

6.2.4 Notwithstanding anything in this Sublease or the Master Lease to the contrary, any cure or grace period set forth in this Sublease or the Master Lease shall not apply to any of the foregoing, and, specifically, if Subtenant fails to timely comply with the foregoing terms of this <u>Section 6.2</u>, then Sublandlord shall have the right to draw upon the Letter of Credit without notice to Subtenant and hold the proceeds as the Security Deposit. If any portion of the Letter of Credit is drawn upon, Subtenant shall, within ten (10) Business Days after written demand therefor, reinstate the Letter of Credit to the amount then required under this Sublease, and Subtenant's failure to do so within ten (10) Business Days shall be a default under this Sublease.

7. INCORPORATION OF MASTER LEASE BY REFERENCE; ASSUMPTION

7.1 Incorporation. Subtenant acknowledges that it has read the Master Lease and is fully familiar with all terms and conditions of the Master Lease. All of the Sections and Exhibits of the Master Lease are incorporated into this Sublease as if fully set forth in this Sublease except that (i) **Paragraphs 1, 3, 4, 5, 8, 10, 12 and 13** of the Summary of Basic Lease Information and **Sections 1.1.1** (except the second (2nd) and fourth (4th) sentences), **1.1.4, 1.3, 1.4, 1.5, 2, 3, 4.1** [General Terms], **4.2.1** [Base Year], **4.2.8.5, 4.2.10** [Tenant's Payment of Certain Tax Expenses], **5.4** [Tenant's Dogs], **8.1** (except the third (3rd), fifth (5th) and sixth (6th) sentences), **10.1** [Indemnification and Waiver], **10.4** [Landlord Insurance], **14.8** [Allowed Shared Occupancy Agreements], **16** [Holding Over], **18** [Subordination], **19.5.2** [Tenant Right to Self Help], **19.5.3** [Landlord Reimbursement], **21** [Letter of Credit], **23.1.2** [Top of Building Signage], **23.1.3, 23.1.4** (except the second (2nd), penultimate and ultimate sentences), **29.5** [Memorandum of Lease], **29.12** [Landlord Exculpation], **29.21** [Notices], **29.23** [Brokers], **29.27** [Confidentiality], **29.33** [Patriot Act and Executive Order 13224], **29.34** [Background Checks], **29.35** [Foreign Employment; Change], **29.44**

[Tenant Competitor], 29.45 [Landlord Representations and Warranties] and 29.46 [Anti-corruption] of the Master Lease and Exhibits B, C [Tenant Work Letter] (except Section 4.2.2.3 and Schedule 4 thereof), D [Notice of Lease Term Dates], H, I [Memorandum of Lease], L-1, L-2 and M [Tenant's Signage] to the Master Lease are deleted in their entirety from the Master Lease as incorporated into this Sublease, (ii) Sections 3 [Expansion Term], 4 [Base Rent], 6 [Expansion Improvements], and 7 [Broker] of the First Amendment are deleted in their entirety from the Master Lease as incorporated into this Sublease, (iii) all representations and warranties made by Master Landlord in the Master Lease are made solely by Master Landlord and not by Sublandlord, (iv) any rights of Sublandlord to extend, expand, contract, cancel or terminate the Master Lease shall not apply to or benefit Subtenant in any manner whatsoever; provided, however, that nothing herein shall prevent Subtenant from exercising its right to terminate this Sublease in accordance manner whatsoever unless provided otherwise in the Master Lease that are designated as personal to Original Tenant and its Permitted Transferees shall not apply to or benefit Subtenant in any manner whatsoever.

Usage of Terms. Except as otherwise provided in or modified by this Sublease, the term "Landlord" as used in the Master Lease shall refer to "Sublandlord" hereunder, the term "Tenant" as used in the Master Lease shall refer to "Subtenant" hereunder, the term "Lease" as used in the Master Lease shall refer to this Sublease, the term "Commencement Date" as used in the Master Lease shall refer to the applicable "Commencement Date" hereunder, the term " Lease Term" as used in the Master Lease shall refer to the "Term" hereunder, the term "Base Rent" as used in the Master Lease shall refer to "Base Rent" hereunder, the term "Additional Rent" as used in the Master Lease shall refer to "Additional Rent" hereunder, the term "Rent" as used in the Master Lease shall refer to "Rent" hereunder, the term "Lease State and the term state a the Master Lease shall refer to the applicable "Expiration Date" hereunder and the term "Permitted Use" as used in the Master Lease shall refer to the "Permitted Use", the term "Base Year" as used in the Master Lease shall refer to the "Sublease Base Year" hereunder, and the term "Tenant Signage" as used in the Master Lease shall refer to "Subtenant's Signage" hereunder. Any reference to the consent or approval of "Landlord" being required under the Master Lease, as incorporated into this Sublease, shall require the consent or approval of both Master Landlord and Sublandlord (except with respect to Alterations (as defined below), which shall only require the approval of Master Landlord). Notwithstanding anything to the contrary contained in this Sublease, Sublandlord shall not be required to (i) provide any of the insurance, services or construction to the Premises or the Project that Master Landlord may have agreed to provide pursuant to the Master Lease, (ii) provide any utilities (including electricity) to the Premises, or invoices or information related to such utilities, that Master Landlord may have agreed to furnish pursuant to the Master Lease (or as required by law), (iii) make any of the repairs that Master Landlord may have agreed to make pursuant to the Master Lease (or as required by law), including any repairs required following a casualty, (iv) take any other action relating to the operation, maintenance, repair, telecommunications capabilities, restoration, rebuilding, alteration, licensing or servicing of the Premises or the Project that Master Landlord may have agreed to provide, furnish, make, comply with, or take, or cause to be provided, furnished, made complied with or taken under the Master Lease, (v) provide any security or security system (including any Tenant's Security System installed by or for the use of Sublandlord under the Master Lease) for the Premises or the Project or (vi) except as set forth in this Sublease, provide Subtenant with any abatement, rebate, credit, allowance or other concession required of Master Landlord pursuant to the Master Lease. Subtenant shall not make any claim against Sublandlord for any damage that may arise by reason of (a) the failure of Master Landlord to keep, observe or perform any of its obligations under the Master Lease or (b) the acts or omissions of Master Landlord or its agents, contractors, employees, invitees or licensees so long as Sublandlord has used commercially reasonable efforts to enforce all of Sublandlord's rights under the Master Lease to the extent necessary for Subtenant's use and occupancy of the Premises.

7.3 <u>Abatements Under Master Lease</u>. Subtenant shall be entitled to receive any abatement of Rent as to the Premises during the Term resulting from any casualty, condemnation or interruption of services that Sublandlord has actually received under the Master Lease, less any expenses incurred by Sublandlord in obtaining such abatement. Sublandlord shall use commercially reasonable efforts to enforce its abatement rights under the Master Lease, including pursuant to **Section 6.6** [Abatement Event] and **Article 11** [Damage and Destruction] of the Master Lease, and shall keep Subtenant reasonably apprised of all such efforts on request of Subtenant.

7.4 <u>Conflicts with Master Lease</u>. If any provisions of this Sublease expressly conflict with any portion of the Master Lease as incorporated herein, the terms of this Sublease shall govern. Subtenant shall assume and perform for the benefit of Sublandlord and Master Landlord all of Tenant's obligations under the Master Lease provisions to the extent expressly incorporated herein.

7.5 Services Under Master Lease. Subtenant shall be entitled to receive all of the work, services, repairs, repairing, restoration, parking passes, the licenses granted, utilities provided, elevator or HVAC services, or the performance of any other obligations required of Master Landlord under the Master Lease with respect to the Premises (except to the extent any such obligations were not incorporated by reference above); provided, however, Sublandlord's sole obligation with respect thereto shall be to request the same from Master Landlord, as requested in writing by Subtenant and at Subtenant's sole cost and expense. If Master Landlord shall default in any of its obligations to Sublandlord with respect to the Premises, Sublandlord will use commercially reasonable efforts to cause Master Landlord to perform and observe such obligations and in connection therewith (and in connection with enforcement of abatement rights as provided in the final sentence of <u>Section 7.3</u>). Sublandlord shall not be obligated to commence any legal, arbitration or audit proceedings against Master Landlord, or utilize any self-help rights, or make any payment of money or other consideration other than as expressly required of Sublandlord under the Master Lease on any failure of Master Landlord or act or grant any consent or approval under the Master Lease, or from any misfeasance or non-feasance of Master Landlord, nor shall the obligations of Subtenant hereunder be excused or abatem in any manner by reason thereof, except as expressly provided in this Sublease.

7.6 <u>Requests for Services From Master Landlord</u>. Subtenant shall cooperate with Sublandlord as may be required to obtain from Master Landlord any such work, services, repairs, repainting, restoration, the provision of utilities, elevator or HVAC services, or the performance of any of Master Landlord's other obligations under the Master Lease, provided that in day-to-day issues, if Master Landlord so agrees, Subtenant shall first request Master Landlord provide the desired service or item and shall only request same from Sublandlord if Master Landlord fails to perform. This Sublease shall at all times during the Term remain subject and subordinate to the Master Lease (and to all matters to which the Master Lease is subject and subordinate) and to all modifications and amendments to the Master Lease.

8. MASTER LEASE.

8.1 <u>Actions under Master Lease</u>. At any time and on reasonable prior notice to Subtenant, Sublandlord can elect to require Subtenant to perform Subtenant's obligations under this Sublease directly to Master Landlord if Master Landlord so consents to same to the extent such consent is required under the Master Lease. Subtenant shall send to Sublandlord from time to time copies of all notices and other communications that Subtenant shall send to and receive from Master Landlord (which notices sent by Subtenant to Master Landlord shall be sent in accordance with Section 29.17 [Notices] of the Master Lease. Subtenant shall not do or permit to be done anything which would constitute a violation or breach of any of the terms, conditions or provisions of the Master Lease or which would cause the Master Lease

to be terminated or forfeited by virtue of any rights of termination or forfeiture reserved by or vested in Master Landlord. During the Term of this Sublease, Sublandlord shall not voluntarily terminate the Master Lease, subject to Sublandlord's right to terminate the Master Lease in the event of a casualty, condemnation, force majeure, or default by Master Landlord under the Master Lease. If the Master Lease terminates in whole or in part, this Sublease shall terminate with respect to the applicable portion of the Premises and the parties shall be relieved from all liabilities and obligations under this Sublease with respect to such portion (for the avoidance of doubt, other than as provided in <u>Section 24.11</u>). Subtenant's time to exercise its right to terminate this Sublease for Casualty (which right shall be subject to Master Landlord's rights under the Master Lease to suspend such termination) shall not exceed eighty percent (80%) of the corresponding time under the Master Lease, and within such period of time Subtenant shall deliver to Sublandlord written notice of whether Subtenant elects to exercise such right.

- 8.2 <u>Sublandlord Representations</u>. Sublandlord hereby represents and warrants to Subtenant as of the Effective Date:
 - (a) the Master Lease is in full force and effect;

(b) the Master Lease attached hereto as Exhibit A is a full and complete copy of the agreement between Master Landlord and Sublandlord with respect to Sublandlord's leasing of the Premises (however Subtenant acknowledges that the attached is redacted to delete all financial information);

- (c) to the actual knowledge of Sublandlord, the Master Lease constitutes the entire agreement between Master Landlord and Sublandlord relating to the lease of the Premises; and
- (d) no default or breach by Sublandlord or, to the actual knowledge of Sublandlord, by Master Landlord exists under the Master Lease as of the Effective Date.

As used in this Sublease, the phrase "actual knowledge of Sublandlord" shall be limited to the actual knowledge of George Kreitem (without personal liability therefor).

ACCEPTANCE OF PREMISES "AS-IS".

Other than Sublandlord's obligation to deliver the FF&E and make the repairs (if any) described in the last sentence of this <u>Section 9</u>, the Premises shall be delivered to Subtenant in "AS-IS" and "WITH ALL FAULTS" condition without any representations and warranties with respect thereto by Sublandlord, its agents, officers, directors, employees, consultants or attorneys. Subtenant tacknowledges and agrees that Sublandlord and its agents, officers, directors, employees, consultants or attorneys. Subtenant tacknowledges and agrees that Sublandlord and its agents, officers, directors, employees, consultants or attorneys. Subtenant tacknowledges and agrees that Sublandlord and its agents, officers, directors, employees, consultants or attorneys. Subtenant tacknowledges and agrees that Sublandlord and its agents, officers, directors, employees, consultants or attorneys. Subtenant tacknowledges and agrees that Sublandlord there in except as expressly set forth in this Sublease. The taking of possession of any Premises Component by Subtenant shall be conclusive evidence that Subtenant cacepts the same "AS-IS" and "WITH ALL FAULTS" and that the Premises are suited for the use intended by Subtenant and are in good and satisfactory condition at the time such possession was taken. Subtenant represents and warrants to Sublandlord that (i) its sole intended use of the Premises is for general office use and related uses to the extent permitted in accordance with the Master Lease (the "**Permitted Use**") and (ii) prior to executing this Sublease it has made such investigations as it deems appropriate with respect to the suitability of the Premises for its intended use and has determined that the Premises are suitable for such intended use. Sublandlord shall have no obligation whatsoever to construct any improvements for Subtenant to repair or refurbish the Premises other than any repairs necessary to repair any material damage caused by Sublandlord's vacation of any Premises Component, within thirty (30) days following Subla

receipt of a written request by Subtenant to do so, Sublandlord and Subtenant shall conduct a joint walk-through of such Premises Component to determine if Sublandlord is required to perform any repairs prior to delivering such Premises Component to Subtenant.

10. USE; SIGNAGE, SUBLANDLORD'S TRADE NAMES; PARKING; ROOF TOP DECK.

10.1 Use. Subtenant agrees that the Premises shall be used by Subtenant (and its permitted assignees and subtenants) solely for the Permitted Use and for no other use, business or purpose whatsoever.

10.2 Signage. Subject to and in accordance with Article 23 [Signs] of the Master Lease and Master Landlord's approval in its sole and absolute discretion, Subtenant shall have the rights to the signage granted to Tenant under the Master Lease (to the extent such rights are incorporated into this Sublease pursuant to <u>Section 7.1</u> above and as otherwise granted in the Master Landlord's Consent) ("Subtenant's Signage"). Subtenant acknowledges and agrees that the Building Top Signage and Exterior Building are personal to Sublandlord and are not transferred hereunder unless provide otherwise in the Master Landlord's Consent) ("Subtenant's Signage"). Subtenant shall have no rights with respect thereto. Subtenant shall be responsible, at its sole cost and expense, for all costs associated with the design, fabrication, permitting, installation, repair, maintenance and replacement of Subtenant's Signage and Exterior Building are personal to Sublandlord has made any representations or warranties regarding the likelihood of Subtenant obtain all permits and approvals required by the City of San Francisco. Subtenant acknowledges and agrees that neither Master Landlord no Sublandlord has made any representations or warranties regarding the likelihood of Subtenant obtaining the requirement to any obtenant's Signage and the failure of Subtenant to obtain such permits or approvals shall not delay the Commencement Date, release Subtenant of any of Subtenant so the hereunder. Subtenant shall remove Subtenant's Signage prior to the scheduled Expiration Date of the applicable Premises Component and/or as required by Master Lease. Prior to the delivery of each Premises Component of First Increment to Subtenant, Sublandlord shall remove its signage with respect to such Premises Component at its sole cost and expense, including repairing any damage to the Building caused by such removal.

10.3 <u>Trade Names</u>. The parties agree that they shall not, without the other party's prior written consent, which consent may be withheld by such party in its sole and absolute discretion, use the names, characters, artwork, designs, trade names, copyrighted materials, trademarks or service marks of the other party or its parent, affiliated or subsidiary companies, employees, directors, shareholders, assigns, successors or licensees (i) in any advertising, publicity or promotion or (ii) in any manner other than expressly in accordance with this Sublease.

10.4 Parking. Subtenant shall have the right, but not the obligation, to sublease all of the parking passes that Sublandlord has the right to lease pursuant to and in accordance with the terms and conditions set forth in the Master Lease. Subtenant shall pay to Sublandlord as Additional Rent any amounts charged by Master Landlord to Sublandlord in connection with the leasing of parking passes or use of the Parking Garage, which amounts are subject to change by Master Landlord from time to time in accordance with the terms and conditions of the Master Lease. Subtenant shall notify Sublandlord on or before November 1 of each year during the Term stating the number of parking passes that Subtenant wishes to lease during the coming year in accordance with Article 28 [Tenant Parking] of the Master Lease and Sublandlord shall notify Master Landlord on or before November 15 of each year of

Sublandlord's election to lease such parking passes on Subtenant's behalf (and, for the avoidance of doubt, such parking passes shall be subleased by Subtenant in accordance herewith).

10.5 <u>Roof Top Deck</u>. For the avoidance of doubt, commencing as of the Commencement Date with respect to the Fourth Floor, Subtenant shall be responsible for the Deck Terrace in accordance with provisions related thereto under the Master Lease.

11. CONFIDENTIALITY.

Confidentiality Requirements. Except as expressly permitted in this Section 11.1, neither party nor its agents, servants, employees, invitees and contractors will, without the prior written consent of 11.1 the other party, disclose any Confidential Information of the other party to any third party. Information will be considered "Confidential Information" of a party if: (i) it is disclosed by the party to the other party in tangible form and is conspicuously marked "Confidential", "Proprietary" or the like; (ii) it is disclosed by the party to the other party in non-tangible form and is identified by such party as confidential, proprietary or the like at the time of disclosure; or (iii) would reasonably be understood, given the nature of the information or the circumstances surrounding its disclosure, to be confidential. In addition, notwithstanding anything in this Sublease to the contrary, the terms of this Sublease (but not its mere existence) will be deemed Confidential Information of each party. Other than the terms and conditions of this Sublease, information will not be deemed Confidential Information hereunder if such information: (a) is known to the receiving party prior to receipt from the disclosing party directly or indirectly from a source other than one known to have an obligation of confidentiality to the disclosing party; (b) becomes known (independently of disclosure by the disclosing party) to the receiving party directly from a source other than one known to have an obligation of confidentiality to the disclosing party; (c) becomes publicly known or otherwise ceases to be secret or confidential, except through a breach of this Sublease by the receiving party; or (d) is independently developed by the receiving party. Each party will secure and protect the Confidential Information of the other party (including the terms of this Sublease) in a manner consistent with the steps taken to protect its own confidential information, but not less than a reasonable degree of care. Each party may disclose the other party's Confidential Information where: (1) the disclosure is required by law or by an order of a court or other governmental body having jurisdiction after giving reasonable notice to the other party with adequate time for such other party to seek a protective order, if reasonably possible; (2) if in the opinion of counsel for such party, disclosure is advisable under any applicable securities laws regarding public disclosure of business information; (3) the disclosure is reasonably necessary and is to that party's or its affiliates' or its actual or prospective lenders' or investors' employees, officers, directors, members, attorneys, accountants, consultants and advisors, or the disclosure is otherwise necessary for a party to exercise its rights and perform its obligations under this Sublease; (4) the disclosure is reasonably necessary for the party to conclude a business transaction; or (5) the disclosure is to Master Landlord, so long as in all cases the disclosure is no broader than reasonably necessary and the person who receives the disclosure agrees prior to receiving the disclosure to keep the information confidential. Each party is responsible for ensuring that any Confidential Information of the other party that the first party discloses pursuant to this Section 11.1 is kept confidential by the person receiving the disclosure. Without limiting the generality of this Section 11.1, neither Subtenant nor Sublandlord will, directly or indirectly, issue any press release regarding this Sublease or any matters set forth in this Sublease, or use either party's name for any commercial purposes or use any of either party's trademarks, in each case, without the express prior written consent of the other party to be granted or withheld in such party's sole and absolute discretion. Each party acknowledges that any breach of this Section 11.1 may cause irreparable harm for which monetary damages are an insufficient remedy and therefore that upon any breach of this Section 11.1 the non-breaching party shall be entitled to appropriate equitable relief without the posting of a bond in addition to whatever other remedies it might have at law or in equity.

11.2 <u>Subtenant Security Requirements</u>. Sublandlord acknowledges the confidential nature of Subtenant's business and shall comply with reasonable security requirements of Subtenant (except in the event of an emergency) when entering the Premises. Sublandlord shall use commercially reasonable efforts to keep any knowledge gained through inspection by Sublandlord or access by Sublandlord to the Premises confidential (provided that Sublandlord shall be permitted to disclose such knowledge as described in clauses (1)—(5) of <u>Section 11.1</u>) and Sublandlord shall not use such knowledge for any other purpose than managing the Premises, and when entering the Premises, Sublandlord shall not unreasonably interfere with the operation of Subtenant's business in the Premises.

12. ASSIGNMENT AND SUB-SUBLETTING.

12.1 Consent Requirements. Subject to Subtenant obtaining the consent of Master Landlord and Sublandlord, which consent by Sublandlord will not be unreasonably withheld, conditioned or delayed, Subtenant shall have the right to assign this Sublease or to sub-sublease the Premises (each, a "**Transfer**") in accordance with the provisions of **Article 14** [Assignment and Subletting] of the Master Lease (to the extent **Article 14** of the Master Lease is incorporated into this Sublease pursuant to <u>Section 7.1</u> above) and this <u>Section 12</u>. Copies of all materials required by **Section 14.1** [Transfers] of the Master Lease, and this <u>Section 12</u> shall be delivered simultaneously to both Master Landlord and Sublandlord, together with Subtenant's request for any such consent. Without limiting the reasons upon which Sublandlord could reasonably withhold its consent if (i) it does not approve the proposed use of the Premises, (ii) it does not approve the creditworthiness or business reputation of the assignee or new sub-subtenant (each, a "**Transfere**") or (iii) such Transfer Could trigger Master Landlord's requter rights under **Section 14.4** [Landlord's Option as to Subject Space] of the Master Lease. In connection with any Transfer, Sublandlord shall have the right to review and approve the current financial statements of any proposed Transferee subject to a commercially reasonable non-disclosure agreement.

12.2 Fees: Sharing of Profits. Subtenant shall pay Sublandlord's reasonable fees and expenses incurred with respect to any proposed Transfer (regardless of whether such Transfer is actually consummated), including reasonable attorneys' fees (at market rates) incurred in reviewing Transfer documentation and required Master Landlord consents thereto, any fees charged by Master Landlord to Sublandlord in connection with any such Transfer (including expenses under Section 14.1 [Transfers] of the Master Lease), and any architects', engineers' and other consultants' fees required for such Transfer, within thirty (30) days following written demand therefor from Sublandlord accompanied by invoices reasonably substantiating such costs. If Sublandlord consents to any Transfer, Subtenant shall pay to Sublandlord fifty percent (50%) of any Transfer Premium (as that term is defined in the Master Lease). Subtenant shall provide Sublandlord with a detailed statement setting forth the calculation of any Transfer Premium Subtenant either has or will derive from such Transfer. In addition, Sublandlord or its representative shall have the right at all reasonable times upon no less than ten (10) Business Days' prior written notice to audit the books and records of Subtenant solely with respect to the calculation of the Transfer Premium. If such inspection reveals that the amount of Transfer Premium paid to Sublandlord was understated by more than five percent (5%), then within thirty (30) days of Subtenant's receipt of the results of such audit, Subtenant shall pay Sublandlord the deficiency and the cost of Sublandlord's audit (not to exceed \$5,000).

12.3 <u>Collection of Rent; Further Assignment/Subletting</u>. If this Sublease is assigned, or if the Premises or any part thereof is Transferred or occupied by anyone other than Subtenant, whether or not Subtenant shall have been granted any required consent, Sublandlord may, after default by Subtenant beyond applicable notice and cure periods, collect rent and other charges from such Transfere or other occupant, and apply the net amount collected to the Base Rent and Additional Rent herein reserved, but no such Transfer, occupancy or collection shall be deemed to be a waiver of the requirements of this

Section 12 or an acceptance of the Transferee or other occupant as the subtenant under this Sublease. The consent by Sublandlord or Master Landlord to a Transfer shall not in any way be construed to relieve Subtenant from obtaining the consent of Master Landlord and Sublandlord to any further Transfer. Such Transfer shall be subject to all of the terms and conditions of the Master Lease and this Sublease, and Subtenant shall remain primarily liable under this Sublease notwithstanding any Transfer.

13. DEFAULTS AND REMEDIES.

131 Defaults and Remedies. Upon any default by Subtenant under this Sublease or under the Master Lease beyond applicable notice and cure periods, Sublandlord shall have all rights and remedies available at law or in equity, including all of the rights and remedies described in the Master Lease, including Article 19 [Defaults; Remedies] of the Master Lease. Without limiting the generality of the foregoing, Sublandlord shall have the rights and remedies provided by California Civil Code Section 1951.2, including the right to terminate Subtenant's right to possession of the Premises and to recover the worth at the time of award of the amount by which the unpaid Base Rent, Additional Rent and other charges for the balance of the Term after the time of award exceed the amount of rental loss for the same period that Subtenant proves could be reasonably avoided, as computed pursuant to subsection (b) of California Civil Code 1951.2; and the rights and remedies provided by California Civil Code Section 1951.4 that allow Sublandlord to continue this Sublease in effect and to enforce all of its rights and remedies under this Sublease, including the right to recover Base Rent, Additional Rent and other charges as they become due, for so long as Sublandlord does not terminate Subtenant's right to possession. Subtenant shall have no right to occupy the Premises or any portion thereof after the expiration or earlier termination of this Sublease. Subtenant shall remove from each Premises Component all of its fixtures and equipment, including the FF&E (to the extent required to be removed under Section 12 below), and any Alterations (to the extent required to be removed under Section 14 below) upon the expiration or earlier termination of this Sublease as to such Premises Component. If Subtenant or any party claiming by, through or under Subtenant holds over in any Premises Component (or any portion thereof) beyond the Expiration Date therefor, Sublandlord may exercise any and all remedies available to it at law or in equity to recover possession of such Premises Component and to recover damages Subtenant shall indemnify, defend and hold Sublandlord harmless from all actual liabilities, claims and damages suffered by Sublandlord (including attorneys' fees) resulting from or occasioned by Subtenant's holding over in any Premises Component (or any portion thereof) beyond the Expiration Date therefor, including consequential damages. For each and every month or partial month that Subtenant or any party claiming by, through or under Subtenant remains in occupancy of all or any of any portion of any Premises Component after the expiration or earlier termination of this Sublease as to such Premises Component or after termination of Subtenant's right to possession of such Premises Component, Subtenant shall pay, as minimum damages and not as a penalty, monthly Base Rent equal to one hundred fifty percent (150%) of the Base Rent applicable to such Premises Component on a per diem basis immediately prior to the date of such expiration or earlier termination of this Sublease or of Subtenant's right of possession. The acceptance by Sublandlord of any lesser sum shall be construed as payment on account and not in satisfaction of damages for such holding over. Within a commercially reasonable period of time following Sublandlord's actual receipt from Master Landlord of an estimate of Holding Over Damages under Article 16 [Holding Over] of the Master Lease, Sublandlord or its third party consultants or other agents shall revise such estimate to account for Sublandlord's Holding Over Damages and deliver such revised version to Subtenant. Subtenant's failure to remove any Alterations required to be removed under Section 14 below or the FF&E required to be removed under Section 28 below or any other items of furniture, equipment, business and trade fixtures, free-standing cabinet work, movable partitions, cabling and other articles of personal property owned or installed by Subtenant from the Premises upon the expiration or earlier termination of this Sublease shall be deemed to be a holding over by Subtenant. Any such property not so removed by Subtenant as required herein shall be deemed abandoned and may be stored, removed, and disposed of by Sublandlord at Subtenant's expense, and

Subtenant waives all claims against Sublandlord for any damages resulting from Sublandlord's retention and/or disposition of such property.

13.2 Notice and Cure Periods. Except as otherwise set forth herein, in the event of a non-monetary default by Subtenant under this Sublease, Subtenant shall have the same notice and cure rights provided for in the Master Lease as Sublandlord would have for a similar default under the Master Lease, except that Subtenant's time to cure shall not exceed eighty percent (80%) of the corresponding time under the Master Lease.

14. ALTERATIONS

Except as provided herein, Subtenant shall not make any improvements, alterations, additions or changes (collectively, "Alterations") in or to the Premises or make changes to locks on doors or add, disturb or in any way change any plumbing, mechanical, electrical, HVAC, life safety or other Building systems without obtaining the prior written consent of Master Landlord. Any Alterations must be done in full compliance with the provisions of the Master Lease and shall be made at Subtenant's sole cost and expense (including any review fees charged by Master Landlord under the Master Lease) and by contractors or mechanics approved by Master Landlord (except as otherwise set forth in the Master Lease). All work with respect to any Alterations shall be performed in a good and workmanlike manner, shall be of a quality equal to or exceeding the then existing construction standards for the Building and shall be constructed in compliance with all plans approved by Master Landlord. Alterations shall be diligently prosecuted to completion to the end that the Premises shall be at all times a complete unit except during the period necessarily required for such work. All Alterations shall be made strictly in accordance with all applicable laws relating thereto including all building codes and regulations and the Americans with Disabilities Act (the "ADA"). In furtherance of the foregoing Subtenant, at Subtenant's sole cost and expense, shall make and complete any and all necessary alterations or upgrades to the Premises and/or the Building arising by reason of Subtenant's Alterations in order to fully comply with the ADA and any life safety requirements applicable to the Premises and the Building. Subtenant, at its sole cost and expense, shall obtain any and all permits and consents of applicable governmental authorities in connection with all Alterations. Subtenant shall be liable to Sublandlord and Master Landlord for the reasonable costs of any improvements to the Building (whether or not on the Premises) which may be required as a consequence of Subtenant's Alterations. Sublandlord hereby reserves the right to require any contractor, subcontractor or materialsman working in or providing materials to the Premises to provide lien waivers and liability insurance covering the Alterations to the Premises in accordance with the terms of the Master Lease. Subtenant shall give Master Landlord and Sublandlord ten (10) days' written notice prior to the commencement of any Alterations and shall allow Master Landlord and Sublandlord to enter the Premises and post appropriate notices to avoid liability to contractors or material suppliers for payment for any Alterations. All Alterations shall remain in and be surrendered with the applicable Premises Component as a part thereof at the applicable Expiration Date, without disturbance, molestation or injury, subject to Master Landlord's rights under the Master Lease to require the removal of Subtenant's Alterations that are Specialty Improvements and restoration of such Premises Component, which removal and restoration shall be performed by Subtenant in accordance with the Master Lease, at Subtenant's sole cost and expense. Subtenant shall request in accordance with Section 8.5 [Landlord's Property] of the Master Lease, and deliver to Sublandlord notice of, Master Landlord's decision with respect to all Subtenant's Alterations that are Specialty Improvements in each Premises Component at least seven (7) months prior to the applicable Expiration Date. Sublandlord in its sole and absolute discretion may deliver to Subtenant notice no less than six (6) months prior to each Expiration Date that Sublandlord elects to require Subtenant not to perform its removal and restoration obligations with respect to any Specialty Improvements within the applicable Premises Component, in which case Subtenant shall continue to maintain in accordance with this Sublease and the Master Lease such Specialty Improvements and surrender the same with the Premises Component as a part thereof as of

the applicable Expiration Date, without disturbance, molestation or injury, and such Premises Component to be restored to its condition prior to performance of such Alterations. In such event, all expenses to remove said Alterations and to so restore the Premises shall be paid by Subtenant. Notwithstanding anything herein to the contrary, Subtenant shall not be required to restore any alterations or other improvements within any Premises Component that were installed or made therein prior to the Commencement Date thereof.

15. INDEMNIFICATION; LIMITATION OF DAMAGES.

15.1 <u>Subtenant's Duty to Indemnify</u>. Except to the extent caused by Sublandlord's gross negligence or willful misconduct, but subject in all events to the waiver of subrogation set forth in <u>Section 17.2</u> below, Subtenant shall indemnify, defend and hold Sublandlord and Master Landlord and their respective directors, officers, agents, employees, licensees or invitees harmless from all claims, damages, losses, liabilities, costs and expenses, including any sums for which Sublandlord may be liable to Master Landlord under any indemnity or hold harmless provision in the Master Lease and reasonable attorneys' fees and costs, arising from: (i) Subtenant's use of the Premises or the conduct of its business or any activity, work, or thing done, permitted or suffered by Subtenant in or about the Premises, (ii) any breach or default in the performance of any obligation to be performed by Subtenant under the terms of this Sublease (or any consents thereto) and (iii) any act, neglect, fault or omission of Subtenant or of its directors, officers, agents, employees, licensees or invitees with respect to Subtenant upon notice from Sublandlord shall defend the same at Subtenant's expense by counsel reasonably approved in writing by Sublandlord. Subtenant or its counsel shall keep Sublandlord fully apprised at all times of the status of such defense and shall not settle same without the written consent of Sublandlord. To the fullest extent permitted by law, Subtenant, as a material part of the consideration to Sublandlord, hereby assumes all risk of and waives all claims against Sublandlord with respect to damage to property or injury to persons in, upon or about the Premises from any cause whatsoever except that which is caused by the failure of Sublandlord of observe any of the terms and conditions of this Sublease or Sublandlord's failure to comply with local, state or federal law where such failure has persisted for an unreasonable period of time after written notice to Sublandlord of ouch failure.

15.2 <u>Limitation of Damages</u>. Under no circumstances shall either party to this Sublease be liable for any indirect, incidental, special or consequential damages arising out of this Sublease, including any damages arising from lost revenues, profits, use or business opportunity regardless of the cause of such damages and whether or not the other party was aware or should have been aware of the possibility of these damages; provided, however, that Subtenant will remain liable for any consequential damages arising out of any holdover of any Premises Component after the expiration of the Term with respect to such Premises Component or termination of this Sublease as to such Premises Component to the extent payable by Sublandlord under the Master Lease or under this Sublease. If the Master Lease is terminated due to a default by Sublandlord hereunder and as a result of such default Master Landlord elects to terminate the Master Lease, then Sublandlord shall be liable to Subtenant (and shall reimburse Subtenant for) all reasonable costs and expenses incurred by Subtenant to relocate from the Premises to a new location.

16. DAMAGE TO SUBTENANT'S PROPERTY.

Master Landlord and Sublandlord and their respective directors, officers and agents shall not be liable for (i) any damage to any property entrusted to employees at the Building or its property managers, (ii) loss or damage to any property by theft or otherwise, (iii) any injury or damage to persons or property resulting from fire, explosion, falling plaster, steam, gas, electricity, water or rain that may leak from any

part of the Building or from the pipes, appliances or plumbing work therein or from the roof, street or sub-surface or from any other place or resulting from dampness or any other cause whatsoever, or (iv) any damage or loss to the business or occupation of Subtenant arising from the acts or neglect of other tenants or occupants of, or invitees to, the Building, except, in any case, with respect to Sublandlord, to the extent caused by the gross negligence or willful misconduct of Sublandlord. Subtenant shall give prompt notice to Sublandlord and Master Landlord in case of fire or accident in the Premises or in the Building or of defects therein or in the fixtures or equipment.

17. SUBTENANT'S INSURANCE.

17.1 <u>Insurance Requirements</u>. Subtenant shall, at its sole cost and expense, maintain throughout the Term any insurance coverage required to be maintained by Sublandlord under the Master Lease, pursuant to the terms and conditions of the Master Lease, with a company authorized to transact business in the jurisdiction where the Premises is located and otherwise in accordance with the terms of the Master Lease. Sublandlord acknowledges and agrees that Subtenant may provide property coverage required under the Master Lease on a "special form" rather than "all risk" basis. Sublandlord and Master Landlord (and such other parties as required by the provisions of the Master Lease) shall be named as additional insurades under such insurance. Subtenant shall provide Sublandlord and Master Landlord with certificates of insurance evidencing the insurance required to be maintained by Subtenant herein prior to the Commencement Date and, upon request, from time to time thereafter. Sublenant further agrees to give not less than thirty (30) days' advance written notice to Sublandlord and Master Landlord (and any other parties named as additional insureds thereon) of any cancellation or reduction of insurance under any such policy.

17.2 Waiver of Subrogation. Sublandlord and Subtenant intend that their respective property loss risks shall be borne by reasonable insurance carriers as provided herein, and Sublandlord and Subtenant hereby agree to look solely to, and seek recovery only from, their respective insurance carriers in the event of a property loss to the extent that such coverage is agreed to be provided hereunder. Subtenant hereby waives on behalf of itself and on behalf of its insurers any and all rights of recovery against Sublandlord, Master Landlord and the directors, officers, employees, agents and representatives of Sublandlord or Master Landlord, by way of subrogation or otherwise, on account of loss or damage occasioned to Subtenant or its property or the properties of others under its control caused by fire or any of the extended coverage risks described hereunder to the extent that such loss or damage is insured under any insurance policy in force at the time of such loss or damage or required to be carried hereunder. If necessary for its effectiveness, Subtenant shall give notice to its insurers on account of damage to the Sublandlord or its property or the properties of others under its control caused by fire or any of the extended coverage risks described herein to the extent that such loss or damage to necessary for its effectiveness, agents and representatives on account of damage to the Sublandlord or its property or the properties of others under its control caused by fire or any of the extended coverage risks described herein to the extent that such loss or damage is insured under any insurance policy in force at the time of such loss or damage or required to be carried hereunder.

18. SERVICES.

Except to the extent expressly provided in this Sublease, including Section 7.3 above, Sublandlord shall not be liable for, and Subtenant shall not be entitled to any abatement of Rent by reason of, (i) the failure to furnish or delay in furnishing any of the services when such failure is caused by accident, breakage, repairs, strikes, lockouts or other labor disturbances or labor disputes of any character, or by any other cause, similar or dissimilar, beyond the reasonable control of Sublandlord or Master Landlord or by the making of any repairs or improvements to the Premises or to the Building or (ii) the limitation, curtailment, rationing or restrictions on use of water, electricity, gas or any other utility servicing the Premises or the Building by any utility or governmental agency; provided, however, that any

abatement of Rent provided to Sublandlord by Master Landlord related to the Premises shall be passed through to Subtenant as and when received by Sublandlord. Subtenant shall not connect any electrical equipment to the Building's electrical distribution system which may overload the electrical capacity of the Building or the Premises.

19. TIME; BUSINESS DAY.

Time is of the essence of this Sublease. As used herein, the term "Business Day" means "business day" as defined in the Master Lease

20. RIGHT TO PERFORM.

If Subtenant shall fail to pay any sum of money required to be paid by it hereunder, or shall fail to perform any other act on its part to be performed hereunder, and such failure shall continue for five (5) Business Days after written notice thereof by Sublandlord (or with respect to any restoration, removal or repair obligations in connection with the surrender of a Premises Component, immediately upon the applicable Expiration Date), Sublandlord may, but shall not be obligated so to do, and without waiving or releasing Subtenant from any obligations of Subtenant, make any such payment or perform any such other act on Subtenant's part to be made or performed as provided in this Sublease. Subtenant shall reimburse Sublandlord for all reasonable costs incurred by Sublandlord in connection with such payment or performance within thirty (30) days following Subtenant's receipt of written demand accompanied by a documented invoice therefor.

21. NON-WAIVER.

Neither the acceptance of Rent nor any other act or omission of Sublandlord or Subtenant at any time or times after the happening of any event authorizing the cancellation or forfeiture of this Sublease shall operate as a waiver of any past or future violation, breach or failure to keep or perform any covenant, agreement, term or condition hereof, or deprive either party of its right to cancel or forfeit this Sublease, upon the notice required by law, at any time that cause for cancellation or forfeiture may exist, or be construed so as to at any future time prevent either party from promptly exercising any other option, right or remedy that it may have under any term or provision of this Sublease.

22. NOTICES.

All notices under this Sublease shall be in writing and addressed to either Sublandlord or Subtenant as follows:

If to Sublandlord:	salesforce.com, inc. The Landmark at One Market, Third Floor San Francisco, California 94105 Attention: General Counsel
with a copy to:	salesforce.com, inc. The Landmark at One Market, Third Floor San Francisco, California 94105 Attention: Executive Vice President of Real Estate
and a copy to:	Paul Hastings LLP 101 California Street, 48th Floor San Francisco, California 94111

Twilio Inc. 375 Beale Street, Suite 300 San Francisco, CA 94105 Attention: General Counsel

or to such addresses as may hereafter be designated by either party in writing. Notwithstanding the foregoing, Sublandlord may also always deliver any notice to Subtenant at the Premises. Notices delivered personally or sent same-day courier will be effective immediately upon delivery to the addressee at the designated address; notices sent by overnight courier will be effective one (1) Business Day after acceptance by the service for delivery; notices sent by mail will be effective two (2) Business Days after mailing.

23. SURRENDER OF PREMISES.

The voluntary or other surrender of this Sublease by Subtenant, or a mutual cancellation hereof, shall not work a merger, and shall, at the option of Sublandlord, operate as an assignment to it of any sub-subleases or sub-subtenancies.

24. GENERAL PROVISIONS.

24.1 Entire Agreement. This Sublease and Exhibits A — C attached hereto contain all of the agreements of the parties with respect to the subject matter hereof, and there are no verbal or other agreements which modify or affect this Sublease. This Sublease and Exhibits A — C attached hereto supersede any and all prior agreements made or executed by or on behalf of the parties hereto regarding the Premises.

24.2 <u>Terms and Headings</u>. The words "Sublandlord" and" Subtenant" include the plural as well as the singular, and words used in any gender include all genders. The titles to sections of this Sublease are not a part of this Sublease and shall have no effect upon the construction or interpretation of any part hereof.

24.3 Successors and Assigns. All of the covenants, agreements, terms and conditions contained in this Sublease shall inure to and be binding upon Sublandlord and Subtenant and their respective permitted successors and assigns.

24.4 Brokers. Subtenant represents and warrants to Sublandlord that, except with respect to CBRE, Inc. ("Subtenant's Broker"), Subtenant has not engaged any broker, finder or other person who would be entitled to any commission or fees in respect of the negotiation, execution or delivery of this Sublease, and Subtenant shall indemnify, defend and hold harmless Sublandlord against any loss, cost, liability or expense incurred by Sublandlord as a result of any claim asserted by any such broker, finder or other person (other than Subtenant's Broker"), Sublandlord against any loss, cost, liability or expense incurred by Sublandlord represents and warrants to Subtenant that, except with respect to CBRE, Inc. ("Sublandlord's Broker"), Sublandlord has not engaged any broker, finder or other person who would be entitled to any commission or fees in respect of the negotiation, execution or delivery of this Sublease, and Sublandlord's Broker"), Sublandlord has not engaged any broker, finder or other person who would be entitled to any commission or fees in respect of the negotiation, execution or delivery of this Sublease, and Sublandlord's Broker"), Sublandlord has not engaged any broker, finder or other person who would be entitled to any commission or fees in respect of the negotiation, execution or delivery of this Sublease, and Sublandlord's Broker"), Sublandlord has not engaged any broker, finder or other person who would be entitled to any commission or fees in respect of the negotiation, execution or delivery of this Sublease, and Sublandlord's Broker"), Sublandlord has not engaged any broker, finder or other person (other than Sublandlord's Broker") on the basis of any arrangements or agreements made or alleged to have been made by or on behalf of Sublandlord.

Sublandlord shall pay all commissions due to Subtenant's Broker and Sublandlord's Broker arising out of this Sublease pursuant to a separate written agreement.

24.5 Liability of Sublandlord; Limitations on Subtenant's Remedies. No officer, director, employee or shareholder of Sublandlord, nor any parent, subsidiary or affiliate of Sublandlord shall have or incur any personal liability whatsoever with respect to this Sublease. With respect to any provision of this Sublease that specifically requires that Sublandlord shall not unreasonably withhold, unreasonably condition or unreasonably delay its consent or approval. Subtenant in no event shall be entitled to make, nor shall Subtenant make, any claim, and Subtenant hereby waives any claim, for any sum of money whatsoever as damages, costs, expenses, attorneys' fees or disbursements, whether affirmatively or by way of setoff, counterclaim or defense, based upon any claim or assertion by Subtenant that Sublandlord has unreasonably withheld, unreasonably conditioned or unreasonably delayed such consent or approval. Subtenant's sole remedy for claimed unreasonable withholding or unreasonably delaying by Sublandlord of its consent or approval shall be an action or proceeding brought and prosecuted solely at Subtenant's own cost and expense to enforce such provision, for specific performance, injunction or declaratory judgment.

24.6 Liability of Subtenant. No officer, director, employee or shareholder of Subtenant, nor any parent, subsidiary or affiliate of Subtenant shall have or incur any personal liability whatsoever with respect to this Sublease.

24.7 Severability. Any provision of this Sublease which shall prove to be invalid, void or illegal shall in no way affect, impair or invalidate any other provision hereof, and the remaining provisions hereof shall nevertheless remain in full force and effect.

24.8 Examination of Sublease. Submission of this instrument for examination or signature by Subtenant does not constitute a reservation of or option to sublease, and it is not effective as a sublease or otherwise unless and until (i) the execution by and delivery to both Sublandlord and Subtenant, and (ii) Master Landlord consents hereto as provided in Section 3 above.

24.9 <u>Recording</u>. Neither Sublandlord nor Subtenant shall record this Sublease or any memorandum hereof without the written consent of the other and any attempt by Subtenant to do the same shall constitute an immediate and uncurable default by Subtenant under this Sublease.

24.10 <u>Authorized Signatory</u>. Both parties hereby represent and warrant to the other party that the person executing this Sublease is a duly authorized representative of the signing party and has full authority to execute and deliver this Sublease.

24.11 Survival of Obligations. All provisions of this Sublease which require the payment of money or the delivery of property after the termination of this Sublease or require Subtenant to indemnify, defend or hold Master Landlord (and its associated parties) harmless shall survive the expiration or earlier termination of this Sublease.

24.12 <u>Counterparts</u>. This Sublease may be executed in one or more counterparts, each of which shall be deemed an original, but all of which when taken together will constitute one and the same instrument. The parties hereto consent and agree that this Sublease may be signed and/or transmitted by facsimile, e-mail of a .pdf document or using electronic signature technology (e.g., via DocuSign or similar electronic signature technology), and that such signed electronic record shall be valid and as effective to bind the party so signing as a paper copy bearing such party's hand-written signature. The parties further consent and agree that (i) to the extent a party signs this Sublease using electronic signature

technology, by clicking "sign", such party is signing this Sublease electronically and (ii) the electronic signatures appearing on this Sublease shall be treated, for purposes of validity, enforceability and admissibility, the same as hand-written signatures.

24.13 OFAC List.

(d)

24.13.1 For purposes of this Sublease, the following terms shall have the following meanings:

(a) "Sanctions Law" means any laws or regulations concerning economic and trade sanctions and restrictions, such as those administered by the United Nations, United States authorities (such as those enforced by the United States Department of the Treasury, the United States Department of Commerce, and/or the United States Department of State), Her Majesty's Treasury, and the European Union. These include but are not limited to laws or regulations governing anti-terrorism, and anti-money laundering activities, and include the United States Bank Secrecy Act, the United States Money Laundering Control Act of 1986, Executive Order No. 13224, and Title 3 of the USA Patriot Act (defined below), and any regulations promulgated under any of them, each as may be amended from time to time.

(b) "Executive Order No. 13224" means Executive Order No. 13224 on Terrorist Financing effective September 24, 2001, and relating to "Blocking Project and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism," as may be amended from time to time.

(c) **"Prohibited Person**" means (i) a person or entity that is listed in, or owned or controlled by a person or entity that is listed in, the Annex to Executive Order No. 13224 or has been subsequently sanctioned under that authority; (ii) a person or entity with whom Subtenant is prohibited from dealing or otherwise engaging in any transaction by any Sanctions Law; or (iii) a person or entity that is named as a "specially designated national and blocked person" on the most current list published by the U.S. Treasury Department's Office of Foreign Assets Control at its official website, http://sanctionssearch.ofac.treas.gov/, or at any replacement website or other official publication of such list, or a legal or natural person or entity owned or controlled by such a listed person.

Law 107-56).

"USA Patriot Act" means the "Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001" (Public

24.13.2 Subtenant is not, nor at any time during the Term will be: (i) in violation of any Sanctions Law; (ii) conducting any business or engaging in activities impermissible under Sanctions Law, including providing support for the proliferation of weapons of mass destruction, narcotics trafficking, the financing of terror, or organized crime, or any transaction or dealing with any Prohibited Person, including the making or receiving of any contribution of funds, goods or services to or for the benefit of any Prohibited Person or any legal or natural person owned or controlled by a Prohibited Person; (iii) dealing in, or otherwise engaging in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224 or other Sanctions Law; (iv) included on any other list whereby the provision of the services contemplated in this agreement would be in violation of Sanctions Law; or (v) engaging in or conspiring to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate any of the prohibited Person natural is not a Prohibited Person. has no reason to believe that it or any of its affiliated parties is the target of an investigation by authorities that could result in it becoming a Prohibited Person or at any time during the Term will it be, a Prohibited Person.

24.14 Appendices and Riders. The following appendices and riders are attached hereto and by this reference made a part of this Sublease:

EXHIBIT A	Master Lease
EXHIBIT B	FF&E
EXHIBIT C	Form of Letter of Credit

INTENTIONALLY DELETED.

25

26. CERTIFIED ACCESS SPECIALIST.

For purposes of Section 1938 of the California Civil Code and to Sublandlord's actual knowledge, Sublandlord hereby discloses to Subtenant, and Subtenant hereby acknowledges, that the Premises have not undergone inspection by a Certified Access Specialist ("CASp"). As required by Section 1938(e) of the California Civil Code, Sublandlord hereby states as follows: "A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction-related accessibility standards within the premises." In furtherance of the foregoing, Sublandlord and Subtenant hereby agree as follows: (i) any CASp inspection requested by Subtenant shall be conducted, at Subtenant's sole cost and expense, by a CASp reasonably approved in advance by Sublandlord; and (ii) the parties' rights and obligations with respect to making any repairs within the Premises to correct violations of construction-related accessibility standards shall be in accordance with the Master Lease (to the extent incorporated herein) and <u>Section 14</u> above. The foregoing verification is included in this Sublease solely for the purpose of complying with California Civil Code Section 1938 and shall not in any manner affect Sublandlord's and Subtenant's respective responsibilities for compliance with construction-related accessibility standards as provided under this Sublease.

27. ANTI-CORRUPTION.

In accordance with the U.S. Foreign Corrupt Practices Act (15 U.S.C. Section 78dd-1, et. seq.), Subtenant affirms that it has not and agrees that it will not, in connection with this Sublease (and any services provided thereunder) offer, promise, agree to make or authorize any corrupt or improper payment or transfer of anything of value, any benefit, or any advantage, directly or indirectly to: (i) any Government Official; (ii) any director, officer, employee, representative or agent of Sublandlord or any of its affiliates; or (iv) any other person or entity if such payment or transfers of or acquiescence in extortion, kickbacks or other unlawful or improper means of obtaining business or any improper advantage. Subtenant shall promptly inform Sublandlord oup obecoming aware of any possible violations of this sublease. Subtenant uses in the performance of Subtenant's obligations under this Sublease. For purposes of this Sublease, "Government Official" means anyone that is, works for, or on the behalf of (a) a national, regional, municipal, or local

government; (b) a department, agency, subsidiary, or branch of a national, regional, municipal, or local government; (c) a government-owned or government-controlled company (for example, a state-owned oil company, bank, airline, hospital, university, etc.); (d) a subsidiary of a government-owned or government-controlled company; (e) a public international organization (for example, the International Monetary Fund, the United Nations, the World Bank, the World Trade Organization, etc.); (f) a member of a royal family; or (g) a political party, political party official, or candidate for political office.

28. FURNITURE, FIXTURES AND EQUIPMENT.

As an economic concession to Subtenant, Subtenant shall be allowed to use all of Sublandlord's furniture, fixtures and equipment located within the Premises as of the applicable Commencement Date as more particularly set forth on <u>Exhibit B</u> attached hereto (collectively, the "FF&E"). The FF&E shall be delivered to Subtenant "AS-IS, WHERE-IS" and Sublandlord shall have no obligation to maintain, repair or replace any of the FF&E; provided, however, that Sublandlord hereby represents and warrants that it currently holds title to the FF&E free and clear of any liens or encumbrances and provided, further that in the event there are any liens or encumbrances against the FF&E. Sublandlord shall be responsible, at its sole cost, for satisfying such liens or encumbrances and Subtenant shall have no obligation with respect thereto. During the Term, Subtenant shall keep the FF&E in good condition and repair and shall not remove the FF&E from the Premises; provided, however, that Subtenant shall be permitted to remove the FF&E from the Premises during any period when Subtenant is renovating or remodeling the Premises. At the expiration of the Term or earlier termination of this Sublease as to each Premises Component, Subtenant shall purchase the FF&E located within such Premises Component from Sublandlord for the sum of One Dollar (\$1.00). Except as set forth in the previous sentence, Subtenant shall surrender each Premises Component with all FF&E removed in accordance with Section 13.1 above.

29. FINANCIAL STATEMENTS.

Subtenant represents and warrants that the financial information provided by Subtenant to Sublandlord regarding Subtenant's financial condition is true, complete and correct in all respects. Subtenant acknowledges that Sublandlord has relied upon the financial information provided by Subtenant in Sublandlord's determination to enter into and execute this Sublease.

30. STORAGE SPACE.

Subtenant acknowledges and agrees that the Premises do not include any Storage Space and to the extent that Sublandlord has leased or licensed any Storage Space under the Master Lease or otherwise Sublandlord anticipates that it will terminate such lease or license on or prior to the Commencement Date of the Second Increment. Upon the reasonable request of Subtenant, Sublandlord shall use good faith efforts to assist Subtenant in connection with any lease or license of storage space in the Building.

[Remainder of page intentionally blank; signatures follow on next page]

SUBLANDLORD:

SALESFORCE.COM, INC., a Delaware corporation

By:	/s/ George Kreitem	
Name:	George Kreitem	
Its:	SVP Real Estate	

SUBTENANT:

TWILIO INC., a Delaware corporation

By: Name:

/s/ Jeff Lawson Jeff Lawson Chief Executive Officer

[Signature Page to Sublease]

Its:

EXHIBIT A

MASTER LEASE

(See Attached)

OFFICE LEASE

This Office Lease (the "Lease"), dated as of the date (the "Effective Date") set forth in Section 1 of the Summary of Basic Lease Information (the "Summary"), below, is made by and between HUDSON RINCON CENTER, LLC, a Delaware limited liability company ("Landlord"), and salesforce.com, inc., a Delaware corporation ("Tenant").

SUMMARY OF BASIC LEASE INFORMATION

TERMS OF LEASE		DESCRIPTION
1.	Effective Date:	December 28, 2012
2.	Premises (<u>Article 1</u>).	
	2.1 Building:	101 Spear Street, San Francisco, California 94105, commonly known as One Rincon. Landlord and Tenant hereby agree that the Building contains a total rentable area of 264,718 rentable square feet of office space, as well as additional retail space.
	2.2 Tranche I Premises:	A total of approximately 93,028 rentable square feet of space, consisting of (i) approximately 74,966 rentable square feet of space located on the third (3 rd) floor of the Building (the "3 rd Floor Premises"), and (ii) approximately 18,062 rentable square feet of space located on the ground floor of the Building (the "Ground Floor Premises") (collectively, the "Tranche I Premises"), as further set forth in Exhibit A-1 to this Lease.
	2.3 Tranche II Premises:	A total of approximately 59,689 rentable square feet of space, consisting of (i) approximately 41,614 rentable square feet of space located on the fifth (5 th) floor of the Building (the "5 th Floor Premises"), and (ii) approximately 18,075 rentable square feet of space located on the fourth (4 th) floor of the Building and commonly known as Suite 450 (the "4 th Floor Premises") (collectively, the "Tranche II Premises"), as further set forth in <u>Exhibit A-2</u> to this Lease.
	2.4 Tranche III Premises:	A total of approximately 83,016 rentable square feet of space, consisting of (i) approximately 76,004 rentable square feet of space located on the second (2 nd) floor of the Building (the " 2nd Floor Premises "), (ii) approximately 2,868 rentable square feet of space located on the second (2 nd) floor of the Building and commonly known as Suite 250 (the " Suite 250 Premises "), and (iii) approximately 4,144 rentable square feet of space located on the second (2 nd) floor as Suite 255 (the " Suite 255 Premises ") (collectively, the " Tranche III Premises "), as further set forth in <u>Exhibit A-3</u> to the Office Lease.

		The term " Portion of the Premises " as used in this Lease shall mean either the Ground Floor Premises, the 2 nd Floor Premises, the Suite 250 Premises, the Suite 255 Premises, the 3 rd Floor Premises or the 5 th Floor Premises, as applicable.
		The term " Tranche " as used in this Lease shall mean the Tranche I Premises, the Tranche II Premises, or the Tranches III Premises, as applicable.
		Landlord and Tenant hereby agree that the Tranche I Premises, the Tranche II Premises, and the Tranche III Premises (collectively, the " Premises ") contains a total rentable area of 235,733 rentable square feet.
3.	Lease Term (<u>Article 2</u>).	
	3.1 Length of Term:	With respect to the Tranche I Premises, fifteen (15) years from the Tranche I Commencement Date (as defined below).
		With respect to the Tranche II Premises, thirteen (13) years from the Tranche II Commencement Date (as defined below).
		With respect to the Tranche III Premises, eleven (11) years from the 2^{nd} Floor Commencement Date (as defined below).
	3.2 Commencement Date:	With respect to the Tranche I Premises, the lease commencement date shall be the later of (A) November 1, 2013 (the " Tranche I Anticipated Delivery Date ") and (B) the date Landlord delivers the Tranche I Premises to Tenant in the "Delivery Condition," as that term is defined in the Tenant Work Letter (as defined below) (the " Tranche I Commencement Date "). Tenant shall be permitted to conduct business in the Tranche I Premises on and after the Tranche I Commencement Date without payment of Base Rent until the Rent Commencement Date with respect to the Tranche I Premises.
		With respect to the Tranche II Premises, the lease commencement date shall be the later of (A) May 1, 2014 (the " Tranche III Anticipated Delivery Date ") and (B) the date Landlord delivers the Tranche II Premises to Tenant in the Delivery Condition (the " Tranche II Commencement Date "). Tenant shall be permitted to conduct business in the Tranche II Premises on and after the Tranche II Commencement Date without payment of Base Rent until the Rent Commencement Date with respect to the Tranche II Premises.
		2

With respect to the 2nd Floor Premises, the lease commencement date shall be the later of (A) August 1, 2014 (the "2nd Floor Premises Anticipated Delivery Date") and (B) the date Landlord delivers the 2nd Floor Premises to Tenant in the Delivery Condition (the "2nd Floor Premises Commencement Date"). Tenant shall be permitted to conduct business in the 2nd Floor Premises on and after the 2nd Floor Premises Commencement Date without payment of Base Rent until the Rent Commencement Date with respect to the 2nd Floor Premises.

With respect to the Suite 250 Premises, subject to the terms of <u>Section 1.1.4</u> below, the lease commencement date shall be the later of (A) August 1, 2015 (the "**Suite 250 Anticipated Delivery Date**") and (B) the date Landlord delivers the Suite 250 Premises to Tenant in the Delivery Condition (the "**Suite 250 Commencement Date**"). Tenant shall be permitted to conduct business in the Suite 250 Premises on and after the Suite 250 Commencement Date without payment of Base Rent until the Rent Commencement Date with respect to the Suite 250 Premises.

With respect to the Suite 255 Premises, subject to the terms of <u>Section 1.1.4</u> below, the lease commencement date shall be the later of (A) May 1, 2017 (the "**Suite 255 Anticipated Delivery Date**") and (B) the date Landlord delivers the Suite 255 Premises to Tenant in the Delivery Condition (the "**Suite 255 Commencement Date**"). Tenant shall be permitted to conduct business in the Suite 255 Premises on and after the Suite 255 Premises.

With respect to each Portion of the Premises, all references in this Lease to the "Anticipated Delivery Date" shall mean and refer to the "Tranche I Anticipated Delivery Date," the "Tranche II Anticipated Delivery Date," the "2nd Floor Premises Anticipated Delivery Date," the "Suite 250 Anticipated Delivery Date," or the "Suite 255 Anticipated Delivery Date," as applicable.

With respect to each Portion of the Premises, the term "Commencement Date" shall mean the lease commencement date applicable to such Portion of the Premises as set forth above.

With respect to each Portion of the Premises, the date that is nine (9) months after the applicable Commencement Date with respect to such Portion of the Premises.

3.3 Rent Commencement Date:

With respect to the Tranche I Premises, if the Tranche I Commencement Date shall be the first day of a calendar month, then the day immediately preceding the fifteenth (15th) anniversary of the Tranche I Commencement Date; or if the Tranche I Commencement Date shall be other than the first day of a calendar month, then the last day of the month in which the fifteenth (15th) anniversary of the Tranche I Commencement Date occurs (the "**Tranche I Lease Expiration Date**").

With respect to the Tranche II Premises, if the Tranche II Commencement Date shall be the first day of a calendar month, then the day immediately preceding the thirteenth (13th) anniversary of the Tranche II Commencement Date shall be other than the first day of a calendar month, then the last day of the month in which the thirteenth (13th) anniversary of the Tranche II Commencement Date occurs (the "Tranche II Lease Expiration Date").

With respect to the 2nd Floor Premises, if the Tranche III Commencement Date shall be the first day of a calendar month, then the day immediately preceding the eleventh (11th) anniversary of the Tranche III Commencement Date; or if the Tranche III Commencement Date shall be other than the first day of a calendar month, then the last day of the month in which the eleventh (11th) anniversary of the Tranche III Commencement Date shall be then the first day of a calendar month, then the last day of the month in which the eleventh (11th) anniversary of the Tranche III Commencement Date occurs (the "Tranche III Lease Expiration Date").

With respect to each of the Suite 250 Premises and the Suite 255 Premises, the Tranche III Lease Expiration Date.

With respect to each Portion of the Premises, the term "Lease Expiration Date" shall mean the lease expiration date applicable to such Portion of the Premises as set forth above.

With respect to each Tranche, two (2) options to renew for a period of five (5) years each, as more particularly set forth in Section 2.2 of this Lease.

4

3.5 Option Terms:

4. Base Rent (<u>Article 3</u>):

4.1 Tranche I Premises:

Year*	Monthly Installment of Base Rent	Annual Rental Rate per Rentable Square Foot
1		Redacted
1		
2		
3		
4		
5		
6		
7		
8		
9		
10		
11		
12		
13		
14		
15		
2 Tranche II Premises:		
Year*	Monthly Installment	Annual Rental Rate per Rentable Square Foot

Lease Year*	of Base Rent	Rentable Square Foot
		Redacted
1		
2		
3		
4		
5		

		Redacted
5		
6		
7		
8		
9		
10		
11		
12		
13		

4.3 2nd Floor Premises:

Lease Year*		Monthly Installment of Base Rent	Annual Rental Rate per Rentable Square Foot
			Redacted
1			
2			
2			
3			
-			
4			
5			
6			
7			
8			
8			
9			
10			
11			
11			

*Subject to the terms of Section 2.1, the first (1st) Lease Year shall commence upon the applicable Commencement Date for the applicable Portion of the Premises (i.e., on the Tranche I Commencement Date for the Tranche I Premises, on the Tranche II Commencement Date for the Tranche II.

**Subject to the terms of Section 1.1.4 below, notwithstanding the Lease Year, on the Rent Commencement Date for the Suite 250 Premises or the Suite 255 Premises, as the case may be, the Annual Rental Rate per Rentable Square Foot for the Suite 250 Premises, or the Suite 255 Premises, as applicable, shall be the then-scheduled Annual Rental Rate per Rentable Square Foot for the 2nd Floor Premises.

Exhibit B attached hereto sets forth the Monthly Installment of Base Rent for each month of the Lease Term assuming that each Portion of the Premises is delivered on the Anticipated Delivery Date.

5	Deer Very (Article 1):	Calar des unes 2014
5.	Base Year (<u>Article 4</u>):	Calendar year 2014.
6.	Tenant's Share (<u>Article 4</u>):	Tranche I Premises: Approximately 35.14%
		Tranche II Premises: Approximately 22.55%.
		Tranche III Premises: Approximately 31.36%, comprised of the following.
		• 2 nd Floor Premises: Approximately 28.71%.
		• Suite 250 Premises: Approximately 1.08%.
		• Suite 255 Premises: Approximately 1.57%.
7.	Permitted Use (<u>Article 5</u>):	General office use and training and any other legally permitted use consistent with a first-class office building. Subject to Landlord review and approval of the specific location and installation details in accordance with the terms of the Tenant Work Letter or <u>Article 8</u> of this Lease, Tenant shall be permitted to have a fully developed private corporate food facility for its employees and provide childcare in the Building.
8.	Letter of Credit (<u>Article 21</u>):	Redacted subject to the terms and conditions of <u>Article 21</u> of this Lease.
9.	Parking Pass Ratio (Article 28):	One (1) unreserved parking pass for every 5,000 rentable square feet of the Premises.
10.	Address of Tenant (Section 29.17):	salesforce.com, inc. The Landmark@One Market, 3 rd Floor San Francisco, California 94105 Attention: <u>General Counsel</u>
		With a copy to:
		salesforce.com, inc. The Landmark@One Market, Third Floor San Francisco, California 94105 Attention: Vice President of Real Estate
		With a copy to:
		Paul Hastings LLP 55 2 nd Street, 24 th Floor San Francisco, California 94105
		7

11.	Address of Landlord (Section 29.17):	Hudson Rincon Center, LLC 11601 Wilshire Boulevard, Suite 1600 Los Angeles, California 90025
		With a copy to:
		Allen Matkins Leck Gamble Mallory & Natsis LLP 1901 Avenue of the Stars, Suite 1800 Los Angeles, California 90067
12.	Brokers (Section 29.23):	<u>Tenant's Broker:</u> Cushman & Wakefield One Maritime Plaza, Suite 900 San Francisco, California 94111
		<u>Landlord's Broker</u> : The CAC Group 255 California Street, 2 nd Floor San Francisco, California 94111
13.	Landlord's Wiring Instructions (Article 3):	See Schedule 1 attached to this Lease.
		8

ARTICLE 1

PREMISES, BUILDING, PROJECT, AND COMMON AREAS

1.1 Premises, Building, Project and Common Areas.

1.1.1 The Premises. Landlord hereby leases to Tenant and Tenant hereby leases from Landlord the premises set forth in <u>Section 2</u> of the Summary (the "Premises"). The outline of the Premises is set forth in <u>Exhibit A</u> attached hereto and each floor or floors of the Premises has approximately the number of rentable square feet as set forth in <u>Section 2</u> of the Summary. The parties hereto agree that the lease of the Premises is upon and subject to the terms, covenants and conditions herein set forth, and Landlord and Tenant each covenant as a material part of the consideration for this Lease to keep and performe each and all of such performance. The parties hereto hereby acknowledge that the purpose of <u>Exhibit A</u> is to show the approximate location of the Premises in the "Building," as that term is defined in <u>Section 1.1.2</u>, below, only, and such Exhibit is not mecant to constitute an agreement, representation or warranty as to the construction of the Premises, the precise area thereof or the specific location of the "Common Areas," as that term is defined in <u>Section 1.1.2</u>, below, or the elements thereof or of the accessways to the Premises or the "Project," as that term is defined in <u>Section 1.1.2</u>, below. Except as specifically set forth in this Lease and in the Tenant Work Letter attached hereto as <u>Exhibit C</u> (the "**Tenant Work Letter**"), Landlord shall not be obligated to provide or pay for any improvement work or services related to the improvement of the Premises. Tenant also acknowledges that neither Landlord nor any agent of Landlord has made any representation or warranty regarding the condition of the Premises, the Building or the Project or with respect to the suitability of any of the foregoing for the conduct of Tenant's business, except as specifically set forth in this Lease and in the Tenant Work Letter.

1.1.2 <u>The Building and The Project</u>. The Premises are a part of the building set forth in <u>Section 2.1</u> of the Summary (the "Building"). The Building is part of a mixed use office, retail, and residential project currently known as "Rincon Center" The term "Project," as used in this Lease, shall mean (i) the Building and the Common Areas, including the underground parking facility (the "Parking Garage"), (ii) the building located adjacent to the Building at 121 Spear Street, San Francisco, California 94105 ("Two Rincon"), and (iii) the land (which is improved with landscaping, the Parking Garage and other improvements) upon which the Building, Two Rincon, and the Common Areas are located.

1.1.3 Common Areas. Tenant shall have the non-exclusive right to use in common with other tenants in the Project, and subject to the Rules and Regulations (as defined below) and Landlord's reasonable access control procedures and systems, those portions of the Project which are provided, from time to time, for use in common by Landlord, Tenant and any other tenants of the Project such as entrances, lobbies, fire vestibules, restrooms, mechanical areas, ground floor corridors, elevators and elevator foyers, the Parking Garage, electrical and janitorial closets, telephone and equipment rooms, loading and unloading areas, the Project's plaza areas, if any, ramps, drives, stairs, and similar access ways and service ways and other common areas and facilities in and adjacent to the Building and the Project (such areas, together with such other portions of the Project designated by Landlord, in its discretion, including certain areas designated for the exclusive use of certain tenants, or to be shared by Landlord and certain tenants, are collectively referred to herein as the "Common Areas." The Common Areas, "as used in this Lease, shall mean the portions of the Project designated as such by Landlord. The term "Building Common Areas," as used in this Lease, shall mean the portions of the Common Areas are maintained and operated shall be at the reasonable discretion of Landlord (but shall at least be consistent with the manner in which the common areas of Comparable Buildings (as defined below) are operated and maintained). Landlord argees that, if Tenant so elects and appoints a representative, then, upon Tenant's written request made not more often than on a quarterly basis, Landlord shall meet and confer with Tenant's representative regarding the manner in which the Common Areas in a manner that would materially affect Tenant's use of or access to the Project and the Common Areas; provided, that, Landlord may not modify the Building Common Areas in a manner that would materially affect Tenant's use of or access to

delayed. Except when and where Tenant's right of access is specifically excluded in this Lease, Tenant shall have the right of access to the Premises, the Building, the Parking Garage, and the Common Areas twentyfour (24) hours per day, seven (7) days per week during the "Lease Term," as that term is defined in <u>Section 2.1</u>, below.

Delivery of Premises. Landlord shall deliver each Portion of the Premises to Tenant in Delivery Condition on the applicable Anticipated Delivery Date set forth in Section 3.2 of the Summary for such Portion of the Premises. If Landlord does not deliver any Portion of the Premises in Delivery Condition within thirty (30) days after the applicable Anticipated Delivery Date for such Portion of the Premises (each, an "Outside Delivery Date"), then Tenant shall receive an abatement of Base Rent with respect to such Portion of the Premises for the period of time equal to (x) for the first sixty (60) days after the applicable Outside Delivery Date, one (1) day of Base Rent abatement for every one (1) full day after the Outside Delivery Date until Landlord delivers such Portion of the Premises to Tenant in Delivery Condition, and (y) after the first sixty (60) days after the applicable Outside Delivery Date (each, a "Second Outside Delivery Date"), two (2) days of Base Rent for every one (1) full day after the Second Outside Delivery Date until Landlord deliver such Portion of the Premises to Tenant in Delivery Condition (the "Base Rent Abatement Credits"). Tenant may immediately apply any accrued Base Rent Abatements Credits against payments of Rent as they become due. The Outside Delivery Date, the Second Outside Delivery Date and the "Blow Out Date" (as that term is defined in this Section 1.1.4 below) shall be extended to the extent of any delay or delays caused by Force Majeure Delays (as defined in Section 29.15, below, provided that neither Landlord's failure to cause the prior tenant of the Tranche I Premises to vacate the same, nor Landlord's failure to obtain any required permits or approvals, unless as a result of a general failure of the applicable governmental agencies in the granting of permits or approvals, shall be deemed to be a Force Majeure Delay for the purposes of this Section 1.1.4). If Landlord does not deliver the Tranche I Premises on or before March 31, 2014 (the "Blow Out Date"), then Tenant shall have the right to deliver a notice to Landlord (a "Blow Out Notice") at any time thereafter electing to terminate this Lease, in which event, (x) this Lease shall terminate, and (y) Landlord shall return any prepaid rent and the L-C (as defined below) to Tenant with five (5) business days. The Blow Out Notice must be delivered by Tenant to Landlord, if at all, not earlier than the Blow Out Date (as the same may be extended pursuant to the terms of this Section 1.1.4). Notwithstanding anything to the contrary set forth in this Section 1.1.4 or elsewhere in this Lease, Tenant hereby acknowledges and agrees that if Landlord delivers the Tranche I Premises to Tenant in Delivery Condition any time after the Blow Out Date, but prior to Tenant's delivery of any Blow Out Notice, then Tenant's right to terminate this Lease pursuant to this Section 1.1.4 shall automatically terminate and be of no further force or effect. Additionally, notwithstanding anything to the contrary set forth in this Section 1.1.4 or elsewhere in this Lease, to the extent Landlord is able to deliver the Suite 250 Premises and/or the Suite 255 Premises to Tenant earlier than the applicable Anticipated Delivery Date set forth in Section 3.2 of the Summary, such applicable Anticipated Delivery Date(s) shall be accelerated accordingly; provided, however, (i) in no event shall the Commencement Date for the Suite 250 Premises and/or the Suite 255 Premises occur prior to the 2nd Floor Premises Commencement Date; and (ii) Landlord shall provide Tenant with at least six (6) months prior notice of the new Anticipated Delivery Date for the Suite 250 Premises and/or the Suite 255 Premises, as the case may be, which date as set forth in Landlord's notice shall then be the "Anticipated Delivery Date" for the Suite 250 Premises and/or the Suite 255 Premises, as applicable, for all purposes of this Lease (e.g. the remedies set forth in this Section 1.1.4 shall apply as if the date set forth in Landlord's notice were the Anticipated Delivery Date).

1.2 <u>Rentable Square Feet of Premises and Building</u>. Notwithstanding anything set forth in this Lease to the contrary, Landlord and Tenant hereby stipulate and agree that the rentable area of the Building and the Premises is as set forth in <u>Section 2</u> of the Summary.

1.3 **Right of First Refusal to Lease Space in Building**. Subject to the terms and conditions of this Section 1.3, below, Landlord hereby grants to the Tenant originally named in this Lease (the "Original Tenant") and its Permitted Transferees (as defined below) an ongoing right of first refusal with respect to any vacant space in the Building, not including any ground floor retail space (the "First Refusal Space"). Notwithstanding the foregoing, such first refusal right of Tenant shall commence only following the expiration or earlier termination of the existing leases (including renewals and extensions pursuant to rights currently existing in such existing leases and exercised strictly in accordance with the terms and conditions set forth in such existing leases) of the First Refusal Space, and such right of first refusal shall also be subordinate to all expansion, first offer, first negotiation and other similar rights which relate to the First Refusal Space and which rights are set forth in leases of space in the Project existing as of the Effective Date and which are exercised strictly in accordance with the terms and conditions

set forth in such existing leases (all such tenants under existing leases of the First Offer Space and other tenants of the Project, collectively, the "Building One Superior Right Holders").

1.3.1 <u>Procedure for Offer</u>. Landlord shall notify Tenant (the "First Refusal Notice") from time-to-time when and if Landlord has agreed or is willing to agree to the fundamental economic terms (including term, rent and other material economic terms and conditions) for lease of the First Refusal Space to a third party (other than a Building One Superior Right Holder) (the "Refusal Terms"). Pursuant to such First Refusal Notice, Landlord shall offer to lease to Tenant the applicable First Refusal Space on the Refusal Terms. The First Refusal Notice shall describe the applicable First Refusal Space, and the applicable Refusal Terms.

1.3.2 <u>Procedure for Acceptance</u>. If Tenant wishes to exercise Tenant's right of first refusal with respect to the First Refusal Space described in the First Refusal Notice, then within ten (10) business days of delivery of the First Refusal Notice to Tenant, Tenant shall deliver notice (a "**First Refusal Acceptance Notice**") to Landlord of Tenant's exercise of its right of first refusal with respect to all of the First Refusal Space described in the First Refusal Notice on all of the Refusal Terms.

1.3.3 Construction in First Refusal Space. Subject to any concessions granted to Tenant in accordance with the Refusal Terms, Tenant shall take such First Refusal Space in its "as is" condition, and the construction of improvements in such First Refusal Space shall comply with the terms of <u>Article 8</u> of this Lease. The terms of the Tenant Work Letter attached to this Lease as <u>Exhibit C</u> shall not apply to the construction of any improvements in such First Refusal Space.

1.3.4 <u>Amendment to Lease</u>. If Tenant timely exercises Tenant's right of first refusal to lease First Refusal Space as set forth herein, Landlord and Tenant shall within thirty (30) days thereafter execute an amendment to this Lease (a "First Refusal Space Amendment") for such First Refusal Space pursuant to this <u>Section 1.3</u>. Tenant's lease of such First Refusal Space shall be upon the express terms set forth in the First Refusal Notice, but otherwise upon the terms and conditions set forth in the Refusal Terms (the "First Refusal Space shall commence upon such commencement date as is set forth in the Refusal Terms (the "First Refusal Commencement Date"), and shall terminate on the Tranche II Lease Expiration Date; provided, however, notwithstanding the foregoing or anything set forth in this Lease to the contrary, the minimum lease term for Tenant's lease of any First Refusal Be five (5) years, and accordingly, if the Tranche II Lease Expiration Date is less than five (5) years from the First Refusal Commencement Date, Tenant's lease of the First Refusal Space will extend beyond Tenant's lease of the Tranche II Premises under this Lease. Any economic concessions set forth in the Refusal Terms shall be equitably adjusted to account for any difference in term set forth in the Refusal Terms and the actual term of Tenant's lease of its right to lease the First Refusal Space; provided, however, an otherwise valid exercise of Tenant's right of first refusal shall be of full force and effect irrespective of whether the First Refusal Space Amendment, if applicable, shall be executed by Landlord and Tenant within thirty (30) days following Tenant's exercise of its right to lease the First Refusal Space; provided, however, an otherwise valid exercise of Tenant's right of first refusal shall be of full force and effect irrespective of whether the First Refusal Space Amendment is tringly signed by Landlord and Tenant.

1.3.5 <u>Termination of First Refusal Right</u>. The rights contained in this <u>Section 1.3</u> shall be personal to the Original Tenant and its "Permitted Transferees" as defined in <u>Section 14.7</u>, below (and not any other assignee, sublessee or other transferee of the Original Tenant's interest in this Lease). The right to lease First Refusal Space as provided in this <u>Section 1.3</u> may not be exercised if, as of the date of the attempted exercise of the expansion option by Tenant, or, at Landlord's option, as of the scheduled date of delivery of such First Refusal Space to Tenant, Tenant is in default under this Lease (beyond any applicable notice and cure periods).

1.3.6 <u>Rejection</u>; <u>Right to Lease</u>. If Tenant fails to timely deliver to Landlord a First Refusal Acceptance Notice, then Landlord shall be free for one hundred twenty (120) days thereafter (for purposes of this <u>Section 1.3.6</u>, the "One Hundred Twenty Day Period") to lease the First Refusal Space identified in the applicable First Refusal Notice on economic terms and conditions that are not more favorable than the Refusal Terms. If Tenant fails to timely deliver to Landlord a First Refusal Acceptance Notice and Landlord either (i) desires to lease the First Refusal Space identified in such First Refusal Notice on terms and conditions more favorable than the Refusal Terms (for purposes of this <u>Section 1.3.6</u>, the "Updated Terms and Conditions") and/or (ii) fails to lease the First Refusal Space within the One Hundred Twenty Day Period, then before entering into a lease for such First

Refusal Space on the Updated Terms and Conditions or after the expiration of the One Hundred Twenty Day Period, Landlord shall again offer the First Refusal Space to Tenant for lease (on the Updated Terms and Conditions (if applicable)) (for purposes of this <u>Section 1.3.6</u>, the "**Updated Availability Notice**"). Tenant shall have ten (10) business days after receipt of the Updated Availability Notice to deliver to Landlord notice of Tenant's acceptance of such First Refusal Space on the Updated Terms and Conditions (if applicable) (for purposes of this <u>Section 1.3.6</u>, the "**Updated Offer Acceptance Notice**"). If Landlord does not receive an Updated Offer Acceptance Notice from Tenant within such ten (10) business day period (or Landlord receives a notice of rejection), then for a period of one hundred twenty (120) days thereafter Landlord shall be free to lease the First Refusal Space on economic terms and conditions that are not more favorable than the Updated Terms and Conditions (if applicable), otherwise Landlord shall again deliver an Updated Availability Notice to Tenant and the terms and conditions of this <u>Section 1.3.6</u> shall apply thereto.

1.4 **Right of First Offer to Lease Space in Two Rincon**. Subject to the terms and conditions of this Section 1.4, below, Landlord hereby grants to the Original Tenant and its Permitted Transferees an ongoing right of first offer with respect to any space that becomes available for lease at Two Rincon (the "First Offer Space"). Notwithstanding the foregoing, such first offer right of Tenant shall commence only following the expiration or earlier termination of the existing leases (including renewals and extensions pursuant to rights currently existing in such existing leases and exercised strictly in accordance with the terms and conditions set forth in such existing leases of the First Offer Space, if any, and such right of first refusal shall also be subordinate to all expansion, first offer, first negotiation and other similar rights which relate to the First Offer Space and which rights are set forth in leases of space in Two Rincon existing as of the Effective Date and which are exercised strictly in accordance with the terms and conditions set forth in such existing leases of the First Offer Space and other tenants of the Project, collectively, the "Building Two Superior Right Holders"). Tenant's right of first offer shall be on the terms and conditions set forth in this Section 1.4.

1.4.1 <u>Procedure for Offer</u>. Landlord shall notify Tenant (the "First Offer Notice") from time to time when the First Offer Space, or any portion thereof, becomes available for lease to third parties, provided that no Building Two Superior Right Holder wishes to lease such space. Pursuant to such First Offer Notice, Landlord shall offer to lease to Tenant the then available First Offer Space. The First Offer Notice shall describe the space so offered to Tenant and shall set forth the "First Offer Rent," as that term is defined in <u>Section 1.4.3</u>, below, and the other economic terms upon which Landlord is willing to lease such space to Tenant (the "First Offer Terms").

1.4.2 Procedure for Acceptance. If Tenant wishes to exercise Tenant's right of first offer with respect to the space described in the First Offer Notice, then within ten (10) business days of delivery of the First Offer Notice to Tenant, Tenant shall deliver written notice to Landlord (the "First Offer Exercise Notice") irrevocably exercising its right of first offer with respect to the entire space described in the First Offer Notice on the terms contained in such notice, except that, if Tenant wishes to dispute Landlord's determination of First Offer Rent (as defined below) set forth in the First Offer Notice, then Tenant's Toffer Notice shall so notify Landlord of such dispute (and in such event the First Offer Rent shall be determined in accordance with <u>Section 2.2.3</u> of the Lease, as if the term "Option Rent" was "First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires (an "Intervening Lease"); provided, however, that Tenant's ongoing right of first offer for such First Offer Space shall thereafter arise only following the expiration or earlier termination of such Intervening Lease (including renewals and extensions pursuant to rights in such leases and exercised strictly in accordance with the terms and conditions set forth in such leases). If Tenant timely exercises its right of first offer but fails to timely object to the First Offer Rent set forth in the First Offer Notice, then Tenant shall be deemed to have disputed Landlord's determination of First Offer Rent set forth in the First Offer Notice, the remaint shall be deemed to have disputed Landlord's determination of First Offer Rent set forth in the First Offer Notice, then Tenant shall be deemed to have disputed Landlord's determination of First Offer Rent set forth in the First Offer Notice, then Tenant shall be deemed to have disputed Landlord's determination of First Offer Rent set forth in the First Offer Notice, then Tenant shall be deemed to have disputed Landlord's determination of First Offer Rent set forth in

1.4.3 <u>First Offer Rent</u>. The rent payable by Tenant for the applicable First Offer Space (the "First Offer Rent") shall be "Fair Market Rent" for such First Offer Space, as determined in accordance with the terms of <u>Section 2.2.3</u> of this Lease as if (i) the term "Option Term" was "First Offer Term" (as defined below), (ii) the term "Option Rent" was "First Offer Rent," (iii) the term "Option Rent Notice," and (iv) the terms "Premises," "Tranche" or "floor" was "First Offer Space."

1.4.4 <u>Construction in First Offer Space</u>. Subject to any concessions granted to Tenant in accordance with the determination of the First Offer Rent for the applicable First Offer Space in accordance with <u>Section 1.4.3</u> above, Tenant shall take such First Offer Space in its "as is" condition, and the construction of improvements in such First Offer Space shall comply with the terms of <u>Article 8</u> of this Lease. The terms of the Tenant Work Letter attached to this Lease as <u>Exhibit C</u> shall not apply to the construction of any improvements in such First Offer Space.

1.4.5 <u>Amendment to Lease</u>. If Tenant timely exercises Tenant's right to lease the applicable First Offer Space as set forth herein, Landlord and Tenant shall promptly thereafter execute a lease amendment for such First Offer Space upon the terms and conditions as set forth in the First Offer Notice and this <u>Section 1.4</u>; provided, however, that an otherwise valid exercise of the such right of first offer shall be fully effective whether or not a lease amendment is executed. Tenant shall commence payment of Rent for such First Offer Space, and the term (the "First Offer Term") of such First Offer Space shall commence upon such commencement date (the "First Offer Commencement Date") as is determined as a part of the determination of the First Offer Rent in accordance with <u>Section 1.4.3</u> above, and shall terminate on the Tranche II Lease Expiration Date; provided, however, notwithstanding the foregoing or anything set forth in this Lease to the contrary, the minimum lease term for Tenant's lease of any First Offer Space shall be five (5) years, and accordingly, if the Tranche II Lease Expiration Date is less than five (5) years from the First Offer Commencement Date, Tenant's lease of the First Offer Space will extend beyond Tenant's lease of the Tranche II Premises under this Lease.

1.4.6 <u>Termination of Right of First Offer</u>. The rights contained in this <u>Section 1.4</u> shall be personal to the Original Tenant and its Permitted Transferees, and may only be exercised by the Original Tenant or a Permitted Transferee (and not any other assignee, sublessee or other transferee of the Original Tenant's interest in this Lease) if not more than fifteen percent (15%) of the rentable area of the Premises (which shall be calculated as if the Premises includes all of the Tranche I Premises, the Tranche II Premises and the Tranches III Premises increased of whether the Commencement Date with respect to the Tranche I Premises, the Tranche II Premises and the Transferees). Tenant shall not have the right to lease any First Offer Space, as provided in this <u>Section 1.4</u>, if, as of the date of the attempted exercise of any right of first offer by Tenant, or as of the scheduled date of delivery of the applicable First Offer Space to Tenant, Tenant is in default under this Lease.

1.4.7 Rejection; Right to Lease. If Tenant fails to timely deliver to Landlord a First Offer Exercise Notice, then Landlord shall be free for one-hundred eighty (180) days thereafter (for purposes of this Section 1.4.7, the "One Hundred Eighty Day Period") to lease the First Offer Space identified in the applicable First Offer Notice, on material economic terms that, on a net effective basis, are not more than five percent (5%) more favorable to the tenant than the terms offered Tenant in the applicable First Offer Notice ("Materially Equivalent Terms"). If Tenant fails to timely deliver to Landlord a First Offer Space evitinn the One Hundred Eighty Day Period or desires to lease the First Offer Space on lease than Materially Equivalent Terms, then before entering into a lease for such First Offer Space after the expiration of the One Hundred Eighty Day Period or on such less than Materially Equivalent Terms, Landlord shall again offer the First Offer Space to framt for lease (for purposes of this Section 1.4.7, the "Updated Availability Notice"), at such less than Materially Equivalent Terms, if applicable. Tenant shall have ten (10) business days after receipt of the Updated Availability Notice to deliver to Landlord notice of Tenant's acceptance of such First Offer Space of this Section 1.4.7, the "Updated Offer Acceptance Notice"). If Landlord does not receive an Updated Offer Acceptance Notice from Tenant within such ten (10) business day period (or Landlord receives a notice of rejection), then for a period of one hundred eighty (180) days thereafter Landlord shall be free to lease the First Offer Space (for purposes of this Section 1.4.7, the "Updated Offer Acceptance Notice"). If Landlord does not receive an Updated Offer Acceptance Notice from Tenant within such ten (10) business day period (or Landlord receives a notice of rejection), then for a period of one hundred eighty (180) days thereafter Landlord shall be free to lease the First Offer Space (for purposes of this Section 1.4.7, shall apply ther

1.4.8 Tenant's Share; Operating Expenses; Management Fees. Operating Expenses (as defined below) with respect to Two Rincon shall include (i) all Direct Expenses (as defined below) attributable solely to the office portions of Two Rincon, (ii) an equitable portion of the Direct Expenses attributable to Two Rincon as a whole, and (iii) an equitable portion of the Direct Expenses attributable to the Project as a whole ("Building Two Operating Expenses"). If Tenant leases space in Two Rincon, then Tenant's Share with respect to Building Two Operating Expenses shall equal the rentable square foot of Tenant's premises in Two Rincon divided

by the total rentable square footage of Two Rincon. If Tenant leases space in Two Rincon, then Building Two Operating Expenses include fees payable by Landlord for management of the Project not in excess of three percent (3%) of gross revenues for the office portions of the Two Rincon only.

1.5 Right of First Offer to Purchase Building.

Right of First Offer. If Landlord intends to Sell (as defined below) the Building or Project, as applicable, Landlord shall first deliver written notice to Tenant (a "Sale Notice"), which notice 1.5.1 shall set forth all material economic terms of the Sale (including, without limitation, transaction structure (e.g., ground lease, sale of fee interest or sale of equity interests), price, deposit(s), due diligence period (not to be less than thirty (30) days), closing date, seller financing (if any) and assumption of Landlord's debt (if any) or sale free and clear of such debt) along with a proposed purchase agreement. For the purposes hereof, "Sale," "Selling" shall mean to sell, transfer, assign or exchange the Building or Project, or grant an option to dispose of or otherwise transfer or convey the membership or other direct or indirect controlling ownership interests in the Landlord or ground lease the Building or Project. Concurrently with Landlord's delivery to Tenant of a Sale Notice, Landlord shall, deliver to Tenant (i) project operating statements (including rentals and other revenues, and Project expenses, including Operating Expenses and Taxes) for the prior two (2) calendar years, and for the most recently completed calendar quarter, and (ii) current rent roll for all leases, licenses or occupancy agreements of the Project (the "Other Leases") (such items in (i) and (ii), above, the "Initial Review Materials"). If, within ten (10) business days following the date of such notification, Tenant notifies Landlord that Tenant is interested in purchasing the Project from Landlord (the "Interest Notice"), then Landlord and Tenant shall diligently negotiate in good faith for thirty (30) days (the "Negotiation Period") from the date upon which Tenant delivers its Interest Notice to agree upon the terms of, and to execute, a purchase agreement with respect to the Building or the Project, as applicable, on the terms and conditions set forth in the Sale Notice. Upon the agreement on the terms of the purchase agreement, and subject to the terms of such purchase agreement, Landlord will provide Tenant with other customary materials and information regarding the Project, including, without limitation, (a) copies of any Other Leases, (b) a copy of Landlord's most recent title policy and property survey, if any, (c) a copy of Landlord's most recent Phase 1 (and Phase 2) environmental reports, if any, (d) such plans and specifications for the Property as Landlord has in its possession and control, and (e) such other information within Landlord's possession and control related to the operation of the Property that Buyer may from time to time reasonably request and which is not proprietary or subject to confidentiality obligations of Seller (the "Due Diligence Materials"). At Tenant's option, prior to the end of the Negotiation Period, Tenant may deliver to Landlord a notice setting forth Tenant's final offer on the price and other material terms at which Tenant would be willing to purchase the Project (the "Final Offer Notice"). Notwithstanding the foregoing, the foregoing terms of this Section 1.5 shall not be applicable to (1) the Sale of the Building or Project to an entity which is Controlled (as defined below) by, Controls or is under common Control with Landlord (any such entity, an "Landlord Affiliate"), or (2) the Sale of the Building or Project together with one or more other commercial office building properties owned by Landlord or any Landlord Affiliate (which shall not include a sale of only the Project (i.e. the Building and Two Rincon)) (any such Sale, a "Portfolio Sale"), or (3) an entity in which Landlord maintains a "Managing Member Interest," as defined below, or an "Investor Member Interest" as defined below (in either case, a "Joint Venture Entity") (any such sale as set forth in items (1), (2) or (3), an "Exempt Sale"). To the extent Landlord enters into an Exempt Sale, and the new owner subsequently elects to Sell the Building or Project, then such new owner shall first deliver a Sale Notice to Tenant, and the parties (i.e., the new owner and Tenant) shall proceed pursuant to the terms of this Section 1.5, provided that for purposes of the foregoing, all references to "Landlord" in this Section 1.5 shall mean the new owner. As used herein a "Managing Member Interest" shall mean an equity interest of not less than ten percent (10%) of the ownership entity, coupled with the right to direct the day to day operations of the Project in a manner consistent with the rights of managing members in real estate joint ventures between sophisticated institutional parties, and an "Investor Member Interest" shall mean an equity interest of not less than twenty-five percent (25%) of the ownership entity, coupled with the right to approve major decisions regarding the ownership of the Project (e.g., decisions to sell or finance) in a manner consistent with the rights of a joint venture capital partner in real estate joint ventures between sophisticated institutional parties. For purposes of this Lease, the term "control" shall mean the ownership of at least fifty percent (50%) of the equity interests in an entity and the right to vote such equity interests in the ordinary course of business.

1.5.2 <u>Rejection; Re-Offer</u>. If Tenant fails to timely deliver to Landlord an Interest Notice, or Landlord and Tenant fail to reach agreement on a purchase agreement prior to expiration of the Negotiation Period,

then, unless Tenant has delivered a Final Offer Notice, Landlord shall be free for one-hundred eighty (180) days thereafter (for purposes of this <u>Section 1.5.2</u>, the "**One Hundred Eighty Day Period**") on any terms Landlord desires. In the event that Tenant has delivered the Final Offer Notice, then, prior to Landlord entering into an agreement to sell the Project to any third party for a purchase price which is less than ninety-five percent (95%) of the purchase price (the "**Sale Offer Notice**"). Concurrently with Landlord's delivery of a Sale Offer Notice, Landlord shall first offer to Sell the Project to Tenant for such lower purchase price (the "**Sale Offer Notice**"). Concurrently with Landlord's delivery of a Sale Offer Notice, Landlord shall deliver to Tenant any updates to the Due Diligence Materials to reflect material changes in the same. Tenant shall accept or reject such offer within five (5) business days after delivery of the Sale Offer Notice. If Tenant fails to accept the offer to purchase the Project to any entity it desires, for a purchase price which is less than ninety-five percent (95%) of the purchase price and entity if desires, for a purchase that of Tenant's rejection or failure to accept the Sale Offer Notice, during the One Hundred Eight Day Period, provided that a new One Hundred Eight Day period shall commence on the date of Tenant's rejection or failure to accept the Sale Offer Notice.

1.5.3 **Participation**. If Tenant fails to timely deliver an Interest Notice, then Landlord shall nevertheless allow Tenant to participate in any "bid process" or other open call for offers to purchase the Building or the Project, as applicable, from the market at large, conducted by Landlord, and in connection therewith shall permit Tenant to submit an offer to purchase the Building or the Project, as applicable, from the market at large, conducted by Landlord, and in connection therewith shall permit Tenant to submit an offer to purchase the Building or the Project, as applicable, like any other third party; provided, however, nothing herein shall require or obligate Landlord to accept any offer submitted by Tenant, and Landlord may reject such offer in its sole and absolute discretion. If a particular Sale does not involve a "bid process" or open call for offers to purchase, then the terms of this Section 1.5.3 shall not apply.

1.5.4 **Termination of Right of First Offer to Purchase.** The rights contained in this Section 1.5 shall be personal to the Original Tenant and its Permitted Transferees. Tenant shall not have the right of first offer to purchase the Building or Project, as applicable, as provided in this Section 1.5, shall be date of the attempted exercise of any right of first offer to purchase by Tenant, Tenant is in default under this Lease, beyond any applicable notice and cure period set forth in this Lease. The terms of this Section 1.5.4 shall terminate upon any Sale that is not an Exempt Sale.

ARTICLE 2

LEASE TERM; OPTION TERM(S)

2.1 Initial Lease Term; Beneficial Occupancy. The terms and provisions of this Lease shall be effective as of the Effective Date. With respect to each Portion of the Premises, the term of this Lease (the "Lease Term") shall be as set forth in Section 3.1 of the Summary, and shall terminate on the applicable Lease Expiration Date set forth in Section 3.2 of the Summary, and shall terminate on the applicable Lease Expiration Date set forth in Section 3.1 of the Summary (as the same may be extended by Tenant in Delivery Condition on or before the Anticipated Delivery Date with respect to such Portion of the Premises, then except as expressly set forth in Section 1.1.4 above, Landlord shall not be subject to any liability for its failure to do so, and such failure shall not affect the validity of this Lease Term; provided, however, that the first Lease Year shall commence on the applicable Lease Term; provided, however, that the first Lease Year shall commence on the first day of the next calendar month; and further provided that the last Lease Year with respect to each Portion of the Premises. At any time during the Lease Term, Landlord may deliver to Tenant a notice in the form as set forth in <u>Exhibit D</u>, attached hereto ("Commencement Letter as to a Portion of the Premises, then Termises, then information set forth therein, which Tenant shall execute and return to Landlord within thirty (30) days of receipt thereof. If Landlord fails to deliver a Commencement Letter as to a Portion of the Premises, then Tenant shall be permitted to prepare and send to Lease there the receipt set forth therein, the right to occupy each Portion of the Premises. The asent the right to occupy each Portion of the Premises, then there receipt from Tenant, the Commencement Letter as to a Portion of the Premises, then Tenant shall be commencement Letter as sent by Tenant shall be deemed to have correctly set forth the matters addressed in it. Notwithstanding any provision to the contrary set forth in this Lease, Ternat shall have the r

during the period between the applicable Lease Commencement Date and the applicable Rent Commencement Date (the "Beneficial Occupancy Period"), provided that (i) Tenant shall have delivered to Landlord satisfactory evidence of the insurance coverage required to be carried by Tenant in accordance with <u>Article 10</u>, below, (ii) Tenant has delivered to Landlord the L-C (as defined below) as required by <u>Article 21</u> below and the first month's Base Rent for the Tranche I Premises pursuant to <u>Article 3</u> below, and (iii) all of the terms and conditions of this Lease shall apply during the Beneficial Occupancy Period, other than Tenant's obligation to pay "Base Rent" (as that term is defined in <u>Article 3</u> below) or any Excess (as defined below) with respect to such Portion of the Premises.

2.2 Option Terms

2.2.1 Option Right. Landlord hereby grants to the Original Tenant and its Permitted Transferees two (2) successive options to extend the Lease Term with respect to one or more floors (or all of the space leased on any particular floor by Tenant) of any Tranche of Premises, each by a period of five (5) years (each, an "Option Term"). Each option to extend shall be exercisable only by "Notice" (as that term is defined in <u>Section 29.17</u> of this Lease) delivered by Tenant to Landlord as provided below, provided that, ("Option Term Exercise Notice"), not earlier than fifteen (15) and not less than twelve (12), months prior to the applicable Lease Expiration Date (or the expiration of the first Option Term, as applicable), <u>provided</u> that as of the date of delivery of such Notice. Tenant is not then in default under this Lease (after receipt of all required notices from Landlord and the expiration of all applicable cure period set forth in this Lease). Tenant shall specify in the Option Term Exercise Notice the floor(s) of the Premises (or portions thereof) that Tenant desires to extend. Upon the proper exercise of such option to extend, and provided that, at Landlord's election, as of the end of the then applicable Lease Term, Tenant is not then in default under this Lease (after receipt of all required notices from Landlord and the expiration of all applicable cure period set forth in this Lease), then the Lease Term shall be extended independently for the applicable floor(s) of the Premises (or portions thereof) for a period of five (5) years.

222 Option Rent. The annual Rent payable during the Option Term shall be equal to the annual Fair Market Value (as hereinafter defined) determined for each Portion of the Premises as of commencement of the applicable Option Term (the "Calculation Date"). "Fair Market Value" for a particular Portion of the Premises shall mean the fair market annual rent (including additional rent and considering any "base year" or "expense stop" applicable thereto) for such Portion of the Premises, taking into account all escalations, at which, as of the Calculation Date, tenants are leasing non-sublease, non-encumbered, nonequity space comparable in size, location and quality to the applicable floor of the Premises (or portion thereof) for a term of five (5) years, in an arm's-length transaction, which comparable space is located in the Building or in Two Rincon, or in other first-class commercial office buildings ("Comparable Buildings") in the North and South San Francisco Financial Districts and the San Francisco SOMA Area (collectively, the "Market Area"), and which comparable transactions (collectively, the "Comparable Transactions") are entered into within the six (6) month period immediately preceding Landlord's delivery of the "Option Rent Notice" (defined below) to Tenant, taking into consideration the following concessions (the "Concessions"): (a) rental abatement concessions, if any, being granted such tenants in connection with such comparable space; (b) tenant improvements or allowances provided or to be provided for such comparable space, taking into account, and deducting the value of, the existing improvements in the applicable portion of the Premises, such value to be based upon the age, condition, design, quality of finishes and layout of the improvements and the extent to which the same can be utilized by a Tenant based upon the fact that the precise tenant improvements existing in the Premises are specifically suitable to Tenant; (c) other monetary concessions being granted such tenants in connection with such comparable space, (d) the rental rate and escalations for the Comparable Transactions, (e) the amount of parking rent per parking permit paid in the Comparable Transactions, if any, (f) operating expense and tax protection granted in such Comparable Transactions such as a base year or expense stop; (g) the size of the floor(s) of the Premises as compared to the size the premises leased in the Comparable Transaction; (h) views, and (i) the brokerage commissions payable in connection with the Comparable Transaction compared to the fact that no broker commissions are payable under this Lease in connection with the Option Term (unless Landlord is paying a brokerage commission in connection with such Option Term). If Tenant timely disputes Landlord's determination of Fair Market Value in accordance with Section 2.2.3 below, then the dispute shall be resolved by arbitration as provided in Section 2.2.3 below.

2.2.3 Determination of Market Rent. In the event Tenant timely and appropriately exercises an option to extend the Lease Term, Landlord shall notify Tenant of Landlord's determination of the Option Rent at

least six (6) months before the applicable Lease Expiration Date. If Tenant, on or before the date which is thirty (30) days following the date upon which Tenant receives written notice containing Landlord's determination of the Option Rent, in good faith objects to Landlord's determination of the Option Rent, or if Tenant irrevocably exercises its right of first offer with respect to the entire space described in such First Offer Notice but objects to Landlord's determination of First Offer Rent, then in either event Landlord and Tenant shall attempt to agree upon the Option Rent (or First Offer Rent, as applicable) using their best good-faith efforts. If Landlord and Tenant shall attempt to agree upon the Option Rent (or Tenant's object to Landlord's determination of First Offer Rent, as applicable) (the "**Outside Agreement Date**"), then each party shall make a separate determination of the Option Rent, or First Offer Rent, as applicable. Agreement Date, and such determinations shall be binding and be submitted to arbitration in accordance with <u>Sections 2.2.3.1</u> through <u>2.2.3.3</u>, below.

2.2.3.1 Landlord and Tenant shall each appoint one arbitrator who shall by profession be a real estate broker who shall have been active over the ten (10) year period ending on the date of such appointment in the leasing, of first-class, high-rise commercial office properties in the Market Area. The determination of the arbitrators shall be limited solely to the issue of whether Landlord's or Tenant's submitted Option Rent (or First Offer Rent, as applicable) is the closest to the actual Option Rent (or First Offer Rent, as applicable), taking into account the requirements of <u>Section 2.2.2</u> of this Lease. Each such arbitrator shall be appointed within thirty (30) days after the Outside Agreement Date. Landlord and Tenant may consult with their selected arbitrators prior to appointment and may select an arbitrator who has previously represented Landlord and/or Tenant, as applicable). The arbitrators so selected by Landlord and Tenant pursuant to this <u>Section 2.2.3.1</u> shall be deemed "Advocate Arbitrators" for purposes of this <u>Section 2.2.3.1</u> if either party fails to timely appoint an Advocate Arbitrator, then the other party may deliver written notice of such failure to the party that failed to timely appoint an Advocate Arbitrator, and if such failure is not cured within five (5) business days following receipt of such written notice, then the Advocate Arbitrator timely appointed by the other party shall determine the Option Rent, then the other party shall have the right to deliver a notice to the failed party expressly referencing this <u>Section 2.2.3.1</u> and the time period for performance hereunder, and if the failed party does not deliver its determination of the Option Rent within three (3) business days following its receipt of such notice, then the determination of the other party shall be deemed to be the Option Rent.

2.2.3.2 The two (2) Advocate Arbitrators so appointed shall be specifically required pursuant to an engagement letter within ten (10) days of the date of the appointment of the last appointed Advocate Arbitrator to agree upon and appoint a third arbitrator (for purposes of this <u>Section 2.2</u>, the "**Neutral Arbitrator**") who shall be qualified under the same criteria set forth hereinabove for qualification of the two Advocate Arbitrators, except that (i) neither the Landlord or Tenant or either parties' Advocate Arbitrator may, directly or indirectly, consult with the Neutral Arbitrator prior or subsequent to his or her appearance, and (ii) the Neutral Arbitrator shall be required to disclose any conflicts to Landlord and Tenant. The Neutral Arbitrator shall be retained via an engagement letter jointly prepared by Landlord's counsel and Tenant's counsel. If the two (2) Advocate Arbitrator. The cost of the arbitrator may agree upon and appoint a difficult or for the arbitrator of the Neutral Arbitrator shall be retained via an engagement letter jointly prepared by Landlord's counsel and Tenant's counsel. If the two (2) Advocate Arbitrator. The cost of the arbitrator shall be paid by Landlord and Tenant equally.

2.2.3.3 Within ten (10) days following the appointment of the Arbitrator, Landlord and Tenant shall enter into an arbitration agreement (for purposes of this Section 2.2, the "Arbitration Agreement") which shall set forth the following:

(a) Each of Landlord's and Tenant's best and final and binding determination of the Option Rent exchanged by the parties pursuant to Section 2.2.3, above;

(b) An agreement to be signed by the Neutral Arbitrator, the form of which agreement shall be attached as an Exhibit to the Arbitration Agreement, whereby the Neutral Arbitrator shall agree to undertake the arbitration and render a decision in accordance with the terms of this Lease, as modified by the

Arbitration Agreement, and shall require the Neutral Arbitrator to demonstrate to the reasonable satisfaction of the parties that the Neutral Arbitrator has no conflicts of interest with either Landlord or Tenant or any of their affiliates;

(c) Instructions to be followed by the Neutral Arbitrator when conducting such arbitration;

Tenant's

(d) That Landlord and Tenant shall each have the right to submit to the Neutral Arbitrator (with a copy to the other party), on or before the date that occurs fifteen (15) days following the appointment of the Neutral Arbitrator, an advocate statement (and any other information such party deems relevant) prepared by or on behalf of Landlord or Tenant, as the case may be, in support of Landlord's or Tenant's respective determination of Option Rent or First Offer Rent, as applicable (the "**Briefs**");

(e) That within five (5) business days following the exchange of Briefs, Landlord and Tenant shall each have the right to provide the Neutral Arbitrator (with a copy to the other party) with a written rebuttal to the other party's Brief (the "**First Rebuttals**"); provided, however, such First Rebuttals shall be limited to the facts and arguments raised in the other party's Brief and shall identify clearly which argument or fact of the other party's Brief is intended to be rebutted;

(f) That within five (5) business days following the parties' receipt of each other's First Rebuttal, Landlord and Tenant, as applicable, shall each have the right to provide the Neutral Arbitrator (with a copy to the other party) with a written rebuttal to the other party's First Rebuttal (the "Second Rebuttals"); provided, however, such Second Rebuttals shall be limited to the facts and arguments raised in the other party's First Rebuttal and shall identify clearly which argument or fact of the other party's First Rebuttal is intended to be rebutted;

(g) The date, time and location of the arbitration, which shall be mutually and reasonably agreed upon by Landlord and Tenant, taking into consideration the schedules of the Neutral Arbitrator, the Advocate Arbitrators, Landlord and Tenant, and each party's applicable consultants, which date shall in any event be within forty-five (45) days following the appointment of the Neutral Arbitrator;

(h) That no discovery shall take place in connection with the arbitration, other than to verify the factual information that is presented by Landlord or Tenant;

(i) That the Neutral Arbitrator shall not be allowed to undertake an independent investigation or consider any factual information other than presented by Landlord or Tenant, except that the Neutral Arbitrator shall be permitted to visit the Project and the buildings containing the Comparable Transactions;

(j) Tenant shall have the right to present oral arguments to the Neutral Arbitrator at the arbitration for a period of time not to exceed three (3) hours (for purposes of this Section 2.2, the "Tenant's Initial Statement");

(k) Following Tenant's Initial Statement, Landlord shall have the right to present oral arguments to the Neutral Arbitrator at the arbitration for a period of time not to exceed three (3) hours (for purposes of this Section 2.2, the "Landlord's Initial Statement");

(1) Following Landlord's Initial Statement, Tenant shall have up to two (2) additional hours to present additional arguments and/or to rebut the arguments of Tenant (for purposes of this <u>Section 2.2</u>, the "**Tenant's Rebuttal Statement**");

(m) Following Tenant's Rebuttal Statement, Landlord shall have up to two (2) additional hours to present additional arguments and/or to rebut the arguments of Tenant;

(n) That, not later than ten (10) business days after the date of the arbitration, the Neutral Arbitrator shall render a decision (the "Ruling") indicating whether Landlord's or

submitted Option Rent is closer to the Neutral Arbitrator's determination of what the Fair Market Rent should be for the applicable floor(s) of the Premises or First Offer Space, as applicable, which Ruling shall set forth the facts and reasons for the decision of the Neutral Arbitrator;

(o) That following notification of the Ruling, Landlord's or Tenant's submitted Option Rent determination, whichever is selected by the Neutral Arbitrator as being closer to the Option Rent shall become the then applicable Option Rent; and

(p) That the decision of the Neutral Arbitrator shall be binding on Landlord and Tenant.

(q) If a date by which an event described in <u>Section 2.2.3.3</u>, above, is to occur falls on a weekend or a holiday, the date shall be deemed to be the next business day. If the parties fail to enter into an Arbitration Agreement within ten (10) days following the appointment of the Neutral Arbitrator, then the arbitration shall nonetheless proceed in accordance with this <u>Section 2.2.3</u> notwithstanding such failure.

2.2.3.4 In the event that the Option Rent (or First Offer Rent, as applicable) shall not have been determined pursuant to the terms hereof prior to the commencement of the Option Term for a Tranche of the Premises (or prior to the First Offer Commencement Date, as the case may be), Tenant shall be required to pay Base Rent then in effect with respect to such Tranche of the Premises the Option Rent (or First Offer Rent, as applicable) initially provided by Landlord to Tenant, and upon the final determination of the Option Rent (or First Offer Rent, as applicable) initially provided by Landlord to Tenant, and upon the final determination of the Option Rent (or First Offer Rent, as applicable) due, and the appropriate party shall make any corresponding payment to the other party within thirty (30) calendar days after the Option Rent for such Tranche of the Premises (or First Offer Rent, as applicable) has finally been determined.

2.2.3.5 Upon final determination of the Option Rent for each Tranche of the Premises, Landlord and Tenant shall execute an amendment reflecting Tenant's exercise of the applicable extension option and the extension of the Lease Term and the Option Rent for such Tranche of the Premises as finally determined.

2.3 Early Termination Right.

2.3.1 <u>Termination Right</u>. Notwithstanding any provision to the contrary set forth in this Lease, but subject to the terms of this <u>Section 2.3</u> below, the Original Tenant and its Permitted Transferees shall have a one-time right (the "**Termination Right**") to terminate this Lease with respect to only the entire 3rd Floor Premises, effective as of a date selected by Tenant within sixty (60) days following the seventh (7th) anniversary of the Tranche III Commencement Date (the "**Termination Date**") by delivering to Landlord prior written notice irrevocably exercising the Termination Right (the "**Termination Notice**"), which Termination Notice shall be delivered no later than twelve (12) months prior to the Termination Date.

2.3.2 <u>Termination Fee</u>. As consideration for and as a condition precedent to such early termination, Tenant shall pay to Landlord the "Termination Fee" (as that term is defined below) concurrently with Tenant's delivery of the Termination Notice. Time is of the essence with respect to Tenant's delivery of the Termination Notice and payment of the Termination Fee. The "**Termination Fee**" shall equal the sum of (i) the unamortized portion, as of the Termination Date, of the "Tenant Improvement Allowance" (as that term is defined in the Tenant Work Letter) for the 3rd Floor Premises, (ii) the unamortized portion, as of the Termination Date, of the "tenant Improvement Allowance" (as that term is defined in the Tenant Work Letter) for the 3rd Floor Premises, (ii) the unamortized portion, as of the Termination Date, of the sing commissions to be paid by Landlord in connection with the 3rd Floor Premises (it being acknowledged and agreed to by Landlord Tenant that the Tenant Improvement Allowance and leasing commissions paid by Landlord in connection with this Lease shall be amortized in the same manner as a fully-amortized loan, using an interest rate of Reducted per annum, over the initial Lease Term with respect to the 3rd Floor Premises, (iii) Base Rent for the 3rd Floor Premises for six (6) months, based on the annual Base Rent rate per rentable square foot of Reducted and (iv) Base Rent for the 3rd Floor Premises for six (6) months, based on the annual Base Rent rate per rentable square foot of Reducted and (iv) Base Rent for the 3rd Floor Premises for six (6) months, based on the annual Base Rent rate per rentable square foot of Reducted and (iv) Base Rent for the 3rd Floor Premises for six (6) months, based on the annual Base Rent rate per rentable square foot of Reducted and (iv) Base Rent for the 3rd Floor Premises for six (6) months, based on the annual Base Rent rate per rentable square foot of Reducted and (iv) Base Rent for the 3rd Floor Premises for six (6) months, based on the annu

2.3.3 Exercise of Termination Right. Provided Tenant properly exercises the Termination Right in accordance with the terms of this Section 2.3, then, effective as of the Termination Date, Tenant's lease of the 3rd Floor Premises shall automatically terminate and be of no further force or effect, and Landlord and Tenant shall be relieved of their respective obligations under this Lease with respect to the 3rd Floor Premises, except for those obligations set forth in this Lease with respect to the period of Tenant's tenancy prior to the Termination Date and such obligations which specifically survive the expiration or earlier termination of this Lease, including, without limitation, the payment by Tenant of all amounts owed by Tenant under this Lease up to and including the Termination Date. Without limitation the payment by Tenant of all amounts owed by Tenant under this Lease up to and including the Termination Date. Without limit ghe generality of the foregoing, Tenant shall vacate and surrender possession of the 3rd Floor Premises to Landlord on such Termination Date, the holdover provisions of <u>Article 16</u> of this Lease shall apply to such space. The Termination Right shall be personal to the Original Tenant and its Permitted Transferees, and may not be exercised by any assignee, sublessee or other Transferee of the Original Tenant's interest in this Lease (other than a Permitted Transferee). Upon the Termination Date, the amount of the Letter of Credit may be reduced by the sum of Redacted so that the L-C deposited with Landlord thereafter shall be in the sum of Redacted . Landlord agrees to authorize such reduction in writing to the issuer of the L-C. With respect to the increases and decreases to the L-C amount as set forth herein, Landlord shall return the original L-C deposited hereunder to Tenant within five (5) business days following Tenant's delivery of any new L-C. If Tenan thas not previously delivered the Termination Notice as of the date that Landlord's claim for damages in such instances

ARTICLE 3

BASE RENT

Commencing on the Rent Commencement Date with respect to each Portion of the Premises, Tenant shall commence paying with respect to such Portion of the Premises (and will thereafter continue to pay with respect to such Portion of the Premises through the Expiration Date applicable to such Portion of the Premises), without prior notice or demand, to Landlord at the address or pursuant to the wiring instructions sat forth in Section 13 of the Summary, or, at Landlord's option, at such other address or wiring instructions as Landlord may from time to time designate in writing, by a check or wire transfer for currency which, at the time of payment, is legal tender for private or public debts in the United States of America, the base rent ("Base Rent") as set forth in Section 4 of the Summary with respect to such Portion of the Premises, payable in equal monthly installments in advance on or before the first day of each and every calendar month during the Lease Term, without any setoff or deduction whatsoever. The Base Rent for the first full month of the Lease Term with respect to the Tranche I Premises following the Rent Commencement Date for the Tranche I Premises shall be paid within ten (10) business days after Tenant's execution of this Lease. If any payment of Rent is for a period which is shorter than one month, the Rent for any such fractional month shall accrue on a daily basis during such fractional month, and shall total an amount equal to the product of (i) a fraction, the numerator of which is the number of days in such fractional month, and (ii) the then-applicable monthly installment of Base Rent. All other payments or adjustments required to be made under the terms of this Lease that require protation on a time basis shall be prorated on the same basis.

ARTICLE 4

ADDITIONAL RENT

4.1 <u>General Terms</u>. In addition to paying the Base Rent specified in <u>Article 3</u> of this Lease, for each Portion of the Premises Tenant shall pay "Tenant's Share" of the annual "Building Office Pool Direct Expenses," as those terms are defined in <u>Sections 4.2.9</u>, and <u>4.3.1</u> of this Lease, as applicable, which are in excess of the amount of Building Office Pool Direct Expenses or Building Retail Pool Direct Expenses applicable to the "Base Year," as that term is defined in <u>Section 4.2.1</u>, below; provided, however, that in no event shall any decrease in Direct Expenses for any Expense Year below Direct Expenses for the Base Year entitle Tenant to any decrease in Base Rent or any credit against sums due under this Lease. Such payments by Tenant, together with any and all other amounts

payable by Tenant to Landlord pursuant to the terms of this Lease, are hereinafter collectively referred to as the "Additional Rent," and the Base Rent and the Additional Rent are herein collectively referred to as "Rent." All amounts due under this <u>Article 4</u> as Additional Rent shall be payable for the same periods and in the same manner as the Base Rent; provided, however, the parties hereby acknowledge that the first monthly installment of Tenant's Share of any "Estimated Excess," as that term is set forth in, and pursuant to the terms and conditions of, <u>Section 4.4.2</u> of this Lease, shall first be due and payable for the calendar month occurring immediately following the expiration of the Base Year. Without limitation on other obligations of Tenant which survive the expiration of the Lease Term, the obligations of Tenant to pay the Additional Rent provided for in this <u>Article 4</u> shall survive the expiration of the Lease Term.

- 4.2 Definitions of Key Terms Relating to Additional Rent. As used in this Article 4, the following terms shall have the meanings hereinafter set forth:
 - 4.2.1 "Base Year" shall mean the period set forth in Section 5 of the Summary

4.2.2 "Building Office Pool Direct Expenses" shall mean those amounts of "Operating Expenses" and "Tax Expenses," as those terms are defined below, allocated to the "office" portion of the Building as provided in Section 4.3.1, below.

- 4.2.3 Intentionally Omitted
- 4.2.4 Intentionally Omitted
- 4.2.5 "Direct Expenses" shall mean "Operating Expenses" and "Tax Expenses."
- 4.2.6 "Expense Year" shall mean each calendar year in which any portion of the Lease Term falls, through and including the calendar year in which the Lease Term expires.

"Operating Expenses" shall mean all costs and amounts of every kind and nature which Landlord pays during any Expense Year because of or in connection with the ownership, 4.2.7 management, maintenance, security, repair or operation of the Project, or any portion thereof. Without limiting the generality of the foregoing, Operating Expenses shall specifically include any and all of the following: (i) the cost of supplying all utilities (but excluding the cost of electricity consumed in the Premises and the premises of other tenants of the Building and any other buildings in the Project (since Tenant is separately paying for the cost of electricity pursuant to Section 6.1.3 below)), the cost of operating, repairing and maintaining the utility, telephone, mechanical, sanitary, storm drainage, and elevator systems, and the cost of maintenance and service contracts in connection therewith; (ii) the cost of licenses, certificates, permits and inspections and the cost of contesting any governmental enactments which may affect Operating Expenses, and the costs incurred in connection with a governmentally mandated transportation system management program or similar program; (iii) the cost of all insurance carried by Landlord in connection with the Project, and the commercially reasonable deductible portion of any insured loss otherwise covered by such insurance; (iv) the cost of landscaping, relamping, and all supplies, tools, equipment and materials used in the operation, repair and maintenance of the Project, or any portion thereof; (v) costs incurred in connection with the parking areas servicing the Project; (vi) fees and other costs, including management fees, consulting fees, legal fees and accounting fees, of all contractors and consultants in connection with the management, operation, maintenance and repair of the Project; (vii) payments under any equipment rental agreements and the fair rental value of any management office space; (viii) wages, salaries and other compensation and benefits, including taxes levied thereon, of the "Property Manager" and persons generally considered to be lower in rank than the position of "Property Manager" engaged in the operation, maintenance and security of the Project; (ix) operation, repair and maintenance of all systems and equipment and components thereof of the Building; (x) the cost of janitorial, alarm, security and other services, repair of wall and floor coverings, ceiling tiles and fixtures in Common Areas, and maintenance and repair of curbs and walkways, and repair to roofs; (xi) amortization of the cost of acquiring personal property used in the maintenance, operation and repair of the Project, or any portion thereof; provided, however, that any capital expenditure shall be amortized with commercially reasonable interest (as determined in accordance with sound real estate management and accounting practices at the time of the expenditure) over its useful life as Landlord shall reasonably determine in accordance with sound real estate management and accounting practices; (xii) the cost of capital improvements (A) that are acquired to cause, in Landlord's good faith judgment, an immediate (i.e.,

commencing within the first year in which costs for such capital improvements are included in Operating Expenses) reduction in other Operating Expenses, amortized over the useful life of such improvements at an annual rate not to exceed the amount of Operating Expenses); or (B) that are incurred due to any Laws first enacted or made applicable to any Portion of the Project or the Common Areas after the applicable Commencement Date; provided, however, that any capital expenditure shall be amortized with commercially reasonable interest (as determined in accordance with sound real estate management and accounting practices at the time of the expenditure) over its useful life as Landlord shall reasonably determine in accordance with sound real estate management and accounting practices (costs under items (A) and (B), above, "Allowed Capital Costs"); (xiii) costs, fees, charges or assessments imposed by, or resulting from any mandate imposed on Landlord by, any federal, state or local government for fire and police protection, trash removal, community services, or other services which do not constitute "Tax Expenses" as that term is defined in <u>Section 4.2.8.1</u>, below; and (xiv) payments under the following covenants, conditions and restrictions (or any other covenants, conditions or restrictions entered into after the Effective Date in accordance with the terms and conditions of this Lease) (collectively, the "REA"): Rincon Center Associates Declaration Establishing Reciprocal Easements and Covenants Running With the Land recorded in the Recorder's Office of San Francisco, California, on June 27, 1988 as Document No. E196829; Declaration Establishing Reciprocal Easements and Covenants Running With the Land recorded in the Recorder's Office of San Francisco, California, on June 28, 2006 as Document No. 2006-1201148-00 ("Second Amendment"); and Third Amendment to Declaration Establishing Reciprocal Easements and Covenants Running With the Land recordeal Easements and Covenants Running on June 28, 2006 as Document No

4.2.7.1 costs, including marketing costs, legal fees, space planners' fees, advertising and promotional expenses, and brokerage fees incurred in connection with the original construction or development, or original or future leasing of the Project (including, without limitation, the cost of construction of the Base Building (as defined in the Tenant Work Letter)), and costs, including permit, license and inspection costs, incurred with respect to the installation of improvements made for tenants or other occupants of the Project or incurred in renovating or otherwise improving, decorating, painting or redecorating space for tenants or other occupants of the Project;

4.2.7.2 except as set forth in items (xi) and (xii) above, depreciation or amortization of the Project and interest, principal payments, debt service, points, fees, penalties or other debt costs on mortgages or other debt instruments;

4.2.7.3 costs for which the Landlord is reimbursed by any tenant or occupant of the Project or by insurance by its carrier (or which would have been reimbursed by Landlord's insurance carrier if Landlord had maintained the insurance required to be carried by Landlord under this Lease) or any tenant's carrier or by anyone else, including, any warrantor (except to the extent of commercially reasonable deductibles);

4.2.7.4 any reserves for bad debts, rent loss, capital items, future Operating Expenses or any other purpose;

4.2.7.5 costs associated with the operation of the business of the partnership or entity which constitutes the Landlord, as the same are distinguished from the costs of operation of the Project (which shall specifically include, but not be limited to, accounting costs associated with the operation of the Project). Costs associated with the operation of the partnership or entity which constitutes the Landlord include costs of partnership accounting and legal matters, costs of defending any lawsuits with any mortgagee (except as the actions of the Tenant may be in issue), costs of selling, syndicating, financing, mortgaging or hypothecating any of

the Landlord's interest in the Project, and costs incurred in connection with any disputes between Landlord and its employees, between Landlord and Project management, or between Landlord and other tenants or occupants, and Landlord's general corporate overhead and general and administrative expenses;

4.2.7.6 the wages and benefits of any employee who does not devote substantially all of his or her employed time to the Project unless such wages and benefits are prorated to reflect time spent on operating and managing the Project; provided, that in no event shall Operating Expenses for purposes of this Lease include the items described in <u>Section 4.2.7.24</u> below;

4.2.7.7 amount paid as ground rental for the Project by the Landlord;

Expenses;

4.2.7.8 overhead and profit increment paid to the Landlord or to subsidiaries or affiliates of the Landlord for services in the Project to the extent the same exceeds the costs of such services rendered by qualified, first-class unaffiliated third parties on a competitive basis;

4.2.7.9 any compensation paid to clerks, attendants or other persons in commercial concessions operated by the Landlord;

4.2.7.10 rentals and other related expenses incurred in leasing air conditioning systems, elevators or other equipment which if purchased the cost of which would be excluded from Operating Expenses as a capital cost, except equipment not affixed to the Project which is used in providing janitorial or similar services and, further excepting from this exclusion such equipment rented or leased to remedy or ameliorate an emergency condition in the Project;

4.2.7.11 all items and services for which Tenant or any other tenant in the Project reimburses Landlord or which Landlord provides selectively to one or more tenants (other than Tenant) without reimbursement;

4.2.7.12 costs, other than those incurred in ordinary maintenance and repair, for sculpture, paintings, fountains or other objects of art;

4.2.7.13 any costs expressly excluded from Operating Expenses elsewhere in this Lease;

4.2.7.14 costs to the extent arising from the negligence or willful misconduct of Landlord or its agents, employees, vendors, contractors, or providers of materials or services;

4.2.7.15 expenses and costs relating in any way whatsoever to the identification, testing, monitoring and control, encapsulation, removal, replacement, repair, or abatement of any Hazardous Substances (as defined below) or mold within the Building or Project;

4.2.7.16 fees payable by Landlord for management of the Project in excess of three percent (3%) of base rent for the office portions of the Building only and Building Office Pool Direct

4.2.7.17 costs, fines, penalties, or interest incurred due to a violation of Laws or the terms and conditions of any lease by Landlord or any other tenant in the Project;

4.2.7.18 costs incurred by Landlord for alterations, additions, and replacements which are considered capital expenditures under generally accepted accounting principles, consistently applied, except for Allowed Capital Costs;

4.2.7.19 to the extent the Common Areas are not in compliance with Laws as of the applicable Commencement Date, then costs of bringing such Common Areas into compliance with building codes, laws, rules, regulations, or any other governmental rules or requirements, including, without limitation, the Americans With Disabilities Act of 1990 and Title 24 of the California Code of Regulations (or its successor);

4.2.7.20 any costs required to be expended by Landlord in order to cause the Premises to be in the applicable Delivery Condition;

4.2.7.21 repairs or other work paid for through condemnation proceeds;

4.2.7.22 repairs, alterations, additions, improvements or replacements made to rectify or correct any defect in the design, materials, or workmanship of the Project;

4.2.7.23 costs incurred by Landlord, including penalties and interest, due to violation by Landlord or any other tenant or occupant of the Building of applicable Laws, the terms and conditions of any lease, ground lease, mortgage or deed of trust, or other covenants, conditions or restrictions encumbering the Building or the Land;

4.2.7.24 salaries, wages, bonuses, and other compensation (including hospitalization, medical, surgical, retirement plan, pension plan, union dues, parking privileges, life insurance, including group life insurance, welfare, and other fringe benefits, and vacation, holidays, and other paid absence benefits) relating to asset managers, leasing agents, promotional directors, officers, directors, or executives of Landlord that are above the rank of "Property Manager";

4.2.7.25 costs incurred in the operation, maintenance and repair of any concession serving the Project (not including parking operations);

4.2.7.26 brokerage commissions, attorneys' and accountants' fees related thereto, loan brokerage fees, closing costs, interest charges and other similar costs incurred in connection with the sale, refinancing, mortgaging, or selling, or change of ownership of the Project;

4.2.7.27 all costs incurred by Landlord in connection with any dispute relating to the Landlord's title to or ownership of the Project or any portion thereof;

4.2.7.28 interest and penalties due to late payments of taxes and utility bills or any other obligations;

4.2.7.29 any damage and repairs covered under any insurance policy carried by, or required to be carried by, Landlord;

4.2.7.30 costs for which Landlord has been compensated by a management fee to the extent that the inclusion of such costs in Operating Expenses would result in a double charge;

4.2.7.31 costs incurred in installing, operating and maintaining any specialty improvement not normally installed, operated and maintained in projects comparable to the Building, including, without limitation, an observatory, luncheon club, or athletic or recreational facilities;

4.2.7.32 costs and expenses of providing after-hours HVAC service to other tenant spaces in the Building;

4.2.7.33 costs and expenses to provide janitorial service, water, gas, fuel, steam, lights, sewer service and other utilities to other tenants or occupants of the Building materially in excess of amounts typically used in connection with ordinary office use;

4.2.7.34 costs for janitorial services for any rentable area in the Premises to the extent Tenant provides such services to the Premises at its own cost;

4.2.7.35 expenses in connection with services or other benefits which are not offered to Tenant or for which Tenant is charged directly but which are provided to another tenant or occupant of

the Building;

occupant of the Building:

4.2.7.36 advertising or promotional expenditures, and the costs of acquiring and installing signs in or on any of the Building identifying the owner of the Building or any other tenant or uilding;

4.2.7.37 costs of repairs or other work resulting from a casualty or occurrence with regard to which either Landlord or Tenant elects to terminate this Lease;

4.2.7.38 costs, expenses, fees or payments under any REA other than as expressly set forth in item (xiv) above; and

4.2.7.39 costs of operation, repair, maintenance and other expenses relating to the portions of the Project used for retail, restaurant and residential use;

4.2.7.40 costs arising from Landlord's charitable or political contributions;

4.2.7.41 any gifts provided to any entity whatsoever, including, but not limited to, Tenant, other tenants, employees, vendors, contractors, prospective tenants and agents;

4.2.7.42 the cost of any magazine, newspaper, trade or other subscriptions;

4.2.7.43 rent for any office space occupied by Project management personnel to the extent the size or rental rate of such office space exceeds the size or fair market rental value of office space occupied by management personnel of the Comparable Buildings in the vicinity of the Building, with adjustment where appropriate for the size of the applicable project;

4.2.7.44 costs of repairs or other work occasioned by fire, windstorm, or other casualty, whether or not Landlord carries such insurance, in excess of \$250,000.00 per Expense Year for the Project (except for any commercially reasonable insurance deductible shall be included in the definition of Operating Expenses);

4.2.7.45 costs relating to the repair of structural portions of the roof, foundations, floors and exterior walls and all structural seismic upgrading costs, in excess of \$250,000.00 per Expense Year for the Project; and

4.2.7.46 cost of any repairs to the Premises, the Building or the Project made necessary by any negligence or willful misconduct of Landlord or any of its assignees, or employees, or their respective agents, representatives, or contractors.

If Landlord is not furnishing any particular work or service (the cost of which, if performed by Landlord, would be included in Operating Expenses) to a tenant who has undertaken to perform such work or service in lieu of the performance thereof by Landlord, Operating Expenses shall be deemed to be increased by an amount equal to the additional Operating Expenses which would reasonably have been incurred during such period by Landlord if it had at its own expense furnished such work or service to such tenant. If the Project is not at least ninety-five percent (95%) occupied during all or a portion of any Expenses for such year to determine the amount of Operating Expenses that would have been incurred had the Project been ninety-five percent (95%) occupied; and the amount so determined shall be deemed to have been the amount of Operating Expenses for such year. In no event shall any adjustments to Operating Expenses in any calendar year result in Landlord receiving from Tenant and other tenants more than one hundred percent (100%) of the cost of the actual Operating Expenses paid by Landlord in any such calendar year. Landlord shall not (i) make a profit by charging items to Operating Expenses that are otherwise also charged separately to others and (ii) subject to Landlord's right to adjust the components of Operating is an amount in excess of what Landlord incurs for the items included in Operating Expenses. In the event and to the extent any portion of the Project is covered by a warranty or service agreement which provides warranty-type protection at any time during the Base Year and is not covered by such warranty or such warranty-type protection at any time during the Base Year and is not covered by such warranty or such warranty-type protection at any time during the Base Year and is not covered by such warranty or such warranty-type protection at any time during the Base Year and is not covered by such warranty or such warranty-type protection at any time during the Base Year and is not covered by such

agreement in a subsequent Expense Year to the same extent, Operating Expenses for the Base Year shall be deemed increased by the amount Landlord would have incurred during the Base Year with respect to the items or matters covered by the subject warranty or warranty-type protection, had such warranty or such service agreement not been in effect during the entire Base Year. Any insurance premium resulting from any new type of insurance is carried by Landlord, so that Tenant shall only be required to pay subsequent increases in such premiums. If a new category of expense is incurred after the Base Year, of a type that is not a category of expense then generally included in operating expenses by a majority of landlords of Comparable Buildings, the first full year's expense for such item shall be added to Operating Expenses for the Base Year, so that Tenant shall only be required to pay subsequent increases in such premiums. If a new category of expense is carried by Landlord to, Force Majeure, boycotts, strikes, conservation surcharges, embargoes or shortages (including utility rate increases) due to extraordinary circumstances, including, but not limited to, Force Majeure, boycotts, strikes, conservation surcharges, embargoes or shortages (the "Temporarily Increased Costs"), provided that, at such time as such Temporarily Increased Costs; provided, that such adjustment to the Base Year shall be appropriately adjusted to remove such Temporarily Increased Costs; provided, that such adjustment to the Base Year shall not result in re-adjustment of Operating Expenses payable for Expense Years which occurred after the Base Year. Landlord may subtract such category from the Base Year commencing with such subsequent year. Landlord agrees that, if Tenant so elects and appoints a representative (the "Tenant Facility Coordinator"), Landlord shall permit the Tenant Facility Coordinator to meet with Landlord on a mutually agreeable regular basis (not more than one (1) time per calendar quarter) to discuss the operations of th

4.2.8 <u>Taxes</u>

4.2.8.1 "**Tax Expenses**" shall mean all federal, state, county, or local governmental or municipal taxes, fees, charges or other impositions of every kind and nature, whether general, special, ordinary or extraordinary, (including, without limitation, real estate taxes, general and special assessments, transit taxes, leasehold taxes or taxes based upon the receipt of rent, including gross receipts or sales taxes applicable to the receipt of rent, unless required to be paid by Tenant, personal property taxes imposed upon the fixtures, machinery, equipment, apparatus, systems and equipment, appurtenances, furniture and other personal property used in connection with the Project, or any portion thereof), which shall be paid during any Expense Year (without regard to any different fiscal year used by such governmental or municipal authority) because of or in connection with the ownership, leasing and operation of the Project, or any portion thereof. If, for any tax fiscal year, the Project is not separately assessed, but is assessed jointly with other property, then Landlord shall equitably apportion such Taxes Expenses for such tax fiscal year based upon allocable tax basis among the properties jointly assessed.

4.2.8.2 If, by Law, any Taxes may be paid in installments (whether or not interest accrues on the unpaid balance), then, for any calendar year, Landlord shall include in the calculation of such Taxes only the amount of the installments (with any interest) due and payable during such year had Landlord selected the longest permissible period of payment.

4.2.8.3 Tax Expenses shall include, without limitation: (i) any tax on the rent, right to rent or other income from the Project, or any portion thereof, or as against the business of leasing the Project, or any portion thereof; (ii) any assessment, tax, fee, levy or charge in addition to, or in substitution, partially or totally, of any assessment, tax, fee, levy or charge previously included within the definition of real property tax, it being acknowledged by Tenant and Landlord that Proposition 13 was adopted by the voters of the State of California in the June 1978 election ("**Proposition 13**") and that assessments, taxes, fees, levies and charges may be imposed by governmental agencies for such services as fire protection, street, sidewalk and road maintenance, refuse removal and for other governmental services formerly provided without charge to property owners or occupants, and, in further recognition of the decrease in the level and quality of governmental services and amenities as a result of Proposition 13, Tax Expenses shall also include any governmental or private assessments or the Project's contribution towards a governmental or private cost-sharing agreement for the purpose of augmenting or improving the quality of services and amenities normally provided by governmental agencies; (iii) any assessment, tax, fee,

levy, or charge allocable to or measured by the area of the Premises or the Rent payable hereunder, including, without limitation, any business or gross income tax or excise tax with respect to the receipt of such rent, or upon or with respect to the possession, leasing, operating, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises, or any portion thereof; (iv) any assessment, tax, fee, levy or charge, upon this transaction or any document to which Tenant is a party, creating or transferring an interest or an estate in the Premises; and (v) all of the real estate taxes and assessments imposed upon or with respect to the Building and all of the real estate taxes and assessments imposed on the land and improvements comprising the Project.

4.2.8.4 Refunds of Tax Expenses shall be credited against Tax Expenses and refunded to Tenant regardless of when received, based on the Expense Year to which the refund is applicable, provided that in no event shall the amount to be refunded to Tenant for any such Expense Year exceed the total amount paid by Tenant as Tax Expenses under this <u>Article 4</u> for such Expense Year. If Tax Expenses for any period during the Lease Term or any extension thereof are increased after payment thereof for any reason, including, without limitation, error or reassessment by applicable governmental or municipal authorities, Tenant shall pay Landlord upon demand Tenant's Share of any such increased Tax Expenses included by Landlord as Tax Expenses pursuant to the terms of this Lease. Notwithstanding anything to the contrary contained in this <u>Section 4.2.8</u>, there shall be excluded from Tax Expenses (i) any excess profits taxes, franchise taxes, gift taxes, capital stock taxes, inheritance and succession taxes, estate taxes, documentary transfer taxes, federal or state income, corporate or capital gains taxes, penalties incurred as a result of Landlord's failure to pay taxes or to file any tax or informational returns and other taxes to the extent applicable to Landlord's general or net income (as opposed to rents, receipts or income attributable to operations at the Project), (ii) any items included as Operating Expenses, and (iii) any items paid by Tenant under <u>Section 4.5</u> of this Lease.

4.2.8.5 Upon written request by Tenant on no more than an annual basis, Landlord agrees to notify Tenant in writing on or before ninety (90) days prior to the final date in which any appeal of Tax Expenses must be filed whether or not it will appeal property taxes for the Project for the tax bill in question. If Landlord elects not to appeal property taxes for the Project which relate to any portion of the Lease Term, then Landlord agrees to cooperate with Tenant should Tenant desire to appeal such property taxes for the Project. If Tenant elects to exercise the option to appeal pursuant to this Section, then Tenant is required to provide Landlord with written notice ("Appeals Notice") at least thirty (30) days prior to the final date in which the appeal must be filed. Upon receipt of the Appeals Notice, Landlord of terminate an any party reasonably approved by Tenant in order to file and prosecute such appeal and thereafter Landlord and the resulting appeal reduces the property taxes for the calendar year in question as compared to the original bill received for such calendar year, then the costs for such appeal (up to the amount of the reduction in Tax Expenses) shall be included as an Operating Expense and passed through to the tenants of the Building. In turn, if the appeal is not successful, or if the cost of such appeal in excess of the amount of the reduction in Tax Expenses. Tenant's failure to timely deliver an Appeals Notice shall waive Tenant's rights to request an appeal of such tax increase or assessment for such calendar year. In addition, Tenant's obligations to reimburse Landlord for the costs of the appeal in excess, of the amount of the reduction in Tax Expenses. Tenant's failure to timely deliver an Appeals Notice shall waive Tenant's rights to request an appeal of such tax increase or assessment for such calendar year. In addition, Tenant's obligations to reimburse Landlord for the costs of the appeal pursuant to this Section, and Tenant's rights to benefit from any reduction in the Tax Expe

4.2.8.6 Notwithstanding anything to the contrary set forth in this Lease, the amount of Tax Expenses for the Base Year and any Expense Year shall be calculated without taking into account any decreases in real estate taxes obtained in connection with Proposition 8, and, therefore, the Tax Expenses in the Base Year and/or an Expense Year may be greater than those actually incurred by Landlord, but shall, nonetheless, be the Tax Expenses due under this Lease; provided that (i) any costs and expenses incurred by Landlord in securing any Proposition 8 reduction shall not be included in Direct Expenses for purposes of this Lease, and (ii) tax refunds under Proposition 8 shall not be deducted from Tax Expenses, but rather shall be the sole property of Landlord and Tenant acknowledge that this <u>Section 4.2.8.6</u> is not intended to in any way affect (A) the inclusion in Tax Expenses of the statutory two percent (2.0%) annual increase in Tax Expenses (as such statutory increase may

be modified by subsequent legislation), or (B) the inclusion or exclusion of Tax Expenses pursuant to the terms of Proposition 13, which shall be governed pursuant to the terms of Sections 4.2.8.1 through 4.2.8.5, above.

4.2.9 "Tenant's Share" shall mean the percentage set forth in <u>Section 6</u> of the Summary.

4.2.10 <u>Tenant's Payment of Certain Tax Expenses</u>. Notwithstanding anything set forth in this Lease to the contrary (including, without limitation, <u>Section 4.2.8.1</u> above), in the event that, at any time prior to the thirtieth (30th) month following the Tranche I Commencement Date (the "**Prop 13 Protection Period**"), any sale, refinancing, or change in ownership of the Building or Project is consummated (specifically excluding, however, a change in ownership to a lender resulting from a foreclosure or a deed-in-lieu of foreclosure), and as a result thereof, and to the extent that in connection therewith, the Building or Project is reassessed (the "**Reasessment**") for real estate tax purposes by the appropriate governmental authority pursuant to the terms of Proposition 13, then the terms and conditions of this <u>Section 4.2.10</u> shall apply to such Reassessment of the Building or Project.

4.2.10.1 The Tax Increase. For purposes of this Section 4.2.10, the term "Tax Increase" shall mean that portion of the Tax Expenses, as calculated immediately following the Reassessment, which is attributable solely to the Reassessment. Accordingly, the term Tax Increase shall not include any portion of the Tax Expenses, as calculated immediately following the Reassessment, which is attributable to (i) the initial assessment of the value of the Project, the base, shall and core of the Building or the tenant improvements located in the Building; (ii) assessments which were pending immediately prior to the Reassessment which assessments were conducted during, and included in, such Reassessment, or which assessments were otherwise rendered unnecessary following the Reassessment; of (iii) the annual inflationary increase of real estate taxes, but not in excess of two percent (2.0%) per annum. The Tax Increase shall also not include any increase in taxes payable with respect to the Building as a result of any modification of the tax code, including, without limitation, and modification or revocation of Proposition 13.

4.2.10.2 Protection. During the Prop 13 Protection Period, Tenant shall not be obligated to pay any portion of the Tax Increase relating to any Reassessment of the Building or Project. After the Prop 13 Protection Period, any portion of the Tax Increase relating to any Reassessment of the Building or Project shall be included in Direct Expenses.

4.3 Allocation of Direct Expenses.

4.3.1 <u>Method of Allocation</u>. The parties acknowledge that the Building is a part of a multi-building project and that the costs and expenses incurred in connection with the Project (i.e. the Direct Expenses) should be shared between certain of the tenants of the Building and the tenants of the other buildings in the Project. In addition, within each building of the Project, certain of the Direct Expenses are allocated among the "office" and "retail" portions of the particular building. Accordingly, as set forth in <u>Section 4.2</u> above, Direct Expenses (which consists of Operating Expenses) are determined annually for the Project as a whole, and a portion of the Direct Expenses, which portion shall be determined by Landlord on an equitable basis, shall be allocated to the office tenants of the Building (as opposed to the tenants of any other building in the Project as a whole, and a sopposed to the retail tenants of the Building), and such portion shall be the "**Building Office Pool Direct Expenses**" for purposes of this Lease. Such Building Office Pool Direct Expenses attributable to the Building as a whole, and (ii) an equitable portion of the Direct Expenses attributable to the Project as a whole. As of the Effective Date, the allocations referenced in clause (iii) above are made in the following percentages (which are subject to change during the Lease Term in accordance with Landlord's reasonable and equitable determinations):

Office Pool of the Building: 46.69% Retail Pool of the Building: 7.91% Office Pool of Two Rincon: 40.19% Retail Pool of Two Rincon: 5.20%

As of the Effective Date, the costs incurred by Landlord in connection with Landlord's obligations under the REA that are to be shared between the residential areas and the commercial areas are allocated as follows: 29.60% of

such costs to the residential pool, and 70.40% to the commercial pool. Tenant acknowledges that certain costs are allocated or shared in different amounts, as reasonably and equitably determined by Landlord.

4.4 <u>Calculation and Payment of Additional Rent</u>. If for any Expense Year ending or commencing within the Lease Term, Tenant's Share of Building Office Pool Direct Expenses for such Expense Year exceeds Tenant's Share of Building Office Pool Direct Expenses applicable to the Base Year, then Tenant shall pay to Landlord, in the manner set forth in <u>Sections 4.4.1</u> and <u>4.4.2</u>, below, and as Additional Rent, an amount equal to the excess (the "Excess").

4.1 Statement of Actual Direct Expenses and Payment by Tenant. Landlord shall give to Tenant on or before June 1 following the end of each Expense Year, a statement (the "Statement") which shall state in reasonably detailed categories the Direct Expenses incurred or accrued for the Base Year and each Expense Year thereafter, as applicable, and which shall indicate the amount of the Excess. Upon receipt of the Statement for each Expense Year commencing or ending during the Lease Term, if an Excess is present, Tenant shall pay, within thirty (30) days after receipt of the Statement, the full amount of the Excess for such Expense Year, less the amounts, if any, paid during such Expense Year as "Estimated Excess," as that term is defined in Section 4.4.2. below (a "Shortfall"), and if Tenant paid more as Estimated Excess than the actual Excess (an "Overpayment"), Tenant shall receive a credit in the amount of such Overpayment against Rent next due under this Lease. The failure of Landlord to timely furnish the Statement for any Expense Year shall not prejudice Landlord or Tenant from enforcing its rights under this <u>Article 4</u>. Even though the Lease Term has expired and Tenant has vacated the Premises, when the final determination is made of Tenant's Share of Direct Expenses for the Expense Year in which this Lease terminates, if an Excess is present, Tenant shall, within thirty (30) days after receipt of the Statement, pay to Landlord such amount, and if an Overpayment occurred, then Landlord shall, within thirty (30) days, deliver a check payable to Tenant in the amount of such Overpayment. The provisions of this <u>Section 4.4.1</u> shall survive the expiration or earlier termination of the Excess Term. Notwithstanding the terms of this <u>Section 4.4.1</u>. Tenant shall not be responsible for any Excess attributable to any Expense Year which is first billed to Tenant more than two (2) years after the end of the applicable Expenses for which Landlord had not previously received an invoice therefor and which are currently

4.4.2 Statement of Estimated Direct Expenses. In addition, prior to May 1 of each calendar year of the Lease Term, Landlord shall give Tenant a yearly expense estimate statement (the "Estimate Statement") which shall set forth in reasonably detailed categories Landlord's good faith reasonable estimate (the "Estimate") of what the total amount of Direct Expenses for the then-current Expense Year and the estimated Excess (the "Estimated Excess") as calculated by comparing the Building Direct Expenses for such Expense Year, which shall be based upon the Estimate, to the amount of Building Direct Expenses for the Base Year. The failure of Landlord to timely furnish the Estimate Statement for any Expense Year shall not preclude Landlord from enforcing its rights to collect any Additional Rent under this Article 4, nor shall Landlord be prohibited from revising any Estimate Statement or Estimated Excess therefore delivered to the extent necessary; provided that Landlord may not revise the Estimate Statement or Estimated Excess more than once in any Expense Year. Thereafter, Tenant shall pay, within thirty (30) days after receipt of the Estimate Statement, a fraction of the Estimated Excess for the then-current Expense Year (reduced by any amounts paid pursuant to the second to last sentence of this <u>Section 4.4.2</u>). Such fraction shall have as its numerator the number of months which have elapsed in such current Expense Year, including the month of such payment, and twelve (12) as its denominator. Until a new Estimated Excess set of the total Estimate Statement is furnished (which Landlord shall have the right to delivered by Landlord to Tenant. Throughout the Lease Term Landlord shall powers and the reviews with respect to Building Direct Expenses in accordance with generally accepted real estate accounting and management practices, consistently applied.

4.5 Taxes and Other Charges for Which Tenant Is Directly Responsible

4.5.1 Tenant shall be liable for and shall pay before delinquency, taxes levied against Tenant's equipment, furniture, fixtures and any other personal property located in or about the Premises. If any such taxes on

Tenant's equipment, furniture, fixtures and any other personal property are levied against Landlord or Landlord's property or if the assessed value of Landlord's property is increased by the inclusion therein of a value placed upon such equipment, furniture, fixtures or any other personal property and if Landlord pays the taxes based upon such increased assessment, which Landlord shall have the right to do regardless of the validity thereof but only under proper protest if requested by Tenant, Tenant shall upon demand repay to Landlord the taxes so levied against Landlord or the proportion of such taxes resulting from such increase in the assessment, as the case may be.

4.5.2 Notwithstanding any contrary provision herein, Tenant shall pay prior to delinquency any (i) rent tax or sales tax, service tax, transfer tax or value added tax, or any other applicable tax on the rent or services herein or otherwise respecting this Lease, or (ii) taxes assessed upon or with respect to the possession, leasing, operation, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises or any portion of the Project, including the Parking Garage.

Landlord's Books and Records. Landlord shall maintain for a period of at least three (3) years following the end of the calendar year to which they pertain complete and accurate books and records of all Direct Expenses. Within one (1) year after receipt of a Statement by Tenant (the "Review Period"), if Tenant disputes the amount of Direct Expenses or Excess set forth in the Statement, an officer of Tenant or an independent, certified public accounting firm, designated and paid for by Tenant on a non-contingency fee basis, may, after reasonable notice to Landlord and at reasonable times, inspect Landlord's records with respect to the Statement at Landlord's offices (an "Audit"), provided that Tenant has paid all amounts required to be paid under the applicable Estimate Statement and Statement, as the case may be (provided that Tenant may pay such amounts under protest). In connection with such inspection, Tenant and Tenant's agents must agree in advance to follow Landlord's reasonable rules and procedures regarding inspections of Landlord's records, and shall execute a commercially reasonable confidentiality agreement regarding such inspection. The inspection shall be completed within sixty (60) days following the date Landlord makes such records available for review. Tenant's failure to dispute the amount of Additional Rent set forth in any Statement within the Review Period shall be deemed to be Tenant's approval of such Statement and Tenant, thereafter, waives the right or ability to dispute the amounts set forth in such Statement. If after such inspection, Tenant still disputes such Additional Rent, a determination as to the proper amount shall be made, Tenant's expense, by an independent certified public accountant who has not represented Landlord or Tenant or any of their affiliates in the preceding five (5) years (the "Accountant") selected by Landlord and subject to Tenant's reasonable approval, and such determination by the Accountant (a "Certification") shall be binding on Landlord and Tenant. If no Certification is performed, and Tenant's audit shows that Direct Expenses was overstated by more than three percent (3%), then the cost of Tenant's accountant shall be paid for by Landlord. If such final determination reflects that Tenant has overpaid Tenant's Share of Direct Expenses for the period in question, then Landlord shall credit such excess to Tenant's next payment of Base Rent and Direct Expenses (and if a Certification was performed, then the cost of the Certification and the Audit shall be paid by Landlord). If such final determination reflects that Tenant has underpaid Tenant's Share of Operating Expenses, Tenant shall promptly pay such additional Direct Expenses to Landlord within thirty (30) days after such determination (and if a Certification was performed, then the cost of the Certification and the Audit shall be paid by Tenant). Tenant hereby acknowledges that Tenant's sole right to inspect Landlord's books and records and to contest the amount of Direct Expenses payable by Tenant shall be as set forth in this Section 4.6, and Tenant hereby waives any and all other rights pursuant to applicable Law to inspect such books and records and/or to contest the amount of Direct Expenses payable by Tenant.

ARTICLE 5

USE OF PREMISES

5.1 <u>Permitted Use</u>. Tenant shall use the Premises solely for the Permitted Use set forth in <u>Section 7</u> of the Summary and Tenant shall not use or permit the Premises or the Project to be used for any other purpose or purposes whatsoever without the prior written consent of Landlord.

5.2 <u>Prohibited Uses</u>. The uses prohibited under this Lease shall include, without limitation, use of the Premises or a portion thereof for (i) offices of any agency or bureau of the United States or any state or political subdivision thereof; or (ii) offices or agencies of any foreign governmental or political subdivision thereof. Tenant further covenants and agrees that Tenant shall not use, or suffer or permit any person or persons to use, the Premises

or any part thereof for any use or purpose contrary to the provisions of the Rules and Regulations, or in violation of the laws of the United States of America, the State of California, or the ordinances, regulations or requirements of the local municipal or county governing body or other lawful authorities having jurisdiction over the Project) including, without limitation, any Environmental Laws (as defined below) (collectively, "Laws"). Notwithstanding the foregoing provisions of this Section 5.2 to the contrary. Tenant need not comply with any Laws so long as Tenant is contesting the validity thereof or the applicability thereof in accordance with the remainder of this Section 5.2. Tenant, at its expense, after notice to Landlord, may contest by appropriate proceedings prosecuted diligently and in good faith, the validity or applicability of any Laws with which Tenant is responsible for compliance hereunder, provided that (a) the condition which is the subject of such contest does not pose a danger to persons or property, (b) the certificate of occupancy or other occupancy permit for the Premises or the Project is neither subject to being suspended nor threatened to be suspended by reason of non-compliance or otherwise by reason of such contest, and (c) Landlord is not subject to criminal penalty or to prosecution for a crime by reason of Tenant's non-compliance or otherwise by reason of such contest. Tenant shall not use or allow the Premises to be used for any unlawful purpose nor shall Tenant cause, maintain or permit any nuisance in, on or about the Premises. Tenant shall comply with the REA; provided, that, Landlord shall not enter into any new REA after the Effective Date which would affect Tenant's use of or access to the Premises, or which would increase Tenant's obligations or decrease Tenant's rights under this Lease, without the approval of Tenant, such approval, not to be unreasonably withheld, conditioned or delayed. Tenant shall not cause or permit any "Hazardous Substance," as that term is defined below, to be kept, maintained, used, stored, produced, generated or disposed of (into the sewage or waste disposal system or otherwise) on or in the Premises by Tenant's agents, employees, contractors, invitees, assignees or sublessees, without first obtaining Landlord's written consent. However, notwithstanding the preceding sentence, Landlord agrees that Tenant may use, store and properly dispose of commonly available household cleaners and chemicals to maintain the Premises and Tenant's routine office operations (such as printer toner and copier toner) and UPS batteries. Tenant shall promptly notify, and shall direct Tenant's agents, employees contractors, invitees, assignees and sublessees to promptly notify, Landlord of any incident in, on or about the Premises, the Building or the Project that would require the filing of a notice under any federal, state, local or quasi-governmental law (whether under common law, statute or otherwise), ordinance, decree, code, ruling, award, rule, regulation or guidance document now or hereafter enacted or promulgated, as amended from time to time, in any way relating to or regulating any Hazardous Substance. As used herein, "Hazardous Substance" means any substance which is toxic, ignitable, reactive, or corrosive and which is regulated by any local government, the State of California, or the United States government. "Hazardous Substance" includes any and all material or substances which are defined as "hazardous waste," "extremely hazardous waste" or a "hazardous substance" pursuant to state, federal or local governmental law. "Hazardous Substance" also includes asbestos, polychlorobiphenyls (i.e., PCB's) and petroleum.

5.3 **Roof Top Deck**. Tenant shall have the exclusive right to use one or more portion(s) of the roof top of the 3^{rd} floor of the Building (i.e., on the level of the 4^{th} floor), which shall contain not more than 3,000 square feet of space in the aggregate within the area shown on **Exhibit E** attached hereto (the "**Deck Terrace**") as a roof top deck, free of charge throughout the Lease Term. Tenant acknowledges that the setbacks applicable to the Deck Terrace are 5 feet on the north, west and east sides, and 32 feet on the south side. Subject to the foregoing, Landlord shall approve the location of the portion(s) of the area shown on **Exhibit E** attached hereto to be used as the Deck Terrace (such approval not to be unreasonably withheld, conditioned or delayed). Tenant may, at Tenant's sole cost and expense and at Tenant's election in its sole and absolute discretion, design, engineer and install any necessary improvements to allow for the use of the Deck Terrace by Tenant, subject to the terms of this <u>Section 5.3</u>, and such improvements shall be deemed "Alterations" additionally subject to the terms of <u>Article 8</u> below. In particular, and without limitation, Landlord may disapprove any aspect of the Deck Terrace. Terrace are 5 for the Building or Premises, or (iv) violates any applicable Laws. Tenant acknowledges that Landlord may require Tenant to install reasonable screening as a condition to Tenant's use of the Deck Terrace. Subject to the terrace is of <u>Article 27</u> below, Landlord shall have the right to enter the Deck Terrace. Tenant shall kee, maintain and repair the Deck Terrace, and and safe condition, and consistent with the appearance of decks or patios of Comparable Buildings. Subject to the terms of this <u>Section 5.3</u> and the load requirements of the roof top, Tenant shall have the right to install decking and railings, and to place and maintain shrubbery and bushes (collectively, the "**Plants**") and furniture ("**Deck Terrace Furniture**") on the Deck Terrace, subject to Landlord's prior

given if Landlord fails to respond within ten (10) days after Tenant's request for such approval from Landlord. Any Deck Terrace Furniture and/or Plants must be reasonably secured to the Deck Terrace. The method by which any such items are secured to the Deck Terrace shall be subject to Landlord's prior approval, which approval shall not be unreasonably withheld, conditioned or delayed, and which approval shall be deemed given if Landlord fails to respond within ten (10) days after Tenant's request for such approval from Landlord. Tenant shall remain solely liable for any liability arising from Tenant's placement of such items on the Deck Terrace, and Landlord shall have no liability in connection therewith, except to the extent attributable to the negligence or willful misconduct of Landlord or its employees. Tenant shall remove any Deck Terrace Furniture and/or Plants upon the expiration or earlier termination of the Lease, and shall return the affected portion of the Deck Terrace to the condition the roof top would have been in had no such Deck Terrace si f the Deck Terrace were part of the Premises. Tenant acknowledges that the use of the Deck Terrace is subject to all applicable Laws and the REA, including, but not limited to, all requirements of the City and County of San Francisco, and the Landmarks Historical Preservation Committee. Landlord agrees to use commercially reasonable efforts, at no cost to Landlord, to assist and support Tenant in obtaining any such required approvals. However, Tenant's failure to obtain such approvals, and resultant failure to be allowed to install all or any portion of the Deck Terrace shall not be a default by Landlord under this Lease, or entite Tenant's notion of the applicable core.

5.4 Tenant's Dogs

5.4.1 In General. Subject to the provisions of this Section 5.4, and such additional reasonable rules and regulations as may be promulgated by Landlord from time to time with respect to the bringing of Dogs into the Project, Tenant shall be permitted to bring non-aggressive, fully domesticated, fully-vaccinated dogs (which dogs are owned by the Tenant or its employees) ("Tenant's Dogs"), into one (1) floor of the Premises designated by Tenant in a written notice delivered to Landlord from time to time (the "Dog Floor"). Tenant's Dogs shall be strictly controlled and supervised at all times by Tenant's employees. No more than twenty five (25) Tenant's Dogs will be allowed in the Premises at any one time. Tenant's Dogs must be on leashes while in any area of the Project outside of the Premises. Tenant's Dogs will be allowed on the Deck Terrace, or on any interior decks or balconies. Within three (3) business days following Tenant's receipt of Landlord's request, Tenant shall provide Landlord with reasonably satisfactory evidence showing that all current vaccinations have been received by the Tenant's Dogs shall not be brought to the Project if such dog is ill or contracts a disease that could potentially threaten the health or wellbeing of any tenant or occupant of the Building (which diseases may include, but shall not be limited to, rabies, leptospirosis and lyme disease). Tenant's Dogs in or about the Project shall be promptly removed and disposed of in trash receptacles designated by Landlord, and any areas of the Project affected by such waste shall be cleaned and otherwise sanitized. Landlord shall have the right to permanently ban any of Tenant's Dogs from the Project if any such dog previously exhibited dangerously aggressive behavior, damages or destroys property in the Project, or violates the specific provisions of this Section 5.4. Notwithstanding any provision to the contrary contained in this Amendment, if Tenant's Dogs are collectively found to be a substantial, repeated nuisance to the Project (an

5.4.2 Dog Access. Except as expressly set forth in this Section 5.4.2. Tenant's Dogs shall not be allowed in any Common Areas of the Building or Project. No Tenant's Dogs shall be allowed in any Building elevators at any time. Tenant may use Stairwell #9 on the P-1 level of the Parking Garage for the purposes of moving Tenant's Dogs from a car in the Parking Garage to such stairwell and from such stairwell to the courtyard and sidewalks of the Project. Tenant's Dogs shall be permitted in the courtyards and on the sidewalks of the Project while in transit to the Premises. From the courtyard and sidewalks of the Project, Tenant's Dogs may enter the Ground Floor Premises is not the Dog Floor, then Tenant's Dogs shall promptly be conveyed from the Ground Floor Premises into Stairwell #2 and from such stairwell to the Dog Floor.

5.4.3 <u>Costs and Expenses</u>. Tenant shall pay to Landlord, within thirty (30) after demand, all reasonable maintenance and repair costs incurred by Landlord that directly result from Tenant's Dogs presence in the

Building, Premises or Project, including, but not limited to, janitorial, waste disposal and landscaping costs and expenses. The indemnification provisions of <u>Article 10</u> of this Lease shall apply to any claims relating to any of Tenant's Dogs.

5.4.4 <u>Rights Personal</u>. The right to bring Tenant's Dogs into the Premises pursuant to this <u>Section 5.4</u> is personal to the Original Tenant and its Permitted Transferees, and may not be exercised by any subtenant or other Transferee.

ARTICLE 6

SERVICES AND UTILITIES

6.1 Standard Tenant Services. Landlord shall provide the following services on all days (unless otherwise stated below) during the Lease Term.

6.1.1 Subject to limitations imposed by applicable Laws, Landlord shall provide heating and air conditioning ("HVAC") between sixty eight (68) degrees and seventy six (76) degrees Fahrenheit during Building Hours. The "Building Hours" shall be from 7:00 A.M. to 6:00 P.M. Monday through Friday, except for the date of observation of New Year's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day, President's Day, and Martin Luther King Day (collectively, the "Holidays"). Landlord may add other Holidays to the extent such Holidays are observed by a majority of the Comparable Buildings. Tenant shall reasonably cooperate with Landlord at all times and abide by all reasonable regulations and requirements that Landlord may reasonably prescribe for the proper functioning and protection of the HVAC, electrical, mechanical and plumbing systems.

6.1.2 Landlord shall furnish electricity (subject to Tenant's obligations to pay for the use thereof as provided in this Lease) to each floor of the Premises for the operation of Tenant's electrical systems as follows: (i) for plug load at a demand load of not less than 4.0 watts per usable square foot per floor of the Premises (excluding electrical wiring and facilities for connection to Tenant's lighting fixtures); and (ii) for the operation of Tenant's lighting at a demand load of not less than 1.0 watts per usable square foot per floor of the Premises.

6.1.3 Tenant shall be responsible to pay directly, and not as a part of Operating Expenses, for the cost of all electricity consumed in the Premises (not including electricity associated with the Building HVAC system). Such payments shall be made directly to the applicable utility, or if Tenant is not able to contract directly with such utility, Tenant shall reimburse Landlord for all costs (without mark-up) of electricity consumed in the Premises (not including electricity associated with the Building HVAC system) based on a separate submeter (the "Submetering Equipment") installed in the Premises by Tenant (the cost of such installation shall be split equally between Landlord and Tenant). Tenant may audit Landlord's readings of the Submetering Equipment and Landlord shall deliver reasonably detailed invoices to Tenant reflecting Landlord's reading of the Submetering Equipment and resulting electricity costs.

6.1.4 Landlord shall replace Building standard lamps, starters and ballasts for Building standard lighting fixtures within the Premises as part of Operating Expenses. At Tenant's request, if Tenant installs lighting fixtures that are not the same as the current Building standard, Landlord will stock lamps, starters and ballasts for Tenant's fixtures, provided that to the extent such non-Building standard items are more expensive than Building standard, Tenant shall be responsible to pay any incremental additional cost.

6.1.5 Landlord shall provide city water from the regular Building outlets for drinking, lavatory and toilet purposes in the Building Common Areas.

6.1.6 Landlord shall provide janitorial services to the Premises and window washing services in accordance with the specification set forth on **Exhibit F** attached hereto, except the dates of the Holidays. Tenant acknowledges that the specification set forth on **Exhibit F** is subject to modification during the Lease Term, provided that in any event the janitorial services provided shall be consistent with those provided in Comparable Buildings.



6.1.7 Landlord shall provide nonexclusive, non-attended automatic passenger elevator service during the Building Hours, and shall have at least one elevator available at all other times. Landlord shall provide nonexclusive freight elevator service subject to reasonable prior scheduling, and provided that Tenant will be responsible for any incremental additional cost of providing any such freight elevator services during non-Building Hours (e.g., any additional security or personnel costs).

6.1.8 Landlord shall provide access control services for the Project in a manner commensurate with Comparable Buildings

6.1.9 Landlord shall provide tepid water for use in the Premises at points of supply, in amounts reasonably appropriate for a general office use.

6.2 **Overstandard Tenant Use**. Tenant's use of electricity shall never exceed the capacity of the feeders to the Project or the risers or wiring installation. If Tenant desires to use HVAC or fan service during hours other than Building Hours, then Tenant shall give Landlord reasonable prior notice of Tenant's desired use and Landlord shall provide such after-hours HVAC to Tenant at an hourly rate reasonably determined by Landlord based on Landlord's actual costs of electricity, engineering costs and incremental depreciation for reduction in the useful life of the Building's HVAC system resulting from such after-hours use. Such service is available on a "zone by zone" basis, with each floor of the Building containing four (4) separate zones. Landlord's current hourly rate for after-hours HVAC is ^{Redacted} per zone. If Tenant uses water in excess of amounts reasonably appropriate for general office use, or if Tenant desires to use other utilities not required to be supplied by Landlord under this Lease, Tenant shall pay to Landlord, upon billing, the cost of such excess consumption, and maintenance of equipment which is installed in order to supply such excess consumption, and the cost of the increased wear and tear on existing equipment caused by such excess consumption; and Landlord maintenance of such additional metering devices.

6.3 Interruption of Use. Tenant agrees that Landlord shall not be liable for damages, by abatement of Rent (except as otherwise provided in Section 6.6) or otherwise, for failure to furnish or delay in furnishing any service (including telephone and telecommunication services), or for any diminution in the quality or quantity thereof, when such failure or delay or diminution is occasioned, in whole or in part, by Force Majeure (as defined below); and such failures or delays or diminution shall never be deemed to constitute an eviction or disturbance of Tenant's use and possession of the Premises or relieve Tenant from paying Rent or performing any of its obligations under this Lease, except as otherwise provided in Section 6.6 or elsewhere in the Lease.

6.4 **Fire Stairs**. Landlord hereby agrees that Landlord shall not prohibit Tenant from using the fire stairs between floors of the Premises for the regular travel of employees between such floors, except to the extent Landlord is otherwise ordered by a governmental entity having jurisdiction over the Premises to so prohibit Tenant from such use. Tenant shall have the right to have access to such fire stairs, provided that Tenant shall cause the door(s) on each floor of the Premises from such fire stairs to be monitored and accessed by means of a security system installed by Tenant pursuant to <u>Section 6.7</u> of this Lease, which security system shall comply with all applicable Laws. Landlord hereby makes no representation to Tenant as to whether or not the use of the fire stairs between floors of the Premises for the regular travel of employees between such floors is allowed under applicable Laws.

6.5 **Rooftop Rights**. In accordance with, and subject to, the terms and conditions set forth in <u>Article 8</u>, below, and this <u>Section 6.5</u>, Tenant shall have the non-exclusive right to use the Building's roof for the installation and maintenance, at Tenant's sole cost and expense, satellite dishes/antennae or other telecommunications devices on the roof of the Building (and reasonable equipment and cabling related thereto), for receiving of signals or broadcasts (as opposed to the generation of transmission of any such signals or broadcasts) (all such equipment is defined collectively as the "**Telecommunications Equipment**"), as well as the installation, use and maintenance of conduit and other related equipment (the "**Ancillary Equipment**"). The manner of installation and the particular specifications of the Telecommunications Equipment and Ancillary Equipment shall be subject to the prior written approval of Landlord, which shall not be unreasonably withheld, conditioned or delayed. Landlord makes no representations or warranties whatsoever with respect to the condition of the roof of

the Building, or the fitness or suitability of the roof of the Building for the installation, maintenance and operation of the Telecommunications Equipment or the Ancillary Equipment, including, without limitation, with respect to the quality and clarity of any receptions and transmissions to or from the Telecommunications Equipment and the presence of any interference with such signals whether emanating from the Building or otherwise. Tenant shall maintain such Telecommunications Equipment and Ancillary Equipment in good condition and repair, at Tenant's sole cost and expense. Tenant shall remove such Telecommunications Equipment and Ancillary Equipment of the Lease, and shall return the affected portion of the roofbop and the Premises to the condition the roofbop and the Premises would have been in had no such Telecommunications Equipment and/or Ancillary Equipment to even see, and shall return the affected portion of the roof shall be used by Tenant only in connection with Tenant's use of the Premises, and Tenant shall not have the right to sublease or license the use of the roof to any third party (except to Permitted Transferees and other Transferees to whom the lease is assigned or portions of the Premises are subleased in accordance with <u>Article 14</u> below). Notwithstanding any such review or approval by Landlord, Tenant shall remain solely liable for any damage to any portion of the roof or noof membrane, specifically including any penetrations. Equipment and Ancillary Equipment shall, in all instances, comply with applicable Laws. Tenant acknowledges that Landlord has engaged a roof-top management company to coordinate the use of the roof, and that Tenant's sight as set forth above shall be subject to compliance with the reasonable and customary procedures instituted by such company, including payment of commercially reasonable installation and use for the roof, and that Tenant's asset forth above shall be subject to compliance with the reasonable and customary procedures institued by such company, inc

6.6 Abatement Event. If (i) Landlord fails to perform the obligations required of Landlord under the terms, covenants and conditions of this Lease, and (ii) such failure causes all or a portion of the Premises to be untenantable and unusable by Tenant (the "Abatement Event") and Landlord has not cured such Abatement Event within five (5) days after the receipt of written notice from Tenant or the day Landlord otherwise becoming aware of such Abatement Event, then Rent payable under this Lease for that portion of the Premises rendered untenantable and unusable and not used by Tenant shall be abated for the period from the commencement of such Abatement Event to the earlier of the date Landlord cures such Abatement Event or the date Tenant recommences the use of such portion of the Premises; provided that if the entire Premises has not been rendered untenantable and unusable by the Abatement Event, the amount of abatement that Tenant is entiled to receive shall be proteed based upon the percentage of the Premises (which shall be based on a ratio of the square feet of rentable area rendered untenantable and unusable. Notwithstanding the foregoing, in the event there is a disruption of services to the Premises. Landlord agrees to promptly use commercially reasonable efforts to resolve such failure of such services.

6.7 <u>Tenant's Security System</u>. Tenant shall be entitled to install a separate security system for the Premises and may include, without limitation, key-card systems, security lighting and video monitoring equipment in the Premises (which may also include card readers in the Common Areas adjacent to the entrances to the Premises, subject to Landlord's reasonable approval of the locations thereof) ("**Tenant's Security System**"), either as an Alteration (pursuant to the terms and conditions of <u>Article 8</u>) or as a part of the initial Tenant Improvements being constructed pursuant to the terms and conditions of the Tenant Work Letter; provided, however, (i) Tenant shall entite that Tenant's Security System shall otherwise be subject to the terms and conditions of Tenant's Security System shall be used to the terms and conditions of the Tenant Work Letter; as applicable. At Tenant's sole cost, Tenant shall be permitted to tie Tenant's Security Equipment into the Building Systems if requested by Tenant provided that (a) Tenant's Security Equipment is compatible with the Building Systems if requested by Tenant for the terms and wells as utilize its own employees or third parties to perform security system of a malfunction. Tenant shall at all times provide Landlord with a contact person who can disarm the security system and who is familiar with the functions of Tenant's Security System in the avent of a malfunction. Tenant shall at all times provide Landlord with a contact person who can disarm the affected portion of the Premises and Building to the condition the Premises and Building would have been in had no such Tenant's Security System had been installed (reasonable wear and tear excepted).

6.8 Supplemental HVAC Units. Notwithstanding anything to the contrary contained in this Lease, at any time during the Lease Term, Tenant shall have the right but not the obligation to install in the Premises, at Tenant's sole cost and expense, subject to the application of the Tenant Improvement Allowance, as applicable, one (1) or more Supplemental HVAC Unit's hall mean a self-enclosed electric heating and cooling unit of the size and tonnage, and having the specifications, approved by Landlord, which approval shall not be unreasonably withheld, conditioned or delayed. Tenant shall have access to and use of the Building's condenser water up to and not to exceed 20 tons maximum per floor (non-aggregated) for such facilities. If Tenant connects to the Building condenser water system, Tenant shall pay Landlord's standard cost for condenser water use (currently Reduced per ton per month). Tenant acknowledges that such charge varies based on monthly variation in the cost of electricity. Notwithstanding anything to the contrary set forth herein, at the end of the Lease Term, at Landlord's option, Tenant shall either: (1) remove, at Tenant's sole cost and expense, any Supplemental HVAC Unit and restore all portions of the Premises and the Building affected by such removal to their condition immediately prior to the installation of such equipment, ordinary wear and tear excepted; or (2) leave any such Supplemental HVAC Unit in place, in which event the Supplemental HVAC Unit shall be the property of Landlord.

6.9 Interior Decks. For purposes of this Lease, the "Interior Decks" shall mean the decks accessible from and forming a part of the Premises which extend into the interior Common Areas of the Building. Tenant may, at Tenant's cost and expense (but subject to applicable of the Tenant Improvement Allowance), modify or replace the plants and furniture on the Interior Decks; provided, that, Landlord shall approve any such modification or replacement of the plants and furniture (such approval not to be unreasonably withheld, conditioned or delayed). If Tenant elects to replace any such furniture, then Landlord, at Landlord's cost and expense, shall remove such existing furniture from the Interior Decks.

ARTICLE 7

REPAIRS

Landlord shall maintain in good condition and operating order and keep in good repair and condition, in a manner commensurate with first-class office buildings in the Financial District of San Francisco, California and in a clean, safe and neat condition, the structural portions of the Building, including the foundation, floor/ceiling slabs, roof structure (as opposed to roof membrane), curtain wall, exterior glass and mullions, columns, beams, shafts (including elevator shafts), stairs, stairwells, elevator cab, men's and women's washrooms, Building mechanical, electrical and telephone closets, and all common and public areas servicing the Building, including the parking areas, landscaping and exterior Project signage (collectively, "Building Structure") and the Base Building mechanical, electrical, life safety, plumbing, sprinkler systems and HVAC systems which were not constructed by Tenant Parties (collectively, the "Building Systems") and the Common Areas. Landlord shall use commercially reasonable efforts to minimize interference with Tenant's business in making any repairs or replacements to the Building on the Premises. Tenant shall, at Tenant's own expense, keep the Premises, including all improvements, fixtures, equipment, interior window coverings, and furnishings therein, and the floor or floors of the Building on which the Premises is located, in good order, repair and condition at all times during the Lease Term, but such obligation shall not extend to the Building Structure and the Building Systems. Excluding Tenant's upstressly set forth in this Lease, Tenant hereby waives any and all rights under and benefits of subSection 1 of Section 1932 and Sections 1941 and 1942 of the California Civil Code or under any similar Law.

ARTICLE 8

ADDITIONS AND ALTERATIONS

8.1 <u>Landlord's Consent to Alterations</u>. Tenant may not make any improvements, alterations, additions or changes to the Premises or any mechanical, plumbing or HVAC facilities or systems pertaining to the Premises (collectively, the "Alterations") without first procuring the prior written consent of Landlord to such Alterations, which consent shall not be unreasonably withheld by Landlord, provided it shall be deemed reasonable for Landlord to withhold its consent to any Alteration which adversely affects the structural portions or the systems or equipment of the Building or is visible from the exterior of the Building. If Landlord disapproves of any

proposed Alterations, Landlord shall respond, in writing, stating the grounds for such disapproval, within ten (10) days after receipt of Tenant's request for approval of the proposed Alterations, provided that, if Landlord fails to respond, Landlord shall be deemed to have approved the proposed Alterations. Notwithstanding the foregoing, Tenant shall be permitted to make Alterations following ten (10) days' notice to Landlord, but without Landlord's prior consent, to the extent that such Alterations do not (i) adversely affect the Building Systems, exterior appearance of the Building, structural integrity or roof of the Building, (ii) require the issuance of a building permit or other governmental approvals, or (iii) cost more than One Hundred Thousand and 00/100 Dollars (\$100,000.00) for a particular job of work (the "**Cosmetic Alterations**"). The construction of the initial improvements to the Premises shall be governed by the terms of the Tenant Work Letter and not the terms of this <u>Article 8</u>. Any Alterations shall comply with any reasonable Landlord's requirements relating to maintaining the current environmental certification of the Building (e.g., any LEED requirements).

8.2 Manner of Construction. Landlord shall approve the contractor who will perform any Alterations (other than Cosmetic Alterations). If Landlord disapproves of any proposed contractor, Landlord shall respond, in writing, stating the grounds for such disapproval, within ten (10) business days after receipt of Tenant's request for approval of the proposed contractor, provided that, if Landlord fails to respond, Landlord shall be deemed to have approved the proposed contractor. Tenant shall construct such Alterations and perform such repairs in a good and workmanlike manner, in material conformance with any and all applicable Laws, all in conformance with the Construction Rules (as defined in the Tenant Work Letter). In the event Tenant performs any Alterations in the Premises which require or give rise to governmentally required charges to the Base Building, then Landlord shall, at Tenant's expense, make such changes to the Base Building. In performing the work of any such Alterations, Tenant shall have the work performed in such manner so as not to eliminate access to the Project or any portion thereof, by any other tenant of the Project, and use commercially reasonable efforts to minimize interference with the business of other tenants in the Project. In addition to Tenant's obligations under <u>Article 9</u> of this Lease, upon completion of any Alterations, Tenant shall deliver to the Project management office a reproducible copy of the "as built" drawings of the Alterations, to the extent applicable, as well as copies of all permits, approvals and other documents issued by any governmental agency in connection with the Alterations.

8.3 **Payment for Tenant Improvements.** If payment is made directly to contractors, Tenant shall comply with Landlord's reasonable requirements for final lien releases and waivers in connection with Tenant's payment for work to contractors. Landlord shall not be entitled to receive an administrative or supervision fee with regard to repairs. Alterations or any other work arising from or related to this Lease except as expressly set forth herein. Tenant shall pay to Landlord, as additional rent, the reasonable costs of Landlord's third party engineers and other third party consultants (but not Landlord's on-site management personnel) which are reasonable required to be engaged by Landlord for review of all plans, specifications and working drawings for the Alterations, within thirty (30) days after Tenant's receipt of invoices from Landlord together with reasonable supporting evidence.

8.4 <u>Construction Insurance</u>. In addition to the requirements of <u>Article 10</u> of this Lease, in the event that Tenant makes any Alterations, prior to the commencement of such Alterations, Tenant shall provide Landlord with evidence that Tenant carries insurance in accordance with <u>Section 4.2.2.3</u> of the Tenant Work Letter attached hereto.

8.5 Landlord's Property. Except as expressly set forth in this Lease, Landlord and Tenant hereby acknowledge and agree that (i) all Alterations, improvements, fixtures, equipment and/or appurtenances which may be installed or placed in or about the Premises (excluding Tenant's removable trade fixtures, furniture or non-affixed office equipment), from time to time, shall be at the sole cost of Tenant and shall be and become part of the Premises and the property of Landlord, and (ii) the "Tenant Improvements" (as that term is defined in Section 2 of the Tenant Work Letter) to be constructed in the Premises pursuant to the terms and conditions of the Tenant Work Letter shall, upon completion of the same, be and become a part of the Premises and the property of Landlord, except that Tenant may remove any Alterations, improvements, fixtures and/or equipment which Tenant can substantiate to Landlord have not been paid for with any Tenant improvement allowance funds provided to Tenant by Landlord, provided Tenant repairs any damage to the Premises and Building caused by such removal. Furthermore, Landlord may, by written notice to Tenant prior to the end of the Lease Term, or given following any earlier termination of this Lease, require Tenant, at Tenant's expense, to remove any Specialty Improvements to the

Premises, and to repair any damage to the Premises and Building caused by such removal; provided, however, if, in connection with its notice to Landlord with respect to any such Specialty Improvements, (x) Tenant requests Landlord's decision with regard to the removal of such Specialty Improvements, and (y) Landlord thereafter agrees in writing to waive the removal requirement with regard to such Specialty Improvements, then Tenant shall not be required to so remove such Specialty Improvements, fails to address the removal requirement with regard to such Specialty Improvements, fails to address the removal requirement with regard to such Specialty Improvements. If Tenant fails to address the removal and/or to repair any damage caused by the removal of any Specialty Improvements, then at Landlord's option, either (A) Tenant shall be deemed to be holding over in the Premises and Rent shall continue to accrue in accordance with the terms of <u>Article 16</u>, below, until such work shall be completed, and/or (B) Landlord may do so and may charge the actual, reasonable out-of-pocket costs thereof to Tenant, which costs shall be paid to Landlord withiout thirty (30) days after receipt of invoice together with reasonable supporting evidence. "<u>Specialty Improvements</u>" means any improvement that are not of a customary general office nature, and shall include, without limitation, the following: (i) rolling files and structural supports, (ii) built-in or high-density file systems, (iii) any supplemental HVAC system installed by Tenant, (v) any improvements which affect the Building Structure, including, without limitation, any data center. Notwithstanding the foregoing, "Specialty Improvements which affect the Building or on behalf of Tenant in the Premises, including, without limitation, any data center. Notwithstanding the foregoing, "Specialty Improvements" shall be ofference rooms or training rooms except to the extent aspects thereof are specifically set forth above. Landlord shall not unreasonable withoh deal is appro

ARTICLE 9

COVENANT AGAINST LIENS

Tenant shall keep the Project and Premises free from any liens or encumbrances arising out of the work performed, materials furnished or obligations incurred by or on behalf of Tenant, and shall protect, defend, indemnify and hold Landlord harmless from and against any claims, liabilities, judgments or costs (including, without limitation, reasonable attorneys' fees and costs) arising out of same or in connection therewith. Tenant shall give Landlord notice at least ten (10) days prior to the commencement of any such work on the Premises (or such additional time as may be necessary under applicable Laws) to afford Landlord the opportunity of posting and recording appropriate notices of non-responsibility. Tenant shall remove any such lien or encumbrance by bond or otherwise within thirty (30) days after notice by Landlord, and if Tenant shall fail to do so, Landlord may pay the amount necessary to remove such lien or encumbrance, without being responsible for investigating the validity thereof. The amount so paid shall be deemed Additional Rent under this Lease payable upon demand, without limitation as to other remedies available to Landlord under this Lease. Nothing contained in this Lease shall authorize Tenant to do any act which shall subject Landlord's title to the Building or Premises to any liens or encumbrances whether claimed by operation of law or express or implied contract.

ARTICLE 10

INDEMNIFICATION AND INSURANCE

10.1 Indemnification and Waiver. Except to the extent arising from the negligence or willful misconduct of the Landlord or the "Landlord Parties" (as that term is defined below), Tenant hereby assumes all risk of damage to property or injury to persons in, upon or about any Portion of the Premises which has been delivered to Tenant from any cause whatsoever and agrees that Landlord, its partners, subpartners and their respective officers, agents, servants, employees, and independent contractors (collectively, "Landlord Parties") shall not be liable for, and are hereby released from any responsibility for, any damage either to person or property or resulting from the loss of use thereof, which damage is sustained by Tenant or by other persons claiming through Tenant. Except to the extent arising from the negligence or willful misconduct of Landlord Parties, Tenant shall indemnify, defend, protect, and hold harmless the Landlord Parties from and against any and all loss, cost, damage,

expense and liability (including without limitation court costs and reasonable attorneys' fees) incurred in connection with or arising from: (a) any causes in, on or about any Portion of the Premises which has been delivered to Tenant; (b) the use or occupancy of the Premises by Tenant or any person claiming under Tenant; (c) any activity, work, or thing done, or permitted or suffered by Tenant in or about the Premises; (d) any negligence or willful misconduct of Tenant or any person claiming under Tenant; (b) the use or occupancy of the premises by Tenant or rany person claiming under Tenant; (c) any activity, work, or thing done, or permitted or suffered by Tenant in or about the Premises; (d) any negligence or willful misconduct of Tenant or any person claiming under Tenant; or the contractors, agents, employees, or visitors of Tenant or any other person ends or about the Project (collectively, "**Tenant Parties**"); (e) any injury or damage to the personal property or other items within the Premises. Except to the extent arising from the negligence or willful misconduct of the Tenant or the Tenant or the Tenant Indemnites (as defined below), Landlord shall indemnify, protect and hold Tenant and its officers, directors, employees, agents, shareholders and affiliates ("**Tenant Indemnites**") harmless from and against any and all liabilities, claims and/or losses of any kind arising, directly or indirectly, entirely or in part, out of the negligence or willful misconduct of Landlord, the Landlord Parties or breach of Landlord's representations and warranties set forth herein or any violation by Landlord of any Law. Each party's agreement to indemnify the other party pursuant to this <u>Section 10.1</u> is not intended and shall not relieve any insurance carrier of its obligations under policies required to be carried by such party pursuant to this Lease, to the extent such policies cover the matters subject to such party's indemnification obligations; nor shall they supersede any inconsistent agreement of the party su

10.2 Tenant's Compliance With Landlord's Fire and Casualty Insurance. If Tenant's conduct or use of the Premises for other than general office use causes any increase in the premium for Landlord's insurance policies then Tenant shall reimburse Landlord for any such increase.

10.3 Tenant's Insurance. Tenant shall maintain the following coverages in the following amounts.

10.3.1 Commercial General Liability Insurance covering the insured against claims of bodily injury, personal injury and property damage (including loss of use thereof) arising out of Tenant's operations, and contractual liabilities for limits of liability not less than:

Bodily Injury and	\$5,000,000 each occurrence
Property Damage Liability	\$5,000,000 annual aggregate
1 9 6 9	60 0
Personal Injury Liability	\$5,000,000 each occurrence
· · · · · · · · · · · · · · · · · · ·	\$5,000,000 annual aggregate 0% Insured's participation
	\$5,000,000 annual aggregate 0 % filsured s participation

10.3.2 Physical Damage Insurance covering (i) all office furniture, business and trade fixtures, office equipment, free-standing cabinet work, movable partitions, merchandise and all other items of Tenant's property on the Premises installed by, for, or at the expense of Tenant, (ii) the "Tenant Improvements," as that term is defined in <u>Section 2.1</u> of the Tenant Work Letter, and (iii) all other improvements, alterations and additions to the Premises. Such insurance shall be written on an "all risks" of physical loss or damage basis, for the full replacement cost value (subject to reasonable deductible amounts) new without deduction for depreciation of the covered items and in amounts that meet any co-insurance clauses of the policies of insurance and shall include coverage for damage or other loss caused by fire or other peril including, but not limited to, vandalism and malicious mischief, theft, water damage of any type, including sprinkler leakage, bursting or stoppage of pipes, and explosion, and providing business interruption coverage for a period of one year.

10.3.3 Worker's Compensation and Employer's Liability or other similar insurance pursuant to all applicable state and local statutes and regulations.

10.3.4 Business Income Interruption for one (1) year plus Extra Expense insurance in such amounts as will reimburse Tenant for actual direct or indirect loss of earnings attributable to the risks outlined in Section 10.3.2 above.

10.4 Landlord Insurance. Throughout the Lease Term, Landlord agrees to maintain (i) fire and extended coverage insurance, and, at Landlord's option, earthquake damage coverage, terrorism coverage, wind and hurricane coverage, and such additional property insurance coverage as Landlord deems appropriate and is customarily carried by other prudent landlords who own similar buildings to the Building in the Financial District of San Francisco, on the insurable portions of Building in an amount not less than the fair replacement value thereof, subject to reasonable deductibles (ii) boiler and machinery insurance amounts and with deductibles that would be considered standard for similar buildings to the Building in the Financial District of San Francisco, and (iii) commercial general liability insurance with a combined single limit coverage of at least \$3,000,000.00 per occurrence. All such insurance shall be obtained from insures which meet the requirements of <u>Section 10.5</u> below. Duly executed certificates showing the material terms for the same, shall be deposited with Tenant on the date Tenant first occupies the Premises and upon renewals of such policies upon written request. Any failure of Landlord to obtain and maintain the insurance policies and coverages required hereunder or failure by Landlord to meet any of the insurance of this Lease beyond applicable notice and cure periods shall constitute a material default hereunder, and Landlord shall be solely responsible for any loss suffered by Tenant as a result of such failure.

10.5 Form of Policies. The minimum limits of policies of insurance required of Tenant under this Lease shall in no event limit the liability of Tenant under this Lease. Such insurance shall (i) name Landlord, and any other party the Landlord so specifies, as an additional insured, including Landlord's managing agent, if any (for liability policies only); (ii) specifically cover the liability assumed by Tenant under this Lease, including, but not limited to, Tenant's obligations under <u>Section 10.1</u> of this Lease; (iii) be issued by an insurance company having a rating of not less than A- VII in A.M. Best's Insurance Guide or which is otherwise reasonably acceptable to Landlord and licensed to do business in the State of California; and (iv) be primary insurance as to all claims thereunder and provide that any insurance carried by Landlord is excess and is non-contributing with any insurance requirement of Tenant. Tenant shall endeavor to cause such policies to provide that said insurance shall lot be canceled or coverage changed unless thirty (30) days' prior written notice (or ten (10) days in the event of non-payment of premium) shall have been given to Landlord and any mortgagee of Landlord. Tenant shall deliver said policy or policies or certificates thereof to Landlord on or before the Tranche I Commencement Date and prior to the expiration dates thereof. In the event Tenant shall fail to procure such insurance, or to deliver such policies or certificate, and such failure continues for more than five (5) business days after written notice, Landlord may, at its option, procure such policies for the account of Tenant, and the cost thereof shall be paid to Landlord within thirty (30) days after delivery to Tenant of bills therefor.

10.6 **Subrogation**. Landlord and Tenant intend that their respective property loss risks shall be borne by reasonable insurance carriers to the extent above provided, and Landlord and Tenant hereby agree to look solely to, and seek recovery only from, their respective insurance carriers in the event of a property loss to the extent that such coverage is agreed to be provided hereunder. The parties each hereby waive all rights and claims against each other for such losses, and waive all rights of subrogation of their respective insurance of subrogation shall not affect the right to the insured to recover thereunder. The parties agree that their respective insurance policies are now, or shall be, endorsed such that the waiver of subrogation shall not affect the right of the insured to recover thereunder, so long as no material additional premium is charged therefor.

10.7 Additional Insurance Obligations. Tenant shall carry and maintain during the entire Lease Term, at Tenant's sole cost and expense, increased amounts of the insurance required to be carried by Tenant pursuant to this <u>Article 10</u> and such other reasonable types of insurance coverage and in such reasonable amounts covering the Premises and Tenant's operations therein, as may be reasonably requested by Landlord, but in no event in excess of the amounts and/or types of insurance then being required by landlords of the Comparable Buildings. Landlord may change the amounts and/or types of insurance required to be expiration of the initial Lease Term of the Tranche I Premises and only one time during each and every five (5)-year period (if any) following the expiration of the initial Lease Term of the Tranche I Premises.

ARTICLE 11

DAMAGE AND DESTRUCTION

11.1 **Repair of Damage to Premises by Landlord**. Tenant shall promptly notify Landlord of any damage to the Premises resulting from fire or any other casualty. If the Premises or any Common Areas serving or providing access to the Premises shall be damaged by a fire or any other casualty (collectively, a "Casualty"), Landlord shall promptly and dilgently, subject to Force Majeure delays and subject to all other terms of this <u>Article 11</u>, restore the Base Building and such Common Areas. Such restoration shall be to substantially the same condition of the Base Building and the Common Areas prior to the Casualty, except for modifications required by Laws. Within sixty (60) days after the occurrence of any Casualty, Landlord shall cause to be delivered to Fenant an estimate (the "Estimate"), prepared by a qualified, independent, experienced and reputable architect and/or general contractor and addressed to Tenant, of the number of days (assuming no Force Majeure delay), measured from the date of the Casualty, that will be required for Landlord upon the occurrence of any damage to the Premises, which repairs shall be romportly inform its insurance carrier of any such damage. Tenant shall promptly notify Landlord upon the occurrence of any damage to the Premises, which repairs shall be made in compliance with the terms of <u>Article 8</u>, above. Landlord shall not be liable for any inconvenience or annoyance to Tenant or its visitors, or injury to Tenant's business resulting in any way from such damage or the repair thereof; provided that Landlord use commercially reasonable efforts to minimize interference with Tenant's business during any such repairs. If such Casualty shall have damaged the Premises or Common Areas necessary to Tenant's occupancy, and the Premises is not occupied by Tenant as a result thereof, then during the time and to the extent the Premises is unfit for occupancy. Tenant's right to rent abatement pursuant to the preceding sentence shall be reasonable for Tenant to terminate all business oreasonab

11.2 Termination Rights. Notwithstanding the terms of Section 11.1 of this Lease, Landlord may elect not to rebuild and/or restore the Premises, Building and/or Project, and instead terminate this Lease, by notifying Tenant in writing of such termination within sixty (60) days after the date of discovery of the damage, such notice to include a termination date giving Tenant sixty (60) days to vacate the Premises, but Landlord may so elect only if one or more of the following conditions is present: (i) according to the Estimate, repairs to the Base Building and Common Areas cannot be completed within twelve (12) months after the date of discovery of the damage (when such repairs are made without the payment of overtime or other premiums); (ii) more than \$1,000,000 of the damage (including any earthquake insurance deductible, but not including any other deductibles) is not covered by Landlord's insurance policies other than as a result of Landlord's failure to maintain insurance required by Section 10.4 above (the "Insurance Shortfall"), provided that if Landlord elects to terminate the Lease based on such Insurance Shortfall, Tenant may elect to pay such Insurance Shortfall itself, in which case Landlord shall not have the right to terminate the Lease on such basis, or (iii) the damage occurs during the last twelve (12) months of the Lease Term and, according to the Estimate, restoration and repair of the Base Building and Common Areas cannot be completed within sixty (60) days after the date of discovery of the damage (when such repairs are made without the payment of overtime or other premiums). Notwithstanding the terms of Section 11.1 of this Lease, Fernand according to the Estimate, restoration and repair of the Base Building and Common Areas cannot be completed within sixty (60) days after the date of discovery of the damage (when such repairs are made without the payment of overtime or other premiums). Notwithstanding the terms of Section 11.1 of this Lease, Fernand according to the Estimate, repairs to t

(ii) the date that is twelve (12) months after the date of discovery of the damage, Tenant shall have the right to terminate this Lease, by notice to Landlord (the "Damage Termination Notice"), effective as of a date set forth in the Damage Termination Notice (the "Damage Termination Date"), which Damage Termination Date may be up to sixty (60) days after delivery of the Damage Termination Notice. Notwithstanding the foregoing, if Tenant delivers a Damage Termination Notice to Landlord, then Landlord shall have the right to suspend the occurrence of the Damage Termination Date for a period ending thirty (30) days after the Damage Termination Date set forth in the Damage Termination Notice by delivering to Tenant, within five (5) business days of Landlord's receipt of the Damage Termination Notice, a certificate of Landlord's contractor's good faith judgment that the repairs shall be substantially completed within thirty (30) days after the Damage Termination Notice shall termination of such thirty-day period, then the Damage Termination Notice shall terminate the course of such thirty-day period. At any time, from time to time, after the date of completed mage, Tenant may request that Landlord inform Tenant of Landlord's reasonable opinion of the date of completion of the repairs and Landlord shall respond to such request within five (5) business days.

11.3 <u>Waiver of Statutory Provisions</u>. The provisions of this Lease, including this <u>Article 11</u>, constitute an express agreement between Landlord and Tenant with respect to any and all damage to, or destruction of, all or any part of the Premises, the Building or the Project, and any statute or regulation of the State of California, including, without limitation, Sections 1932(2) and 1933(4) of the California Civil Code, with respect to any rights or obligations concerning damage or destruction in the absence of an express agreement between the parties, and any other statute or regulation, now or hereafter in effect, shall have no application to this Lease or any damage or destruction to all or any part of the Premises, the Building or the Project.

11.4 Date of Casualty. The terms and provisions of this <u>Article 11</u> shall apply to any damage to the Building or the Project caused as a result of fire or other casualty, regardless of whether such damage occurs during, or prior to the commencement of, the initial Lease Term.

11.5 Parking Garage. If Tenant is prevented from using, and does not use the Parking Garage as a result of any damage by fire or other casualty, and Tenant is then still occupying and using the Premises for the Permitted Use, then Landlord shall, upon written request from Tenant, use commercially reasonable efforts to provide Tenant with reasonable alternate parking (based on the rentable square footage of the Premises that Tenant is then occupying and using for the Permitted Use), which alternate parking shall include, if outside the Project, a commercially reasonable shuttle service from such alternate parking location to the Project, until such time as the damage to the Parking Garage is substantially completed.

ARTICLE 12

NONWAIVER

No provision of this Lease shall be deemed waived by either party hereto unless expressly waived in a writing signed thereby. The waiver by either party hereto of any breach of any term, covenant or condition herein contained shall not be deemed to be a waiver of any subsequent breach of same or any other term, covenant or condition herein contained. The subsequent acceptance of Rent hereunder by Landlord's shall not be deemed to be a waiver of any preceding breach by Tenant of any term, covenant or condition herein contained. The subsequent acceptance of Rent hereunder by Landlord's knowledge of such preceding breach at the time of acceptance of such Rent. No acceptance of a lesser amount than the Rent herein stipulated shall be deemed a waiver of Landlord's right to receive the full amount due, nor shall any endorsement or statement on any check or payment or any letter accompanying such check or payment be deemed an accord and satisfaction, and Landlord may accept such check or payment without prejudice to Landlord's right to receive the full amount due. No receipt of monies by Landlord from Tenant after the termination of this Lease shall in any way alter the length of the Lease Term or of Tenant's right of possession hereunder, or after the giving of any notice shall reinstate, continue or extend the Lease Term or affect any notice given Tenant prior to the receipt of such monies, it being agreed that after the service of notice or the commencement of a suit, or after final judgment for possession of the Premises, Landlord may receive and collect any Rent due, and the payment of said Rent shall not waive or affect said notice,

suit or judgment. No payment of Rent by Tenant after a breach by Landlord shall be deemed a waiver of any breach by Landlord.

ARTICLE 13

CONDEMNATION

If the whole or any part of the Premises, Building or Project shall be taken by power of eminent domain or condemned by any competent authority for any public or quasi-public use or purpose, or if any adjacent property or street shall be so taken or condemned, or reconfigured or vacated by such authority in such manner as to require the use, reconstruction or remodeling of any part of the Premises, Building or Project, or if Landlord shall grant a deed or other instrument in lieu of such taking by eminent domain or condemnation, Landlord shall have the option to terminate this Lease effective as of the date possession is required to be surrendered to the authority if more than twenty-five percent (25%) of the rentable square feet of the Premises is taken, or if access to the Premises is substantially impaired, in each case for a period in excess of one hundred eighty (180) days, Tenant shall have the option to terminate this Lease effective as of the date possession is required to be surrendered to the authority for any compensation because of such taking and Landlord shall be entitled to the entire award or payment in connection therewith, except that Tenant shall have the right to file any separate claim available to Tenant for any taking of Tenant's personal property and fixtures belonging to Tenant and removable by Tenant upon expiration of the Lease Term pursuant to the terms of this Lease shall not be so terminated, the Rent shall be proportionately abated in proportion to the ratio that the amount of rentable square feet of the Premises taken bears to the total rentable square feet of the Premises. Tenant shall not because feet of the Premises. Tenant hereby waives any and all rights it might otherwise have pursuant to Section 1265.130 of The California Code of Civil Procedure. Notwithstanding anything to the contrary contained in this <u>Article 13</u>, in the event of a temporary taking of all or any portion of the tamount of rentable square feet of the Premises. Landlord shall be entitled to receive the en

ARTICLE 14

ASSIGNMENT AND SUBLETTING

14.1 **Transfers**. Tenant shall not, without the prior written consent of Landlord, assign, mortgage, pledge, hypothecate, encumber, or permit any lien to attach to, or otherwise transfer, this Lease or any interest hereunder, permit any assignment, or other transfer of this Lease or any interest hereunder by operation of law, sublet the Premises or any part thereof, or enter into any license or concession agreements or otherwise permit the occupancy or use of the Premises or any part thereof by any persons other than Tenant and its employees and contractors (all of the foregoing are hereinafter sometimes referred to collectively as "**Transfers**" and any person or entity to whom any Transfer is made or sought to be made is hereinafter sometimes referred to as a "**Transfere**"). If Tenant desires Landlord's consent to any Transfer, Tenant shall notify Landlord in writing, which notice (the "**Transfer Notice**") shall include (i) the proposed effective date of the Transfer, which shall not be less than fifteen (15) business days nor more than one hundred eighty (180) days after the date of delivery of the Transfer Notice, (ii) a description of the portion of the Premises to be transferred (the "**Subject Space**"), (iii) a calculation of the "Transfer Premium," as that term is defined in <u>Section 14.3</u> below, in connection with such Transfer and the name and address of the proposed Transfere, and a copy of all existing executed and/or proposed documentation pertaining to the proposed Transfer, including all existing operative documents to be executed to evidence such Transfer or owner thereof. Any Transfer made without Landlord's option, constitute a default by Tenant under this Lease. Whether or not Landlord consents to any proposed Transfer, Tenant shall pay Landlord's reasonable

review and processing fees, as well as any reasonable professional fees (including, without limitation, attorneys', accountants', architects', engineers' and consultants' fees) incurred by Landlord, not to exceed Redacted for a particular Transfer, within thirty (30) days after written request by Landlord.

14.2 Landlord's Consent. Landlord shall not unreasonably withhold, condition or delay its consent to any proposed Transfer of the Subject Space to the Transferee on the terms specified in the Transfer Notice. If Landlord disapproves of any proposed assignment or sublease, Landlord's response shall explain the grounds for such disapproval. If Landlord fails to make such election within seven (7) business days after Landlord's receipt of the notice from Tenant, Tenant may deliver a written "reminder notice" requesting Landlord's failure to make such election within thre (3) business days after such reminder notice shall be deemed to be Landlord's consent to the proposed assignment or sublease. Without limitation as to other reasonable grounds for withholding consent, the parties hereby agree that it shall be reasonable under this Lease and under any applicable Law for Landlord to withhold consent to any proposed Transfer if the intended use of the Premises by the assignee or sublesse is not permitted by Laws or is not a Permitted Use. The parties hereby agree that it shall only be reasonable under this Lease and under any applicable Law for Landlord to withhold consent to any proposed Transfer if the following apply:

- 14.2.1 With respect to Tenant's request for Landlord's approval of a subletting of this Lease by Tenant:
 - 14.2.1.1 The Transferee intends to use the Subject Space for purposes which are not permitted under this Lease;

14.2.1.2 The Transferee is associated with services, products or ideologies of a sexual, provocative, political, or similarly controversial nature which is inconsistent with tenants in

14.2.1.3 The Subject Space is in excess of 15,000 rentable square feet, upon such Transfer, Tenant will be subleasing more than 50% of the Premises, and the Transferee is not a party of reasonable financial worth and/or financial stability in light of the responsibilities to be undertaken in connection with the Transfer on the date consent is requested; and

14.2.1.4 The sublessee (or any affiliate of the sublessee) is then negotiating with Landlord or has negotiated with Landlord within the previous three (3) months to become a tenant of the office space portion of the Project, or is a current tenant or subtenant within the office portion of the Building or Project, provided, however, Landlord must be able to accommodate such sublessee with available inventory in the Project similar to the Subject Space (including the same ceiling height and similar size floor plates). In the event Tenant seeks to enter into a sublease, then Tenant may provide written notice to Landlord of its desire to sublease a portion of the Premises and Landlord and Tenant agree, in good faith, to discuss Tenant's sublease and whether Tenant shall have the right to approach any current tenants or subtenants in the Project; provided, however, Landlord may elect that Tenant shall not have the right to contact any current tenant or subtenants of the Project for Tenant's subleasing of the Premises if Landlord reasonably believes it can accommodate such tenant's or subtenant's space needs in the Project with currently or imminently available inventory in the Project similar to the Subject Space (including the same ceiling height and similar floor plates).

14.2.2 With respect to Tenant's request for Landlord's approval of an assignment of this Lease by Tenant:

14.2.2.1 The conditions set forth in Sections 14.2.1.1 and 14.2.1.2, above;

14.2.2.2 The Transferee is not a party of reasonable financial worth and/or financial stability in light of the responsibilities to be undertaken in connection with the Transfer on the

date consent is requested; and

the Comparable Buildings;

14.2.2.3 The assignee (or any affiliate of the assignee) is then negotiating with Landlord or has negotiated with Landlord within the previous three (3) months to become a tenant of the office space portion of the Project, or is a current tenant or subtenant within the office portion of the Building or Project, provided, however, Landlord must be able to accommodate such assignee with available inventory in the Project similar to the Premises (including the same ceiling height and similar size floor plates). In the event Tenant seeks to assign the Lease, then Tenant may provide written notice to Landlord of its desire to assign the Lease and Landlord and Tenant agree, in good faith, to discuss Tenant's assignment and whether Tenant shall have the right to approach any current tenants or subtenants in the Project; provided, the Project for Tenant's assignment of its Lease if Landlord reasonably believes it can accommodate such tenant's assignee to the Project similar to the Project with currently or imminently available inventory in the Project similar to the Premises (including the same ceiling height and similar size floor plates).

If Landlord consents to any Transfer pursuant to the terms of this Section 14.2. Tenant may within six (6) months after Landlord's consent, but not later than the expiration of said six (6) month period, enter into such Transfer of the Premises or portion thereof, provided that if there are any changes in the terms and conditions from those specified in the Transfer Notice such that Landlord would initially have been entitled to refuse its consent to such Transfer under this Section 14.2 then Tenant shall again submit the Transfer to Landlord for its approval. If Tenant or any proposed Transfere claims that Landlord has unreasonably withheld or delayed its consent under this Section 14.2 or otherwise has breached or acted unreasonably under this <u>Article 14</u>, Tenant shall have all of its rights and remedies at law and in equity, provided that Tenant hereby waives any right at law or equity to terminate this Lease, on its own behalf and, to the extent permitted under all applicable Laws, on behalf of the proposed Transferee.

14.3 Transfer Premium. If Landlord consents to a Transfer, as a condition thereto which the parties hereby agree is reasonable, Tenant shall pay to Landlord fifty percent (50%) of any "Transfer Premium," as that term is defined in this <u>Section 14.3</u>, received by Tenant from such Transferee. "Transfer Premium" shall mean all rent, additional rent or other consideration payable by such Transfere in connection with the Transfer Premium shall be calculated, after first deducting the reasonable expenses incurred by Tenant for (i) any changes, alterations and improvements to the Premises in connection with the Transfer Premium shall be calculated, after first deducting the reasonable expenses incurred by Tenant for (i) any changes, alterations and improvements to the Premises in connection with the Transfer Premium shall be calculated, after first deducting the reasonable expenses incurred by Tenant for (i) any changes, alterations and improvements to the Premises in connection with the Transfer ter term by tenant in separately demising the portion of the Premises ("Transfer Costs"). "Transfer Premium" shall also include, but not be limited to, key money, bonus money or other cash consideration paid by Tenant to Transfere to Tenant to Transfere in connection with such Transfer, and any payment in excess of fair market value for services rendered by Tenant to Transfere or for assets, fixtures, inventory, equipment, or furniture transferred by Tenant to Transfere in connection with such Transfer. Tenant shall first recoup all Transfer Costs from the Transfere before any Transfer Premium must be paid to Landlord. Notwithstanding anything set forth herein to the contrary, in no event shall Tenant be required to pay a Transfer Premium in connection with a Transfer Pursuant to <u>Section 14.7</u> of this Lease below.

14.4 Landlord's Option as to Subject Space. Notwithstanding anything to the contrary contained in this Article 14, in the event Tenant contemplates a Transfer of one or more full floors of the Premises (or of all of the Premises located on a particular floor of the Building), for substantially all of the remainder of the Lease Term (i.e., would leave less than one (1) month of the Lease Term following the expiration of such Transfer), Tenant shall give Landlord notice (the "Intention to Transfer Notice") of such contemplated Transfer (whether or not the contemplated Transfer contemplated Transfer swhich Tenant intends to Transfer Notice shall specify the portion of and amount of rentable square feet of the Premises which Tenant intends to Transfer (the "Contemplated Transfer Space"), the contemplated de of commencement of the Contemplated Transfer (the "Contemplated Transfer (the "Contemplated Transfer (the addition to Transfer Notice is delivered to Landlord pursuant to this Section 14.4 in order to allow Landlord to elect to recapture the Contemplated Transfer Space for the term set forth in the Intention to Transfer Notice. Thereafter, Landlord shall have the option, by giving written notice to Tenant within ten (10) business days after receipt of any Intention to Transfer Notice, to recapture the Contemplated Transfer Space. Such recapture shall cancel and terminate this Lease with respect to such Contemplated Transfer

Space as of the Contemplated Effective Date. In the event of a recapture by Landlord, if this Lease shall be canceled with respect to less than the entire Premises, the Rent reserved herein shall be prorated on the basis of the number of rentable square feet retained by Tenant in proportion to the number of rentable square feet contained in the Premises, and this Lease as so amended shall continue thereafter in full force and effect, and upon request of either party, the parties shall execute written confirmation of the same. If Landlord declines, or fails to elect in a timely manner, to recapture such Contemplated Transfer Space under this <u>Section 14.4</u>, then, subject to the other terms of this <u>Article 14</u>, for a period of nine (9) months (the "Nine Month Period,") commencing on the last day of such ten (10) business day period, Landlord shall not have any right to recapture the Contemplated Transfer Space with respect to any Transfer made during the Nine Month Period, provided that any such Transfer is on material economic terms that, on a net effective basis, are not more than five percent (5%) more favorable to the subtenant than the terms set forth in the Intention to Transfer Notice, and provided further that any such Transfer of such Contemplated Transfer Space consummated within such Nine Month Period), Tenant shall again be required to submit a new Intention to Transfer Notice to Landlord with respect any contemplated Transfer, as provided above in this <u>Section 14.4</u>.

14.5 Effect of Transfer. If Landlord consents to a Transfer, (i) the terms and conditions of this Lease shall in no way be deemed to have been waived or modified, (ii) such consent shall not be deemed consent to any further Transfer by either Tenant or a Transfere, (iii) Tenant shall deliver to Landlord, promptly after execution, an original executed copy of all documentation pertaining to the Transfer in form reasonably acceptable to Landlord, (iv) Tenant shall furnish upon Landlord's request a complete statement, certified by an independent certified public accountant, or Tenant's chief financial officer, setting forth in detail the computation of any Transfer Premium Tenant has derived and shall derive from such Transfer, and (v) no Transfer relating to this Lease or agreement entered into with respect thereto, whether with or without Landlord's consent, shall relieve Tenant or any guarantor of the Lease from any liability under this Lease, including, without limitation, in connection with the Subject Space. Notwithstanding anything to the contrary set forth in this Lease, the original Tenant named in this Lease shall not be bound by any amendment or agreement which is not expressly executed and delivered by the original Tenant named in this Lease, and Landlord is not authorized or entitled to rely upon any assignee or subtenant to the contrary, whether or not the same is a Permitted Transfere. Landlord or its authorized representatives shall have the right at all reasonable times to audit the books, records and papers of Tenant relating to any Transfer, and shall have the right to make copies thereof. If the Transfer Premium respecting any Transfer shall be found understated, Tenant shall, within thirty (30) days after demand, pay the deficiency, and if understated by more than two percent (2%), Tenant shall pay Landlord's costs of such audit. In connection with such audit, Landlord and Landlord's agents must agree in advance to follow Tenant's reasonable rules and procedures regarding inspections of Tenant

14.6 Occurrence of Default. Any Transfer hereunder shall be subordinate and subject to the provisions of this Lease, and if this Lease shall be terminated during the term of any Transfer, Landlord shall have the right to: (i) treat such Transfer as cancelled and repossess the Subject Space by any lawful means, or (ii) require that such Transfere attorn to and recognize Landlord as its landlord under any such Transfer. If Tenant shall be in default under this Lease beyond all applicable notice and cure periods expressly set forth in this Lease, Landlord is hereby irrevocably authorized, as Tenant's agent and attorney-in-fact, to direct any Transfere to make all payments under or in connection with the Transfer directly to Landlord (which Landlord shall apply towards Tenant's obligations under this Lease) until such default is cured. Such Transferee shall be deemed a vaiver of any provision of this <u>Article 14</u> or the approval of any Transfere to be performed or observed under this Lease, whether thereofore or thereafter accruing. In no event shall Landlord's enforcement of any provision of this Lease against any Transfere be deemed a waiver of Landlord's right to enforce any term of this Lease against Tenant or any other person. If Tenant's obligations hereunder have been guaranteed, Landlord's consent to any Transfer shall be offective unless the guarantor also consents to such Transfere.

14.7 Deemed Consent Transfers. Notwithstanding anything to the contrary contained in this Lease, (A) an assignment or subletting of all or a portion of the Premises to an affiliate of Tenant (an entity which is Controlled by, Controls, or is under common Control with Tenant), (B) a sale of corporate shares of capital stock in

Tenant in connection with an initial public offering of Tenant's stock on a nationally-recognized stock exchange, (C) an assignment of the Lease to an entity which acquires all or substantially all of the stock or assets of Tenant, or (D) an assignment of the Lease to an entity which is the resulting entity of a merger or consolidation of Tenant during the Lease Term (any such assignee, sublessee, licensee or occupant described in items (A) through (D) of this <u>Section 14.7</u> hereinafter referred to as a "**Permitted Transferee**") provided that (i) Tenant notifies Landlord no more than ten (10) business days after the effective date of any such assignment or sublease and promptly supplies Landlord with any documents or information reasonably requested by Landlord regarding such transfere as set forth above, (ii) no assignment relating to this Lease, whether with or without Landlord's consent, shall relieve Tenant from any liability under this Lease, and, in the event of an assignment of Tenant's entire interest in this Lease, the liability of Tenant and such Transferee shall be joint and several.

14.8 <u>Allowed Shared Occupancy Agreements</u>. Notwithstanding any contrary provision of this <u>Article 14</u>, the Original Tenant and its Permitted Transferees shall have the right, without the receipt of Landlord's consent and without payment to Landlord of the Transfer Permium or counting towards Landlord's right to recapture space under <u>Section 14.4</u> above, to permit the occupancy of up to 20,000 RSF of space in the Premises, in the aggregate, to any individual(s) or entity that has an ongoing business relationship with Tenant (other than the dual occupancy of the Premises), which occupancy may include the use of a corresponding interior support area and other portions of the Premises which shall be common to Tenant and the permitted occupants, on and subject to the following conditions: (i) each individual or entity shall be of a character and reputation consistent with the quality of the Building and the Project; (ii) no individual or entity shall occup a separately demised portion of the Premises or which contains an entrance to such portion of the Premises other than the primary entrance to the Premises; and (iii) such occupancy shall not be a subterfuge by Tenant to avoid its obligations under this Lease or the restrictions on Transfers pursuant to this <u>Article 14</u>. Tenant shall provide notice to Landlord of any occupancy under this <u>Section 14.8</u> within ten (10) business days after the commencement thereof. Tenant shall promptly supply Landlord with any documents or information reasonably requested by Landlord regarding the identity of any such individuals or entities. Any occupancy permitted under this <u>Section 14.8</u> shall not be deemed a Transfer under this <u>Article 14</u>. Notwithstanding the foregoing, no such occupancy shall relieve Tenant from any liability under this Lease.

ARTICLE 15

SURRENDER OF PREMISES; OWNERSHIP AND REMOVAL OF TRADE FIXTURES

15.1 <u>Surrender of Premises</u>. No act or thing done by Landlord or any agent or employee of Landlord during the Lease Term shall be deemed to constitute an acceptance by Landlord of a surrender of the Premises such intent is specifically acknowledged in writing by Landlord. The delivery of keys to the Premises to Landlord or any agent or employee of Landlord shall not constitute a surrender of the Premises or effect a termination of this Lease, whether or not the keys are thereafter retained by Landlord, and notwithstanding such delivery Tenant shall be entitled to the return of such keys at any reasonable time upon request until this Lease shall have been properly terminated. The voluntary or other surrender of this Lease by Tenant, whether accepted by Landlord or not, or a mutual termination hereof, shall not work a merger, and at the option of Landlord shall operate as an assignment to Landlord of all subleases or subtenancies affecting the Premises or terminate any or all such sublessees or subtenancies.

15.2 <u>Removal of Tenant Property by Tenant</u>. Upon the expiration of the Lease Term, or upon any earlier termination of this Lease, Tenant shall, subject to the provisions of this <u>Article 15</u>, quit and surrender possession of each Portion of the Premises to Landlord in as good order and condition as when Tenant took possession, reasonable wear and tear and repairs due to casualty or condemnation, or which are specifically made the responsibility of Landlord hereunder excepted, and subject to Tenant's obligation to remove any Specialty Tenant Improvements to the extent required by this Lease. Tenant shall not have any obligation to repair or replace any carpeting, or to repaint any painted areas of the Premises upon such surrender. Upon such expiration or termination, in addition to Tenant's obligations under <u>Section 29.31</u>, below, except as otherwise set forth in this Lease, Tenant shall, without expense to Landlord, remove or cause to be removed from the Premises all debris and rubbish, and such items of furniture, equipment, business and trade fixtures, free-standing cabinet work, movable partitions and other articles of personal property owned by Tenant or installed or placed by Tenant at its expense in the Premises, and such similar articles of any other persons claiming under Tenant, as Landlord may, in its sole

discretion, require to be removed, and Tenant shall repair at its own expense all damage to the Premises and Building resulting from such removal.

ARTICLE 16

HOLDING OVER

If Tenant holds over in any Portion of the Premises after the expiration of the Lease Term with respect to such Portion of the Premises or earlier termination thereof, with or without the express or implied consent of Landlord, such tenancy shall be from month-to-month only, and shall not constitute a renewal hereof or an extension for any further term, and in such case Base Rent shall be payable at a monthly rate equal to one hundred fifty percent (150%) of the Base Rent applicable during the last rental period of the Lease Term under this Lease with respect to such Tranche. Such month-to-month tenancy shall be subject to every ortion ad greement contained herein. Tenant's payment of Rent as set forth in this <u>Article 16</u> shall be proportionately reduced for every Portion of the Premises vacated by Tenant during any holdover such that Tenant shall only pay Rent for those Portion(s) of the Premises actually occupied by Tenant during any holdover. Nothing contained in this <u>Article 16</u> shall be construed as consent by Landlord to any holding over by Tenant, and Landlord expressly reserves the right to require Tenant to surrender possession of the Premises to Landlord as provided in this Lease upon the expiration or other termination of this Lease. The provisions of this <u>Article 16</u> shall be clease to any other rights or remedies of Landlord provided herein or at law. If Tenant fails to surrender any Portion of the Premises within sixty (60) days after the termination or expiration of this Lease, from all loss, costs (including reasonable attorneys' fees) and liability resulting from any claims made by such succeeding tenant or occupant founded upon such failure to surrender and any lost profits to Landlord resulting therefrom. Tenant shall have the right to request that Landlord provide to Tenant a written notice setting forth Landlord's good faith estimate of the maximum amount of consequential damages (including loss of profits, loss of business days after receipt of such request, Landlord shall provide Tenan

ARTICLE 17

ESTOPPEL CERTIFICATES

Within ten (10) business days following a request in writing by Landlord, Tenant shall execute, acknowledge and deliver to Landlord an estoppel certificate (or provide written comments to any proposed certificate delivered by Landlord), which, as submitted by Landlord, shall be in the form of <u>Exhibit G</u>, attached hereto indicating therein any exceptions thereto that may exist at that time. If Tenant fails to timely execute, acknowledge and deliver such estoppel certificate which written notice must state in bold and all caps "FAILURE TO RESPOND TO THIS WRITTEN NOTICE WITHIN FIVE (5) BUSINESS DAYS AFTER RECEIPT HEREOF SHALL CONSTITUTE ACCEPTANCE OF AN ESTOPPEL CERTIFICATE". If Tenant fails to execute and deliver such estoppel certificate delivered by Landlord's second written request therefor, such failure shall constitute an acceptance of the Premises and an acknowledgment by Tenant that statements included in the estoppel certificate received from Landlord's second written comments to any prospective mortgage or purchaser of all or any portion of the Project. If Tenant provides written comments to any scopel certificate (or provide written comments to any scopel certificate (or provide written comments to any such certificate delivered by Landlord) within a five (5) business day period following the receipt of Landlord's second written request therefor, such failure shall constitute an acceptance of the Premises and an acknowledgment by Tenant that statements included in the estoppel certificate received from Landlord, then Tenant shall have five (5) business days following receipt of a revised estoppel certificate (or provide written comments to any such certificate may be relied upon by any prospective mortgage or purchaser of all or any portion of the Project. If Tenant provides written comments to any schowledge and deliver to Landlord such revised estoppel certificate (or provide written comments to any such revised estoppel certificate (or provide written comments to any such certificate delivered by La

to timely execute, acknowledge and deliver such estoppel certificate (or provide written comments to any proposed certificate delivered by Landlord).

Landlord shall, within ten (10) business days after receipt of a written request from Tenant, execute and deliver (or provide written comments to any proposed certificate delivered by Tenant) to Tenant a statement certifying: (a) the date of commencement and expiration of each Portion of the Premises; (b) the fact that this Lease is unmodified and in full force and effect (or, if there have been modifications to this Lease, that this lease is in full force and effect, as modified, and stating the date and nature of such modifications); (c) the date to which the rent and other sums payable under this Lease been paid; (d) the fact that there are no current defaults under this Lease by either Landlord or Tenant except as specified in Landlord's statement; and (e) such other factual matters as may be reasonably requested by Tenant. Landlord irrevocably agrees that if Landlord fails to execute and deliver such certificate which written notice must state in bold and all caps "FAILURE TO RESPOND TO THIS WRITTEN NOTICE WITHIN FIVE (5) BUSINESS DAYS AFTER RECEIPT HEREOF SHALL CONSTITUTE AN EVENT OF DEFAULT." If Landlord fails to execute and correct, without exception. Landlord agrees that any statement delivered pursuant to this paragraph may be relied upon by any assignee, lender, subtenant or investor of Tenant. If Landlord provides written comments to any estoppel certificate for the statements included in the estoppel certificate estoppel certificate (or provide written comments to any stoppel certificate (or provide written comments to any estoppel certificate to execute, acknowledge and deliver such revised estoppel certificate (or provide written comments to any estoppel certificate to execute, acknowledge and deliver such revised estoppel certificate (or provide written comments to any stoppel certificate (or provide written comments to any stoppel certificate (or provide written comments to any stoppel certificate (or provide written comments to any proposed estificate to execute, acknowledge and deliver such r

At any time during the Lease Term, Landlord may require Tenant to provide Landlord with a current financial statement and financial statements of the two (2) years prior to the current financial statement year. Such statements shall be prepared in accordance with generally accepted accounting principles and, if such is the normal practice of Tenant, shall be audited by an independent certified public accountant. Landlord agrees that Tenant's financial statements are deemed to be Tenant's confidential information. Landlord hereby agrees to maintain Tenant's financial statements as proprietary and confidential and will take reasonable measures to avoid disclosure and unauthorized use of the financial statements (including, without limitation, measures at least as stringent as it takes to protect its own confidential information of a similar nature). Landlord agrees not to disclose Tenant's financial statements to any third party other than as-needed to any lender, prospective lender, or purchaser and to Landlord's attorneys, accountants, investment advisors and similar business advisors, provided that in the case of each such disclosure: (a) such disclosures are for bona fide business purposes related to the Project; (b) all such third parties have signed a commercially reasonable non-disclosure agreement prior to receiving any of Tenant's financial statements; and (c) in case of any breach by any such third party of the applicable non-disclosure agreement, Landlord shall not be liable to Tenant for such breach, but Landlord shall reasonable gonerate with Tenant, at Tenant's sole cost and expense, to enforce the terms of such non-disclosure agreement, including, without limitation, assigning Landlord's rights to enforce such non-disclosure agreement, including without function apply to the extent the financial statements: (i) are required to be disclosed by applicable laws after giving reasonable notice to Tenant to allow Tenant's financial statements and (v) in a set of any breach by any such third party of the

controls Tenant or with entities which are otherwise Affiliates of Tenant), then Tenant's obligation to provide Landlord with a copy of its most recent current financial statement shall be deemed satisfied.

ARTICLE 18

SUBORDINATION

Concurrently with the execution and delivery of this Lease, Landlord has delivered to Tenant a fully executed subordination, non-disturbance and attornment agreement (the "Current SNDA") from Landlord's Current Lender (as defined below). Subject to the terms of the Current SNDA, or any Future SNDA Agreement (as defined below), this Lease shall be subject and subordinate to all present and future ground or underlying leases of the Building or Project and to the lien of any mortgage, trust deed or other security document now or hereafter in force against the Building or Project or any part thereof, if any, and to all renewals, extensions, modifications, consolidations and replacements thereof (collectively, "Security Documents"), and to all advances made or hereafter to be made upon the security of such mortgages or trust deeds, unless the holders of such mortgages, trust deeds or other encumbrances, or the lessors under such ground lease or underlying leases, require in writing that this Lease be superior thereto. Tenant's agreement to subordinate this Lease to any Security Document shall be subject to Tenant's receipt of a Future SNDA Agreement (as hereinafter defined). A "Future SNDA Agreement" shall mean a commercially reasonable agreement between Landlord, Tenant and the holder of such Security Document reasonably acceptable to Tenant, which provides, among other things, that, so long as Tenant is paying the rent due under the Lease and is not otherwise in default under the Lease beyond any applicable cure period, its right to possession and the other terms of the Lease shall remain in full force and effect. Any Future SNDA Agreement shall expressly provide that (i) upon any party succeeding to the interest of Landlord under this Lease, such party shall be liable for the performance of the "Landlord Work," and any "Seismic Work" required to be performed by Landlord under this Lease, (ii) the provisions of this Lease regarding any casualty or condemnation shall control notwithstanding any conflicting provisions in the Security Documents, and (iii) either (A) that upon any party succeeding to the interest of Landlord under this Lease, such party shall be liable for the payment of any unpaid portion of the Tenant Improvement Allowance, or (B) that Tenant's right of set-off pursuant to Section 19.5.3 below for failure to pay the Tenant Improvement Allowance or Section 29.23 below for failure to pay brokerage commissions shall remain in effect and upon any party succeeding to the interest of Landlord under this Lease, any accumulated amounts which Tenant may have to set-off against rent and other amounts due under this Lease shall continue in full force and effect and be binding against such party and Tenant shall have the continued right to set-off such amounts against rent and other amounts due under this Lease. Notwithstanding the foregoing, in lieu of the Future SNDA Agreement providing the express recognitions set forth in items (i) and (iii), above, Landlord may deliver to Tenant either (1) a guaranty binding upon a "Qualified Guarantor," as defined below, guaranteeing the performance and/or payment of such obligations, or (2) an unconditional, irrevocable standby letter of credit in the form and from an issuing bank meeting the requirements of the L-C to be delivered by Tenant to Landlord pursuant to Article 21 of this Lease, and all of the other provisions of Article 21 below (other than the provisions regarding reductions in the L-C Amount) shall apply to such letter of credit (with such changes as necessary to reflect that the letter of credit is being deliver by Landlord to Tenant), in an amount equal to the sum of the then undisbursed amount, if any, of the reasonably estimated cost of any then remaining Landlord Work and Seismic Work, any remaining Improvement Allowance and any remaining unpaid commissions owed to Tenant's Broker, as protection for Tenant's ability to receive the benefits of such work and the payment of the Improvement Allowance pursuant to the terms of the Work Letter and for payment of commissions owed to Tenant's broker, and if the party succeeding to the interest of Landlord under this Lease fails to disburse the Tenant Improvement Allowance or to pay the commissions owed to Tenant's Broker then Tenant may enforce such guaranty or draw on the letter of credit, as applicable, in the amount and to the extent Tenant would have been able to off-set such amount pursuant to the terms of Sections 19.5.3 or 29.23 below, respectively. Landlord shall additionally use commercially reasonable efforts to cause any Future SNDA to contain an agreement that Tenant's Right of First Offer to Purchase shall survive any foreclosure, deed-in-lieu of foreclosure of other exercise of remedies under any Security Documents (but the failure of such Future SNDA to contain such agreement shall not be a failure of the requirement to provide a Future SNDA). As used herein, a "Qualified Guarantor" shall mean a person or entity that has a total "net worth," as determined pursuant to generally accepted accounting principles (GAAP), in excess of \$650 Million, and "shareholder surplus," as determined by GAAP, in excess of \$250 Million. Tenant covenants and agrees in the event any proceedings are brought for the foreclosure of any such mortgage or deed in lieu thereof (or if any ground lease is terminated), to attorn, to the lienholder or purchaser or any successors thereto upon any such foreclosure sale

or deed in lieu thereof (or to the ground lessor), if so requested to do so by such purchaser or lienholder or ground lessor, and to recognize such purchaser or lienholder or ground lessor as the lessor under this Lease, provided such lienholder or purchaser or ground lessor shall agree to accept this Lease and not disturb Tenant's occupancy, so long as Tenant timely pays the rent and observes and performs the terms, covenants and conditions of this Lease to be observed and performed by Tenant. Landlord's interest herein may be assigned as security at any time to any lienholder. Tenant shall, within ten (10) business days of request by Landlord, provide written comments to any proposed Future SNDA Agreement delivered by Landlord, Landlord's nuclease to be observed and performed by Tenant. Landlord's nuclease to be asserved and performed by the terms, covenants and conditions of this Lease to be observed and performed by Tenant. Landlord's interest herein may be assigned as security at any time to any lienholder. Tenant shall, within ten (10) business days of request by Landlord, may provide to Tenant a second written request with respect to such Future SNDA Agreement (or provide written notice must state in bold and all caps "FAILURE TO RESPOND TO THIS WRITTEN NOTICE WITHIN FIVE (5) BUSINESS DAYS AFTER RECEIPT HEREOF SHALL CONSTITUTE AN EVENT OF DEFAULT". If Tenant fails to execute such Future SNDA Agreement (or provide written comments to any proposed future SNDA Agreement delivered by Landlord) with such five (5) business day period shall be deemed to be a material event of default hereunder. Tenant waives the provisions of any current or future Law which may give or purport to give Tenant any right or election to terminate or otherwise adversely affect this Lease and the obligations of the Tenant hereunder in the event of any process future SNDA Agreement (cor provide written comments to any such revised Future SNDA Agreement (or provide written comments to any such revised Future SNDA Agreement to execute,

ARTICLE 19

DEFAULTS; REMEDIES

19.1 Events of Default. The occurrence of any of the following shall constitute a default of this Lease by Tenant:

19.1.1 Any failure by Tenant to pay any Rent or any other charge required to be paid under this Lease, or any part thereof, when due unless such failure is cured within five (5) business days after notice; or

19.1.2 Except where a specific time period is otherwise set forth for Tenant's performance in this Lease, in which event the failure to perform by Tenant within such time period shall be a default by Tenant under this Section 19.1.2, any failure by Tenant to observe or perform any other provision, covenant or condition of this Lease to be observed or performed by Tenant where such failure continues for thirty (30) days after written notice thereof from Landlord to Tenant; provided that if the nature of such default is such that the same cannot reasonably be cured within a thirty (30) day period, Tenant shall not be deemed to be in default if it diligently commences such cure within such period and thereafter diligently proceeds to rectify and cure such default; or

19.1.3 Abandonment (as defined in California Civil Code Section 1951.3) of the Premises by Tenant.

The notice periods provided herein are in lieu of, and not in addition to, any notice periods provided by Law.

19.2 <u>Remedies Upon Default</u>. Upon the occurrence of any event of default by Tenant, Landlord shall have, in addition to any other remedies available to Landlord at law or in equity (all of which remedies shall be distinct, separate and cumulative), the option to pursue any one or more of the following remedies, each and all of which shall be cumulative and nonexclusive, without any notice or demand whatsoever.

19.2.1 Terminate this Lease, in which event Tenant shall immediately surrender the Premises to Landlord, and if Tenant fails to do so, Landlord may, without prejudice to any other remedy which it may have for possession or arrearages in rent, enter upon and take possession of the Premises and expel or remove Tenant and any other person who may be occupying the Premises or any part thereof, without being liable for prosecution or any claim or damages therefor; and Landlord may recover from Tenant the following:

19.2.1.1 The worth at the time of award of any unpaid rent which has been earned at the time of such termination; plus

19.2.1.2 The worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that Tenant proves could have been reasonably avoided; plus

19.2.1.3 The worth at the time of award of the amount by which the unpaid rent for the balance of the Lease Term after the time of award exceeds the amount of such rental loss that Tenant proves could have been reasonably avoided; plus

19.2.1.4 Any other amount necessary to compensate Landlord for all the detriment proximately caused by Tenant's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom; and

19.2.1.5 At Landlord's election, such other amounts in addition to or in lieu of the foregoing as may be permitted from time to time by applicable Law.

The term "rent" as used in this Section 19.2 shall be deemed to be and to mean all sums of every nature required to be paid by Tenant pursuant to the terms of this Lease, whether to Landlord or to others. As used in Section 19.2.1.1 and 19.2.1.2, above, the "worth at the time of award" shall be computed by allowing interest at the Interest Rate. As used in Section 19.2.1.3 above, the "worth at the time of award" shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of San Francisco at the time of award plus one percent (1%). Notwithstanding the foregoing, Landlord's damages in the event of any event of default by Tenant shall not be increased beyond the amounts calculated as provided above based on any construction costs paid or incurred by Landlord, any brokerage commissions or allowances paid by Landlord to Tenant (including the Tenant Improvement Allowance) or any free rent provided to Tenant (or any rent attributable to any period between the satisfaction of the Delivery Condition for any Portion of the Premises).

19.2.2 Landlord shall have the remedy described in California Civil Code Section 1951.4 (lessor may continue lease in effect after lessee's breach and abandonment and recover rent as it becomes due, if lessee has the right to sublet or assign, subject only to reasonable limitations). Accordingly, if Landlord does not elect to terminate this Lease on account of any default by Tenant, Landlord may, from time to time, without terminating this Lease, enforce all of its rights and remedies under this Lease, including the right to recover all rent as it becomes due.

19.2.3 Landlord shall at all times have the rights and remedies (which shall be cumulative with each other and cumulative and in addition to those rights and remedies available under Sections 19.2.1 and 19.2.2, above, or any law or other provision of this Lease), without prior demand or notice except as required by applicable Law, to seek any declaratory, injunctive or other equitable relief, and specifically enforce this Lease, or restrain or enjoin a violation or breach of any provision hereof.

19.3 Subleases of Tenant. Whether or not Landlord elects to terminate this Lease on account of any default by Tenant, as set forth in this <u>Article 19</u>, Landlord shall have the right to terminate any and all subleases, licenses, concessions or other consensual arrangements for possession entered into by Tenant and affecting the Premises or may, in Landlord's sole discretion, succeed to Tenant's interest in such subleases, licenses, concessions or arrangements. In the event of Landlord's election to succeed to Tenant's interest in any such subleases, licenses, concessions or arrangements, Tenant shall, as of the date of notice by Landlord of such election, have no further right to or interest in the rent or other consideration receivable thereunder.

19.4 Efforts to Relet. No re-entry or repossession, repairs, maintenance, changes, alterations and additions, reletting, appointment of a receiver to protect Landlord's interests hereunder, or any other action or omission by Landlord shall be construed as an election by Landlord to terminate this Lease or Tenant's right to possession, or to accept a surrender of the Premises, nor shall same operate to release Tenant in whole or in part

from any of Tenant's obligations hereunder, unless express written notice of such intention is sent by Landlord to Tenant. Tenant hereby irrevocably waives any right otherwise available under any Law to redeem or reinstate this Lease.

19.5 Landlord Default

19.5.1 Notice and Cure. Landlord shall be in default hereunder if Landlord shall fail to pay any sums payable by Landlord hereunder within ten (10) business days after the date such sum is due. Notwithstanding anything to the contrary set forth in this Lease, Landlord shall be in default in the performance of any other obligation required to be performed by Landlord pursuant to this Lease if Landlord fails to perform such obligation within thirty (30) days after the receipt of notice from Tenant specifying in detail Landlord's failure to perform; provided, however, if the nature of Landlord's obligation is such that more than thirty (30) days are required for its performance, then Landlord shall not be in default under this Lease if it shall commence such performance within such thirty (30) day period and thereafter diligently pursues the same to completion. Upon any such default by Landlord under this Lease, Tenant may, except as otherwise specifically provided in this Lease to the contrary, exercise any of rights and remedies set forth herein or otherwise provided at law or in equity.

19.5.2 Tenant Right to Self Help. Notwithstanding anything to the contrary contained herein, if an event or circumstance which requires the action of Landlord with respect to repair, maintenance or provision of services to the Building or any portion thereof is an Emergency Situation (as defined below) or will have a material, adverse impact on Tenant's ability to conduct its business in the Premises (an "Adverse Condition"), including, for example, any failure to provide (or cause to be provided) electricity, HVAC, chilled water or elevator access to the Premises, and Landlord fails to commence corrective action within a reasonable period of time, given the circumstances, after the receipt of such notice, but in any event not later than ten (10) business days after receipt of such notice, then Tenant may proceed to take the required action upon delivery of an additional five (5) business days ontice to Landlord specifying that Tenant is taking such required action (provided, however, that the initial ten (10) business day notice shall not be required in the event of an "Emergency Situation" as that term is defined herein, but rather the ten (10) business day notice shall be replaced by a two (2) business day period, and no second notice shall be required). For purposes hereof, the term "Emergency Situation" shall mean a situation which poses an imminent threat: (x) to the physical well-being of persons at the Building, (y) of material damage to Tenant's personal property in the Premises. If Tenant exercises its right to remedy an Adverse Condition, as provided above, then Landlord shall reimburse the actual reasonable cost thereof together with interest at the Interest Rate within thirty (30) days following Tenant's delivery of invoice together with reasonable supporting evidence. If Tenant undertakes any action pursuant to this paragraph, Tenant shall (a) proceed in accordance with all applicable Laws; (b) retain to effect such actions only such reputable contractors and suppliers as are duly licensed in the

19.5.3 Landlord Reimbursement. Subject to any express limitations set forth herein to the contrary, the remedies provided for in this Lease are in addition to all other remedies available to Tenant at law or in equity by statute or otherwise. Without limiting the generality of the foregoing, if Landlord fails to pay or reimburse Tenant when due any amount owed to Tenant under this Lease, including, without limitation: (i) Tenant fails to receive any Improvement or other amounts owed to Tenant in connection with construction of the Tenant Improvements after submission of all required documents and information as required by this Lease; (ii) Landlord fails to reimburse Tenant in connection with the exercise of Tenant's rights pursuant to Sections 19.5.1 and 19.5.2 above; or (iii) monetary damages awarded to Tenant in any arbitration proceeding or by a court in any legal proceeding, then Tenant shall have the right, in addition to all other remedies available to Tenant at law or in equity by statute or otherwise, to either off-set against Rent due under this Lease or otherwise abate payments of Base Rent, Operating Expenses or other Rent an amount equal to (a) the applicable amounts owed to Tenant plus (b) interest on the amounts owed to Tenant from the date incurred until such off-set off-set against Rent due under this Lease into (10) business days prior to exercising its right of off-set (the "Offset Notice"). If Landlord does not deliver a detailed written objection to Tenant within ten (10) business days after receipt of such Offset Notice, then Tenant shall be entitled to deduct from Rent payable by Tenant under this Lease,

the amount set forth in such Offset Notice. If, however, Landlord delivers to Tenant, within ten (10) business days after receipt of the Offset Notice, a good faith written objection to the payment of such amounts, setting forth with reasonable particularity Landlord's reasons for its claim that such action did not have to be taken by Landlord pursuant to the terms of this Lease or that the charges are excessive (in which case Landlord shall pay the amount it contends would not have been excessive), then Tenant shall not then be entitled to such deduction from Rent; provided, however, that Tenant may proceed to claim a default by Landlord, and may submit such claim to arbitration under the commercial arbitration rules of JAMS (and Landlord and Tenant) to arbitration of such matter by JAMS and the determination of such arbitration shall be final and binding upon both Landlord and Tenant). If Tenant prevails with regard to such claim, the amount of the award (which shall include interest Rate from the time of each expenditure by Tenant until the date Tenant neceives such amount by payment or offset and attorneys' fees and related costs) may be deducted by Tenant from the Rent next due and owing under this Lease.

ARTICLE 20

COVENANT OF QUIET ENJOYMENT

Landlord covenants that Tenant, on paying the Rent, charges for services and other payments herein reserved and on keeping, observing and performing all the other terms, covenants, conditions, provisions and agreements herein contained on the part of Tenant to be kept, observed and performed, shall, during the Lease Term, peaceably and quietly have, hold and enjoy the Premises subject to the terms, covenants, conditions, provisions and agreements hereof without interference by any persons lawfully claiming by or through Landlord. The foregoing covenant is in lieu of any other covenant express or implied.

ARTICLE 21

LETTER OF CREDIT

21.1 Delivery of Letter of Credit. Within ten (10) business days following the Effective Date, Tenant shall deliver to Landlord, as protection for the full and faithful performance by Tenant of all of its obligations under this Lease and for all losses and damages Landlord may suffer as a result of any breach or default by Tenant under this Lease, an unconditional, irrevocable, standby letter of credit (the "L-C") in the amount set forth in Section 8 of the Summary (the "L-C Amount"), substantially in the form attached hereto as Exhibit H, payable in the City of San Francisco, California, running in favor of Landlord, drawn on a bank (the "Bank") reasonably approved by Landlord and at a minimum having a long term issuer credit rating from Standard and Poor's Professional Rating Service (the "Credit Rating Threshold"), and otherwise conforming in all respects to the requirements of this Article 21, including, without limitation, all of the requirements of <u>Section 21.2</u> below, all as set forth more particularly hereinbelow. As of the Effective Date, Landlord approves any of the following to be the Bank: (i) Wells Fargo, (ii) Bank of America, (iii) Silicon Valley Bank or (iv) Deutsche Bank. Tenant shall pay all expenses, points and/or fees incurred by Tenant in obtaining and maintaining the L-C. In the event of an assignment by Tenant of its interest in the Lease (and irrespective of whether Landlord's consent is required for such assignment), the acceptance of any replacement or substitute letter of credit by Landlord from the assignee shall be subject to Landlord's prior written approval, in Landlord's reasonable supporting evidence. Provided Tenant has not previously been in material default under this Lease (after receipt of all required notices from Landlord and the expiration of all applicable cure period set forth in this Lease), then the L-C Amount may be reduced beginning as of the expiration of the tenth (10th) Lease Year to equal Redacted Landlord agrees to authorize such reduction in writing to the is

21.2 In General. The L-C shall be "callable" at sight, permit partial draws and multiple presentations and drawings, and be otherwise subject to the Uniform Customs and Practices for Documentary Credits (1993-Rev), International Chamber of Commerce Publication #500, or the International Standby Practices-ISP 98, International Chamber of Commerce Publication #590. Tenant further covenants and warrants as follows:

21.2.1 Landlord Right to Transfer. The L-C shall provide that Landlord, its successors and assigns, may, at any time and without notice to Tenant and without first obtaining Tenant's consent thereto, transfer (one or more times) all or any portion of its interest in and to the L-C to another party, person or entity, in connection with the assignment by Landlord of its rights and interests in and to this Lease or separate from this Lease if such assignment is to Landlord's lender. In the event of a transfer of Landlord's interest in the Building, Landlord shall transfer the L-C, in whole or in part, to the transferee and thereupon Landlord shall, without any further agreement between the parties, be released by Tenant from all liability therefor from and after such date, and it is agreed that the provisions hereof shall apply to every transfer or assignment of the whole or any portion of said L-C to a new landlord. In connection with any such transfer of the L-C by Landlord, Tenant shall, at Tenant's transfer and processing fees in connection therewith.

21.2.2 <u>No Assignment by Tenant</u>. Tenant shall neither assign nor encumber the L-C or any part thereof. Neither Landlord nor its successors or assigns will be bound by any assignment, encumbrance, attempted assignment or attempted encumbrance by Tenant in violation of this <u>Article 21</u>.

21.2.3 <u>Replenishment</u>. If, as a result of any drawing by Landlord on the L-C pursuant to its rights set forth in <u>Section 21.3</u> below, the amount of the L-C shall be less than the L-C Amount, Tenant shall, within ten (10) business days thereafter, provide Landlord with (i) an amendment to the L-C restoring such L-C to the L-C Amount or (ii) additional L-Cs in an amount equal to the deficiency, which additional L-Cs shall comply with all of the provisions of this <u>Article 21</u>, and if Tenant fails to comply with the foregoing, notwithstanding anything to the contrary contained in <u>Section 19.1</u> above, the same shall constitute an incurable default by Tenant under this Lease (without the need for any additional notice and/or cure period).

21.2.4 <u>Renewal; Replacement</u>. If the L-C expires earlier than the date (the "LC Expiration Date") that is sixty (60) days after the expiration of the Lease Term, Tenant shall deliver a new L-C or certificate of renewal or extension to Landlord at least sixty (60) days prior to the expiration of the L-C then held by Landlord, without any action whatsoever on the part of Landlord, which new L-C shall be irrevocable and automatically renewable through the LC Expiration Date upon the same terms as the expiring L-C or such other terms as may be acceptable to Landlord in its reasonable discretion. In furtherance of the foregoing, Landlord and Tenant agree that the L-C shall contain a so-called "evergreen provision," whereby the L-C will automatically be renewed unless at least sixty (60) days 'prior written notice of non-renewal is provided by the issuer to Landlord; provided, however, that the final expiration date identified in the L-C, beyond which the L-C shall not automatically renew, shall not be earlier than the LC Expiration Date.

21.2.5 Bank's Financial Condition. If, at any time during the Lease Term, the Bank's long term credit rating is reduced below the Credit Rating Threshold ("Bank Credit Threat"), then Landlord shall have the right to require that Tenant obtain from a different issuer a substitute L-C that complies in all respects with the requirements of this <u>Article 21</u>, and Tenant's failure to obtain such substitute L-C within thirty (30) days following Landlord's written demand therefor (with no other notice or cure or grace period being applicable thereto, notwithstanding anything in this Lease to the contrary) shall entitle Landlord, or Landlord's then managing agent, to immediately draw upon the then existing L-C in whole or in part, without notice to Tenant, as more specifically described in <u>Section 21.3</u> below.

21.3 <u>Application of Letter of Credit</u>. Tenant hereby acknowledges and agrees that Landlord is entering into this Lease in material reliance upon the ability of Landlord to draw upon the L-C as protection for the full and faithful performance by Tenant of all of its obligations under this Lease and for all losses and damages Landlord may suffer as a result of any breach or default by Tenant under this Lease. Landlord, or its then managing agent, shall have the right to draw down an amount up to the face amount of the L-C if any of the following shall have occurred or be applicable: (A) such amount is due to Landlord under the terms and conditions of this Lease, or (B) Tenant has field a voluntary petition under the U. S. Bankruptey Code or any state bankruptey code (collectively, "**Bankruptey Code**"), or (C) an involuntary petition has been filed against Tenant under the Bankruptey Code, or (D) the Bank has notified Landlord that the L-C will not be renewed or extended through the LC Expiration Date, or (E) a Receivership (as such term is defined in <u>Section 21.6.1</u> below) has occurred and Tenant has failed to comply with the requirements of <u>Section 21.6</u> below, as applicable. If Tenant shall breach any

provision of this Lease or otherwise be in default hereunder or if any of the foregoing events identified in Sections 21.3(B) through (E) shall have occurred, Landlord may, but without obligation to do so, and without notice to Tenant, draw upon the L-C, in part or in whole, and the proceeds may be applied by Landlord (i) to cure any breach or default of Tenant, (ii) against any Rent payable by Tenant under this Lease that is not paid when due and/or (iii) to pay for all losses and damages allowable under Law as a result of any breach or default by Tenant under this Lease. The use, application or retention of the L-C, or any portion thereof, by Landlord shall not prevent Landlord from exercising any other right or remedy provided by this Lease or by any applicable Law, it being intended that Landlord shall not first be required to proceed against the L-C, and shall not operate as a limitation on any recovery to which Landlord may otherwise be entitled. Tenant agrees not to interfere in any way with payment to Landlord of the proceeds of the L-C, regardless of whether any dispute exists between Tenant and Landlord as to Landlord's right to draw upon the L-C. No condition or term of this Lease shall be deemed to render the L-C conditional to justify the issuer of the L-C in failing to honor a drawing upon such L-C in a timely manner. Tenant agrees and acknowledges that (1) the L-C constitutes a separate and independent contract between Landlord and the Bank, (2) Tenant is not a third party beneficiary of such contract, and (3) in the event Tenant becomes a debtor under any chapter of the Bankruptcy Code, neither Tenant, any trustee, nor Tenant's bankruptcy estate shall have any right to restrict or limit Landlord's claim and/or rights to the L-C and/or the proceeds thereof by application of Section 502(b)(6) of the U. S. Bankruptcy Code or otherwise.

21.4 Security Deposit Laws. Landlord and Tenant acknowledge and agree that in no event or circumstance shall the L-C or any renewal thereof or any proceeds thereof, or any Interim Cash Deposit, be subject to the terms of California Civil Code Section 1950.7 or any other Laws applicable to security deposits in the commercial context which conflict with the express terms and conditions of this Lease.

21.5 Proceeds of Draw. In the event Landlord draws down on the L-C pursuant to Section 21.3(D) or (E) above, the proceeds of the L-C may be held by Landlord and applied by Landlord against any Rent payable by Tenant under this Lease that is not paid when due and/or to pay for all losses and damages that Landlord has suffered as a result of any breach or default by Tenant under this Lease. Any unused proceeds shall constitute the property of Landlord and need not be segregated from Landlord's other assets. Landlord agrees that the amount of any proceeds of the L-C received by Landlord, and not (a) applied against any Rent payable by Tenant under this Lease that was not paid when due or (b) used to pay for any losses and/or damages suffered by Landlord as a result of any breach or default by Tenant under this Lease (the "Unused L-C Proceeds"), shall be paid by Landlord to Tenant (x) upon receipt by Landlord of a replacement L-C in the full L-C Amount, which replacement L-C shall comply in all respects with the requirements of this <u>Article 21</u>, or (y) within five (5) business days after the LC Expiration Date; provided, however, that if prior to the LC Expiration Date a voluntary petition is filed by Tenant's creditors, under the Bankruptey Code, then Landlord shall not be obligated to make such payment in the amount of the Unused L-C Proceeds until either all preference issues relating to payments under this Lease have been resolved in such bankruptey or reorganization case or such bankruptey or reorganization case has been dismissed.

21.6 Bank Placed Into Receivership.

21.6.1 <u>Bank Placed Into Receivership</u>. In the event the Bank is placed into receivership or conservatorship (any such event, a "Receivership") by the Federal Deposit Insurance Corporation or any successor or similar entity (the "FDIC"), then, effective as of the date such Receivership occurs, the L-C shall be deemed to not meet the requirements of this <u>Article 21</u>, and, within ten (10) business days following Landlord's notice to Tenant of such Receivership (the "LC Replacement Notice"), Tenant shall (i) replace the L-C with a substitute L-C from a different issuer that complies in all respects with the requirements of this <u>Article 21</u> or (ii) in the event Tenant demonstrates to Landlord that Tenant is reasonably unable to obtain a substitute L-C from a different issuer, that complies in all respects with the requirements of this <u>Article 21</u> within the foregoing ten (10) business day period, deposit with Landlord cash in the L-C Amount (the "Interim Cash Deposit"); provided, however, that, in the case of the foregoing sub-clause (ii), Tenant shall continue to diligently pursue the replacement of the L-C from a different issuer that complies in all respects with the requirements of this <u>Article 21</u>, and upon Landlord's receipt and acceptance of such replacement L-C, Landlord shall return to Tenant the Interim Cash Deposit, with no obligation on the part of Landlord to pay any interest thereon. If Tenant fails to comply in any respect with the requirements of this <u>Article 21</u>, then, notwithstanding anything in this Lease to the contrary.

Landlord shall have the right to declare Tenant in default of this Lease for which there shall be no notice or grace or cure periods being applicable thereto other than the aforesaid ten (10) business day period), in which event, Landlord shall have the right to pursue any and all remedies available to it under this Lease and at law, including, without limitation, treating any Receivership as a Bank Credit Threat and exercising Landlord's remedies under <u>Section 21.2.5</u> above, to the extent possible pursuant to then existing FDIC policy. Tenant shall be responsible for the payment of any and all costs incurred with the review of any replacement L-C (including without limitation Landlord's reasonable attorneys' fees), which replacement is required pursuant to this Section or is otherwise requested by Tenant within thirty (30) days after receipt of invoice together with reasonable supporting evidence.

21.6.2 Interim Cash Deposit. During any period that Landlord remains in possession of the Interim Cash Deposit (any such period, a "Deposit Period"), it is understood by the parties that such Interim Cash Deposit shall be held by Landlord as security for the full and faithful performance of Tenant's covenants and obligations under this Lease. The Interim Cash Deposit shall not constitute an advance of any Rent, an advance payment of any other kind, nor a measure of Landlord's damages in case of Tenant's default. If, during any such Deposit Period, Tenant defaults with respect to any provisions of the Lease, as amended, including, but not limited to, the provisions relating to the payment of Rent, the removal of property and the repair of resultant damage, then Landlord may but shall not be required to, from time to time, without notice to Tenant and without waiving any other remedy available to Landlord, use the Interim Cash Deposit, or any portion of it, to the extent necessary to cure or remedy such default or failure or to compensate Landlord for all damages sustained by Landlord resulting from Tenant's default or failure to comply fully and timely with its obligations pursuant to the Lease, as amended. Tenant shall immediately pay to Landlord on demand any amounts oa applied in order to restore the Interim Cash Deposit to its original amount, and Tenant's failure to immediately do so shall constitute a default under the Lease, as amended. In the event Landlord is in possession of the Interim Cash Deposit at the expiration or termination of the Lease, as amended, its is in compliance with the covenants and obligations with respect to the Interim Cash Deposit are those of a debtor and not a trustee. Landlord shall neture to Tenant the Interim Cash Deposit, less any amounts deducted by Landlord's obligations with respect to the Interim Cash Deposit in whole or in part, to the transfere on any time be entitled to interest on the Interim Cash Deposit. In the event of a transfer of Landlord's interest in the Buil

ARTICLE 22

INTENTIONALLY OMITTED

ARTICLE 23

<u>SIGNS</u>

23.1 Building Signage Rights

23.1.1 Interior Main Elevator Lobby Signage. So long as Tenant occupies at least 100,000 rentable square feet of the Premises, Tenant shall have the non-exclusive right to one (1) sign on the west wall of the interior north elevator lobby of the Building, and one (1) sign on the east wall of the interior south elevator lobby of the Building (collectively, the "Lobby Signage"), as set forth on <u>Schedule 23</u> attached hereto. So long as Tenant occupies at least 200,000 rentable square feet of the Premises, such Lobby Signage shall be exclusive to Tenant (provided that Landlord shall continue to have the right to install reasonable signs for the ground floor retail tenants of the Building). At such time as Tenant occupies at least 100,000 rentable square feet but less than 200,000

rentable square feet of the Premises, if requested by Landlord in order to provide signage to another tenant of the Building, Tenant shall remove one (1) of Tenant's signs comprising the Lobby Signage, provided that Tenant shall have no obligation to remove such sign if at such time Tenant only has one (1) sign comprising the Lobby Signage.

23.1.2 <u>Top of Building Signage</u>. So long as Tenant occupies at least 150,000 rentable square feet of the Premises, Tenant shall have the non-exclusive right to install one (1) sign at the top of the exterior of the Building (the "Building Top Signage"). So long as Tenant occupies at least 200,000 rentable square feet of the Premises, such Building Top Signage shall be exclusive to Tenant.

23.1.3 <u>Exterior Building Signage</u>. So long as Tenant continues to occupy at least 150,000 rentable square feet of the Premises, Tenant shall have the non-exclusive right to install two (2) signs on the exterior of the Building (the Building Top Signage (if any) shall count as one (1) of such signs on the exterior of the Building)) (the "Exterior Building Signage"). So long as Tenant continues to occupy at least 200,000 rentable square feet of the Premises, such Exterior Building Signage shall be exclusive to Tenant.

23.1.4 <u>Tenant's Signage</u>. The Lobby Signage, Building Top Signage, Exterior Building Signage, and "Ground Floor Premises Signage," as defined below, are referred to collectively herein as the "**Tenant Signage**". Tenant acknowledges that the Tenant Signage is subject to all applicable Laws and the REA, including, but not limited to, all requirements of the City and County of San Francisco, and the Landmarks Historical Preservation Committee (collectively, the "**Signage Requirements**"). Landlord agrees to use commercially reasonable efforts, at no cost to Landlord, to assist and support Tenant in obtaining any such required approvals. However, Tenant's failure to be all overally approveds to install all or any portion of the Tenant Signage shall not be a default by Landlord under this Lease, or entitle Tenant to any compensation or damages whatsoever. Landlord hereby approves Tenant's name and logo and the graphics, content, materials, color, design, lettering, lighting, illumination, and specifications of Tenant's Signage not set forth on **Exhibit M** attached hereto should Tenant desire to use such name, logo and specifications set forth therein. Landlord shall approve any graphics, content, materials, color, design, lettering, lighting, illumination, and specifications of Tenant's Signage not set forth on **Exhibit M** attached hereto, such approval not to be unreasonably withheld, conditioned or delayed. Tenant acknowledges that specific dimensions and other specifications of the Lobby Signage remain subject to Landlord's review and approval as provided in <u>Section 23.14</u>, below. Tenant shall keep the Tenant Signage in good condition and repair at all times, and at Tenant's sole cost and expense. Upon the expiration or earlier termination of this Lease (or upon any earlier termination of Tenant's rights hereunder), Tenant shall be responsible, at Tenant's sole cost and expense, for the removal of the Tenant Signage and the repair of any damage resulting therefrom, including, without limitation, repairing and/or

23.2 **Full Floors**. If any Portion of the Premises comprise an entire floor of the Building, then Tenant, at its sole cost and expense (subject to application of the Tenant Improvement Allowance), may install identification signage anywhere in such floor of the Premises including in the elevator lobby of such floor of the Premises, provided that such signs must not be visible from the exterior of the Building.

23.3 Multi-Tenant Floors. Tenant, at its sole cost and expense (subject to application of the Tenant Improvement Allowance), may install identification signage anywhere in the 2nd floor elevator lobby of the Building on or after the 2nd floor Premises Commencement Date, provided that such signs must not be visible from the exterior of the Building, provided further, that unless and until Tenant leases the entire 2nd floor of the Building, Landlord may maintain directional signage identifying other tenants of the 2nd floor of the Building in the 2nd floor elevator lobby. Tenant, at its sole cost and expense (subject to application of the Tenant Improvement Allowance), may install identification signage anywhere in the 4th floor elevator lobby of the Building on or after the 4th Floor Premises Commencement Date, provided that such signs must not be visible from the exterior of the Building, provided further, that unless and until Tenant leases the entire 4th floor of the Building on or after the 4th floor of the Building may maintain signage displaying its name and logo in the 4th floor elevator lobby.

23.4 <u>Ground Floor Premises</u>. Tenant shall be permitted to install signage (the number and appearance of which shall be subject to Landlord's prior approval, which approval shall not be unreasonably withheld, conditioned or delayed) in the interior of the Ground Floor Premises which is not affixed to the windows of the

Ground Floor Premises, but which is visible from the exterior of the Ground Floor Premises through such windows (the "Ground Floor Premises Signage").

23.5 <u>Prohibited Signage and Other Items</u>. Any signs, notices, logos, pictures, names or advertisements which are installed in violation of this <u>Article 23</u> may be removed by Landlord after notice and the expiration of applicable cure periods at the sole expense of Tenant.

23.6 Definition. For purposes of this Lease, the term "sign" and "signage" shall mean signs, designs, monuments, logos, banners, projected images, pennants, decals, advertisements, pictures, notices, lettering, numerals or graphics.

23.7 Occupancy. As used in this <u>Article 23</u>, any reference to "occupancy" shall mean that when the total rentable square footage of the Premises then subleased by Tenant to parties other than Permitted Transferees is subtracted from the total rentable square footage of the Premises that Tenant leases or has committed to lease (i.e. rentable square footage of the Premises shall be calculated as if the Premises includes all of the Tranche I Premises, the Tranche II Premises and the Tranches III Premises prior to the Commencement Date with respect to the Tranche I Premises, the Tranche II Premises and the Tranches III Premises) that the resulting number meets or exceeds the relevant threshold.

23.8 <u>Transfer of Right to Tenant's Signage</u>. The Building Top Signage and Exterior Building shall be personal to the Original Tenant and any Permitted Transferee, and may not be transferred to any other party. The Lobby Signage may be transferred only to a Transferee that has assumed all of Tenant's interest under this Lease, or that is subleasing more than one full Portion of the Premises (i.e., all of any of Tranche I, Tranche II or Tranche III), and to no other party.

ARTICLE 24

COMPLIANCE WITH LAW

Tenant shall not do anything or suffer anything to be done in or about the Premises or the Project which will in any way conflict with any Law now in force or which may hereafter be enacted or promulgated. At its sole cost and expense, Tenant shall promptly comply with all such applicable Laws which relate to (i) Tenant's use of the Premises, or (ii) the Alterations or the Tenant Improvements in the Premises. Should any standard or regulation now or hereafter be imposed on Landlord or Tenant by a state, federal or local governmental body charged with the establishment, regulation and enforcement of occupational, health or safety standards for employers, employees, landlords or tenants, then Tenant shall, at its sole cost and expense, comply promptly with such standards or regulations as it relates to the Premises or Tenant use of the Premises. The judgment of any court of competent jurisdiction or the admission of Tenant in any judicial action, regardless of whether Landlord is a party thereto, that Tenant has violated any of said governmental measures, shall be conclusive of that fact as between Landlord and Tenant. Landlord shall comply with all applicable Laws relating to the Base Building, provided that compliance with such applicable Laws is not the responsibility of Tenant under this Lease. Landlord shall be permitted to include in Operating Expenses any costs or expenses incurred by Landlord under this <u>Article 24</u> to the extent consistent with the terms of <u>Section 4.2</u>, above.

ARTICLE 25

LATE CHARGES

If any installment of Rent or any other sum due from Tenant shall not be received by Landlord or Landlord's designee within five (5) business days after written notice, then Tenant shall pay to Landlord a late charge equal to three percent (3%) of the overdue amount. The late charge shall be deemed Additional Rent and the right to require it shall be in addition to all of Landlord's other rights and remedies hereunder or at law and shall not be construed as liquidated damages or as limiting Landlord's remedies in any manner. In addition to the late charge described above, any Rent or other amounts owing hereunder which are not paid within ten (10) business days after the date they are due shall be in interest from the date when due until paid at the "Interest Rate." Notwithstanding the foregoing, the first late payment by Tenant in any twelve (12) month period shall not be subject to such interest

if Tenant pays the applicable amount within ten (10) business days after Tenant's receipt of written notice from Landlord that said amount is due. For purposes of this Lease, the "Interest Rate" shall be an annual rate equal to the lesser of (i) seven percent (7%) per annum, and (ii) the highest rate permitted by applicable Law.

ARTICLE 26

LANDLORD'S RIGHT TO CURE DEFAULT; PAYMENTS BY TENANT

26.1 Landlord's Cure. All covenants and agreements to be kept or performed by Tenant under this Lease shall be performed by Tenant at Tenant's sole cost and expense and without any reduction of Rent, except to the extent, if any, otherwise expressly provided herein. If Tenant shall fail to perform any obligation under this Lease, and such failure shall continue in excess of the time allowed under Section 19.1, above, unless a specific time period is otherwise stated in this Lease, Landlord may, but shall not be obligated to, make any such payment or perform any such act on Tenant's part without waiving its rights based upon any default of Tenant and without releasing Tenant from any obligations hereunder.

26.2 <u>Tenant's Reimbursement</u>. Except as may be specifically provided to the contrary in this Lease, Tenant shall pay to Landlord, upon delivery by Landlord to Tenant of statements equal to expenditures reasonably made and obligations incurred by Landlord in connection with the remedying by Landlord of Tenant's defaults pursuant to the provisions of <u>Section 26.1</u> within thirty (30) days after receipt of invoice together with reasonable supporting evidence. Tenant's obligations under this <u>Section 26.2</u> shall survive the expiration or sooner termination of the Lease Term.

ARTICLE 27

ENTRY BY LANDLORD

Landlord reserves the right at all reasonable times during Tenant's normal business hours on business days and upon at least twenty-four (24) hours prior notice to Tenant (except in the case of an emergency) to enter the Premises to (i) inspect them; (ii) show the Premises to prospective purchasers, or to current or prospective mortgagees, ground or underlying lessors or insurers or during the last twelve (12) months of the Lease Term with respect to each Tranche, show such Tranche to prospective tenants; (iii) post notices of nonresponsibility; or (iv) alter, improve or repair the Premises or the Building, or for structural alterations, repairs or improvements to the Building or the Building's systems and equipment. Landlord shall use commercially reasonable efforts to minimize interference with the operation of Tenant's business at the Premises during any entry by Landlord unto the Premises. Notwithstanding anything to the contrary contained in this Article 27. Landlord may enter the Premises at any reasonable time to (A) perform services required of Landlord, including janitorial service; (B) take possession due to any breach of this Lease in the manner provided herein; and (C) subject to Section 26.1 above, perform any covenants of Tenant which Tenant fails to perform. Landlord may make any such entries without the abatement of Rent, except as otherwise provided in this Lease, and may take such reasonable steps as required to accomplish the stated purposes. For each of the above purposes, Landlord shall at all times have a key with which to unlock all the doors in the Premises, excluding Tenant's vaults, safes and Secured Areas (as defined below). In an emergency, Landlord shall have the right to use any means that Landlord may deem proper to open the doors in and to the Premises. Any entry into the Premises by Landlord in the manner hereinbefore described shall not be deemed to be a forcible or unlawful entry into, or a detainer of, the Premises, or an actual or constructive eviction of Tenant from any portion of the Premises. No provision of this Lease shall be construed as obligating Landlord to perform any repairs, alterations or decorations other than those repairs, alterations or decorations that Landlord has expressly agreed to perform under this Lease. Tenant may reasonably restrict access by any visitor whom Landlord intends to bring onto the Premises who is, or may reasonably be suspected by Tenant to be, a competitor of Tenant. Tenant shall at all times, except in the case of emergencies, have the right to escort Landlord or any Landlord Parties while the same are in the Premises. Notwithstanding anything to the contrary set forth in this Article 27, Tenant may designate in writing certain reasonable areas of the Premises as "Secured Areas" should Tenant require such areas for the purpose of securing certain valuable property or confidential information. In connection with the foregoing, Landlord shall not enter such Secured Areas except in the event of an emergency. Landlord need not clean any area designated by Tenant as a Secured Area and shall only maintain or repair such secured areas to the extent (1) such repair or maintenance is required in order to maintain and repair the Base Building; (2) as required by applicable

Law, or (3) in response to specific requests by Tenant and in accordance with a Schedule reasonably designated by Tenant, subject to Landlord's reasonable approval. Access to the Premises by Landlord shall be in accordance with the security, safety and confidentiality requirements that Tenant may reasonably adopt from time to time, including, without limitation, a requirement that persons (including Landlord or Landlord Parties) having access to the Premises shall sign and deliver to Tenant a confidentiality and nondisclosure agreement in form and content reasonably acceptable to Tenant.

ARTICLE 28

TENANT PARKING

Tenant shall have the right, but not the obligation, to rent from Landlord, commencing on the Tranche I Commencement Date, a number of parking passes, not to exceed the amount of parking passes set forth in Section 9 of the Summary (as determined for the entire Premises, which shall be calculated as if the Premises includes all of the Tranche I Premises, the Tranche II Premises and the Tranches III Premises irrespective of whether or not the Commencement Date with respect to the Tranche I Premises, the Tranche II Premises and the Tranches III Premises has occurred but excluding the rentable square footage of any portion thereof still being leased by other tenants of the Building), on a monthly basis throughout the Lease Term, which parking passes shall pertain to the Parking Garage. To the extent Tenant elects to rent any parking passes from Landlord, Tenant shall pay to Landlord for automobile parking passes on a monthly basis at Landlord's prevailing rate which shall not exceed the fair market rate being charged in Comparable Buildings, or in comparable parking garages or facilities in the area of the Comparable Buildings. Landlord shall provide visitor parking in the Parking Garage for Tenant's visitors at Landlord's prevailing rate which shall not exceed the fair market rate being charged in Comparable Buildings. In addition, Tenant shall be responsible for the full amount of any taxes imposed by any governmental authority in connection with the renting of such parking passes by Tenant or the use of the Parking Garage by Tenant. Tenant's continued right to use the parking passes is conditioned upon Tenant abiding by all reasonable rules and regulations which are prescribed from time to time for the orderly operation and use of the Parking Garage where the parking passes are located, including any sticker or other identification system established by Landlord, Tenant's cooperation in seeing that Tenant's employees and visitors also comply with such reasonable rules and regulations. Landlord specifically reserves the right to change the size, configuration, design, layout and all other aspects of the Parking Garage at any time and Tenant acknowledges and agrees that Landlord may, without incurring any liability to Tenant and without any abatement of Rent under this Lease, from time to time temporarily close-off or restrict access to the Parking Garage for purposes of permitting or facilitating any such construction, alteration or improvements; provided that (i) Landlord shall use commercially reasonable efforts (without any obligation to engage overtime labor or commence any litigation) to minimize the extent and duration of any resulting interference with Tenant's parking rights and (ii) Landlord shall provide reasonable alternative parking for Tenant during the continuation of any work which causes the Parking Garage to close for more than three (3) consecutive business days. Landlord may delegate its responsibilities hereunder to a parking operator in which case such parking operator shall have all the rights of control attributed hereby to the Landlord. The parking passes rented by Tenant pursuant to this Article 28 are provided to Tenant solely for use by Tenant's own personnel and such passes may not be transferred, assigned, subleased or otherwise alienated by Tenant without Landlord's prior approval except to an Permitted Transferred or another assignee or subtenant consented to by Landlord in accordance with Article 14 above. Tenant may validate visitor parking by such methods as the Landlord may establish, at the validation rate from time to time generally applicable to visitor parking. Any election by Tenant to rent any parking passes from Tenant must be made, if at all, by Tenant's delivery of written notice to Landlord within six (6) months after the Tranche I Commencement Date. In addition, Tenant may on or before November 15 of each calendar year of the Lease Term, commit to lease, for at least one (1) year, all or a portion of Tenant's Parking Pass Ratio of parking passes in the Parking Garage commencing on the following January 1 (the "Parking Election Notice"). Thereafter, Landlord may lease the parking passes in the Parking Garage not committed to by Tenant to anyone Landlord desires for a period not to extend beyond January 1 of the following calendar year. If Tenant does not deliver a Parking Election Notice in any particular year, Tenant shall be deemed to have elected to lease no parking passes for the following year

ARTICLE 29

MISCELLANEOUS PROVISIONS

29.1 Terms; Captions. The words "Landlord" and "Tenant" as used herein shall include the plural as well as the singular. The necessary grammatical changes required to make the provisions hereof apply either to corporations or partnerships or individuals, men or women, as the case may require, shall in all cases be assumed as though in each case fully expressed. The captions of Articles and Sections are for convenience only and shall not be deemed to limit, construe, affect or alter the meaning of such Articles and Sections.

29.2 Binding Effect. Subject to all other provisions of this Lease, each of the covenants, conditions and provisions of this Lease shall extend to and shall, as the case may require, bind or inure to the benefit not only of Landlord and of Tenant, but also of their respective heirs, personal representatives, successors or assigns, provided this clause shall not permit any assignment by Tenant contrary to the provisions of Article 14 of this Lease.

29.3 <u>No Air Rights</u>. No rights to any view or to light or air over any property, whether belonging to Landlord or any other person, are granted to Tenant by this Lease. If at any time any windows of the Premises are temporarily darkened or the light or view therefrom is obstructed by reason of any repairs, improvements, maintenance or cleaning in or about the Project, the same shall be without liability to Landlord and without any reduction or diminution of Tenant's obligations under this Lease.

29.4 Transfer of Landlord's Interest. Tenant acknowledges that Landlord has the right to transfer all or any portion of its interest in the Project or Building and in this Lease, and Tenant agrees that in the event of any such transfer, Landlord shall automatically be released from all liability under this Lease arising from and after the date of transfer and Tenant agrees to look solely to such transferee for the performance of Landlord's obligations hereunder arising from and after the date of transfer and such transferee to be performed by Landlord, including the return of any security deposit and the L-C, and Tenant shall attorn to such transferee.

29.5 <u>Memorandum of Lease</u>. Concurrently with the execution and delivery of this Lease, the parties shall promptly execute and Landlord shall record, at its sole cost and expense, a short form memorandum in substantially the form attached hereto as <u>Exhibit I</u>. Within ten (10) business days after Landlord's written request following the expiration or earlier termination of this Lease, Tenant shall execute and deliver to Landlord in recordable form, a quitclaim deed designating Landlord as the transferee.

29.6 Landlord's Title. Landlord's title is and always shall be paramount to the title of Tenant. Nothing herein contained shall empower Tenant to do any act which can, shall or may encumber the title of Landlord.

29.7 Relationship of Parties. Nothing contained in this Lease shall be deemed or construed by the parties hereto or by any third party to create the relationship of principal and agent, partnership, joint venturer or any association between Landlord and Tenant.

29.8 <u>Application of Payments</u>. Landlord shall have the right to apply payments received from Tenant pursuant to this Lease, regardless of Tenant's designation of such payments, to satisfy any obligations of Tenant hereunder, in such order and amounts as Landlord, in its sole discretion, may elect.

29.9 <u>Time of Essence</u>. Time is of the essence with respect to the performance of every provision of this Lease in which time of performance is a factor.

29.10 <u>Partial Invalidity</u>. If any term, provision or condition contained in this Lease shall, to any extent, be invalid or unenforceable, the remainder of this Lease, or the application of such term, provision or condition to persons or circumstances other than those with respect to which it is invalid or unenforceable, shall not be affected thereby, and each and every other term, provision and condition of this Lease shall be valid and enforceable to the fullest extent possible permitted by Law.

29.11 <u>No Warranty</u>. In executing and delivering this Lease, Tenant has not relied on any representations, including, but not limited to, any representation as to the amount of any item comprising Additional Rent or the amount of the Additional Rent in the aggregate or that Landlord is furnishing the same services to other tenants, at all, on the same level or on the same basis, or any warranty or any statement of Landlord which is not set forth herein or in one or more of the exhibits attached hereto.

29.12 Landlord Exculpation. The liability of Landlord or the Landlord Parties to Tenant for any default by Landlord under this Lease or arising in connection herewith or with Landlord's operation, management, leasing, repair, renovation, alteration or any other matter relating to the Project or the Premises shall be limited solely and exclusively to the interest of Landlord in the Project, including any sales, insurance, condemnation or rental proceeds received by Landlord or the Landlord Parties in connection with the Project, Building or Premises. Neither Landlord, nor any of the Landlord Parties shall have any personal liability therefor, and Tenant hereby expressly waives and releases such personal liability on behalf of itself and all persons claiming by, through or under Tenant. The limitations of liability contained in this Section 29.12 shall inure to the benefit of Landlord's and the Landlord Parties' present and future partners, beneficiaries, officers, trustees, shareholders, agents and employees, and their respective partners, hereirs, successors and assigns. Under no circumstances shall any present or future partner of Landlord is a partnership), or trustee or beneficiary (if Landlord or any partner of Landlord is a trust), have any liability for the performance of Landlord's obligations under this Lease.

29.13 Entire Agreement. It is understood and acknowledged that there are no oral agreements between the parties hereto affecting this Lease and this Lease constitutes the parties' entire agreement with respect to the leasing of the Premises and supersedes and cancels any and all previous negotiations, arrangements, brochures, agreements and understandings, if any, between the parties hereto or displayed by Landlord to Tenant with respect to the subject matter thereof, and none thereof shall be used to interpret or construe this Lease, including, without limitation that certain Letter of Intent dated November 15, 2012 between the parties and that certain Confidentiality Agreement dated March 26, 2012 between the parties. None of the terms, covenants, conditions or provisions of this Lease can be modified, deleted or added to except in writing signed by the parties hereto.

29.14 <u>Right to Lease</u>. Subject to <u>Section 29.44</u> below, Landlord reserves the absolute right to effect such other tenancies in the Project as Landlord in the exercise of its sole business judgment shall determine to best promote the interests of the Building or Project. Tenant does not rely on the fact, nor does Landlord represent, that any specific tenant or type or number of tenants shall, during the Lease Term, occupy any space in the Building or Project.

29.15 <u>Force Majeure</u>. Any prevention, delay or stoppage due to strikes, lockouts, labor disputes, acts of God, inability to obtain services, labor, or materials or reasonable substitutes therefor, governmental actions, civil commotions, fire or other casualty, and other similar causes beyond the reasonable control of the party obligated to perform, except with respect to the obligations imposed with regard to Rent and other charges to be paid pursuant to this Lease (collectively, a "Force Majeure"), notwithstanding anything to the contrary contained in this Lease, shall excuse the performance of such party for a period equal to any such prevention, delay or stoppage and, therefore, if this Lease specifies a time period for performance of an obligation of either party, that time period shall be extended by the period of any delay in such party's performance caused by a Force Majeure. This Section 29.15 shall not apply to or govern the obligations of the parties under <u>Section 1.1.4</u> above.

29.16 <u>Waiver of Redemption by Tenant</u>. Tenant hereby waives, for Tenant and for all those claiming under Tenant, any and all rights now or hereafter existing to redeem by order or judgment of any court or by any legal process or writ, Tenant's right of occupancy of the Premises after any termination of this Lease.

29.17 Notices. All notices, demands, statements, designations, approvals or other communications (collectively, "Notices") given or required to be given by either party to the other hereunder or by law shall be in writing, shall be (A) sent by United States certified or registered mail, postage prepaid, return receipt requested ("Mail"), (B) delivered by a nationally recognized overnight courier, or (C) delivered personally. Any Notice shall be sent, transmitted, or delivered, as the case may be, to the appropriate address set forth in Sections 10 and 11 of the Summary, or to such other place as Tenant or Landlord may from time to time designate in a Notice to the other party. Any Notice will be deemed given (i) three (3) days after the date it is posted if sent by Mail, (ii) the date the overnight courier delivery is made, or (iii) the date personal delivery is made or attempted to be made. If Tenant is

notified of the identity and address of Landlord's mortgagee or ground or underlying lessor, Tenant shall give to such mortgagee or ground or underlying lessor written notice of any default by Landlord under the terms of this Lease by registered or certified mail. Notwithstanding anything to the contrary contained herein, if any breach of security in the Premises shall occur, then Landlord shall promptly (based on the nature of such breach) provide notice to Tenant via telephone call and e-mail to the following: (a) via telephone to (650) 653-4600 and (b) via email to gocenter@salesforce.com or such other number(s) and/or email address(es) as Tenant shall from time to time notify Landlord in writing.

29.18 Joint and Several. If there is more than one Tenant, the obligations imposed upon Tenant under this Lease shall be joint and several.

29.19 <u>Authority</u>. If Tenant is a corporation, trust or partnership, each individual executing this Lease on behalf of Tenant hereby represents and warrants that Tenant is a duly formed and existing entity qualified to do business in California and that Tenant has full right and authority to execute and deliver this Lease and that each person signing on behalf of Tenant is authorized to do so. In such event, Tenant shall, within ten (10) days after execution of this Lease, deliver to Landlord satisfactory evidence of such authority and, if a corporation, upon demand by Landlord, also deliver to Landlord satisfactory evidence of (i) good standing in Tenant's state of incorporation and (ii) qualification to do business in California.

29.20 <u>Attorneys' Fees</u>. In the event that either Landlord or Tenant should bring suit for the possession of the Premises, for the recovery of any sum due under this Lease, or because of the breach of any provision of this Lease or for any other relief against the other pursuant to this Lease, then all costs and expenses, including reasonable attorneys' fees, incurred by the prevailing party therein shall be paid by the other party, which obligation on the part of the other party shall be deemed to have accrued on the date of the commencement of such action and shall be enforceable.

29.21 <u>Governing Law; WAIVER OF TRIAL BY JURY</u>. This Lease shall be construed and enforced in accordance with the Laws of the State of California. IN ANY ACTION OR PROCEEDING ARISING HEREFROM, LANDLORD AND TENANT HEREBY CONSENT TO (I) THE JURISDICTION OF ANY COMPETENT COURT WITHIN THE STATE OF CALIFORNIA, (II) SERVICE OF PROCESS BY ANY MEANS AUTHORIZED BY CALIFORNIA LAW, AND (III) TO THE EXTENT PERMITTED BY LAW, IN THE INTEREST OF SAVING TIME AND EXPENSE, TRIAL WITHOUT A JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY EITHER OF THE PARTIES HERETO AGAINST THE OTHER OR THEIR SUCCESSORS IN RESPECT OF ANY MATTER ARISING OUT OF OR IN CONNECTION WITH THIS LEASE, THE RELATIONSHIP OF LANDLORD AND TENANT, TENANT'S USE OR OCCUPANCY OF THE PREMISES, AND/OR ANY CLAIM FOR INJURY OR DAMAGE, OR ANY EMERGENCY OR STATUTORY REMEDY.

29.22 <u>Submission of Lease</u>. Submission of this instrument for examination or signature by Tenant does not constitute a reservation of, option for or option to lease, and it is not effective as a lease or otherwise until execution and delivery by both Landlord and Tenant.

29.23 Brokers. Landlord and Tenant hereby warrant to each other that they have had no dealings with any real estate broker or agent in connection with the negotiation of this Lease, excepting only the real estate brokers or agents specified in Section 12 of the Summary (the "Brokers"), and that they know of no other real estate broker or agent who is entitled to a commission in connection with this Lease. Landlord shall pay the Brokers pursuant to the terms of separate commission agreements (the "Written Agreements") between Landlord and the Brokers. If Landlord does not make payment to Tenant's Broker pursuant to the terms of separate commission agreements (the "Written Agreements") between Landlord and the Brokers. If Landlord does not make payment to Tenant's Broker pursuant to the terms of the Written Agreement with Tenant's Broker and fails to make such payment within the applicable notice and cure period set forth in such Written Agreement, then Tenant may send a written notice to Landlord of such failure and if Landlord fails to pay Tenant's Broker within thirty (30) days following receipt of such notice, Tenant may, at its option, upon written notice to Landlord, make the payment of the amount then due and owing to Tenant's Broker pursuant to the terms and conditions of the Written Agreement with Tenant's Broker, in which event such payment amount (plus interest at the Interest Rate from the date such amount was owed until such off-set occurs) shall be credited against the payment(s) of Rent next due and owing under the Lease, on a monthly basis, until such amount is fully exhausted; provided, however, that Landlord shall have the right, in good faith, to notify Tenant in writing within thirty (30) days

following Landlord's receipt of Tenant's notice that the amounts described in Tenant's notice have been previously paid by Landlord, and upon Landlord providing reasonable evidence thereof to Tenant, Tenant shall not be entitled to offset such amount against Rent; provided, however, that if Tenant disputes such payment by Landlord, then Tenant may submit such claim to arbitration under the commercial arbitration rules of JAMS (and Landlord and Tenant hereby submit to arbitration of such matter by JAMS and the determination of such arbitrator shall be final and binding upon both Landlord and Tenant). If Tenant prevails with regard to such claim, Tenant may, at its option, upon written notice to Landlord, make the payment of the amount then due and owing to Tenant's Broker pursuant to the terms and conditions of the Written Agreement with Tenant's Broker, in which event such payment amount (plus interest at the Interest Rate from the date such amount was owed until such off-set occurs) shall be credited against the payment(s) of Rent next due and owing under the Lease, on a monthly basis, until such amount is fully exhausted. Any amounts so paid by Tenant to Tenant's Broker and offset from Rent shall no longer be owed from Landlord to Tenant's Broker pursuant to the terms of the Written Agreement. Each party agrees to indemnify and defend the other party against and hold the other party harmless from any and all claims, demands, losses, liabilities, lawsuits, judgments, costs and expenses (including without limitation reasonable attorneys' fees) with respect to any leasing commission or equivalent compensation alleged to be owing on account of any dealings with any real estate broker or agent, other than the Brokers, occurring by, through, or under the indemnifying party; provided, however, Tenant's indemnification obligations hereunder shall nevertheless apply to any amounts claimed to be owing to Tenant's Broker which were offset from Tenant's Rent obligations as set forth herein.

29.24 <u>Independent Covenants</u>. This Lease shall be construed as though the covenants herein between Landlord and Tenant are independent and not dependent and Tenant hereby expressly waives the benefit of any statute to the contrary and agrees that if Landlord fails to perform its obligations set forth herein, Tenant shall not be entitled to make any repairs or perform any acts hereunder at Landlord's expense or to any setoff of the Rent or other amounts owing hereunder against Landlord except as expressly set forth herein.

29.25 <u>Project or Building Name and Signage</u>. Landlord shall have the right at any time to change the name of the Project or Building as Landlord may, in Landlord's sole discretion, desire. Tenant shall not use the name of the Project or Building or use pictures or illustrations of the Project or Building in advertising or other publicity or for any purpose other than as the address of the business to be conducted by Tenant in the Premises, without the prior written consent of Landlord.

29.26 <u>Counterparts</u>. This Lease may be executed in counterparts with the same effect as if both parties hereto had executed the same document. Both counterparts shall be construed together and shall constitute a single lease. Landlord or Tenant may deliver this Lease to the other party via electronic fax or other electronic means. Fax or electronic signatures shall be as valid and binding upon the parties as are original ink signatures.

29.27 <u>Confidentiality</u>. Except as expressly permitted in this <u>Section 29.27</u>, neither party nor its agents, servants, employees, invitees and contractors will, without the prior written consent of the other party, disclose any Confidential Information of the other party to any third party. Information will be considered "**Confidential Information**" of a party if either: (a) it is disclosed by the party to the other party in tangible form and is conspicuously marked "Confidential," "Proprietary" or (b) it is disclosed by one party to the other party in non-tangible form and is identified as confidential at the time of disclosure. In addition, notwithstanding anything in this Lease to the contrary, the terms of this Lease (but not its mere existence) will be deemed Confidential Information for ach party. Other than the terms and conditions of this Lease (but not its mere existence) will be deemed Confidential information of a source other than one having an obligation of confidentiality to the disclosing party; (ii) becomes known (independently of disclosure by the disclosing party) to the receiving party directly or indirectly from a source other than one having an obligation of confidentiality to the disclosing party; (iii) becomes known or otherwise ceases to be secret or confidential, except through a breach of this Lease by the receiving party. The terms and conditions of this Lease will cease being confidential if, and only to the extent that, they become publicly known, except through a breach of this Lease) in a manner consistent with the steps taken to protect its own trade secrets and confidential information, but not less than a reasonable degree of care. Each party and siles less in a reasonable degree of care. Each party and solution where: (A) the disclosure is required by applicable Law or by an order of a court or other governmental body having jurisdiction after giving reasonable notice to the other

party with adequate time for such other party to seek a protective order; (B) if in the opinion of counsel for such party, disclosure is advisable under any applicable securities laws regarding public disclosure of business information; (C) the disclosure is reasonably necessary and is to that party's or its affiliates' employees, officers, directors, attorneys, accountants, consultants and other advisors, or to Landlord's mortgage lender and its counsel, or the disclosure is otherwise necessary for a party to exercise its rights and perform its obligations under this Lease, so long as in all cases the disclosure is no broader than necessary and the party who receives the disclosure to receiving the disclosure to keep the information confidential; or (D) the disclosure is reasonably necessary for a party to conclude a business transaction. Each party is responsible for ensuring that any Confidential Information of the other party that the first party discloses pursuant to this <u>Section 29.27</u> is kept confidential by the person

29.28 Intentionally Omitted

29.9 Building Renovations. It is specifically understood and agreed that Landlord has made no representation or warranty to Tenant and has no obligation and has made no promises to alter, remodel, improve, renovate, repair or decorate the Premises, Building, or any part thereof and that no representations respecting the condition of the Premises or the Building have been made by Landlord to Tenant except as specifically set forth herein or in the Tenant Work Letter. However, Tenant hereby acknowledges that Landlord may during the Lease Term renovate, improve, alter, or modify (collectively, the "Renovations") the Project, the Building and/or the Premises including without limitation the parking structure, the retail portion of the Project, common areas, systems and equipment, roof, and structural portions of the same, which Renovations may include, without limitation, (i) installing sprinklers in the Building acommon areas and tenant spaces, (ii) modifying the common areas and tenant spaces to comply with applicable laws and regulations, including regulations relating to the physically disabled, seismic conditions, and building safety and security, and (iii) installing new floor covering, lighting, and wall coverings in the Common Areas. In connection with any Renovations, Landlord may, among other things, erect scaffolding or other necessary structures in the Building, temporarily limit or eliminate access to portions of the Common Areas, or perform work in the Building, which work may create noise, dust or leave debris in the Building. Tenant hereby agrees that such Renovations and Landlord's actions in connection with such Renovations shall in no way constitute a constructive eviction of Tenant ne, except as expressly set forth in <u>Section 6.6</u>, above, entitle Tenant to any abatement of Rent. Landlord shall use commercially reasonable efforts to minimize interference with Tenant's business in making any Renovations.

29.30 No Violation. Tenant hereby warrants and represents that neither its execution of nor performance under this Lease shall cause Tenant to be in violation of any agreement, instrument, contract or Law, by which Tenant is bound.

29.31 <u>Communications and Computer Lines</u>. Tenant may install, maintain, replace, remove or use any communications or computer wires, cables and related devices (collectively the "Lines") at the Project in or serving the Premises, provided that: (i) Tenant shall obtain Landlord's prior written consent not to be unreasonably withheld, conditioned or delayed, use an experienced and qualified contractor approved in writing by Landlord (such approval not to be unreasonably withheld, conditioned or delayed), and comply with all of the other provisions of <u>Articles 8</u> and 2 of this Lease, (ii) an acceptable number of spare Lines and space for additional Lines shall be maintained for existing and future occupants of the Project, as determined in Landlord's reasonable opinion, (iii) the Lines therefor (including riser cables) shall be appropriately insulated to prevent excessive electromagnetic fields or radiation, and shall be surrounded by a protective conduit reasonably acceptable to Landlord, (iv) any new or existing Lines servicing the Premises shall comply with all applicable Laws, (v) as a condition to permitting the installation of new Lines, Landlord may require that Tenant remove any Lines located in or serving the Premises which are installed in violation of these provisions. Landlord shall use commercially reasonable efforts to maintain throughout the Lease Term multiple independent fiber providers at the Building available to Tenant. Upon written request by Tenant, Landlord shall grant upon commercially reasonable terms, a license coterminous with the Lease Term more reputable telecommunications utility provider(s) designated by Tenant (any such utility, an "Approved Fiber Provider"), to permit any such Approved Fiber Provider is conduit and fiber into the Building from locations outside the Building, (b) to provide connectivity from the Building is Main Point of Entry ("MPOE") to the Premises and (c) to permit any such Approved Fiber Provider

to maintain and operate such conduit and fiber in the Premises and Building. Tenant acknowledges that Landlord has engaged the services of a riser maintenance contractor (the "**Riser Manager**") to oversee the use of the MPOE and the Building risers, and that any installation of conduit or fiber in the Building shall be required to be performed by the Riser Manager. In addition, to the extent that Tenant or any Approved Fiber Provider requires the installation of equipment in the MPOE, a commercially reasonable MPOE use fee may be charged. Tenant shall be granted a license coterminous with the Lease Term to locate telecommunications equipment in the MPOE. Subject to coordination with the Riser Manager Tenant shall have access to the MPOE twenty-four (24) hours per day, seven (7) days per week, subject to Landlord's reasonable access control procedures. Upon written request by Tenant, Landlord shall grant upon commercially reasonable terms, a license coterminous with the Lease Term to one or more providers of distributed cellular and wireless antenae(s) ("**Distributed Antennae**") to install equipment in the Building. Tenant shall be granted a license coterminous with the Lease Term to install Distributed Antennae in the Building.

29.32 Development of the Project

29.32.1 <u>Subdivision</u>. Landlord reserves the right to further subdivide all or a portion of the Project; provided such subdivision does not adversely affect Tenant's rights or obligations under this Lease or affect Tenant's access to or use of the Premises. Tenant agrees to execute and deliver, at no cost to Tenant within ten (10) business days following demand by Landlord and in the form reasonably requested by Landlord, any additional documents needed to conform this Lease to the circumstances resulting from such subdivision.

29.32.2 The Other Tenant Improvements. If portions of the Project or property adjacent to the Project (collectively, the "Other Tenant Improvements") are owned by an entity other than Landlord, Landlord, Landlord, at its option, may enter into an agreement with the owner or owners of any or all of the Other Tenant Improvements to provide (i) for reciprocal rights of access and/or use of the Project and the Other Tenant Improvements, (ii) for the common management, operation, maintenance, improvement and/or repair of all or any portion of the Project and the Other Tenant Improvements, (iii) for the use or improvements and the operating expenses and taxes for the Other Tenant Improvements to the Project, and (iv) for the use or improvement of the Other Tenant Improvements and/or excavation of the Other Tenant Improvements and/or the Project in connection with the improvement, construction, and/or excavation of the Other Tenant Improvements and/or the Project; provided that, Tenant shall have the right to approve any such agreement, such approval not to be unreasonably withheld, conditioned or delayed. Nothing contained herein shall be deemed or construed to limit or otherwise affect Landlord's right to convey all or any portion of the Project or any other of Landlord's rights described in this Lease.

29.32.3 <u>Construction of Project and Other Tenant Improvements</u>. Tenant acknowledges that portions of the Project and/or the Other Tenant Improvements may be under construction following Tenant's occupancy of the Premises, and that such construction may result in levels of noise, dust, obstruction of access, etc. which are in excess of that present in a fully constructed project. Tenant hereby waives any and all rent offsets or claims of constructive eviction which may arise in connection with such construction. Landlord shall use commercially reasonable efforts to minimize interference with Tenant's business in performing any such construction.

29.33 <u>Patriot Act and Executive Order 13224</u>. Tenant represents, warrants and covenants that each party that (other than through the passive ownership of interests traded on a recognized securities exchange) constitutes, owns, controls, or is owned or controlled by Tenant, any guarantor hereof or any subtenant of Tenant is not, and at no time during the Lease Term will be, (i) in violation of any applicable laws relating to terrorism or money laundering, or (ii) among the parties identified on any list compiled pursuant to Executive Order 13224 for the purpose of identifying suspected terrorists or on the most current list published by the U.S. Treasury Department Office of Foreign Assets Control at its official website, http://www.treas.gov/ofac/tllsdn.pdf or any replacement website or other replacement official publication of such list.

29.34 Background Checks

29.34.1 <u>Background Investigation Requirements</u>. As a precondition to Tenant providing any of Landlord's employees, independent contractors, vendors, agents, subcontractors, and/or invitees ("**Personnel**") with badge (unescorted) access to the Premises, Landlord shall either permit Tenant to conduct on such Personnel a



criminal history check and verification of education, employment history, Social Security Number and legal right to work, as described below (collectively referred to as "Background Investigation"), or perform its own Background Investigation at Tenant's cost (provided that if a particular vendor already performs a similarly detailed Background Investigation and provides Landlord or Tenant with the results thereof, then Tenant shall rely on the results of such prior Background Investigation and not perform its own investigation). Tenant shall not have the right to require a Background Investigation to the extent that such an investigation is not allowed by any applicable union contracts. The Background Investigation shall include the following:

29.34.1.1 federal and state check for felony and misdemeanor criminal convictions in all locations where the assigned employee has resided, has been employed, has attended school or has applied for credit in the immediately preceding seven (7) years, including a criminal database check of information from all fifty states for federal and state convictions, a check for outstanding warrants and a check for pending felony charges in all such locations, provided that statewide county searches shall be performed in all states where such search mechanism is available without requiring specialized data (such as fingerprints or DNA);

29.34.1.2 a check of U.S. Government Specially Designated National and export denial lists, including criminal records search in the National Criminal Database, an Office of Foreign Assets Control (OFAC) check, a check against the Bureau of Industry and Security Denied Persons List, a check against the Office of Inspector General (OIG) Exclusion List, the FDA Debarment List (Drug Product Applications), and a check against the General Services Administration (GSA) Excluded Parties List;

	29.34.1.3	an all states check of available national and state sex offender registries;
	29.34.1.4	the name to which individual's Social Security Number is attributed shall be verified;
	29.34.1.5	the individual's identity shall be verified by an independent identity check by passport or other similar government document;
he conified.	29.34.1.6	the individual's citizenship, most recent country of permanent residence, and legal right to work in the jurisdiction in which the employee will be performing services shall
be verified;		
	29.34.1.7	confirmation of academic and professional qualifications, including the highest diploma, degree or certificate earned; and
	29.34.1.8	employment history, including dates of employment, employers and positions held, each as provided by the individual and/or as set forth on the individual's resume shall be

verified for at least the previous seven (7) years of employment and military service, or less if the individual was a full-time student during that period. 29.35 <u>Foreign Employment; Change</u>. For any period of time encompassed in the foregoing background check requirement when the employee was resident outside of the United States, such background

checking shall be conducted by an investigative agency approved by Tenant, and in a manner approved by Tenant, at Tenant's sole cost and expense. Such requirements shall be subject to change by Tenant from time to time.

29.35.1 Individual Consent. Prior to Tenant conducting a Background Investigation, Landlord shall secure from Landlord's Personnel such individual's written consent to perform the Background Investigation specified above and to disclose to Tenant, upon request, the results thereof.

29.35.2 Prohibited Assignments. Landlord agrees that Tenant may refuse to grant unescorted access to Landlord's Personnel if such individual:

29.35.2.1 has been convicted of a felony or misdemeanor (or the equivalent thereof under relevant non-US law), or for whom a warrant is outstanding, or for whom a felony or misdemeanor charge is currently pending, or is on a US Government Specially Designated National, the FDA Debarment List (Drug Product Applications), or export denial list. The foregoing shall not apply to a minor traffic violation other than reckless driving, hit and run, driving to endanger, vehicular homicide, or other criminal offense involving gross negligence, recklessness, intentional or willful misconduct while operating a motor vehicle), to a conviction that has been legally expunged, or to a conviction for a misdemeanor that occurred while the employee was under the age of twenty-one years or was more than seven (7) years in the past;

- 29.35.2.2 does not have the legal right to work in the jurisdiction in which the Premises is located;
- 29.35.2.3 or for whom there is a significant deviation between the information reported by the individual and results of the background check.

29.36 <u>Rules and Regulations</u>. Tenant agrees to comply with all rules and regulations of the Building and the Project imposed by Landlord as set forth on <u>Exhibit J</u> attached hereto (the "Rules and Regulations"). To the extent of any conflict between the terms and conditions of this Lease and the terms and conditions set forth in the Rules and Regulations, the terms and conditions of this Lease shall control. Landlord shall enforce the Rules and Regulations in a non-discriminatory manner.

29.37 <u>Standards of Performance and Approvals</u>. Unless otherwise provided in this Lease, whenever approval, consent or satisfaction (collectively, an "approval") is required of a party pursuant to this Lease or an Exhibit hereto, such approval shall not be unreasonably withheld or delayed. Unless provision is made for a specific time period, approval (or disapproval) shall be given within thirty (30) days after receipt of the request for approval. Nothing contained in this Lease shall limit the right of a party to act or exercise its business judgment in a subjective manner with respect to any matter as to which it has been (i) specifically granted such right, or (ii) granted the right to act in its sole discretion or sole judgment, whether "objectively" reasonable under the circumstances, and any such exercise shall not be deemed inconsistent with any covenant of good faith and fair dealing implied by law to be part of this Lease.

29.38 Business Day. For purposes hereof, "business say" shall be all calendar days except Saturdays and Sundays, federal holidays and holidays observed by national banks in the State of California.

29.39 <u>Waiver of Consequential and Special Damages</u>. Neither party shall be liable to the other party for any special or consequential damages, loss of profits, loss of rents or other revenues (other than Rent due under this Lease), loss of business opportunity, loss of goodwill or loss of use resulting from the failure of such party to meet its obligations under the Lease, except as expressly set forth in <u>Article 16</u>, above regarding a holdover in the Premises. The parties acknowledge and agree that if Landlord is required to abate the rent of another tenant at the Project under the terms and conditions of such tenant's lease, or as required by applicable law, as the result of any Alteration constructed by or on behalf of Tenant, or in connection with any repair or maintenance performed by or no behalf of Tenant, which interferes with such tenant's use of its premises, then such rental abatement shall not be deemed consequential damages, loss of pusites of postful or loss of use within the limitation set forth in the preceding sentence.

29.40 <u>Limitation of Tenant's Liability</u>. Under no circumstances shall any present or future partner, member, stockholder, trustee, beneficiary, officer, director, employee or agent of Tenant have any personal liability for the performance of Tenant's obligations under this Lease.

29.41 Landlord's Waiver of Security Interest in Tenant's Personal Property. Landlord hereby acknowledges and agrees that any and all of Tenant's movable furniture, furnishings, trade fixtures and equipment at the Premises ("Tenant's Property") may be financed by a third-party lender or lessor (an "Equipment Lienor"), and Landlord hereby (a) waive any rights to Tenant's Property to the lien of any Equipment Lienor, and (b) agrees to recognize the rights of any such Equipment Lienor, subject to and in accordance with a commercially reasonable waiver agreement to be entered into by and between Landlord and the Equipment Lienor following request by Tenant.

29.42 <u>No Continuous Operation</u>. Notwithstanding any provision of this Lease to the contrary, Tenant shall (a) not be required to occupy or to continuously operate the Premises, and Tenant shall have the right to cease operations (whether or not Tenant vacates the Premises) without same constituting a default by Tenant under this Lease provided Tenant continues to pay Rent and perform its other obligations under this Lease, and (b) have the right to remain open for business only on the days and during the hours Tenant determines is commercially practical.

29.43 No Relocation Rights. Landlord shall have no rights to relocate Tenant without Tenant's prior written approval, which may be withheld for any or no reason in Tenant's sole discretion.

29.44 <u>Tenant Competitor</u>. Provided that this Lease is then in full force and effect, Landlord agrees that Landlord shall not, without the prior written consent of Tenant, enter into any lease, license or other agreement relating to the Building (each, an "Occupancy Agreement") with any Tenant Competitor (hereinafter defined) or permit any tenant, subtenant, licensee or other occupant of the Building under an Occupancy Agreement to assign its lease, license or other agreement for space in the Building or sublet any portion of its premises to a Tenant Competitor. For purposes hereof, the term "Tenant Competitor" shall mean any person, corporation, limited liability company, association, trust or partnership which (i) controls, is controlled by or is under common control with such entity or (ii) which results from a merger or consolidation with such entity or (iii) which succeeds to the business and assets of such entity. Once during each 12-month period during the Term, Tenant shall be entitled to change any of the entities listed above upon thirty (30) days' prior written notice to Landlord, but the list shall in no event exceed ten (10) entities at any given time. Any change in the entities listed above shall be effective only on a prospective basis, and Landlord shall not be liable to Tenant (or aprecipt of Tenant's notice adding such entity to the list of Tenant Competitors. Notwithstanding anything herein to the contrary, Landlord shall not be deemed to have violated the terms of this Section if any tenant or occupant of the Building reges or consolidates with or into, or acquires or acquires or acquires or such assets for the section of this Section 29.44 shall apply with respect to Two Rincon and the term "Building" (as used in this <u>Section 29.44</u>) shall include Two Rincon.

29.45 <u>Landlord Representations and Warranties</u>. To induce Tenant to execute this Lease, and in addition to the other representations and warranties of Landlord contained in this Lease, Landlord warrants and represents that:

29.45.1 If Landlord is a corporation or limited liability company, Landlord hereby covenants and warrants that Landlord is a duly authorized and existing corporation or limited liability company, that Landlord has and is qualified to do business in the State, that the corporation or limited liability company has full right and authority to enter into this Lease, and that each person signing on behalf of the corporation is authorized to do so.

29.45.2 As of the Effective Date, no Security Documents encumber the Project or any portion thereof except those in favor of Wells Fargo Bank, N.A. as trustee for the registered holders of JP Morgan Chase Commercial Mortgage Securities Trust, Commercial Mortgage Pass-Through Certificates, Series 2011-C4, by and through its Master Servicer and attorney in fact, Midland Loan Services, a division of PNC Bank ("Landlord's Current Lender").

29.45.3 Neither its execution of nor performance under this Lease shall cause Landlord to be in violation of any agreement, instrument, contract, or Law by which Landlord is bound.

29.45.4 The only Building One Superior Rights Holders are those tenants set forth on **Exhibit L-1** attached hereto.

29.45.5 The only Building Two Superior Rights Holders are those tenants set forth on **Exhibit L-2** attached hereto.

29.45.6 As of the Effective Date, no person or entity (except Tenant) currently has any right to purchase, ground lease or otherwise take any ownership interest in the Project.

29.45.7 General office use does not invalidate or otherwise violate a requirement or condition of any fire, extended coverage or any other insurance policy carried by Landlord covering the Project or any portion thereof, or the property located therein.

29.46 <u>Anti-corruption</u>. It is the intent of the parties that no payments or transfers of anything of value shall be made which have the purpose or effect of public or commercial bribery, acceptance of or acquiescence in extortion, kickbacks, or other unlawful or improper means of obtaining business or any improper advantage. The parties shall comply with all applicable international anti-corruption laws, such as the Foreign Corrupt Practices Act 15 U.S.C. § 78dd-1, et seq.

29.47 <u>Storage Space</u>. Tenant shall have the right (but not the obligation) to lease up to 13,000 square feet of space in the Building currently designated as storage space (the "**Storage Space**") for the Lease Term, provided that Tenant may terminate any particular designated Storage Space so leased on not less than thirty (30) days prior notice. Tenant shall elect to lease any such space on or before the Tranche I Commencement Date. The Storage Space shall be leased by Tenant at a rental rate equal to ^{Reducted} per square foot per year, which rate shall increase annually to be ^{Reducted} of the prior year's rate. All Storage Space leased by Tenant shall be in a condition reasonable suitable for use as storage space and Tenant shall be fully responsible for repairing any damage to the Storage Space resulting from or relating to Tenant's use thereof. Tenant shall comply with such reasonable rules and regulations set forth in <u>Article 10</u>, above, shall apply to Tenant's use and occupancy of any such Storage Space.

[remainder of page intentionally left blank; signature page follows]

IN WITNESS WHEREOF, Landlord and Tenant have caused this Lease to be executed as of the Effective Date.

"Landlord"

HUDSON RINCON CENTER, LLC, a Delaware limited liability company

- Rincon Center Commercial, LLC, a Delaware limited liability company, By: its sole member
 - Hudson Pacific Properties, L.P., a Maryland limited partnership, its sole member By:
 - Hudson Pacific Properties, Inc., a Maryland corporation, By: its general partner

By:	/s/ Howard Stern
Name:	Howard Stern
Title:	President

"Tenant"

72

salesforce.com, inc., a Delaware corporation

By:	/s/ Burke Norton	
Name:	Burke Norton	
Title:	Executive Vice President	
By:	/s/ Graham V. Smith	
By: Name:	/s/ Graham V. Smith Graham V. Smith	

<u>EXHIBIT B</u>

RENT SCHEDULE

Redacted

EXHIBIT C

TENANT WORK LETTER

This Tenant Work Letter shall set forth the terms and conditions relating to the construction of the Premises. This Tenant Work Letter is essentially organized chronologically and addresses the issues of the construction of the Premises, in sequence, as such issues will arise during the actual construction of the Premises. All references in this Tenant Work Letter to Articles or Sections of "this Lease" shall mean the relevant portions of <u>Articles 1</u> through <u>29</u> of the Office Lease to which this Tenant Work Letter is attached as <u>Exhibit B</u> and of which this Tenant Work Letter forms a part, and all references in this Tenant Work Letter to Sections of "this Lease" shall mean the relevant portion of <u>Sections 1</u> through <u>5</u> of this Tenant Work Letter. All initially capitalized words not separately defined herein shall have the meaning given to such words in the Lease.

SECTION 1

BASE, SHELL AND CORE; DELIVERY OF THE PREMISES; BUILDING STRUCTURE

1.1 Base, Shell and Core. Landlord has constructed, at its sole cost and expense, the base, shell, and core (i) of the Premises, and (ii) of the floors of the Building on which the Premises are located (collectively, the "Base, Shell, and Core" or the "Base Building"). The Base, Shell and Core shall be delivered by Landlord to Tenant in their presently existing, "as-is" condition, except as otherwise expressly provided in this Tenant Work Letter.

1.2 Delivery of the Premises. Landlord shall, at Landlord's sole cost, complete the work items set forth in Schedule 1 attached hereto (collectively, the "Landlord Work"). Landlord shall perform the Landlord Work in a good and workmanlike manner, and, to the extent necessary for Tenant to pull any necessary construction permits or for Tenant to legally occupy the Premises for the Permitted Use, in accordance with applicable Laws. The "Delivery Condition" shall mean as to each Portion of the Premises (a) Landlord's Work in such Portion of the Premises has been substantially completed in accordance with applicable Laws to the extent required in order to allow Tenant to obtain a certificate of occupancy, or its legal equivalent, for such Portion of the Premises for the Permitted Use; (b) all of the Building Systems serving the applicable Laws to the extent required in order for Landlord to maintain any certificate of occupancy issued for the Building or Premises, and (d) such Portion of the Premises has been delivered to Tenant vacant and in broom clean condition.

1.3 Building Structure. Landlord shall be responsible to complete, at Landlord's sole cost and expense, upgrades to the Base Building necessitated by the overall size of the Tenant Improvements required by California and San Francisco Building Code Sections 3404.7, 3401.8 and 1604.11 in effect as of the date hereof (the "Structural Code"). To the extent any specific elements of the Tenant Improvements (e.g., slab openings, stairwells, etc.) require modification of the existing structural systems of the Building that would not otherwise be required by the Structural Code, Tenant shall complete such modifications at Tenant's sole cost and expense. Landlord and Tenant shall mutually cooperate in good faith with each other in connection with the concurrent construction and completion of any structural upgrades and/or modifications pursuant to this Section 1.3 and the Tenant Improvements. (the "Seismic Work").

SECTION 2

TENANT IMPROVEMENTS

1

2.1 <u>Tenant Improvement Allowance</u>. Tenant shall be entitled to a one-time improvement allowance (the "**Tenant Improvement Allowance**") in the amount of Redacted per rentable square foot of the Premises (i.e., Redacted for the costs relating to the initial design and construction of the improvements (the "**Tenant Improvements**"). Notwithstanding the foregoing or anything to the contrary set forth in this Tenant Work Letter, if Tenant notifies Landlord in writing that (i) Tenant desires to perform any cosmetic alterations or other upgrades to

the restrooms on the second (2^{nd}) floor of the Building, and (ii) in connection with such alterations or upgrades, Tenant desires to perform certain work with respect to such restrooms that would otherwise be Landlord Work under the terms of this Tenant Work Letter (i.e., upgrades to the second (2^{nd}) floor restrooms that are necessary in order to pull any required construction permits or for Tenant to legally occupy the 2^{nd} Floor Premises for the Permitted Use), then in addition to the Tenant Improvement Allowance, Tenant shall be entitled to obtain reimbursement from Landlord for the actual and reasonable costs incurred by Tenant in performing any Landlord Work in the second (2^{nd}) floor restrooms promptly upon Landlord's receipt from Tenant of any invoices and/or other reasonable supporting documentation. Tenant shall be entitled to aggregate the Tenant Improvement Allowance in any Portion of the Premises as Tenant may determine in its sole discretion; <u>provided, however</u>, in no event shall Tenant apply less than fifty percent (50%) of the portion of the Tenant Improvement Allowance applicable to any Portion of the Premises to such Portion of the Premises. Notwithstanding the foregoing or any contrary provision of this Lease, all Tenant Improvements shall be deemed Landlord's property under the terms of the Lease. Any unused portion of the Tenant Improvement Allowance remaining as of the date which is eighteen (18) months after the Suite 255 Commencement Date (the "**Outside Draw Date**"), shall remain with Landlord and Tenant shall have no further right thereto; provided, that if as of the Outside Draw Date there remains any unused portion of the Tenant Improvement Allowance then such unused portion shall, subject to the limitations set forth in <u>Section 2.3</u> below, automatically without delivery of a Rent Credit Notice (as defined below) be applied as a credit against the payment(s) of Rent next due under the Lease thereafter.

2.2 Disbursement of the Tenant Improvement Allowance.

2.2.1 Tenant Improvement Allowance Items. Except as otherwise set forth in this Tenant Work Letter, the Improvement Allowance shall be disbursed by Landlord only for the following items and costs (collectively the "Improvement Allowance Items"):

2.2.1.1 Payment of the fees of the "Architect," "Historic Architect" and the "Engineers," as those terms are defined in <u>Section 3.1</u> of this Tenant Work Letter, and payment of the actual and reasonable out-of-pocket engineer or consultant fees incurred by Landlord in connection with the review of the "Construction Drawings," as that term is defined in <u>Section 3.1</u> of this Tenant Work Letter, by third party engineering consultant;

- 2.2.1.2 The payment of plan check, permit and license fees relating to construction of the Tenant Improvements;
- 2.2.1.3 The cost of construction of the Tenant Improvements including, without limitation, testing and inspection costs, contractors' fees and general conditions;

2.2.1.4 The cost of any changes in the Base Building when such changes are required by the Construction Drawings, such cost to include all direct architectural and/or engineering fees and expenses incurred in connection therewith;

2.2.1.5 The cost of any changes to the Construction Drawings or Tenant Improvements required by all applicable building codes (the "Code");

- 2.2.1.6 The cost of the "Coordination Fee," as that term is defined in <u>Section 4.2.2.1</u> of this Tenant Work Letter;
- 2.2.1.7 Sales and use taxes; and
- 2.2.1.8 The cost of installation of cabling, Lines and Supplemental HVAC Unit(s) (if any).

2.2.2 Disbursement of Tenant Improvement Allowance. During the construction of the Tenant Improvements, Landlord shall make monthly disbursements of the Tenant Improvement Allowance for Tenant Improvement Allowance Items and shall authorize the release of monies as follows.

2.2.2.1 <u>Monthly Disbursements</u>. On or before the twentieth (20th) day of each calendar month, during the construction of the Tenant Improvements (or such other date as Landlord may designate), Tenant may deliver to Landlord: (i) a request for payment of the "Contractor," as that term is defined in <u>Section 4.1.1</u> of this Tenant Work Letter, approved by Tenant, in a form to be provided by Landlord, showing the schedule, by trade, of percentage of completion of the Tenant Improvements in the Premises, detailing the portion of the work completed and the portion not completed; (ii) invoices from all of "Tenant's Agents," as that term is defined in <u>Section 4.1.2</u> of this Tenant Work Letter, for labor rendered and materials delivered to the Premises and services provided to Tenant; and (iii) executed mechanic's liens which shall comply with the appropriate provisions of California Civil Code Sections 8132, 8134, 8136 and 8138; which lien releases from all of Tenant in payment of the lesser of: (A) the amounts so requested by Tenant, as set forth in this <u>Section 2.2.2.1</u>, above, less a ten percent (10%) retention (the aggregate amount of such retentions to be known as the "**Final Retention**"), and (B) the balance of any remaining available portion of the Tenant's payment request. Tenant shall, at Tenant's clection, have the right to submit applications for disbursement of the Tenant Improvement Allowance (including the Final Retention) premises, the Suite 250 Premises, respectively) or for the retire Premises. Tenant shall be entitied to deliver requests for disbursements pursuant to this <u>Section 2.2.2.1</u> so of the section 2.2.2.1 so often than the right os submit applications for disbursement of the Tenant Improvement Allowance (including the Final Retention). Landlord's payment of such amounts shall not be deemed Landlord's approval or acceptance of the work furnished or materials supplied as set forth in Tenant's payment request. Tenant shall, at Tenant's clection, have the right to submit applications for disb

2.2.2.2 Final Retention. Subject to the provisions of this Tenant Work Letter, a check payable to Tenant for the Final Retention with respect to the Tenant Improvements performed to each Tranche (or, as to Tranche III, each of the 2nd Floor Premises, the Suite 250 Premises or the Suite 255 Premises, respectively) shall be delivered by Landlord to Tenant within forty-five (45) days following the date upon which the following have been satisfied (i) Tenant delivers to Landlord (a) paid invoices for all Tenant Improvements and related costs for which the Tenant Improvement Allowance is to be dispersed, (b) signed permits for all Tenant S Agents who are entitled to file mechanic's liens which shall comply with the 2nd Floor Premises, the Suite 250 Premises or the Suite 255 Premises, respectively), (c) executed mechanic's lien releases from all of Tenant's Agents who are entitled to file mechanic's liens which shall comply with the appropriate provisols of California Civil Code Sections 8132, 8134, 8136 and 8138; which lien releases shall be conditional with respect to the requested payment amounts and unconditional with respect to payment amounts previously disbursed by Landlord, (ii) the Architect delivers to Landlord, certificate of Substantial Completion" for the applicable Tranche (or, as to Tranche III, in the 2nd Floor Premises, the Suite 250 Premises, respectively) in a form reasonably acceptable to Landlord, certifying that the construction of the Tenant Improvements in such Tranche (or, as to Tranche III, or the 2nd Floor Premises, the Suite 250 Premises or the Suite 255 Premises, respectively) in both paper and electronic forms (including, as-built drawings, and final record CADD files for the associated plans, warranties and guarantees from all contractors, subcontractors and material suppliers, an independent air balance report, and O&M manuals) (the "**Close Out Package**"), and (iv) a certificate of occupancy, a temporary certificate of occupancy or its equivalent is issued to Tenant with respect to suc

2.2.2.3 Other Terms. Landlord shall only be obligated to make disbursements from the Tenant Improvement Allowance to the extent costs are incurred by Tenant for Improvement Allowance Items. Except as set forth in the Lease, all Improvement Allowance Items for which the Tenant Improvement Allowance has been made available shall be deemed Landlord's property under the terms of this Lease.

2.3 <u>Application of Tenant Improvement Allowance Toward Rent</u>. Any time after the Tranche I Delivery Date, Tenant shall be entitled to deliver notice (the "**Rent Credit Notice**") to Landlord that Tenant has elected, in its sole and absolute discretion, to apply an amount of the Tenant Improvement Allowance, not to exceed Redacted per rentable square foot, toward the payment of Rent. Upon receipt of the Rent Credit Notice, such amount shall automatically be deducted from the amount of the available Tenant Improvement Allowance, and shall be applied by Landlord toward the next payment of Rent due under the Lease.

SECTION 3

CONSTRUCTION DRAWINGS

Selection of Architect/Construction Drawings. Tenant shall retain the architect/space planner designated by Tenant and reasonably approved by Landlord (the "Architect") to prepare the "Construction Drawings," as that term is defined in this Section 3.1. Landlord hereby approves Gensler as the Architect if selected by Tenant in Tenant's sole and absolute discretion. If Tenant desires to use an architect/space planner other than Gensler, then Landlord shall approve such architect/space planner, such approval not to be unreasonably withheld, conditioned or delayed. Landlord shall approve or reasonably disapprove any architect/space planner proposed by Tenant within five (5) business days and Landlord's failure to respond within such five (5) business day period shall be deemed to be Landlord's approval of the proposed architect/space planner. Notwithstanding the foregoing or anything to the contrary set forth in this Tenant Work Letter, to the extent Tenant is required or elects to engage the services of a historic architect, then Tenant shall retain the historic architect designated by Landlord (the "Historic Architect"). Tenant shall retain Chubb/Redhawk to prepare all plans and engineering working drawings relating to the lifesafety and sprinkler work in the Premises (the "LifeSafety Consultant"), and Tenant shall retain Nabih Youseff to prepare all plans and engineering working drawings relating to the structural work in the Premises (the "Structural Engineer"). Landlord shall have no right to approve any consultants, engineers, subcontractors or laborers retained by Tenant in construction of the Tenant Improvements other than the Architect, the Historic Architect (if any), the Contractor (as defined below), the LifeSafety Consultant, and the Structural Engineer. The plans and drawings to be prepared by Architect and any structural, mechanical, electrical, plumbing, or HVAC consultants retained by Tenant and the LifeSafety Consultant (collectively, the "Engineers") shall be known collectively as the "Construction Drawings." The Construction Drawings may be prepared and submitted to Landlord independently for each Portion of the Premises, and Landlord shall review the Construction Drawings as so submitted. All Construction Drawings shall comply with the drawing format and specifications set forth in the Construction Rules (if any), and shall be subject to Landlord's reasonable approval, which shall not be withheld except in the case of a "Design Problem," as that term is defined below Tenant and Architect shall verify, in the field, the dimensions and conditions as shown on the relevant portions of the Base Building plans, and Tenant and Architect shall be solely responsible for the same, and Landlord shall have no responsibility in connection therewith. Landlord's review of the Construction Drawings as set forth in this Section 3, shall be for its sole purpose and shall not imply Landlord's review of the same, or obligate Landlord to review the same, for quality, design, Code compliance or other like matters. Accordingly, notwithstanding that any Construction Drawings are reviewed by Landlord or its space planner, architect, engineers and consultants, and notwithstanding any advice or assistance which may be rendered to Tenant by Landlord or Landlord's space planner, architect, engineers, and consultants, Landlord shall have no liability whatsoever in connection therewith and shall not be responsible for any omissions or errors contained in the Construction Drawings. A "Design Problem" is defined as, and will be deemed to exist if such Tenant Improvements may (i) affect the exterior appearance of the Building; (ii) materially adversely affect the Building Structure; (iii) materially adversely affect the Building Systems; or (iv) fail to comply with applicable Laws.

3.2 <u>Final Space Plan</u>. Tenant shall supply Landlord with four (4) hard copies of its final space plan for each Portion of the Premises, along with other renderings or illustrations reasonably required by Landlord, to allow Landlord to understand Tenant's design intent, for the Premises before any architectural working drawings or engineering drawings have been commenced, and concurrently with Tenant's delivery of such hard copies, Tenant shall send to Landlord via electronic mail one (1).pdf electronic copy of such final space plan. The final space plan for each Portion of the Premises (each, a "**Final Space Plan**") shall include a layout and designation of all offices, rooms and other partitioning, their intended use, and equipment to be contained therein. Landlord shall not disapprove any portion of the Final Space Plan which (i) is logically consistent with any Final Space Plan previously approved by Landlord for any Portion of the Premises or (ii) is substantially similar to those improvement depicted on <u>Schedule 2</u> attached hereto. Landlord shall advise Tenant within five (5) business days after Landlord's receipt of a Final Space Plan a Design Problem exists with respect to such Final Space Plan, otherwise Landlord shall approve such Final Space Plan. If Landlord fails to timely respond as to whether a Design Problem exists, Tenant may deliver to Landlord a second written request for such approval, which written notice must state in bold and all caps, "FAILURE TO RESPOND TO THIS WRITTEN NOTICE WITHIN THREE (3) BUSINESS DAYS AFTER RECEIPT HEREOF SHALL CONSTITUTE APPROVAL OF THE FINAL SPACE PLAN." If Landlord fails to advise Tenant whether a Design Problem exists with respect to the Final Space Plan." If Landlord fails to advise Tenant whether a Design Problem exists with respect to the Final Space Plan." If Landlord fails to advise Tenant whether a Design Problem exists with respect to the Final Space Plan." If Landlord fails to advise Tenant whether a Design Problem exists with respect to the Final Spac

day period, then Landlord will be deemed to have approved the applicable Final Space Plan. If Landlord advises Tenant that a Design Problem exists with respect to a Final Space Plan, Tenant shall cause the applicable Final Space Plan to be revised to correct such Design Problem. Landlord shall approve any re-submittal of a Final Space Plan within three (3) business days after receipt thereof and Landlord's failure to respond within such three (3) business day period shall be deemed to be Landlord's approval of such Final Space Plan.

3.3 <u>Final Working Drawings</u>. Upon Landlord's approval (or deemed approval) of a Final Space Plan, Tenant shall cause the Architect and the Engineers to complete the architectural and engineering drawings for the applicable Portion of the Premises, and Architect shall compile a fully coordinated set of architectural, structural, mechanical, electrical and plumbing working drawings in a form which is complete to allow subcontractors to bid on the work and to obtain all applicable permits (collectively, the **"Final Working Drawings**") and shall submit the same to Landlord's approval. Tenant shall supply Landlord with four (4) hard copies of the Final Working Drawings, and concurrently with Tenant's delivery of such hard copies, Tenant shall send to Landlord via electronic mail one (1). pdf electronic copy of such Final Working Drawings. Landlord shall not disapprove any portion of the Final Working Drawings which is logically consistent with either (i) the Final Space Plan corresponding to such Final Working Drawings, (ii) the Final Working Drawings for any other Portion of the Premises or (iii) is substantially similar to those improvement depicted on <u>Schedule 2</u> attached hereto. Landlord shall approve such Final Working Drawings, If Landlord fails to timely respond as to whether a Design Problem exists, Tenant may deliver to Landlord a second written request for such approval, which written notice must state in bold and all caps, "FAILURE TO RESPOND TO THIS WRITTEN NOTICE WITHIN THREE (3) BUSINESS DAYS AFTER RECEIPT HEREOF SHALL CONSTITUTE APPROVAL OF THE FINAL WORKING Drawings a Design Problem exists with nespect to any Final Working Drawings. If Landlord advises Tenant whether the Final Working Drawings a Design Problem exists within such three (3) business day period, then Landlord will be deemed to have approved the applicable Final Working Drawings. If Landlord fails to advise Tenant that a Design Problem exists with respect to any Final Working Drawings. If Landlord shall approve any re-submittal of Fina

3.4 <u>Approved Working Drawings</u>. The Final Working Drawings shall be approved by Landlord in accordance with <u>Section 3.3</u> above prior to the commencement of construction of each Portion of the Premises by Tenant. The Final Working Drawings as approved by Landlord for each Portion of the Premises shall be referred to herein as the "**Approved Working Drawings**". Tenant may submit the Final Working Drawings to the appropriate municipal authorities for all applicable building permits prior to Landlord's approval thereof. Tenant hereby agrees that neither Landlord nor Landlord's consultants shall be responsible for obtaining any building permit or certificate of occupancy for the Premises and that obtaining the same shall be Tenant's responsibility; provided, however, that Landlord shall cooperate with Tenant in executing permit applications and performing other ministerial acts reasonably necessary to enable Tenant to obtain any such permit or certificate of occupancy. No changes, modifications or alterations in the Approved Working Drawings ("**Changes**") may be made without the prior written consent of Landlord, which consent may not be unreasonably withheld; provided, that Tenant shall have the right to modify the Approved Working Drawings unducted shall not disapprove any requested Change (a "**Change Request**") to a Portion of the Premises which Landlord has previously approved with respect to another Portion of the Premises (unless such Change Request, if a Design Problem with respect to such Change Request, otherwise Landlord shall advise Tenant within five (5) business days after Landlord's receipt of a Change Request, if a Design Problem exists, Tenant may deliver to Landlord a second written request, which written notice must state in bold and all caps, "FAILURE TO RESPOND TO THIS WRITTEN NOTICE WITHIN THREE (3) BUSINESS DAYS AFTER

RECEIPT HEREOF SHALL CONSTITUTE APPROVAL OF THE CHANGE REQUEST." If Landlord fails to advise Tenant whether a Design Problems exists within such three (3) business day period, then Landlord will be deemed to have approved the applicable Change Request. If Landlord advises Tenant that a Design Problem exists with respect to a Change Request, Tenant shall cause the applicable Change Request to be revised to correct such Design Problem. Landlord shall approve any re-submittal of a Change Request within three (3) business days after receipt thereof and Landlord's failure to respond within such three (3) business day period shall be deemed to be Landlord's approval of such Change request.

SECTION 4

CONSTRUCTION OF THE TENANT IMPROVEMENTS

4.1 Tenant's Selection of Contractors

4.1.1 <u>The Contractor</u>. A general contractor ("**Contractor**") shall be retained by Tenant to construct the Tenant Improvements. Landlord hereby approves any of the general contractors set forth on <u>Schedule 3</u> hereto as the Contractor if selected by Tenant in Tenant's sole and absolute discretion. If Tenant desires to use a general contractor not set forth on <u>Schedule 3</u> hereto, then Landlord shall approve such general contractor, such approval not to be unreasonably withheld, conditioned or delayed. Landlord shall approve or reasonably disapprove any general contractor proposed by Tenant within five (5) business days and Landlords failure to respond within such five (5) business day period shall be deemed to be Landlord's approval of the proposed general contractor.

4.1.2 <u>Tenant's Agents</u>. All subcontractors and laborers used by Tenant together with the Contractor, Architect, and any other consultants retained by Tenant shall be referred to herein collectively, as "Tenant's Agents". Tenant's Agents must be union.

4.2 Construction of Tenant Improvements by Tenant's Agents.

4.2.1 <u>Construction Contract; Cost Budget</u>. Tenant hereby agrees that Tenant's construction contract and general conditions with Contractor (the "**Contract**") shall contain commercially reasonably warranties and indemnifications that inure to Landlord's benefit. With respect to any Tranche, prior to the commencement of the construction of the Tenant Improvements, and after Tenant has accepted all bids for the Tenant Improvements, Tenant shall provide Landlord with a detailed breakdown, by trade, of the final costs to be incurred or which have been incurred, as set forth more particularly in <u>Sections 2.2.1.1</u> through 2.2.1.8, above (the "**Final Costs**"). Prior to the commencement of construction of the Tenant Improvement, Interest forth more particularly in <u>Sections 2.2.1.1</u> through 2.2.1.8, between the amount of the Final Costs and the amount of the Tenant Improvement Allowance (less any portion thereof already disbursed by Landlord, or in the process of being disbursed by Landlord, on or before the commencement of construction of the Tenant Improvements, and subject to the terms of <u>Section 2.1</u> above with respect to the second (2nd) floor restrooms). If there is an Over Allowance Amount, Tenant shall pay a fraction of each amount to be disbursed by Landlord pursuant to the terms of this Tenant Work Letter, the numerator of which fraction shall equal the Over Allowance Amount, and the denominator of which fraction shall equal the Final Costs, and such payment by Tenant shall be a condition to Landlord's obligation to pay any amounts of the Tenant Improvement Allowance. Notwithstanding anything set forth in this Tenant Work Letter to the contrary, construction of the Tenant Improvements is all not commence in any Tranche until Tenant has procured and delivered to Landlord a copy of all applicable building permits for the applicable Tenant Improvements.

4.2.2 Tenant's Agents.

4.2.2.1 <u>Landlord's General Conditions for Tenant's Agents and Tenant Improvement Work</u>. Tenant's and Tenant's Agent's construction of the Tenant Improvements shall comply with the following: (i) the Tenant Improvements shall be constructed in accordance with the Approved Working Drawings (as modified by any Changes); and (ii) Tenant shall abide by Landlord's construction rules a copy of which is attached hereto as <u>Schedule 4</u> (the "Construction Rules"). To the extent of any inconsistency between the terms and conditions set forth in this Lease or this Tenant Work Letter and the Construction Rules, the terms and conditions set forth in this

Lease and this Tenant Work Letter shall control. In addition to any third party review fees payable as provided in <u>Section 2.2.1.1</u>, above, Tenant shall pay a logistical coordination fee (the "**Coordination Fee**") to Landlord in an amount equal to the product of (i) Redacted and (ii) the total number of rentable square feet in the Premises, which Coordination Fee shall be for all services relating to the coordination of the construction of the Tenant Improvements and shall be deducted from the Tenant Improvement Allowance on a per Portion of the Premises basis.

4.2.2.2 <u>Requirements of Tenant's Agents</u>. Contractor shall guarantee to Tenant and for the benefit of Landlord that the Tenant Improvements shall be free from any defects in workmanship and materials for a period of not less than one (1) year from the date of completion thereof. Contractor shall be responsible for the replacement or repair, without additional charge, of all work done or furnished in accordance with its contract that shall be come defective within one (1) year after the completion of the work performed by Contractor. All such warranties or guarantees as to materials or workmanship of or with respect to the Tenant Improvements shall be contained in the Contract on subcontract and shall be written such duarantees or warranties shall incre to the benefit of both Landlord and Tenant, as their respective interests may appear, and can be directly enforced by either. Tenant covenants to give to Landlord any assignment or other assurances which may be necessary to effect such right of direct enforcement.

4.2.2.3 Insurance Requirements

4.2.2.3.1 General Coverages. All of Tenant's Agents shall carry worker's compensation insurance covering all of their respective employees, and shall also carry public liability insurance, including property damage, all with limits, with companies as are required to be carried by Tenant as set forth in this Lease.

4.2.2.3.2 Special Coverages. Tenant shall carry "Builder's All Risk" insurance in commercially reasonable amounts covering the construction of the Tenant Improvements, it being understood and agreed that the Tenant Improvements shall be insured by Tenant pursuant to this Lease immediately upon completion thereof. Such insurance shall be in amounts and shall include such extended coverage endorsements as may be reasonably required by Landlord including, but not limited to, the requirement that all of Tenant's Agents shall carry excess liability and Products and Completed Operation Coverage insurance, each in amounts not less than \$5,000,000 per incident, \$5,000,000 in aggregate, and in form and with companies as are required to be carried by Tenant as set forth in this Lease.

4.2.2.3.3 General Terms. Certificates for all insurance carried pursuant to this Section 4.2.2.3 shall be delivered to Landlord before the commencement of construction of the Tenant Improvements and before the Contractor's equipment is moved onto the site. Tenant's Agents shall endeavor to cause all such policies of insurance must contain a provision that the company writing said policy will give Landlord thirty (30) days prior written notice (ten (10) days in the event of non-payment of premium) of any cancellation or lapse of the effective date or any reduction in the amounts of such insurance. Tenant's Agents shall maintain all of the foregoing insurance coverage in force until the Tenant Improvements are fully completed and accepted by Landlord, except for any Products and Completed Operation Coverage insurance required by Landlord, which is to be maintained for ten (10) years following completion of the work and accepted by Landlord and Tenant. All policies carried under this Section 4.2.2.3 shall insurance, accept Workers' Compensation, maintained by Tenant's Agents shall preclude subrogation claims by the insurer against anyone insurade thereunder. Such insurance shall provide that it is primary insurance as respects the owner and that any other insurance maintained by owner is excess and noncontributing with the insurance required hereunder.

4.2.3 Governmental Compliance. The Tenant Improvements shall comply in all respects with Laws, as may apply according to the rulings of the controlling public official, agent or other person.

4.2.4 Inspection by Landlord. During construction, upon at least one (1) business day's prior notice, Landlord shall have the right to inspect the Tenant Improvements at all reasonable times, provided however, that Landlord's failure to inspect the Tenant Improvements shall in no event constitute a waiver of any of Landlord's rights hereunder nor shall Landlord's inspection of the Tenant Improvements of the Same. Should Landlord disapprove any portion of the Tenant Improvements as not being in accordance with the Approved Working Drawings, Landlord disapproved. Any defects or deviations in, and/or disapproval by Landlord of, the Tenant Improvements shall be rectified by Tenant at no expense to Landlord, provided however, that in the event Landlord determines that a defect or deviation exists or disapproves of any matter in connection with any portion of the Tenant Improvements and such defect, deviation or matter materially adversely affects the mechanical, electrical, plumbing, heating, ventilating and air conditioning or life-safety systems of the Building, the structure or exterior appearance of the Building, following notice to Tenant and Tenant's right to curve the same within the time periods set forth in the Lease, Landlord may, take such action as Landlord deems necessary, at Tenant's expense and without incurring any liability on Landlord's part, to correct any such defect, deviation and/or matter, including, without limitation, causing the cessation of performance of the construction of the Tenant Improvements until such time as the defect, deviation and/or matter is corrected to Landlord's satisfaction.

4.2.5 <u>Meetings</u>. Commencing upon the execution of the Lease, Tenant shall hold weekly meetings at a reasonable time, with the Architect and the Contractor regarding the progress of the preparation of Construction Drawings and the construction of the Tenant Improvements, which meetings shall be held at a location designated by Tenant, and Landlord and/or its agents shall receive prior notice of, and shall have the right to attend, all such meetings. In addition, minutes shall be taken at all such meetings, a copy of which minutes shall be promptly delivered to Landlord.

4.3 Notice of Completion; Record Set of As-Built Drawings; Close-Out Package.

4.3.1 <u>Notice of Completion</u>. Within fifteen (15) days after completion of construction of the Tenant Improvements, Tenant shall cause a Notice of Completion to be recorded in the office of the Recorder of the county in which the Building is located in accordance with Section 8182 of the Civil Code of the State of California or any successor statute, and shall furnish a copy thereof to Landlord upon such recordation. If Tenant fails to do so, Landlord may execute and file the same on behalf of Tenant as Tenant's agent for such purpose, at Tenant's sole cost and expense.

4.3.2 <u>Record Set of As-Built Drawings</u>. At the conclusion of construction of each Portion of the Premises, (i) Tenant shall cause the Architect and Contractor (A) to update the Approved Working Drawings for such Portion of the Premises as necessary to reflect all changes made to the Approved Working Drawings during the course of construction, (B) to certify to the best of their knowledge that the "recordset" of as-built drawings for such Portion of the Premises are true and correct, which certification shall survive the expiration or termination of this Lease, and (C) to deliver to Landlord two (2) sets of hard copies of such record set of drawings, and two (2) CAD discs containing such drawings, within ninety (90) days following issuance of a certificate of occupancy for the Premises, and (ii) Tenant shall deliver to Landlord the Close Out Package.

SECTION 5

MISCELLANEOUS

5.1 <u>Tenant's Representative</u>. Tenant has designated Mark Schrader as its sole representative with respect to the matters set forth in this Tenant Work Letter (whose e-mail address for the purposes of this Tenant Work Letter is mark.schrader@cushwake.com and phone number is (415) 268-1373) (with a copy by e-mail to wleung@salesforce.com), who, until further notice to Landlord, shall have full authority and responsibility to act on behalf of the Tenant as required in this Tenant Work Letter.

5.2 Landlord's Representative. Landlord has designated Dan Wright as its sole representative with respect to the matters set forth in this Tenant Work Letter, who, until further notice to Tenant, shall have full authority and responsibility to act on behalf of the Landlord as required in this Tenant Work Letter.

5.3 <u>Time of the Essence in This Work Letter</u>. Unless otherwise indicated, all references herein to a "number of days" shall mean and refer to calendar days. If any item requiring approval is timely disapproved by Landlord, the procedure for preparation of the document and approval thereof shall be repeated until the document is approved by Landlord.

5.4 <u>Miscellaneous Costs</u>. During the course of construction of the Tenant Improvements and Tenant's initial move into the Premises, Landlord shall provide, and neither Tenant nor Tenant's Agents nor the Contractor or subcontractors retained by Tenant to construct the Tenant Improvements shall be charged for reasonable, non-exclusive use of (i) temporary HVAC during the Building Hours, or (ii) the use of freight elevators, standard Building security (if any), temporary electricity and utilities, and/or loading docks (all as may be provided or made available by Landlord during Building Hours). During construction of the Tenant Improvements, Landlord shall use commercially reasonable efforts to provide Tenant with priority use of the Building's freight elevators during hours other than during Building Hours; provided that Tenant shall give Landlord reasonable prior notice of such intended after-hours use of the freight elevators, and such after-hours use shall be at Tenant's sole cost and expense.

5.5 <u>Delays</u>

5.5.1 <u>Commencement Date Delays</u>. The Commencement Date with respect to each Portion of the Premises shall occur as provided in <u>Section 2.1</u> of the Lease, provided that the Commencement Date with respect to each Portion of the Premises shall be extended by the number of days of "Commencement Date Delay," as that term is defined, below. As used herein, the term "**Commencement Date Delay**" shall mean only a "Force Majeure Delay" or a "Landlord Caused Delay," as those terms are defined below in this <u>Section 2.5.1</u> of this Tenant Work Letter. As used herein, the term "**Force Majeure Delay**" shall mean only an actual delay resulting from an event of Force Majeure, as that term is defined in <u>Section 29.15</u> of the Lease. As used in this Tenant Work Letter, "**Landlord Caused Delay**" shall mean actual delays to the extent resulting from (i) the failure of Landlord to timely approve or disapprove any Construction Drawings; (ii) delays due to the acts or failures to act of Landlord Parties with respect to payment of the Tenant Improvement Allowance (except as otherwise allowed under this Tenant Work Letter); (iii) any other act or omission of Landlord or any Landlord Party in violation of the Lease or this Tenant Work Letter which actually delays in construction of the Tenant Improvements due to failure of the Base Building to be constructed in accordance with applicable Laws.

5.5.2 Determination of Lease Commencement Date Delay. If Tenant contends that a Commencement Date Delay has occurred, Tenant shall notify Landlord in writing of (i) the event which constitutes such Commencement Date Delay and (ii) the date upon which such Commencement Date Delay is anticipated to end. If such actions, inaction or circumstance described in the Notice set forth in (i) above of this Section 5.5.2 of this Tenant Work Letter (the "**Delay Notice**") are not curred by Landlord within one (1) business day of Landlord's receipt of the Delay Notice and if such action, inaction or circumstance otherwise qualify as a Commencement Date Delay, then a Commencement Date Delay shall be deemed to have occurred commencing as of the date of Landlord's receipt of the Delay Notice and ending as of the date such delay ends.

SCHEDULE 1 TO EXHIBIT C

LANDLORD'S WORK

Landlord shall, at Landlord's sole cost and expense, construct, supply, furnish, and/or install the following items and work:

1. <u>Restrooms</u>. Upgrade the restrooms on the multi-tenant floors of the Premises (i.e., the ground floor, the second (2nd) floor, and the fourth (4th) floor) to be in compliance with applicable Laws to the extent necessary for Landlord to pull any required construction permits or for Tenant to legally occupy such Portion of the Premises for the Permitted Use.

2. Core. Update the existing Building shafts and base building rooms to be in compliance with applicable Laws to the extent necessary for Landlord to pull any required construction permits or for Tenant to legally occupy the Premises for the Permitted Use.

3. <u>Fire Smoke Dampers</u>. To the extent required by applicable Laws, upgrade existing electrical FSDs to pneumatic controls. To the extent Landlord desires to remove an existing FSD that is no longer required by applicable Laws, Landlord shall complete all shaft construction necessary to fill the hole. Alternatively, Landlord may elect to fix close all existing dampers in the walls that are not required to be rated.

APPROVED IMPROVEMENTS

SCHEDULE 2 TO EXHIBIT C

SCHEDULE 3 TO EXHIBIT C

APPROVED CONTRACTORS

SCHEDULE 4 TO EXHIBIT C

CONSTRUCTION RULES

CONSTRUCTION RULES & REGULATIONS

Rincon Center is a mixed use Project. As such it is comprised of office, retal and residential tenants. Rincon Center is also accessible to the public. It is expressly understood that construction shall at no time interfere with the building operations or the use, enjoyment and occupancy of any portion of the Project by any tenant, resident, employee, visitor, etc. (Users). Our goal is to minimize inconvenience to the Users while construction is in progress. Your full cooperation is required in adhering to the following procedures while performing construction work at this Project.

Building management reserves the right at any time to amend these rules and to make future rules required for the safety, protection and maintenance of the Project, the operation and preservation of the good order thereof, and the protection and comfort of the Users. Repeated and flagmant violations of these Construction Rules and Regulations will result in Contractors being baunde from the building and job site. If you have any questions, please contact the Building Management.

General contractors, subcontractors, suppliers, material men, and their employees and anyone working for or on their behalf (Contractors), shall be immediately advised of the following building rules and regulations: concerning their proper conduct within the Project. It is the Contractors' represensibility to exame that it has reseal and understands these rules and regulations. Ignorance of these rules and regulations is not a waiver of liability or responsibility.

GENERAL

All work to be performed shall be approved by Building Management prior to commencement of the job. All Contractors shall employ union labor.

All work shall comply with all applicable federal, state and local codes and ordinances Contractors shall obtain and pay for all required fees, permits and inspections.

Contractors shall furnish to Building Management the following items prior to the commencement of any work: • Copies of all permits and permit drawings • Project schedule • Lust of all subcontractors • Certificates of Insurance for all contractors/subcontractors • Emergency telephone list for all contractors/subcontractors

Contractors shall be responsible for job site security including their materials, equipment and tools and that of their subcontractors. All gang boxes, toolboxes and other containers are subject to impection when moved in or out of the Project.

No Project systems shall be turned off, disengaged, or otherwise affected by Contractors without the approval and supervision of Building Management. Said systems shall include but shall not

be limited to automatic sprinklers, electrical circuits, lighting, emergency lighting, life safety systems, smoke detectors and dometric water supply. All penetrations of concrete tabs thall be subject to x-ray prior to coring. Locations must be approved in advance by Building Management.

Contractors shall ensure that all corridors, exits, elevator lobbies and telephone/electrical closets are not cluttered or blocked. Contractors shall provide walk-off mats at the exit from the construction area and entrance to the elevators.

Doors to all work areas, including mechanical and electrical closets, shall remain closed at all times.

Contractors shall be responsible for the protection of their work and all areas adjacent to their work. Contractors shall maintain the work area in a safe and sanitary condition at all times. No radius or other audio devices are permitted.

BUILDING ACCESS

- Contractors will be responsible for notifying the Building Management, in advance, of all work to be performed in the Project and scheduling access of subcontractors. Contractors will complete "Building Access Request" forms as needed to fulfill this requirement. Contractors who are not placed on the Building Access Schedule will not be permitted access to the Project.
- 2. All workers will be required to sign-in at the Security Console and obtain a visitors badge. A visitors badge shall be worn at all times while working in the Project.

BEFORE CONSTRUCTION BEGINS

- The General Contractor shall meet with the Building Management to review the construction schedule, building access, special requirements and the Construction Rules and Regulations.
- 2 The General Contractor will provide a list of all subcontractors assigned to the job including company name, contact person, and license number, to Building Management for review and approval at least five (5) building Management at least five (5) building Management and a least five (5) building Management and a start five (5) building Management and the start five (5) building the start five (5) building Management and the start five (5) building the start five (5) building Management and the start five (5) building Management (5
- Contractors are required to provide a certificate of insurance and additional insured endorsement to Building Management prior to starting any work in the Project.

- Contractors shall not park their vehicles in the loading dock except for loading and unloading purposes only.
- 5. Towenty-four (24) hour notice must be given to the Building Management for any fire or life safety testing or taking the system off line. Contractors will be responsible for cost and fees of supervision by Building Management. Contractors thall immediately notify the Building Management when testing is complete or the system is ready to be placed on line. If the life safety system is to be off line overnight. Contractors thall be responsible for providing trained personnel for "fire watch".

6. No radios are allowed on the job site.

- No tobacco chewing, smoking or sunflower seeds are permitted on the job site or in the Project.
- 8. The Building Management will furnish keys to the electrical and mechanical rooms upon request. All keys will be signed out at the beginning of each work shift and signed back in at the end of each work shift, unless otherwise agreed to in advance by Building Management.

9. Normal building hours are:

Monday - Friday 6:00 am. - 6:00 pm. Saturday/Sunday 9:00 am. - 6:00 pm.

Any work that creates excessive noise or disruption shall be scheduled at the following times:

One Rincon One: Monday - Friday 6:00 pm. - 6:00 am. Saturday/Sunday 6:00 pm. - 9:00 am.

Note: The Atrium in Rincon One is frequently used for banquets and special functions on nights and weekends. Contractors access will be adjusted accordingly. Rincon Two:

Monday - Friday 7:00 ani. - \$:00 ani. / 6:00 pm - \$:00 pm. Saturday/Sunday 7:00 ani. - 8:00 pm.

Any work that results in excessive noise (i.e. core dralling and/or roto hammering), odors or dust must be completed during these botrs (subject to the operating hours of Project Users these boors may be adjusted as needed to avoid disruptions). All after hours work must be coordinated with the Building Management twenty four (24) hours prior to performing asid work. Contractors needing to work on weekends will provide the Building Management with a list of contractors to bedould to work. This list should

include the number of employees, the company, and the estimated hours the contractors will be working.

- Contractors shall not secure, enable, test, start or open any life safety, mechanical, electrical, domentic water, condenser water, and chilled water or hot water systems without prior approval of Building Management. Authorization will be given on a caseby-case basis.
- 12. Contractors must provide Material Safety Data Sheets to the Building Management for every substance being used on the construction site prior to the start of construction. Contractor shall also maintain these sheets at the job site.
- Contractors are required to provide and make available fire extinguishers on the job site during construction.
- 14. Any Contractors found guilty of rudeness, use of profamity, or lack of courtesy to a building tenant, visitor, or employee will be immediately ejected from the Project and will not be allowed to return.
- 15. Graffiti or vandalism will not be tolerated. Any Contractors caught in the set shall be immediately removed from the Project and will not be allowed to return. Any expenses associated with the removal or repair resulting from the graffiti or vandalism will be at the Contractors expense.
- 16. Rubber or polyurethane wheels are required on all material handling equipment transporting materials across carpet, granite, matble or stone surfaces. Floor protection, such as Masonite, is required to protect flooring surfaces.
- 17. Clothing shall be appropriate for countruction trade involved, i.e. no shorts, sandals, which would be unask for the employee. Clothing containing words, symbols or other forms of communication considered offensive or in bid taste by Building Management shall not be allowed on the Project. Proper skety enginement shall be required as determined by the contractor, i.e. safety glasses, goggles, respirators, etc.

DEMOLITION

- Demolition shall be done during non business hours and care shall be taken not to disturb the Users of the Project. Demolition debris shall be removed using the designated freight elevator and taken out of the Project through the loading dock.
- Prior to the start of construction, Contractors shall cover light fixtures and window blinds with plastic bags. During demoliton, Contractors shall use plastic or other similar protective coverings to keep dat and duru containsel in the areas being demolished. All portions of the Project on the travel soute for debris removal shall be protected, including floors, wills, doors and doorgambe.

3. All common areas of the Project, including doors, frames, walls, floors, ceilings, plants and artwork shall be covered with the proper protective materials during demolition and construction and returned to original condition prior to the start of each work day (7:00 AM), unless agreed to in advance.

MATERIAL STOCKING

- Material deliveries shall be accomplished before business hours. The Building Management shall be notified in advance of such material stocking.
- Material shall be transported in the designated freight elevator. Contractors are responsible to protect the doorjands, floors, ceilings and thresholds in the common areas. If materials are transported by use of stativells all appropriate wall and floor surfaces, railings, door frames and doors shall be protected.
- 3. Trucks may park in the loading dock only long enough to unload materials.
- 4. Delivery of construction materials to the job site shall be made through the loading dock and not through the ground floor lobby and main entrance. If this is not feasible, permission to deviate from this rule shall be obtained from Building Management.

PARKING

There is no free parking available at Rincon Center. All contractors wishing to park in the garage shall pay the standard parking rates. All parking arrangements shall be made with the Building Management, Chief Engineer, and Garage Supervisor.

LOADING DOCK

- LOADING DOCK 1. The loading dock is available for deliveries the following times: Monday-Friday 6:00 am. -11:00 am. 2:00 pm. -4:00 pm.
- All deliveries are limited to twenty (20) minutes. Deliveries needing longer than 20
 minutes must be scheduled after hours with the Building Management, Chief Engineer
 and the Garage Supervisor.
- 3. The height limit of the loading dock is 12'0".

ELEVATORS

Use of the freight elevators on "independent" for tool or material delivery shall be scheduled with Building Management. In the event the freight elevators(s) require padding and protections, the Contractors shall schedule the same with Building Management. Contractor must contact Building Management at least 24 hours ahead of time.

The dimensions and capacities of the freight elevators follow:

One Rincon Center: Cab: 65" x 80" x 112" Door Opening: 41" x 102" Capacity: 3,500 Ibs.

Two Rincon Center: Cab: 65" x 80" x 112" Door Opening: 41" x 102" Capacity: 3,500 Ibs

CONSTRUCTION

- No one shall be allowed to endanger the Project, its premises, or its Users in any manner whattoever. In the event that a similation occurs which threatens the Project or its Users in any manner, Contractors must take tepts to correct the hazardous condition, in the event that the Contractors personnel fail to correct the hazardous condition, the Building Management reserves the right to immediately take steps to correct the situation at Contractors expense.
- No gasoline operated devises, i.e.: concrete saws, coring machines, welding machines, etc., shall be permitted within the building premises. All work requiring such devises shall be electrically operated.
- All gas and oxygen canisters shall be properly chained and supported to eliminate all potential hazards. 3.
- 4. Contractors are responsible for turning off the lights and securing the premises.
- 5. All doors between the premises under construction and the Project common areas must be closed at all times except for ingress and egress purposes. Contractors must hang temporary doors in door cuts into corridors as soon as these door cuts are made.
- temportry doors in door cats mice certaiors as soon as mere door cats are mose. Contractors are responsible to keep the Project clean and tidy and including daily cleamp. Throughout the course of the day, corridors, elevators, elevator lobbies and electrical and jamicrail closets cannot be cluttered or blocked. Where dust and debits have fallen in Project common areas or construction splits have occurred Centractors must immediately clean the area. Failure to keep the job site, Project, public areas, building and exterior areas clean will result in a back charge, including a fee for managing the clean up work and processing the appropriate paper work. Contractors shall provide a mog, mog bucket, broom and vacuum on the job site to be immediately used to remove dirt and debits in the common areas. 6.
- Contractors shall place a carpet remnant immediately inside all exits from the premises under construction for construction workers to use to wipe their shoes every time they exit the job site.

- 8. If any windows are removed for a debris chute or other purposes they shall be replaced if rain is a possibility and the rain can enter the building through this opening. Particular attention must be given to the possibility of rain at night or on weekends.
- Contractors shall use freight elevators. Passenger elevators shall not be used for construction purposes.
- CLEAN UP
- Contractors shall provide a construction dumpster at the job site in a location to be designated by Building Management. Contractors shall monitor and clean the area around the dumpster periodically during the workshy and at the end of the workshy. Particular care must be taken so that no nails or other sharp objects are left on the area surrounding the dumpster or on the travel path to the dumpster.
- The job site, Project and building common areas must be kept clean and tidy at all times. Contractors must remove all trash and related items when they are finished with their jobs.
- 3. Restrooms are provided only for convenience and it is a privilege for construction workers to use the restroom. Cantractors are responsible for its employees and subcontractors two of the restrooms. Failure to keep the restrooms in the condition provided by the Building Management to terainst and to clean up data in Building Management performing clean up and Centractors will be back charged for these costs. Repeated violations of this rule will result in the revocation of this privilege.
- 4. The cleaning of brushes and other tools may only occur in the slop sink in the janitor's closet. Contractors shall keep this room in a tidy and clean condition and clean up after each use.
- 5. Trash from lunches must be removed from the work area each day.
- Unused construction material may be stored within the space under construction, but only if done so in an orderly manner.
- All work performed in occupied tenant spaces must be cleaned by Contractors prior to their leaving the job or at the end of the business day. If Building Management is required to perform additional cleanup (initial and/or follow-up), it will be done at Contractors expense.

PUNCH LIST

 If the tenant has occupied the premises, the Contractors highest priority is to be courteous to tenant's employees, respect their privacy and disrupt their work as little as possible.

- Any material, paint or other product used in completing the punch list shall not be stored in the premises once the tenant has accepted possession of the premises.
- 3. Protective coverings shall be used to cover the improvements and tenant's personal property as necessary. Drop cloths shall be used when any sanding of thest rock or painting is being done. All sheet rock dust shall be immediately wiped off the walls and base and vacuamed off the carpeting.
- 4. Contractors work schedule to complete the peach list shall be arranged to least durupt the teaant's business operations and must be approved in advance by the teaant and Buildang Management. All activities creating excessive noise and dust must be done after normal business howr.
- The General Contractor shall cause the completion of the punch list to occur as quickly as possible. Any unnecessary delays may result in the Building Management completing the punch list stems and back charging the General Contractor.

Rincon Center LEED Requirements:

The following items should have no added cost to the project and the architects, contractors, and subcontractors should be aware of when bidding. The following should be included in ALL Building Construction Guidelines.

- Adhesives and sealants must have VOC content below SCAQMD Rule #1168 or BAAQMD Reg 8, Rule 51.
 Paints and coatings must meet or exceed the VOC and chemical component limits of Green Seal GS-11 requirements.
 Non-carpet flooring is Floor Score-certified and constitutes a minimum of 25% of the finitided floor area.
 Carpet meets the requirements of Carpet and Rug Institute (CRJ) Green Label Plus Indoor Are Quality Test Program.
 Carpet cuckion meets the requirements of the CRI Green Label Testing Program.
 Composite panels and agnifiber products must contain no added tree-formaldehyde result.

- resins. o 75% of construction waste (exclude FF&E, mechanical, and plumbing equipment) diverted from landfill.

The items below are related to different building materials (for example, wall studs, insulation, doors, drywall and ceiling panels) and could potentially have an impact on cost. The following should be included as add alternates (or deducts) when pricing.

- Contain at least 10% post-consumer or 20% post-industrial material.
 Contain at least 70% material ashraped from off-site or outside the organization.
 Contain at least 70% material subraged from co-site, through an internal organizational materials and equipment reuse program.
 Contain at least 50% rapidly renewable materials.

 Contain at least 50% materials harvested and processed or extracted and processed within
 S00 males of the project.
 Consist of at least 50% FSC-Certified wood. Rincon Center Construction Rules and Regulations Page 9

<u>EXHIBIT D</u>

NOTICE OF LEASE TERM DATES

Re: Office Lease dated , 20 (the "Lease"), by and between , a ("Landlord"), and , a ("Tenant"), for approximately rentable square feet of space (the "Premises"), located on the [() floors] and [suites ()] of that certain office building located at , , (the "Building").

Gentlemen:

In accordance with the Office Lease (the "Lease"), we wish to advise you and/or confirm as follows:

1. Tenant has accepted [floors () through ()] and/or [suites ()] of the Building.

2. The Lease Term with respect to [floors () through ()] and/or [suites ()] of the Building shall commence on or has commenced on (the "Applicable Lease Commencement Date") for a term of ending on .

 $\label{eq:constraint} 3. \quad \mbox{Rent with respect to [floors () through ()] and/or [suites ()] of the Building commenced on \ , in the amount of \ . } \\$

4. If the Applicable Lease Commencement Date is other than the first day of the month, the first billing will contain a pro rata adjustment. Each billing thereafter, with the exception of the final billing, shall be for the full amount of the monthly installment as provided for in the Lease.

5. Capitalized terms used herein that are defined in the Lease shall have the same meaning when used herein.

If the provisions of this letter correctly set forth our understanding, please so acknowledge by signing at the place provided below on the enclosed copy of this letter and returning the same to Landlord.

"Landlord":

a

,

By: Name: Its:

Agreed to and Accepted as of , 20.

"Tenant":

, а

By: Name: Its:

<u>EXHIBIT E</u>

ROOF DECK

EXHIBIT F

JANITORIAL SPECIFICATIONS

HUDSON RINCON CENTER, LLC SERVICE AGREEMENT

EXHIBIT A CONTRACT DUTIES

GROUND FLOOR LOBBIES

A. NIGHTLY

1. UNCARPETED FLOORS Hard-surfaced floors are to be dust mopped, using a treated mop to remove all loces dift and then damp mopped with a pH neutral cleaning solution if needed. The floor, when dry, will be even in appearance and show no streaking from cleaning efforts. If rain mats are in use because of inclement weather, they will be vacuumed and spot cleaned as necessary.

2. CARPETED FLOORS Carpoted floors will be vace spot cleaned as necessary. umed daily, including ance each Sunday, and

3. WALLS AND DOORS All doors, door saddles, and jambs will be thoroughly cleaned as necessary to remove all finger-marks, smudges and spils. Stone valls are to be dust wipod with micro-fiber cleth and if needed cleaned with a pH neutral cleaning solution. Wipe wood panel walls with a treated dust clob.

LOBEY GLASS All glass windows, doors and directory board glass will be cleaned with a non-abrasive glass cleaner and left in a bright condition free of streaks and dust.

6. <u>ELEVATOR CAB DOORS, TRACKS, FLOORS AND THRESHOLDS</u> Elevator doors and wood panel cab walls will be damp-wiped and left in a clean condition free of all dust and streaks. Elevator saddles will be wiped clean and all dilt and debris removed from door tracks, using a vacuum crevice attachment. Spills and smudges will be cleaned so that the saddles and tracks are left in a bright, clean condition. All bright metal work will be polished with appropriate materials, including the use of a buffing tool and polishing compounds as required. Elevator floors are to be dust mopped, using a treated mop to remove all loose dit and then, if needed damp mopped with a pin entrat cleaning solution. Stone elevator walls are to be dust wiped with micro-fiber cloth and if needed cleaned

-12-

CONTRACTOR INITIAL

with pH neutral cleaning solution. If rain mats are in place because of inclement weather, mats will be vacuumed.

<u>DUSTING</u> All horizontal surfaces, including furniture tops and ledges within reach, will be dusted using treated cleaners appropriate for each surface. No feather dusters will be allowed.

7. <u>MISCELLANEOUS</u> Signs, sign standards and security podium will be wiped clean and left free of finger-marks and smudges. Bright metal work will be polished with appropriate metal cleaners. Spot clean all interior architectural metal finishes and granite wall surfaces.

8. <u>PLAZA</u> Police public area planters and plaza, removing any trash and debris that may accumulate.

B. WEEKLY

<u>WINDOWS</u>
Window frames (interior and exterior) will be dusted and wiped down on both horizontal
and vertical surfaces to as 8' height, removing all dust and spots.

C. MONTHLY

1. HIGH DUSTING All high dusting beyond the reach of normal day to day dusting will be done monthly. 2. <u>RAIN MATS</u> Rain Mats will be thoroughly steamed cleaned using extraction system monthly during the rainy season, and rolled and stored after inclement weather.

3. ARCHITECTURAL ALUMINUM FINISHES Thoroughly clean all chrome and architectural aluminum fixtures.

WINDOW FRAMES Thoroughly clean all window frames, vertical and horizontal surfaces, with the appropriate cleaning solution, wipe dry, leaving a spot free surface.

E. ANNUALLY

1. <u>RAIN MATS</u> Roln Mats will be thoroughly stearned cleaned using an extraction system after the rainy season ends but in no event later than May 31, each year.

-13-

CONTRACTOR Initials: 97-

III. PUBLIC AREAS ABOVE GROUND FLOOR

A. NIGHTLY

1. <u>CARPETED FLOORS</u> Hallways, corridors and public areas are to be thoroughly vacuumed on a nightly basis.

<u>UNCARPETED FLOORS</u> All hard-surfaced floors are to be mopped with a treated dust mop and maintained as needed to preserve a uniformly clean appearance. Edges and corners are to be treated on an as needed basis.

3. <u>WALLS</u> Walls will be spot-cleaned to remove all smudges, stains and hand marks, on an as needed basis using only clean water or mild cleaning solution. If a cleaning solution is used, the wall will be rinsed with clear water. No abrasive cleaners of any kind or cleaning solutions containing bleach will be used at any time without first obtaining permission from the Manager.

4. <u>DOORS AND JANBS</u> Doors and jambs will be spot-cleaned to remove any hand marks, stains, spills or smudges on an as needed basis. Rinse with clear water and dry. When completed, Joors and jambs shall have a uniform clean appearance. Dust ard remove debris from all metal door threshholds.

5. MISCELLANEOUS GLASS AND METAL WORK All glass and metal accessories, including signs, door hardware, frames, etc. will be wiped clean and left in a uniformly clean condition.

<u>DUSTING</u> Dust furniture, accessories, ledges and other horizontal surfaces using a treated dust coth. No feather duster will be allowed. Spot cleaning will be completed as necessary.

7. PLANTERS Police all public area planers; remove any trash and debris that may accumulate.

-14-

8. DRINKING FOUNTAINS Clean and sanitize nightly wipe dry leaving no spots.

9. WALK-OFF MATS Service all walk-off mats as necessary.

CONTRACTOR Initials:

B. WEEKLY

1. <u>CARPETED FLOORS</u> All carpeted floors will be vacuumed wall to wall on a weekly basis.

UNCARPETED FLOORS
 All uncarpeted floors are to be swept clean and thoroughly wet mopped to
 maintain a uniform clean appearance. Special attention will be given to
 insure the edges, comers, small niches and areas behind doors are
 cleaned as well.

C. MONTHLY

1. <u>CARPETED FLOORS</u> All carpeted floor edges will be vacuumed using an edging attachment. Care should be taken when vacuuming area rugs to avoid damaging the edges.

D. QUARTERLY

1. UNCARPETED FLOORS All hard surfaced floors will be wet mopped, allowed to dry and then machine spray

E. SEMI-ANNUALLY

1. UNCARPETED FLOORS All hard surfaced facers are to be specially treated where year, at intervals with the quarterly irotament described above at the first interval the floors are to be scrubbed and waxed and left in a unform bright condition. All finish spills and splatshes will be forors are to be stripped and waxed and left in a uniform bright condition. All finish spills and splatshes will be completely memory dimensionarily and splatshes.

and spearses and a spearse and

F. ANNUALLY

1. <u>CEILING LIGHTING</u> Clean light diffusers; remove fingerprints from fixtures, ceiling and grid.

-15-

CONTRACTOR Initials:

IV. RESTROOMS

A. NIGHTLY

1. FLOORS, BRIGHT-WORK, AND METAL FIXTURES Floors will be swept clean and wet-mopped using a germicidal detargent containing no deodcrants. All watermarks and stains will be wiped from walls, partitions, light switches, and metal fixtures. All bright-work, including mirrors, will be cleaned or polished using only non-abrasive/non-actic cleaning materials.

2. CERAMIC FIXTURES Scour, wash and disinfect all basins, shower stalls, toilet bowls and urinals with a germicidal detargent solution free of any deodorants, including marbio and file walls near the urinals. Special care will be taken to insure that areas difficult to access, such as the underside of toilet bowls and urinals, will be cleaned to prevent the building up of calcium and iron oxide deposits. Wash both sides of all toilet seats with germicidal solution free of any deodorants. No abrasive or acidic cleaning materials will be used. Leave all surfaces spxt free.

3. URINAL MODESTY SCREENS and TOILET PARTITIONS Damp-wipe urinal modesty screens with germicidal solution free of any deodorants. Surfaces are to dry with a uniform appearance, free of any streaks or smudges. No abrasive or acidic cleaning materials will be used. Leave all surfaces spot free.

4. <u>RESTROOM DOORS</u> All restroom doors will be damp-wiped to remove any hand marks from door and door hardware. No abrasive or acidic cleaning materials will be used. Leave all surfaces spot free.

5. <u>GENERAL</u> It is the intention of these specifications to keep restrooms thoroughly clean and not to use disinfectant to mask coors. All disinfectants will be deedorant free.

6. <u>COMPOST REMOVAL</u> Remove all waste paper from compost receptacles to designated areas. All compost receptacles are to be inbroughly cleaned and wasted and compost liners installed when solled. All compost liners shall conform to Landlord's composing program.

-16-

7. TRASH REMOVAL

CONTRACTOR Initials

Remove all refuse including solied sanitary napkins from sanitary waste receptacles to designated areas. All sasitary waste receptacles are to be thoroughly cleaned and washed and new liners isstalled. All liners shall conform to Landlord's waste program.

8. PRODUCT DISPENSERS Fill tollet tissue and paper towel dispensers and holders, seat cover containers, soap dispensers and sanitary napkin machines with Owner's stock nightly. Contractor shall not leave extra or partially used rolls of toilet paper in restrooms. Care should be taken to inspect dispensing fixtures to insure they are operating properly. Report any deficiencies to the Manager.

9. FLOOR DRAINS Add one cup of wate: to all restroom floor drains.

B. WEEKLY

<u>DUSTING</u> Dust the top edges of partitions, ledges, mirrors, HVAC diffusers, and return air grills and other horizontal surfaces, including vents at bottom of walls.

2. <u>EXPOSED PLUMBING</u> Damp wipe all exposed plumbing (P-traps under sinks) leaving dust free

<u>PRODUCT DISPENSERS</u> Collect coins from sanitary napkin and tampon machines and deliver proceeds wrapped in coin rolls to Owner's representative.

D. QUARTERLY

1. FLOORS Thoroughly clean and reseal all ceramic tile floors using approved sealers.

V. TENANT AREAS

A. NIGHTLY

 <u>CARPETED FLOCRS</u> All hallways, conference rooms, corridors and high traffic areas to be vacuumed on a nishty basis. All edges will be swept to remove dust and then vacuumed. Offices and cubicles will be vacuumed on an as needed -17-

CONTRACTOR Initials

basis, moving all light furniture such as chairs and rolling cabinets if necessary. All furniture will be returned to its original position. Vacuum under desks as needad. Spot clean as necessary.

2. UNCARPETED FLOORS All hard-surfaced floors will be dust-mopped using a treated dust mop, moving all light furniture. All furniture will be returned to its original position. Spot-clean as needed to remove spill and smudges with damp cloth or mop and return to uniformly clean appearance.

3. <u>DUSTING</u> Using a freated dust cloth, wipe all furniture tops. Wipe clean telephones, lamps, and other accessories. Papers, files, etc. left on desk or furniture tops <u>shail not be moved</u> to allow for dusting. Dust only those areas which are clear of papers and files.

4. <u>FURNITURE AND ACCESSORIES</u> On an as needed basis vipe file cabinets, kelephones, furniture and accessories with a water dampened doth tr remove spille, enrudges and streaks. Use a mild detargent cleaning solution only if necessary. On an as needed basis sankize all telephone receivers. Roturn chairs and wateb baskets to their proper positions.

5. <u>DOORS, JAMBS AND WALLS</u> On an as needed basis, ill doors, jambs, walls and window mullions and glass partitions will be spot-deniated to nervoue sheaks, smudges, hand marks and spalls. Give particular attention to areas such as doors, jambs and windows where it is reasonable to export hand marks will be presert. Dust and nervow debris from all metal door thresholds.

6. TRASH REMOVAL AND TRASH LINERS All trash from wastexaskets and trash barrels or other trash, which is identified as such, by signs or notices, will be removed from the premises and deposited in the designated areas for trash. Trash liners will be replaced as necessary but in no event less than weekly. Clean and sanitize trash, recycling and compost containers as required. Owner's composting and recycling program shall be adhered to and supported at all times.

<u>RECYCLING</u> All recyclables from recycling containers or other recycling, which is identified as such, by signs or notices, will be removed from the premises and deposited in the designated areas for recycling. This includes recycling containers located at tenant employee desks as required by Owner's desk-aller recycling program.

8. COMPOSTING

-18-

CONTRACTOR Initials:

All compostables from compost containers typically located at conference rooms, kitchens, kitchenettes, lunch rooms, break rooms and coffee bars will be removed from the premises and deposited in the designated areas for composting. Compost liners will be replaced nightly.

9. DRINKING FOUNTAINS Clean, sanitize and polish drinking fountains.

B. WEEKLY

1. FURNITURE Wipe with treated dust cloth horizontal surfaces of furniture and using the appropriate polishing materials and equipment polish to a uniform shiny appearance. Wipe with a treated dust cloth all sides of furniture and legs on furniture on an as needed basis. Wipe all remaining surfaces, including window sills, which are not dusted during the nightly dusting. Dust all vinyl base.

2. <u>CARPETED FLOORS</u> All carpeted floors will be vacuumed on a weekly basis.

UNCARPETED FLOORS On a weekly basis floors, and in addition to the nightly service, uncarpeted floors are to be given special attention to insure that edges, corners and small niches and areas behind doors are cleaned as well. Floor should dry free of any streaks or smudges. Dust all virity base.

4. THRESHOLDS Clean and polish all metal door thresholds.

<u>DUSTING</u> Dust all horizontal surfaces including horizontal window blinds, windowsills, air conditioner return grilles, tops of partitions and picture frames.

C. MONTHLY

<u>HIGH DUSTING</u> All hoticontal surfaces on furriture, ledges, wainscot, picture frames, wall hangings, etc., that are beyond the neach of normal rightly dusting, will be dusted using a treated dust cloth, All ceiling vents, vents located high on the walls or in ceilings, and light fixtures will be dusted using a treated dust cloth or mop.

-19-

2. <u>VACUUM</u> Vacuum all upholstered furniture.

3. CARPETED FLOORS All carpet edges will be vacuumed with an edging attachment. All carpeted floor areas that are not accessible with standard floor vacuums, but are easily visible will be vacuumed with portable vacuums. For example, desk wells, areas around planters and spaces between furniture. Move all plastic carpet protectors and thoroughly vacuum under and around all desk furniture.

4. WASTE RECEPTACLES Wash all waste receptacles.

D. QUARTERLY

<u>UNCARPETED FLOORS</u> Shower-scrub and machine spray buff, or otherwise recondition all resilient or composition flooring to provide a level of appearance equivalent to a completely refinished floor.

2. PLASTIC CHAIR PROTECTORS Wash all plastic chair pads.

<u>HORIZONTAL WINDOW BLINDS</u> All horizontal window blinds shall be dusted on both sides with a treated dusting tool.

E. ANNUALLY

- 1. <u>DUSTING</u> Dust ceiling surfaces other than acoustical ceiling material.
- 2. <u>HORIZONTAL WINDOW BLINDS</u> Wipe down all vertical blirds at exterior windows as recommended by manufacturer.

VI. BASEMENT AREAS ELEVATOR LOBBYS ONLY (FREIGHT AND PUBLIC)

-20-

A. NIGHTLY

1. <u>UNCARPETED FLOORS</u> All public areas are to be swept using sweeping compound.

2. WALLS, DOORS AND JAMBS

CONTRACTOR Initials:

On an as needed basis, all walls, doors and jambs will be spot-cleaned to remove all hand marks, smudges, streaks and spills.

B. WEEKLY

1. <u>UNCARPETED FLOORS</u> Shower-scrub and machine spray buff, or otherwise recondition all resilient or composition flooring to provide a level of appearance equivalent to a completely refinished floor.

<u>DUSTING</u> All horizontal surfaces, including equipment and furniture, will be dusted with a treated dust cloth.

VII. BUILDING EXTERIOR

A. NIGHTLY

1. SIDEWALKS AND RELATED AREAS Inspections of sidewalks and related areas will be made daily and any spills will be cleaned with a sponge and appropriate cleaning material.

STREET GUTTERS (MARKET AND FREMONT STREETS) Inspect gutters and remove large pieces of trash, broken glass, nails and othor debris.

EXTERIOR FACADE Inspect building's exterior facade for graffiti, spills, smudges and, if found, clean with appropriate materials. Any spill, smudge or graffiti that cannot be cleaned thoroughly shall be reported to the Manager.

B. WEEKLY

1. <u>NEWS VENDING MACHINES AND MAILBOXES</u> News vending machines located around the building in various locations will have the tops and sides wiped clean with a mild detergent. Additionally, vending machines will be aligned with each other, relative to the curb, in an orderly manner.

2. STREET GUTTERS (MARKET AND FREMONT STREETS) -21-

CONTRACTOR Initials:

In addition to removing large pieces of trash or other debris, sweep gutters so that accumulated dirt and small debris is removed and deposited in the building's designated trash areas.

3. <u>SIDEWALKS</u> Machine scrub or steam clean all exterior sidewalks and garage entrances weekly. Contractor can phase work throughout the week provided that sidewalks maintain a uniformly clean appearance throughout the week. Contractor to take care to remove all water and residue on exterior surfaces of building effor washing is completed.

C. MONTHLY

1. <u>WINDOW FRAMES, LEDGES</u> Sweep with soft-bristed brush all horizontal window frames and ledges on the exterior of building.

D. QUARTERLY

WINDOW FRAMES AND LEDGES
Clean with clear water, or a mild detrigent solution if necessary, all horizontal window
finmes and ledges on the setterior of the building. Care will be taken to buch up the
glass panes if the cleaning of the window frame causes any emulgae or streaks.

VIII. STAIRWELLS

A. <u>NIGHTLY</u> 1. All stairwells will be policed.

B. MONTHLY . 1. All stainvells will be cleaned using a treated dust mop. The walls will be spot-cleaned as nocessary.

IX. JANITOR CLOSETS, ENGINEERING AREA, AND STORAGE AREAS

-12-

A. <u>NIGHTLY</u> 1. All janitot's closets, mop sinks, storage rooms or areas, restrooms, lunchrooms and work areas, if applicable, provided by Manager for use by Contractor personnel, will be kept clean and ordering at all times. Mog sinks and the area immediately adjacent will be cleaned immediately after each

CONTRACTOR Initials:

use. Mops shall not be left in water or buckets. Restrooms will be maintained in the same condition as the public restrooms. Service areas will be spot cleaned as necessary. Service area floors should be swept with a broom daily and dust mopped with a treated mop no less than once per week. Finished floors will be stripped and refinished. Empty trash from Engineering and Garage Manager's Offices.

B. WEEKLY 1. Demp map all composition floors in staneoroms. Decodorize and disinfect as required.

High dusting of these areas including all pipes, ducis, conduit, ventilation diffusors and grills and mechanical, electrical equipment exposed beneath the hung ceilings outside the mechanical room equipment.

X. DAY SERVICES

A. <u>DAILY</u> 1. The day janitors shall cover all janitorial problems and requests that arise during the day and shall perform the duties listed hereafter and any additional duties as may be requested by the Manager.

Police main lobby and entrance to building: trash pickup, ashtrays, spills, clean glass in the lobby doors (both main doors and garage lobby doors) and vacuuming of rain mats when in use.

- 3. Police and maintain elevator cabs.
- 4. Install and remove rain mats as necessary.

Respond and assist in the removal and cleanup of all water leakage in building common areas, restrooms and tenant suites.

Police lavatories at least twice per day to check for cleanliness and adequacy of paper supplies. Fill product dispensers as required.

Police and maintain exterior of building four times per day. This is to include the sidewalks and plaza areas.

Replace lamps in light fixtures as required. If lamp replacement does not correct the lighting problem, notify building engineering staff of the problem.

-23-

CONTRACTOR Initials:

B. <u>WEEKLY</u>
 1. Clean and polish stardpipe fire hose connections on the exterior of building.
 2. Clean entrance glass doors, inside and out.

CONTRACTOR Initials

-24-

EXHIBIT G

FORM ESTOPPEL

FORM OF TENANT'S ESTOPPEL CERTIFICATE

The undersigned, as Tenant under that certain Office Lease (the "Lease"), made and entered into as of , 2012 by and between as Landlord, and the undersigned, as Tenant, for Premises [on floors ()] and [in suites ()] of the office building located []], certifies as follows as of the date hereof (and initially capitalized terms used and not defined herein shall have the meanings set forth in the Lease):

1. Attached hereto as Exhibit B is a true and correct copy of the Lease and all amendments and modifications thereto. The documents contained in Exhibit B represent the entire agreement between the parties as to the Premises.

2. The Lease Term for each Portion of the Premises commenced on the date set forth on Exhibit A, and the Lease Term for each Portion of the Premises expires on the date set forth on Exhibit A, subject to renewal in accordance with the Lease, and the undersigned has no option to terminate or cancel the Lease or to purchase all or any part of the Premises, the Building and/or the Project, except as set forth in the Lease.

Base Rent commenced on

4. The Lease is in full force and effect and has not been modified, supplemented or amended in any way except as provided in Exhibit B.

5. Tenant has not transferred, assigned, or sublet any portion of the Premises nor entered into any license or concession agreements with respect thereto except as follows:

6. All monthly installments of Base Rent, all Additional Rent and all monthly installments of estimated Additional Rent have been paid through . The current monthly installment of Base Rent is \$

7. To Tenant's actual knowledge without duty of inquiry, Landlord is not in default under the Lease, except as follows:

8. No rental has been paid more than thirty (30) days in advance and no security has been deposited with Landlord except as provided in the Lease.

9. To Tenant's actual knowledge without duty of inquiry, there are no existing defenses or offsets, or claims or any basis for a claim, that the undersigned has against Landlord, except as follows:

10. If Tenant is a corporation or partnership, Tenant is a duly formed and existing entity qualified to do business in California and Tenant has full right and authority to execute and deliver this Estoppel Certificate and each person signing on behalf of Tenant is authorized to do so.

11. There are no actions pending against the undersigned under the bankruptcy or similar laws of the United States or any state.

12. To Tenant's actual knowledge without duty of inquiry, all tenant improvement work to be performed by Landlord under the Lease has been completed in accordance with the Lease and all reimbursements and allowances due to the undersigned under the Lease in connection with any tenant improvement work have been paid in full, except as follows:

13. Tenant has no options or rights to renew or extend the term of the Lease, except as follows: . . Tenant has no expansion options, rights of first refusal or rights of first offer to lease additional space in the Project, except as set forth in the Lease. Tenant has no options, rights of first refusal or rights of first offer to purchase the Premises, the Building, the Project, or any part thereof and/or the land on which the Premises are situated, except as follows:

The undersigned acknowledges that this Estoppel Certificate may be delivered to Landlord or to a prospective mortgagee or prospective purchaser, and acknowledges that said prospective mortgagee or prospective purchaser will be relying upon the statements contained herein in making the loan or acquiring the property of which the Premises are a part and that receipt by it of this certificate is a condition of making such loan or acquiring such property.

Executed at on the day of , 20.

"Tenant	».			
а			,	
By:	Its:			
2				

<u>EXHIBIT H</u>

FORM LETTER OF CREDIT

ISSUANCE DATE: xxxxx

BENEFICIARY: XXXX

APPLICANT: xxxx

IRREVOCABLE STANDBY LETTER OF CREDIT NO.

DEAR SIR OR MADAM:

BY ORDER OF OUR CLIENT, XXXXX WE, DEUTSCHE BANK AG, NEW YORK BRANCH HEREBY ISSUE IN FAVOR OF XXXXX OUR IRREVOCABLE STANDBY LETTER OF CREDIT NO. , FOR AN AMOUNT NOT TO EXCEED IN THE AGGREGATE US\$XXXXX.00 (XXXXXX and 00/100 UNITED STATED DOLLARS) EFFECTIVE IMMEDIATELY AND EXPIRING AT OUR OFFICE ON XXXXXX.

FUNDS UNDER THIS LETTER OF CREDIT ARE AVAILABLE TO YOU ON DEMAND UPON RECEIPT OF EACH OF THE FOLLOWING:

- 1. DRAFT DRAWN AT SIGHT ON DEUTSCHE BANK AG, NEW YORK BRANCH, FOR THE AMOUNT OF THE DRAWING MARKED "DRAWN UNDER DEUTSCHE BANK AG, NEW YORK BRANCH LETTER OF CREDIT NO. XXXXXX".
- 2. A DATED STATEMENT, SIGNED BY AUTHORIZED REPRESENTATIVE OF XXXXX (SIGNED AS SUCH) STATING ANY OF THE FOLLOWING:

"THE UNDERSIGNED HEREBY CERTIFIES THAT THE LANDLORD, EITHER (A) UNDER THE LEASE (DEFINED BELOW), OR (B) AS A RESULT OF THE TERMINATION OF SUCH LEASE, HAS THE RIGHT TO DRAW DOWN THE AMOUNT OF USD IN ACCORDANCE WITH THE TERMS OF THAT CERTAIN OFFICE LEASE DATED [Insert Lease Date], AS AMENDED (COLLECTIVELY, THE "LEASE"), OR SUCH AMOUNT CONSTITUTES DAMAGES OWING BY THE TENANT TO BENEFICIARY RESULTING FROM THE BREACH OF SUCH LEASE BY THE TENANT THEREUNDER, OR THE TERMINATION OF SUCH LEASE, AND SUCH AMOUNT REMAINS UNPAID AT THE TIME OF THIS DRAWING."

OR

"THE UNDERSIGNED HEREBY CERTIFIES THAT WE HAVE RECEIVED A WRITTEN NOTICE OF [Insert Bank Name]'S ELECTION NOT TO EXTEND ITS STANDBY LETTER OF CREDIT NO. AND HAVE NOT RECEIVED A REPLACEMENT LETTER OF CREDIT WITHIN AT LEAST SIXTY (60) DAYS PRIOR TO THE PRESENT EXPIRATION DATE."

OR

"THE UNDERSIGNED HEREBY CERTIFIES THAT BENEFICIARY IS ENTITLED TO DRAW DOWN THE FULL AMOUNT OF LETTER OF CREDIT NO. AS THE RESULT OF THE FILING OF A VOLUNTARY PETITION UNDER THE U.S. BANKRUPTCY CODE OR A STATE BANKRUPTCY CODE BY THE TENANT UNDER THAT CERTAIN OFFICE LEASE DATED [Insert Lease Date], AS AMENDED

(COLLECTIVELY, THE "LEASE"), WHICH FILING HAS NOT BEEN DISMISSED AT THE TIME OF THIS DRAWING."

OR

"THE UNDERSIGNED HEREBY CERTIFIES THAT BENEFICIARY IS ENTITLED TO DRAW DOWN THE FULL AMOUNT OF LETTER OF CREDIT NO. AS THE RESULT OF AN INVOLUNTARY PETITION HAVING BEEN FILED UNDER THE U.S. BANKRUPTCY CODE OR A STATE BANKRUPTCY CODE AGAINST THE TENANT UNDER THAT CERTAIN OFFICE LEASE DATED [Insert Lease Date], AS AMENDED (COLLECTIVELY, THE "LEASE"), WHICH FILING HAS NOT BEEN DISMISSED AT THE TIME OF THIS DRAWING."

It is a condition of this Letter of Credit that it shall be automatically extended without amendment for one (1) year from the expiry date hereof or any future expiry date unless at least SIXTY (60) days prior to such expiration date WE SEND NOTICE TO YOU BY hand deliverY OR OVERNIGHT courier, at the address stated above, that we elect not to extend this Letter of Credit for any such additional period.

PARTIAL DRAWINGS AND MULTIPLE PRESENTATIONS MAY BE MADE UNDER THIS LETTER OF CREDIT, PROVIDED, HOWEVER, THAT EACH SUCH DEMAND THAT IS PAID BY US SHALL REDUCE THE AMOUNT AVAILABLE UNDER THIS LETTER OF CREDIT.

THIS LETTER OF CREDIT IS TRANSFERABLE. ANY TRANSFER REQUEST MUST BE PRESENTED TO US WITH THE ATTACHED FORM TOGETHER WITH THE ORIGINAL LETTER OF CREDIT. TRANSFERS TO DESIGNATED FOREIGN NATIONALS AND/OR SPECIALLY DESIGNATED NATIONALS ARE NOT PERMITTED AS BEING CONTRARY TO THE U.S. TREASURY DEPARTMENT OR FOREIGN ASSETS CONTROL REGULATIONS. ALL TRANSFER CHARGES ARE FOR THE ACCOUNT OF THE BENEFICIARY.

WE ENGAGE WITH YOU THAT DRAFTS DRAWN UNDER AND IN CONFORMITY WITH THE TERMS AND CONDITIONS OF THIS CREDIT WILL BE DULY HONORED ON PRESENTATION IF PRESENTED ON OR BEFORE THE EXPIRATION AT OUR COUNTERS AT 60 WALL STREET, 25th FLOOR, NEW YORK, NY 10005 ATTN.: TRADE AND RISK SERVICES DEPARTMENT.

PRESENTATION OF A DRAWING UNDER THIS LETTER OF CREDIT MAY BE MADE ON OR PRIOR TO THE THEN CURRENT EXPIRATION DATE HEREOF BY HAND DELIVERY, COURIER SERVICE, OR OVERNIGHT MAIL.

IN THE EVENT THAT THE ORIGINAL OF THIS LETTER OF CREDIT IS LOST, STOLEN, MUTILATED, OR OTHERWISE DESTROYED, WE HEREBY AGREE TO ISSUE A DUPLICATE ORIGINAL HEREOF UPON RECEIPT OF OUR STANDARD AFFIDAVIT OF THE LOSS, THEFT, MUTILATION, OR OTHER DESTRUCTION OF THE ORIGINAL HEREOF.

THIS CREDIT IS SUBJECT TO THE UNIFORM CUSTOMS AND PRACTICE FOR DOCUMENTARY CREDITS (2007 REVISION), INTERNATIONAL CHAMBER OF COMMERCE - PUBLICATION 600.

VERY TRULY YOURS,

DEUTSCHE BANK AG, NEW YORK BRANCH

ATTACHMENT "A" TO LETTER OF CREDIT NO. 839BGC0600161 REQUEST FOR TRANSFER OF LETTER OF CREDIT IN ITS ENTIRETY

Deutsche Bank AG, New York Branch 60 Wall Street 25th Floor New York, NY 10005-2858 Atth: Trade and Risk Services

Re: Deutsche Bank AG, New York Branch Irrevocable Standby Letter of Credit No.

For value received, the undersigned beneficiary hereby irrevocably transfers to:

NAME OF TRANSFEREE ADDRESS OF TRANSFEREE

CITY, STATE/COUNTRY ZIP

(hereinafter, the "transferee") all rights of the undersigned beneficiary to draw under above letter of credit, in its entirety.

bank

By this transfer, all rights of the undersigned beneficiary in such Letter of Credit are transferred to the transferee and the transferee shall have the sole rights as beneficiary hereof, including sole rights relating to any amendments, whether increases or extensions or other amendments and whether now existing or hereafter made. All amendments are to be advised directly to the transferee without necessity of any consent of or notice to the undersigned beneficiary.

The original of such letter of credit is returned herewith, and we ask you to endorse the transfer on the reverse thereof, and forward it directly to the transferee with your customary notice of transfer.

In payment of your transfer commission in amount equal to the lesser of (i) \$1,500.00 and (ii) the greater of (A) ½% of the amount transferred and (B) \$500.00.

Select one of the following: o we enclose a cashier's/certified check o we have wired funds to you through o we authorize you to debit our account #

with you, and in addition thereto, we agree to pay you on demand any expenses which may be incurred by you in connection with this transfer

Very truly yours, [BENEFICIARY NAME]

Authorized Signature

Attachment "A"

The signature(s) of with title(s) as stated conforms to those on file with us; are authorized for the execution of such instrument; and the beneficiary has been approved under our bank's Customer Identification Program. Further, pursuant to Section 326 of the USA Patriot Act and the applicable regulations promulgated thereunder, we represent and warrant that the undersigned bank: (i) is subject to a rule implementing the anti-money laundering compliance program requirements of 31 U.S.C. section 5318(h); (ii) is regulated by a Federal functional regulator [as such term is defined in 31 C.F.R. section 103.120(a) (2)]; and (iii) has a Customer Identification Program that fully complies with the requirements of the regulations.

(Signature of Authenticating Bank)	(Name of Bank)		
(Printed Name/Title)	(Date)		
	Confirmation of Authenticatin	(i) ag Bank's signature performed by:	FOR BANK USE ONLY
	Date:	Time:	a.m./p.m.
	4		

RECORDING REQUESTED BY AND WHEN RECORDED MAIL TO:

Paul Hastings LLP 55 Second Street San Francisco, California 94105

(Space Above This Line is For the Recorder's Use Only)

This document is not subject to documentary transfer tax because the term of the lease does not exceed 35 years.

MEMORANDUM OF LEASE

THIS MEMORANDUM OF LEASE (this "<u>Memorandum</u>") is made and entered into as of December , 2012, by and between HUDSON RINCON CENTER, LLC, a Delaware limited liability company ("<u>Landlord</u>"), and salesforce.com, inc., a Delaware corporation ("<u>Tenant</u>").

WITNESSETH:

WHEREAS, Landlord owns that certain real property in the City and County of San Francisco, State of California commonly known as One Rincon, located at 101 Spear Street, San Francisco, California 94105 and Two Rincon, located at 121 Spear Street, San Francisco, California 94105, being more particularly described on Exhibit A attached hereto and made a part hereof (the "Project");

WHEREAS, Landlord and Tenant entered into that certain unrecorded Lease, dated as of the date hereof (the "Lease"), whereby Tenant leased a portion of the Project from Landlord being more particularly described in the Lease (the "Premises"); and

WHEREAS, Landlord and Tenant desire to evidence the Lease in the official records maintained by the City and County of San Francisco, California by this Memorandum.

NOW, THEREFORE, for good and sufficient consideration acknowledged in the Lease, Landlord has demised, leased and let unto Tenant the Premises, as follows:

Section 1. Defined Terms. Initially capitalized terms used but not defined herein shall have the meanings set forth in the Lease.

Section 2. <u>Term</u>. The Lease Term shall be approximately fifteen (15) years, as set forth in the Lease, commencing on the Lease Commencement Date with respect to the Tranche I Premises. Subject to the terms and conditions set forth in the Lease, Tenant has two (2) consecutive options to extend the Lease Term for each of the Tranche I Premises, the Tranche II Premises and the Tranche III Premises, for five (5) years each.

Section 3. Right of First Refusal (Lease). Subject to the terms and conditions set forth in the Lease, Tenant has a right of first refusal to lease additional space in a portion of the Project commonly known as "One Rincon," located at 101 Spear Street, San Francisco, California ("Building One").

Section 4. Right of First Offer (Lease). Subject to the terms and conditions set forth in the Lease, Tenant has a right of first offer to lease additional space in a portion of the Project commonly known as "Two Rincon," located at 121 Spear Street, San Francisco, California ("Building Two").

Section 5. Right of First Offer (Purchase). Subject to the terms and conditions set forth in the Lease, Tenant has a right of first offer to purchase Building One and the Project, as applicable.

Section 6. Tenant Restrictions. The Lease includes certain restrictions on leasing of space within both Building One and Building Two to Tenant Competitors and Competitor Affiliates, as more particularly set forth in the Lease.

Section 7. Lease Incorporation; Purpose of Memorandum. This Memorandum is subject to all conditions, terms and provisions of the Lease, which agreement is hereby adopted and made a part hereof by reference to the same, in the same manner as if all the provisions thereof were set forth herein in full. This Memorandum has been executed for the purpose of recordation in order to give notice of all of the terms, provisions and conditions of the Lease, and is not intended, and shall not be construed, to define, limit, or modify the Lease. This Memorandum is not a complete summary of the Lease, nor shall any provisions of the Lease.

Section 8. Conflict. In the event of a conflict between the terms of the Lease and this Memorandum, the Lease shall prevail. Reference should be made to the Lease for a more detailed description of all matters contained in this Memorandum.

Section 9. Additional Documents. Upon any termination of the Lease, Landlord and Tenant agree to execute and deliver such documents as are reasonably required and are reasonable approved by the parties to terminate and release this Memorandum of record.

Section 10. Counterparts. This Memorandum may be executed in as many counterparts as may be deemed necessary and convenient, and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed an original, but all such counterparts shall constitute one and the same instrument.

[SIGNATURES APPEAR ON FOLLOWING PAGE]

IN WITNESS WHEREOF, Landlord and Tenant have executed this Memorandum effective as of the date first written above.

LANDLORD:	a Delav	vare lir	nited lia	ter, LLC, iability company
	By:	a Del		ter Commercial, LLC, limited liability company, nber
		By:	a Mary	on Pacific Properties, L.P., ryland limited partnership, le member
			By:	Hudson Pacific Properties, Inc., a Maryland corporation, its general partner
				By: Name: Title:
TENANT:	salesfor a Delav		n, inc., orporatio	
	By: Name: Its:			
	3			

County of			
On Da	before	me, Here Insert Name and Title of Officer	
personally	appeared		
		Name(s) of Signer(s)	
		,	
to the with authorized upon beha	nin instrument and ack d capacity(ies), and that alf of which the persor	f satisfactory evidence to be the person(s) whose name(s) is/are subscribed nowledged to me that he/she/they executed the same in his/her/their t by his/her/their signature(s) on the instrument the person(s), or the entity s) acted, executed the instrument.	
	is true and correct.	RJURY under the laws of the State of California that the foregoing	Place Notary Seal Above
WITNESS	S my hand and official	seal.	
Signature			
Signature		Signature of Notary Public	
		4	4

County of				
On Date	before me,	Here Insert Name and Title of Officer		
personally appear	red			
		Name(s) of Signer(s)		
		,		
to the within instr authorized capaci upon behalf of wh I certify under PE	rument and acknowledged to ty(ies), and that by his/her/th hich the person(s) acted, exe ENALTY OF PERJURY und	evidence to be the person(s) whose name(s) is/are subscribed one that he/she/they executed the same in his/her/their heir signature(s) on the instrument the person(s), or the entity cuted the instrument. er the laws of the State of California that the foregoing		
paragraph is true	and correct.		Place Notary Seal Above	
WITNESS my ha	nd and official seal.			
Signature	Sign	ature of Notary Public		
		5		

EXHIBIT A TO MEMORANDUM OF LEASE

LEGAL DESCRIPTION OF THE PROJECT

That certain real property situated in the State of California, County of San Francisco, described as follows:

RINCON ONE PARCEL:

PARCEL A:

PARCEL THREE, AS SHOWN ON THE PARCEL MAP SHOWING THE SUBDIVISION OF AIRSPACE FILED JUNE 23, 1988, IN BOOK 37 OF PARCEL MAPS, AT PAGES 67-71, INCLUSIVE, OFFICIAL RECORDS OF THE CITY AND COUNTY OF SAN FRANCISCO.

BEING 100 VARA BLOCK NO. 321.

LOT 023, BLOCK 3716.

PARCEL B

AS AN APPURTENANT TO PARCEL A, REFERRED TO ABOVE, EASEMENTS OVER PARCELS ONE AND FOUR, AS SUCH PARCELS ARE SHOWN ON THE PARCEL MAP SHOWING THE SUBDIVISION OF AIRSPACE, FILED JUNE 23, 1988, IN BOOK 37 OF PARCEL MAPS, AT PAGES 67-71, INCLUSIVE, OFFICIAL RECORDS OF THE CITY AND COUNTY OF SAN FRANCISCO.

BEING 100 VARA BLOCK NO. 321.

AND AS SUCH EASEMENTS ARE PROVIDED FOR IN THAT CERTAIN DECLARATION ESTABLISHING RECIPROCAL EASEMETNS AND COVENANTS RUNNING WITH THE LAND DATED AS OF JUNE 27, 1988 AND RECORDED JUNE 27, 1988 IN BOOK E624, PAGE 1132, AS INSTRUMENT NO. E196829, AS SUPPLEMENTED BY THAT CERTAIN DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED MARCH 30, 1999 AND RECORDED MARCH 31, 1999 IN BOOK H353, PAGE 505, AS INSTRUMENT NO. 99-G541539-00, AND AS AMENDED BY THAT CERTAIN SECOND AMENDMENT TO DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED MARCH 30, 1999 AND RECORDED MARCH 31, 1999 IN BOOK H353, PAGE 505, AS INSTRUMENT NO. 99-G541539-00, AND AS AMENDED BY THAT CERTAIN SECOND AMENDMENT TO DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED JUNE 23, 2006 AND RECORDED JUNE 28, 2006 IN BOOK J171, PAGE 0021, AS INSTRUMENT NO. 2006-I201148-00, AND AS FURTHER AMENDED BY THAT CERTAIN THIRD AMENDMENT TO DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED JUNE 28, 2007 AND RECORDED JUNE 28, 2007 AND RECORDED JUNE 28, 2007 AND RECORDED JUNE 11, 2007, AS INSTRUMENT NO. 2007-I400751, IN THE OFFICIAL RECORDS OF THE CITY AND COUNTY OF SAN FRANCISCO.



RINCON TWO PARCEL:

PARCEL A:

PARCEL ONE, AS SHOWN ON THE PARCEL MAP SHOWING THE SUBDIVISION OF AIRSPACE FILED JUNE 23, 1988, IN BOOK 37 OF PARCEL MAPS, AT PAGES 67-71, INCLUSIVE, OFFICIAL RECORDS OF THE CITY AND COUNTY OF SAN FRANCISCO.

BEING 100 VARA BLOCK NO. 321.

LOT 021, BLOCK 3716.

PARCEL B:

AS AN APPURTENANT TO PARCEL A, REFERRED TO ABOVE, EASEMENTNS OVER PARCELS TWO, THREE AND FOUR, AS SUCH PARCELS ARE SHOWN ON THE PARCEL MAP SHOWING THE SUBDIVISION OF AIRSPACE FILED JUNE 23, 1988, IN BOOK 37 OF PARCEL MAPS, AT PAGES 67-71, INCLUSIVE, OFFICIAL RECORDS OF THE CITY AND COUNTY OF SAN FRANCISCO.

BEING 100 VARA BLOCK NO. 321.

AND AS SUCH EASEMENTS ARE PROVIDED FOR IN THAT CERTAIN DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED AS OF JUNE 27, 1988 AND RECORDED JUNE 27, 1988 IN BOOK E624, PAGE 1132, AS INSTRUMENT NO. E196829, AS SUPPLEMENTED BY THAT CERTAIN DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED MARCH 30, 1999 AND RECORDED MARCH 31, 1999 IN BOOK H353, PAGE 505, AS INSTRUMENT NO. 99-G541539-00, AND AS AMENDED BY THAT CERTAIN SECOND AMENDMENT TO DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED MARCH 30, 1999 AND RECORDED MARCH 31, 1999 IN BOOK H353, PAGE 505, AS INSTRUMENT NO. 99-G541539-00, AND AS AMENDED BY THAT CERTAIN SECOND AMENDMENT TO DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED JUNE 23, 2006 AND RECORDED JUNE 28, 2006 IN BOOK J171, PAGE 0021, AS INSTRUMENT NO. 2006-I201148-00, AND AS FURTHER AMENDED BY THAT CERTAIN THIRD AMENDMENT TO DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND COVENANTS RUNNING WITH THE LAND DATED JUNE 28, 2007 AND RECORDED JUNE 28, 2007 AND RECORDED JUNE 21, 2007-1400751, IN THE OFFICIAL RECORDS OF THE CITY AND COUNTY OF SAN FRANCISCO.

<u>EXHIBIT J</u>

RULES AND REGULATIONS

Tenant shall faithfully observe and comply with the following Rules and Regulations. Landlord shall not be responsible to Tenant for the nonperformance of any of said Rules and Regulations by or otherwise with respect to the acts or omissions of any other tenants or occupants of the Project. In the event of any conflict between the Rules and Regulations and the other provisions of this Lease, the latter shall control.

- 1. Tenant shall bear the cost of any lock changes or repairs required by Tenant. Two keys will be furnished by Landlord for the Premises.
- 2. All doors opening to public corridors shall be kept closed at all times except for normal ingress and egress to the Premises.

3. Landlord reserves the right to close and keep locked all entrance and exit doors of the Building during such hours as are customary for comparable buildings in the San Francisco, California area, provided, however, that Tenant shall be provided access to the Project Building and Common Areas twenty-four (24) hours per day, seven (7) days per week. Tenant, its employees and agents must be sure that the doors to the Building are securely closed and locked when leaving the Premises if it is after the normal hours of business for the Building. Any tenant, its employees, agents or any other persons entering or leaving the Building at any time when it is so locked, or at any time that is considered to be after normal business hours for the Building, may be required to sign the Building register. Access to the Building may be refused unless the person seeking access has proper identification or has a previously arranged pass for access to the Building. Landlord will furnish passes to persons for whom Tenant requests same in writing. In case of invasion, mob, riot, public excitement, or other commotion, Landlord reserves the right to temporarily prevent access to the Building or the Project during the continuance thereof by any means it deems appropriate for the safety and protection of life and property.

4. Landlord shall have the right to reasonably prescribe the weight, size and position of all safes and other heavy property brought into the Building and also the times and manner of moving the same in and out of the Building. Safes and other heavy objects shall, if considered necessary by Landlord, stand on supports of such thickness as is necessary to properly distribute the weight. Landlord will not be responsible for loss of or damage to any such safe or property in any case. Any damage to any part of the Building, its contents, occupants or visitors by moving or maintaining any such safe or other property shall be the sole responsibility and expense of Tenant.

5. The requirements of Tenant will be attended to only upon application at the management office for the Project or at such office location designated by Landlord. Employees of Landlord shall not perform any work or do anything outside their regular duties unless under special instructions from Landlord.

6. No sign, advertisement, notice or handbill shall be exhibited, distributed, painted or affixed by Tenant on any part of the Premises (to the extent the same can be seen from outside the Premises) or the Building without the prior written consent of the Landlord.

7. The toilet rooms, urinals, wash bowls and other apparatus shall not be used for any purpose other than that for which they were constructed, and no foreign substance of any kind whatsoever shall be thrown therein. The expense of any breakage, stoppage or damage resulting from the violation of this rule shall be borne by the tenant who, or whose servants, employees, agents, visitors or licensees shall have caused same.

8. Except for vending machines intended for the sole use of Tenant's employees and invitees, no vending machine shall be installed, maintained or operated upon the Premises without the written consent of Landlord.

Tenant shall not bring into or keep within the Project, the Building or the Premises any firearms, birds or aquariums.

10. Landlord reserves the right to exclude or expel from the Project any person who, in the reasonable judgment of Landlord, is intoxicated or under the influence of liquor or drugs, or who shall in any manner do any act in violation of any of these Rules and Regulations.

11. Tenant shall participate in recycling programs to the extent required by Law or LEED certification.

9

12. Tenant shall store all its trash and garbage within the interior of the Premises. No material shall be placed in the trash boxes or receptacles if such material is of such nature that it may not be disposed of in the ordinary and customary manner of removing and disposing of trash and garbage in San Francisco, California without violation of any Law governing such disposal. All trash, garbage and refuse disposal shall be made only through entry-ways and elevators provided for such purposes at such times as Landlord shall designate.

13. Tenant shall comply with all safety, fire protection and evacuation procedures and regulations reasonably established by Landlord or any governmental agency.

14. No awnings or other projection shall be attached to the outside walls of the Building without the prior written consent of Landlord, and no curtains, blinds, shades or screens shall be attached to or hung in, or used in connection with, any window or door of the Premises other than as approved by Landlord (such approval not to be unreasonably withheld, conditioned or delayed). Neither the interior nor exterior of any windows shall be coated or otherwise sunscreened without the prior written consent of Landlord (such approval not to be unreasonably withheld, conditioned or delayed).

15. The sashes, sash doors, skylights, windows, and doors that reflect or admit light and air into the halls, passageways or other public places in the Building shall not be covered or obstructed by Tenant, nor shall any bottles, parcels or other articles be placed on the windowsills.

16. Tenant must comply with applicable "NO-SMOKING" ordinances and all related, similar or successor ordinances, rules, regulations or codes. If Tenant is required under the ordinance to adopt a written smoking policy, a copy of said policy shall be on file in the office of the Building. In addition, no smoking of any substance shall be permitted within the Project except in specifically designated outdoor areas. Within such designated outdoor areas, all remnants of consumed cigarettes and related paraphernalia shall be deposited in ash trays and/or waste receptacles. No cigarettes shall be extinguished and/or left on the ground or any other surface of the Project. Cigarettes shall be extinguished only in ashtrays. Furthermore, in no event shall Tenant, its employees or agents smoke tobacco products or other substances (x) within any interior areas of the Project, or (y) within two hundred feet (200') of the main entrance of the Building or the main entrance of any of the adjacent buildings, or (z) within seventy-five feet (75') of any other entryways into the Building.

17. Tenant shall not use in any space or in the public halls of the Building, any hand trucks except those equipped with rubber tires and rubber side guards.

18. No auction, liquidation, fire sale, going-out-of-business or bankruptcy sale shall be conducted in the Premises without the prior written consent of Landlord.

19. No tenant shall use or permit the use of any portion of the Premises for living quarters, sleeping apartments or lodging rooms.

20. Tenant shall install and maintain, at Tenant's sole cost and expense, an adequate, visibly marked and properly operational fire extinguisher next to any duplicating or photocopying machines or similar heat producing equipment, which may or may not contain combustible material, in the Premises.

21. Landlord will enforce the foregoing Rules and Regulations in a non-discriminatory manner and to the extent that it declines to enforce any of the foregoing Rules and Regulations with respect to any other tenant in the building, Landlord will not be entitled to enforce such Rules and Regulations with respect to Tenant.

22. To the extent that there is any inconsistency between the foregoing Rules and Regulations and the other terms of this Lease, the terms of this Lease will prevail.

INTENTIONALLY OMITTED

EXHIBIT M

TENANT'S SIGNAGE

OFFICE LEASE

Hudson Rincon Center, LLC,

a Delaware limited liability company,

as Landlord,

and

salesforce.com, inc.,

a Delaware corporation

as Tenant.

TABLE OF CONTENTS

i

ARTICLE 1	PREMISES, BUILDING, PROJECT, AND COMMON AREAS
ARTICLE 2	LEASE TERM; OPTION TERM(S)
ARTICLE 3	BASE RENT
ARTICLE 4	ADDITIONAL RENT
ARTICLE 5	USE OF PREMISES
ARTICLE 6	SERVICES AND UTILITIES
ARTICLE 7	REPAIRS
ARTICLE 8	ADDITIONS AND ALTERATIONS
ARTICLE 9	COVENANT AGAINST LIENS
ARTICLE 10	INDEMNIFICATION AND INSURANCE
ARTICLE 11	DAMAGE AND DESTRUCTION
ARTICLE 12	NONWAIVER
ARTICLE 13	CONDEMNATION
ARTICLE 14	ASSIGNMENT AND SUBLETTING
ARTICLE 15	SURRENDER OF PREMISES; OWNERSHIP AND REMOVAL OF TRADE FIXTURES
ARTICLE 16	HOLDING OVER
ARTICLE 17	ESTOPPEL CERTIFICATES
ARTICLE 18	SUBORDINATION
ARTICLE 19	DEFAULTS; REMEDIES
ARTICLE 20	COVENANT OF QUIET ENJOYMENT
ARTICLE 21	LETTER OF CREDIT
ARTICLE 22	INTENTIONALLY OMITTED
ARTICLE 23	SIGNS
ARTICLE 24	COMPLIANCE WITH LAW
ARTICLE 25	LATE CHARGES
ARTICLE 26	LANDLORD'S RIGHT TO CURE DEFAULT; PAYMENTS BY TENANT
ARTICLE 27	ENTRY BY LANDLORD
ARTICLE 28	TENANT PARKING
ARTICLE 29	MISCELLANEOUS PROVISIONS

SCHEDULES

SCHEDULE 1 — LANDLORD'S WIRING INSTRUCTIONS

SCHEDULE 23 — INTERIOR MAIN ELEVATOR LOBBY SIGNAGE

EXHIBITS

- A-1 OUTLINE OF TRANCHE I PREMISES
- A-2 OUTLINE OF TRANCHE II PREMISES
- A-3 OUTLINE OF TRANCHE III PREMISES
- B RENT SCHEDULE
- C TENANT WORK LETTER
- D FORM OF NOTICE OF LEASE TERM DATES
- E ROOF DECK

Page

- F JANITORIAL SPECIFICATIONS
- G FORM ESTOPPEL
- H FORM LETTER OF CREDIT
- I MEMORANDUM OF LEASE
- J RULES AND REGULATIONS
- K INTENTIONALLY OMITTED
- L-1 BUILDING ONE SUPERIOR RIGHTS HOLDERS
- L-2 BUILDING TWO SUPERIOR RIGHTS HOLDERS
- M TENANT'S SIGNAGE

Page

			_
2nd]	Floor Premises		
	itement Event		
	ountant		
	litional Rent		
	vocate Arbitrators		
	rations		
	illary Equipment		
	itration Agreement		
	e Rent		
	e Rent Abatement Remedy		
	e Year		
	eficial Occupancy Period		
	w Out Date		
	w Out Notice		
Brie	fs		
Bro	kers		
Buil	lding		
	lding Common Areas,		
Buil	lding Office Pool Direct Expenses		
	lding Structure		
	lding Systems		
	ualty		
	nmon Areas		
	templated Effective Date		
	templated Transfer Space		
cont			
	metic Alterations		
	nage Termination Date		
	nage Termination Notice		
	mate		
	mate Statement		
	mated Excess		
Exc			
	ense Year t Offer Space		
	t Rebuttals		
	t Refusal Notice		
	t Refusal Space		
	t Refusal Space Amendment		
	ze Majeure		
	und Floor Premises		
	ardous Substance		
	ding Over Damages		
	idays		
HV			
	ntion to Transfer Notice		
	rest Rate		
	dlord		
	dlord Parties		
Lan	dlord's Initial Statement		
		iii	

Page(s)

	Page(s)
Lease	1
Lease Term	15
Lease Year	15
Mail	63
Managing Member Interest	14
Market Area	16
Neutral Arbitrator	17
Nine Month Period	46
Notices	63
Operating Expenses	21
Option Term	16
Original Tenant	10
Other Tenant Improvements	67
Outside Agreement Date	17
Outside Agreenen Date	17
	1,9
Premises	1,9
Project	
Project Common Areas,	9
Proposition 13	26
Reassessment	28
Renovations	66
Rent	21
Review Period	30
Rincon Center	9
Ruling	18
Second Rebuttals	18
Statement	29
Subject Space	43
Suite 250 Premises	1
Suite 255 Premises	1
Summary	1
Superior Right Holders	11
Tax Expenses	26
Telecommunications Equipment	34
Tenant	1
Tenant Parties	39
Tenant Termination Date	
	19
Tenant Termination Notice	19
Tenant Termination Right	19
Tenant Work Letter	9
Tenant's Initial Statement	18
Tenant's Rebuttal Statement	18
Tenant's Share	28
Termination Fee	19
Tranche I Premises	1
Tranche II Premises	1
Tranche III Premises	1
Transfer Notice	43
Transfer Premium	45
Transferee	43
Transfers	43
iv	

FIRST AMENDMENT TO OFFICE LEASE

This FIRST AMENDMENT TO OFFICE LEASE ("First Amendment") is made and entered into as of April 20, 2016, by and between HUDSON RINCON CENTER, LLC, a Delaware limited liability company ("Landlord"), and salesforce.com, inc., a Delaware corporation ("Tenant").

<u>RECITALS</u>:

A. Landlord and Tenant are parties to the Office Lease dated December 28, 2012 (the "Lease"), whereby Tenant leases space (the "Existing Premises") containing approximately 235,733 rentable square feet of space ("RSF") in the office building known as "One Rincon" located at 101 Spear Street, San Francisco, California 94105 (the "Building").

B. Tenant desires to expand the Existing Premises to include Suite 400 in the Building, containing approximately 23,683 RSF on the 4th floor of the Building (the "**Expansion Premises**"), as shown on **Exhibit A** attached hereto, and to make other modifications to the Lease, and in connection therewith, Landlord and Tenant desire to amend the Lease as hereinafter provided.

$\underline{A} \underline{G} \underline{R} \underline{E} \underline{E} \underline{M} \underline{E} \underline{N} \underline{T}$:

NOW, THEREFORE, in consideration of the foregoing recitals and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

1. Capitalized Terms. All capitalized terms when used herein shall have the same meaning as is given such terms in the Lease unless expressly superseded by the terms of this First Amendment.

2. Modification of Premises. Effective as of the date that is the later of (i) November 1, 2016, and (ii) the date that the Expansion Premises is delivered to Tenant in the condition required by this First Amendment (the "Expansion Commencement Date") Tenant shall lease from Landlord and Landlord shall lease to Tenant the Expansion Premises. Consequently, effective upon the Expansion Commencement Date, the Existing Premises shall be increased to include the Expansion Premises. The Expansion Premises shall be deemed to be a part of the "Tranche II Premises", as defined in <u>Section 2.3</u> of the Summary to the Lease. Landlord and Tenant hereby acknowledge that such addition of the Expansion Premises to the Tranche II Premises shall, effective as of the Expansion Commencement Date, (a) increase the size of the total Premises to approximately 259,416 RSF. The Existing Premises and the Expansion Premises may hereinafter collectively be referred to as the "Premises".

3. Expansion Term. The term of Tenant's lease of the Expansion Premises (the "Expansion Term") shall commence on the Expansion Commencement Date and shall expire

coterminously with Tenant's Lease of the Tranche II Premises on the "Tranche II Lease Expiration Date" as defined in Section 3.4 of the Summary to the Lease, unless sooner terminated as provided in the Lease, as hereby amended.

4. Base Rent.

4.1. Existing Premises. Notwithstanding anything to the contrary in the Lease as hereby amended, Tenant shall continue to pay Base Rent for the Existing Premises in accordance with the terms of the Lease.

4.2. **Expansion Premises**. Commencing on the Expansion Commencement Date and continuing throughout the Expansion Term, Tenant shall pay to Landlord monthly installments of Base Rent for the Expansion Premises as follows:

riod During pansion Term	Annual Base Rent	Monthly Installment of Base Rent	Annual Rental Rate per RSF
Expansion Commencement Date		Redacted	
– October 31, 2017			
November 1, 2017 –			
October 31, 2018			
November 1, 2018 – October 31, 2019			
000000 51, 2019			
November 1, 2019 –			
October 31, 2020			
November 1, 2020 –			
October 31, 2021			
November 1, 2021 –			
October 31, 2022			
November 1, 2022 –			
October 31, 2023			
November 1, 2023 –			
October 31, 2024			
November 1, 2024 –			
October 31, 2025			
November 1, 2025 –			
October 31, 2026			
November 1, 2026 –			
April 30, 2027			
	2		

On or before the Expansion Commencement Date, Tenant shall pay to Landlord the Base Rent payable for the Expansion Premises for the first full month of the Expansion Term.

5. Tenant's Share of Building Direct Expenses

5.1. Existing Premises. With respect to the Existing Premises, Tenant shall continue to pay Tenant's Share of Building Office Pool Direct Expenses in accordance with the terms of <u>Article 4</u> of the Lease.

5.2. Expansion Premises. Except as specifically set forth in this Section 5.2, commencing on the Expansion Commencement Date, Tenant shall pay Tenant's Share of Building Office Pool Direct Expenses in connection with the Expansion Premises in accordance with the terms of <u>Article 4</u> of the Lease, provided that with respect to the calculation of Tenant's Share of Building Direct Expenses in connection with the Expansion Premises, the following shall apply:

- 5.2.1 Tenant's Share shall equal 8.9465%; and
- 5.2.2 the Base Year shall be the calendar year 2017.

Expansion Improvements

6.

6.1. **Expansion Allowance**. Landlord hereby grants Tenant an allowance (the "**Expansion Allowance**") in the amount of Redacted per RSF of the Expansion Premises (i.e., Redacted for the 23,683 RSF of the Expansion Premises) to be used for the construction of improvements in the Expansion Premises, which construction and use of the Expansion Allowance shall be on all of the terms and conditions of the Tenant Work Letter attached to the Lease as **Exhibit C** (the "**Tenant Work Letter**"), provided that, with respect to the Expansion Premises, the following shall apply.

- 6.1.1 The Tenant Improvement Allowance referenced in the Tenant Work Letter shall be deemed to refer to the Expansion Allowance.
- 6.1.2 Landlord shall deliver the Expansion Premises to Tenant in its currently existing, "as-is" condition, and Landlord shall not be required to perform any "Landlord Work" or any other work related to the "Delivery Condition" as defined in the Tenant Work Letter with respect to the Expansion Premises.
- 6.1.3 The "Outside Draw Date" with respect to the Expansion Allowance shall be the date that is eighteen (18) months after the Expansion Commencement Date (expected to be on or about May 1, 2018).
- 6.2. As-Is. Except as specifically set forth herein and in the Tenant Work Letter, Landlord shall not be obligated to provide or pay for any improvement work or services

related to the improvement of the Expansion Premises, and Tenant shall accept the Expansion Premises in its presently existing, "as-is" condition. Notwithstanding the foregoing, (a) all of the Building Systems serving the Expansion Premises shall be in good order and operating condition; (b) the Common Areas (including, without limitation, the paths of travel and accessways to the Expansion Premises) shall be in compliance with applicable Laws to the extent required in order for Landlord to maintain any certificate of occupancy issued for the Building or Expansion Premises, and (c) the Expansion Premises shall be delivered to Tenant vacant and in broom clean condition. For purposes of Section 1938 of the California Civil Code, Landlord hereby discloses to Tenant, and Tenant hereby acknowledges, that the Project, Building and Premises have not undergone inspection by a Certified Access Specialist (CASp).

7. **Broker**. Landlord and Tenant hereby warrant to each other that they have had no dealings with any real estate broker or agent in connection with the negotiation of this First Amendment other than CBRE, Inc. (the "Broker"), and that they know of no other real estate broker or agent who is entitled to a commission in connection with this First Amendment. Each party agrees to indemnify and defend the other party against and hold the other party harmless from and against any and all claims, demands, losses, liabilities, lawsuits, judgments, and costs and expenses (including, without limitation, reasonable attorneys' fees) with respect to any leasing commission or equivalent compensation alleged to be owing on account of the indemnifying party's dealings with any real estate broker or agent, other than the Broker, occurring by, through, or under the indemnifying party. The terms of this Section 7 shall survive the expiration or earlier termination of the term of the Lease, as hereby amended.

8. Parking. The Parking Pass Ratio set forth in Section 9 of the Summary to the Lease shall apply to the Expansion Premises effective as of the Expansion Commencement Date.

9. No Further Modification. Except as set forth in this First Amendment, all of the terms and provisions of the Lease shall apply with respect to the Expansion Premises and shall remain unmodified and in full force and effect.

[Signatures appear on the following page.]

EXHIBIT A OUTLINE OF EXPANSION PREMISES

IN WITNESS WHEREOF, Landlord and Tenant have caused this First Amendment to be executed as of the date first set forth above.

"Landlord"

HUDSON RINCON CENTER, LLC, a Delaware limited liability company

- By: Rincon Center Commercial, LLC, a Delaware limited liability company, its sole member
 - By: Hudson Pacific Properties, L.P., a Maryland limited partnership, its sole member
 - By: Hudson Pacific Properties, Inc., a Maryland corporation, its general partner

By:	/s/ Mark T. Lammas		
Name:	Mark T. Lammas		
Title:	Chief Operating Officer Chief Financial Officer & Treasurer		

"Tenant"

salesforce.com, inc., a Delaware corporation

 By:
 /s/ Amy E. Weaver

 Name:
 Amy E. Weaver

 Title:
 EVP & General Counsel

EXHIBIT B

FF&E

(See Attached)

B-1

EXHIBIT C

FORM OF LETTER OF CREDIT

IRREVOCABLE STANDBY LETTER OF CREDIT NUMBER

ISSUE DATE: , 2018

ISSUING BANK:

ISSUING BANK: SILICON VALLEY BANK 3003 TASMAN DRIVE 2ND FLOOR, MAIL SORT HF210 SANTA CLARA, CALIFORNIA 95054

BENEFICIARY:

SALESFORCE.COM, INC. THE LANDMARK AT ONE MARKET, THIRD FLOOR SAN FRANCISCO, CA 94105 ATTN: CORPORATE TREASURER

APPLICANT:

TWILIO INC 375 BEALE ST, 3RD FLOOR SAN FRANCISCO, CALIFORNIA 94105 AMOUNT: US\$ (AND XX/100 U.S. DOLLARS)

EXPIRATION DATE:

PLACE OF EXPIRATION: ISSUING BANK'S COUNTERS AT ITS ABOVE ADDRESS

LADIES AND GENTLEMEN:

WE HEREBY ISSUE IN THE FAVOR OF SALESFORCE.COM, INC., A DELAWARE CORPORATION (THE "<u>BENEFICIARY</u>") OUR IRREVOCABLE TRANSFERABLE LETTER OF CREDIT NO. FOR THE ACCOUNT OF [], A [] ("[]"), FOR THE SUM OF U.S. \$[] (THE "<u>LETTER OF CREDIT</u>"), WHICH SUM IS AVAILABLE AGAINST THE BENEFICIARY'S SIGHT DRAFT(S) DRAWN ON US AND ACCOMPANIED BY A STATEMENT SIGNED BY BENEFICIARY WHICH STATEMENT SHALL READ AS FOLLOWS:

"WE HEREBY CERTIFY THAT THE BENEFICIARY IS ENTITLED TO DRAW UPON THIS LETTER OF CREDIT IN THE AMOUNT OF THE DRAFT SUBMITTED HEREWITH PURSUANT TO THAT CERTAIN SUBLEASE BETWEEN BENEFICIARY, AS SUBLANDLORD, AND [], AS SUBTENANT, AS THE SAME MAY HAVE BEEN AMENDED OR ASSIGNED."

C-1

IT IS A CONDITION OF THIS LETTER OF CREDIT THAT IT WILL BE AUTOMATICALLY EXTENDED WITHOUT AMENDMENT FOR PERIODS OF ONE YEAR FROM THE PRESENT OR ANY FUTURE EXPIRATION DATE. IN THE EVENT WE DO NOT EXTEND THIS LETTER OF CREDIT, WE SHALL NOTIFY YOU IN WRITING BY CERTIFIED MAIL, RETURN RECEIPT REQUESTED OR OVERNIGHT COUREIR SERVICE AT THE ABOVE LISTED ADDRESS, AT LEAST SIXTY (60) DAYS PRIOR TO THE THEN PRESENT EXPIRATION DATE.

IN THE EVENT THAT WE NOTIFY YOU THAT WE ELECT NOT TO EXTEND THIS LETTER OF CREDIT, YOU MAY DRAW HEREUNDER BY MEANS OF YOUR DRAFT EXECUTED BY BENEFICIARY WITHOUT PRESENTATION OF THE FOREGOING STATEMENT OR ANY ADDITIONAL DOCUMENTATION. IN NO EVENT SHALL THIS LETTER OF CREDIT BE AUTOMATICALLY EXTENDED BEYOND SEPTEMBER 28, 2028.

THIS LETTER OF CREDIT IS TRANSFERABLE IN WHOLE BUT NOT IN PART ONE OR MORE TIMES, BUT IN EACH INSTANCE ONLY TO A SINGLE BENEFICIARY AS TRANSFEREE AND FOR THE THEN AVAILABLE AMOUNT, ASSUMING SUCH TRANSFER TO SUCH TRANSFEREE WOULD BE IN COMPLIANCE WITH THEN APPLICABLE LAW AND REGULATION, INCLUDING BUT NOT LIMITED TO THE REGULATIONS OF THE U.S. DEPARTMENT OF TREASURY AND U.S. DEPARTMENT OF COMMERCE. AT THE TIME OF TRANSFER, THE ORIGINAL LETTER OF CREDIT AND ORIGINALS OR COPIES OF ALL AMENDMENTS, IF ANY, TO THIS LETTER OF CREDIT MUST BE SURRENDERED TO US AT OUR ADDRESS INDICATED IN THIS LETTER OF CREDIT TOGETHER WITH OUR TRANSFER FORM ATTACHED HERETO AS EXHIBIT A DULY EXECUTED. APPLICANT SHALL PAY OUR TRANSFER FEE OF ¼ OF 1% OF THE TRANSFER AMOUNT (MINIMUM US\$250.00 AND MAXIMUM US\$5,000.00) UNDER THIS LETTER OF CREDIT. EACH TRANSFER SHALL BE EVIDENCED BY EITHER (1) OUR ENDORSEMENT ON THE REVERSE OF THE LETTER OF CREDIT AND WE SHALL FORWARD THE ORIGINAL OF THE LETTER OF CREDIT SO ENDORSED TO THE TRANSFEREE OR (2) OUR ISSUING A REPLACEMENT LETTER OF CREDIT TO THE TRANSFEREE ON SUBSTANTIALLY THE SAME TERMS AND CONDITIONS AS THE TRANSFERRED LETTER OF CREDIT (IN WHICH EVENT THE TRANSFERRED LETTER OF CREDIT SHALL HAVE NO FURTHER EFFECT).

PARTIAL DRAWINGS AND MULTIPLE PRESENTATIONS ARE AUTHORIZED UNDER THIS LETTER OF CREDIT.

WE HEREBY AGREE THAT DRAFTS DRAWN IN ACCORDANCE WITH THE TERMS STIPULATED HEREIN WILL BE DULY HONORED UPON PRESENTATION AND DELIVERY OF DOCUMENTS AS SPECIFIED VIA OVERNIGHT COURIER SERVICE IF PRESENTED TO: SILICON VALLEY BANK, 3003 TASMAN DRIVE, 2ND FLOOR, MAIL SORT HF210, SANTA CLARA, CALIFORNIA 95054, ON OR BEFORE OR ANY AUTOMATICALLY EXTENDED EXPIRATION DATE. IF A CONFORMING DEMAND FOR PAYMENT IS PRESENTED BEFORE 10:00 A.M. PACIFIC TIME, PAYMENT SHALL BE MADE TO YOU OF THE AMOUNT DEMANDED IN IMMEDIATELY AVAILABLE FUNDS NOT LATER THAN 5:00 P.M. PACIFIC TIME ON THE FOLLOWING BUSINESS DAY. IF A CONFORMING DEMAND FOR PAYMENT IS PRESENTED AFTER 10:00 A.M. PACIFIC TIME, PAYMENT SHALL BE MADE TO YOU OF THE AMOUNT DEMANDED IN IMMEDIATELY AVAILABLE FUNDS NOT LATER THAN 5:00 P.M. PACIFIC TIME ON THE SECOND BUSINESS DAY. AS USED HEREIN, THE TERM "BUSINESS DAY" MEANS A DAY ON WHICH WE ARE OPEN AT OUR ABOVE ADDRESS IN SANTA CLARA, CALIFORNIA TO CONDUCT OUR LETTER OF CREDIT BUSINESS.

C-2

EXCEPT SO FAR AS IS OTHERWISE STATED, THIS IRREVOCABLE LETTER OF CREDIT IS SUBJECT TO THE INTERNATIONAL STANDBY PRACTICES, INTERNATIONAL CHAMBER OF COMMERCE, PUBLICATION NO. 590 ("ISP98"). AS TO MATTERS NOT COVERED BY ISP98, THIS LETTER OF CREDIT SHALL BE SUBJECT TO AND GOVERNED BY THE LAWS OF THE STATE OF CALIFORNIA.

VERY TRULY YOURS,

AUTHORIZED SIGNATURE

AUTHORIZED SIGNATURE

CONSENT TO SUBLEASE AGREEMENT

THIS CONSENT TO SUBLEASE AGREEMENT (this "Agreement") is made as of September 25, 2018, by and among HUDSON RINCON CENTER, LLC, a Delaware limited liability company ("Landlord"), SALESFORCE.COM, INC., a Delaware corporation ("Tenant"), and TWILIO INC., a Delaware corporation ("Subtenant").

RECITALS:

A. Reference is hereby made to that certain Office Lease dated as of December 28, 2012 (the "**Original Lease**"), as amended by that certain First Amendment to Office Lease dated as of April 20, 2016 (the "**First Amendment**"), whereby Tenant leases approximately 259,416 rentable square feet of space (the "**Premises**") of the office building known as "One Rincon" and located at 101 Spear Street, San Francisco, California 94105 (the "**Building**"). The Original Lease and the First Amendment are together referred to herein as the "**Lease**."

B. Pursuant to the terms of <u>Article 14</u> of the Original Lease, Tenant has requested Landlord's consent to that certain Sublease dated as of August 30, 2018, by and between Tenant and Subtenant (the "**Sublease**"), with respect to a subletting by Subtenant of the entire Premises, on the terms more particularly described in the Sublease (the "**Sublet Premises**"). A copy of the Sublease is attached hereto as <u>Exhibit A</u>. Landlord is willing to consent to the Sublease on the terms and conditions contained herein.

C. All defined terms not otherwise expressly defined herein shall have the respective meanings given in the Lease.

<u>AGREEMENT</u>:

1. Landlord's Consent. Landlord hereby consents to the Sublease; provided however, notwithstanding anything contained in the Sublease to the contrary, such consent is granted by Landlord only upon the terms and conditions set forth in this Agreement. The Sublease is subject and subordinate to the Lease. Landlord shall not be bound by any of the terms, covenants, conditions, provisions or agreements of the Sublease, except to the extent otherwise expressly agreed to in this Agreement. Notwithstanding the foregoing, Landlord hereby consents to the following:

1.1 Cosmetic Alterations. Subtenant shall be permitted to make Cosmetic Alterations to the Sublet Premises pursuant and subject to the terms of Section 8.1 of the Original Lease.

1.2 Decend Consent Transfers. Notwithstanding any provision to the contrary set forth in the Lease, Subtenant shall have the right, without the receipt of Landlord's consent, to assign Subtenant's entire interest in the Sublease pursuant to clause (B), (C), or (D) of Section 14.7 of the Original Lease, and any such assignee described in clause (B), (C) or (D) of Section 14.7 of the Original Lease shall be deemed a Permitted Transferee, provided that (i) Subtenant notifies Landlord no more than ten (10) business days after the effective date of any such assignment and promptly supplies Landlord with any documents or information reasonably requested by Landlord regarding such transferee, [ii] no assignment of the Sublease pursuant to this Section 1.2 shall relieve Subtenant from any liability under the Sublease, and (iii) the liability of Subtenant and such Permitted Transferee shall be joint and several.

2. Non-Release of Tenant; Further Transfers. Neither the Sublease nor this consent thereto shall release or discharge Tenant from any liability, whether past, present or future, under the Lease or alter the primary liability of Tenant to pay the rent and perform and comply with all of the obligations of Tenant to be performed under the Lease (including the payment of all bills rendered by Landlord for charges incurred by Subtenant for services and materials supplied to the Sublet Premises). Neither the Sublease nor this consent thereto shall be construed as a waiver of Landlord's right to consent to (x) any further subletting either by Tenant

or by Subtenant, or (y) any assignment by Tenant of the Lease or assignment by Subtenant of the Sublease, or (z) any portion of the Sublet Premises being used or occupied by any other party, in each case to the extent such consent is required under the Lease. Landlord may consent to subsequent sublettings and assignments of the Lease or the Sublease or any amendments or modifications thereto without notifying Tenant nor anyone else liable under the Lease and without obtaining their consent. No such action by Landlord shall relieve such persons from any liability to Landlord or otherwise with regard to the Sublet Premises.

3. Relationship With Landlord. Tenant hereby assigns and transfers to Landlord, pursuant to the express terms and conditions of this Section 3 and conditioned thereon, Tenant's interest in the Sublease and all rentals and income arising therefrom. Landlord, by consenting to the Sublease, agrees that Tenant may receive, collect and enjoy the rents accruing under the Sublease; provided, however, in the event Tenant shall be in Default (beyond any applicable notice and cure periods) in the performance of its obligations to Landlord under the Lease, Landlord may, at its option by notice to Tenant, elect to receive and collect, directly from Subtenant, all rent and any other sums owing under the Sublease, to the extent of the rent and any other sums due under the Lease, all as further set forth in Section 3.1, below. In the event that the Lease shall be terminated, Landlord may, at its option by notice to Tenant, either (i) terminate the Sublease, or (ii) elect to succeed to Tenant's interest in the Sublease and cause Subtenant to attorn to Landlord, as further set forth in Section 3.2, below.

3.1 Landlord's Election to Receive Rents. Landlord shall not, by reason of the Sublease, nor by reason of the collection of rents or any other sums from Subtenant pursuant to Section 3, above, be deemed liable to Subtenant for any failure of Tenant to perform and comply with any obligation of Tenant, and Tenant hereby irrevocably authorizes and directs Subtenant, upon receipt of any written notice from Landlord stating that a Default exists under the Lease in the performance of Tenant's obligations under the Lease (beyond any applicable notice and cure periods), to pay to Landlord the rents and any other sums due under the Lease. Tenant agrees that Subtenant shall have the right to rely upon any such statement and request from Landlord, and that Subtenant shall pay any such rents and any other sums due under the Lease. Tenant agrees that Subtenant shall have the right to rely upon any such statement and request from Leandlord, and that Subtenant shall not have any right or claim against Subtenant for any such rents or any other sums so paid by Subtenant to Landlord. Landlord shall credit Tenant with any rent or other sums received by Landlord under such assignment but the acceptance of any payment on account of rent from Subtenant as the result of any such Default shall in no manner whatsoever be deemed an attornment by Landlord to Subtenant to Landlord, be deemed a waiver by Landlord of any provision of the Lease or serve to release Tenant from any liability under the terms, covenants, conditions, provisions or agreements under the Lease. Notwithstanding the foregoing, any other payment of rent from Subtenant directly to Landlord, regardless of the circumstances or reasons therefor, shall in no manner whatsoever be deemed an attornment by Subtenant to Landlord in the absence of a specific written agreement signed by Landlord to such an effect.

3.2 Landlord's Election of Attornment. In the event Landlord elects, at its option, to cause Subtenant to attorn to Landlord pursuant to <u>Section 3(ii)</u>, above, Landlord shall undertake the obligations of Tenant under the Sublease from the time of the exercise of the option, but Landlord shall not (i) be liable for any prepayment of more than one month's rent or any security deposit paid by Subtenant (except to the extent actually received by Landlord), (ii) be liable for any previous act or omission of Tenant under the Lease or for any other defaults of Tenant under the Sublease, (iii) be subject to any defenses or offsets previously accrued which Subtenant may have against Tenant, or (iv) be bound by any changes or modifications made to the Sublease without the written consent of Landlord.

4. Transfer Premium. Landlord acknowledges and agrees that any amount paid by Subtenant under the Sublease for gross receipts or rental tax paid by Tenant to the City and County of San Francisco shall be excluded from the calculation of the Transfer Premium under and as defined in the Lease (i.e., such amounts shall not constitute rent, additional rent or other consideration payable by Subtenant in connection with the Transfer).

General Provisions

5

5.1 <u>Consideration for Sublease</u>. Tenant and Subtenant represent and warrant that there are no additional payments of rent or any other consideration of any type payable by Subtenant to Tenant with regard to the Sublet Premises other than as disclosed in the Sublease.

5.2 Brokerage Commission. Tenant and Subtenant covenant and agree that under no circumstances shall Landlord be liable for any brokerage commission or other charge or expense in connection with the Sublease and Tenant and Subtenant agree to protect, defend, indemnify and hold Landlord harmless from the same and from any cost or expense (including but not limited to attorneys' fees) incurred by Landlord in resisting any claim for any such brokerage commission.

5.3 **Recapture**. This consent shall in no manner be construed as limiting Landlord's ability to exercise its rights to recapture any portion of the Premises, as set forth in Section 14.4 of the Original Lease, in the event of a proposed future sublease or assignment of such portion of the Premises that triggers such rights of Landlord under Section 14.4 of the Original Lease.

5.4 <u>Controlling Law</u>. The terms and provisions of this Agreement shall be construed in accordance with and governed by the laws of the State of California.

5.5 Binding Effect. This Agreement shall be binding upon and inure to the benefit of the parties hereto, their heirs, successors and assigns. As used herein, the singular number includes the plural and the masculine gender includes the feminine and neuter.

5.6 Captions. The paragraph captions utilized herein are in no way intended to interpret or limit the terms and conditions hereof; rather, they are intended for purposes of convenience only.

5.7 <u>Partial Invalidity</u>. If any term, provision or condition contained in this Agreement shall, to any extent, be invalid or unenforceable, the remainder of this Agreement, or the application of such term, provision or condition to persons or circumstances other than those with respect to which it is invalid or unenforceable, shall not be affected thereby, and each and every other term, provision and condition of this Agreement shall be valid and enforceable to the fullest extent possible permitted by law.

5.8 <u>Attorneys' Fees</u>. If either party commences litigation against the other for the specific performance of this Agreement, for damages for the breach hereof or otherwise for enforcement of any remedy hereunder, the parties hereto agree to and hereby do waive any right to a trial by jury and, in the event of any such commencement of litigation, the prevailing party shall be entitled to recover from the other party such costs and reasonable attorneys' fees as may have been incurred.

5.9 Insurance. Notwithstanding anything to the contrary contained in the Sublease, Subtenant acknowledges and agrees that it will maintain throughout the Term any insurance coverage required to be maintained by Tenant under the Lease (and, for the avoidance of doubt, and notwithstanding any provision to the contrary in the Sublease, Subtenant will maintain property coverage required under the Lease on an "all risk" basis, as required by the Lease) and the proceeds of all such insurance shall be used and applied in accordance with the terms of the Lease.

[remainder of page intentionally left blank; signature page follows]

IN WITNESS WHEREOF, the parties have executed this Consent to Sublease Agreement as of the day and year first above written.

LANDLORD:

HUDSON RINCON CENTER, LLC, a Delaware limited liability company

- Rincon Center Commercial, LLC, a Delaware limited liability company By:
- Its: Sole Member
 - Hudson Pacific Properties, L.P., a Maryland limited partnership Sole Member By: Its:
 - - By: Hudson Pacific Properties, Inc.,
 - a Maryland corporation Its:
 - General Partner
 - /s/ Mark T. Lammas Mark T. Lammas Chief Operating Officer, Chief Financial Officer & Treasurer By: Name: Title:

TENANT:

SALESFORCE.COM, INC., a Delaware corporation

By:	/s/ George Kreitem
Name:	George Kreitem
Title:	Senior Vice President Real Estate

SUBTENANT:

TWILIO INC.,

а	Delaware	corporation

By:	/s/ Jeff Lawson
Name:	Jeff Lawson
Title:	Chief Executive Officer

SUBLEASE [SEE ATTACHED]

EXHIBIT A

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE SECURITIES EXCHANGE ACT OF 1934, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Jeffrey Lawson, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Twilio Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 8, 2018

/s/ JEFFREY LAWSON Jeffrey Lawson

Chief Executive Officer (Principal Executive Officer)

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE SECURITIES EXCHANGE ACT OF 1934, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Lee Kirkpatrick certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Twilio Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 8, 2018

/s/ LEE KIRKPATRICK Lee Kirkpatrick

Chief Financial Officer (Principal Accounting and Financial Officer)

CERTIFICATION OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

Pursuant to the requirement set forth in Rule 13a-14(b) of the Securities Exchange Act of 1934, as amended, (the "Exchange Act") and Section 1350 of Chapter 63 of Title 18 of the United States Code (18 U.S.C. §1350), Jeffrey Lawson, Chief Executive Officer of Twilio Inc. (the "Company"), and Lee Kirkpatrick, Chief Financial Officer of the Company, each hereby certifies that, to the best of his knowledge:

1. The Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2018, to which this Certification is attached as Exhibit 32.1 (the "Periodic Report"), fully complies with the requirements of Section 13(a) or Section 15(d) of the Exchange Act; and

2. The information contained in the Periodic Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 8, 2018

/s/ JEFFREY LAWSON

Jeffrey Lawson Chief Executive Officer (Principal Executive Officer)

/s/ LEE KIRKPATRICK Lee Kirkpatrick

Chief Financial Officer (Principal Accounting and Financial Officer)